CWI Tracts

Managing Editors

K.R. Apt (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.M. Schumacher (CWI, Amsterdam) N.M. Temme (CWI, Amsterdam)

Executive Editor

M. Bakker (CWI Amsterdam, e-mail: Miente.Bakker@cwi.nl)

Editorial Board

W. Albers (Enschede)
M.S. Keane (Amsterdam)
J.K. Lenstra (Eindhoven)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
A.J. van der Schaft (Enschede)
H.J. Sips (Delft, Amsterdam)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

CWI

P.O. Box 94079, 1090 GB Amsterdam, The Netherlands Telephone + 31 - 20 592 9333 Telefax + 31 - 20 592 4199 URL http://www.cwi.nl

CWI is the nationally funded Dutch institute for research in Mathematics and Computer Science.

Polling Systems

S.C. Borst

1991 Mathematics Subject Classification: 60K25 (Queuing Theory), 68M20 (Performance evaluation; queuing; scheduling), 90B22 (Queues and service) ISBN 90 6196 467 9 NUGI-code: 811

Copyright C1996, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

Preface

This monograph is devoted to the mathematical analysis of polling systems. Over the years, polling systems have found a wide range of applications in the performance evaluation of communication systems, computer networks, and manufacturing systems. This monograph provides a generalization and unification of the exact distributional results for the standard polling model, studies several optimization issues, and contains a detailed analysis of various extensions. One of the main extensions concerns *multiple-server* polling systems, which have received remarkably little attention in the literature so far.

The results presented in this monograph were obtained in the course of my PhD research at the Center for Mathematics & Computer Science (CWI) in Amsterdam. I am grateful to the Netherlands Organization for the Mathematics of Operations Research (LNMB) for supporting my research project.

Many people have contributed to this monograph in many ways. It has been a great pleasure and privilege to have Onno Boxma as my advisor. I have strongly benefited from his expertise and have appreciated his constant encouragement. My joint research with Marco Combé, Hanoch Levy, and Rob van der Mei has also been enjoyable and rewarding.

I am indebted to Ger Koole, Jacques Resing, and Rob van den Berg for their helpful comments. Finally, my thanks go to Adri Steenbeek, Peter de Waal, and Lieke van den Eersten-Schultze for their technical assistance.

Murray Hill, August 1996

Sem Borst

ii

CONTENTS

.

Preface					
1	Introduction				
	1.1	Background and motivation	1		
	1.2	Applications of polling models	3		
	1.3	Model description	5		
	1.4	Analysis of polling systems	13		
	1.5	Optimization of polling systems	18		
	1.6	Overview of the monograph $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	22		
2	Decomposition properties and pseudo-conservation laws in polling				
	models				
	2.1	Introduction	27		
	2.2	Queue length decomposition	29		
	2.3	Work decomposition	31		
	2.4	A queueing system with a customer collection mechanism \ldots	34		
3	Polling systems with zero and non-zero switch-over times				
	3.1	Introduction	43		
	3.2	Model description	45		
	3.3	The joint queue length distribution at various epochs	45		
	3.4	The joint queue length distribution at polling epochs	48		
	3.5	Marginal queue lengths and waiting times	52		
	3.6	Computational aspects	55		
4	A pseudo-conservation law for a polling system with a dormant				
	server		59		
	4.1	Introduction	59		
	4.2	Model description	61		
	4.3	A pseudo-conservation law	62		
	4.4	A comparison between the dormant and the non-dormant server			
		case	69		

iii

5	Ag	lobally gated polling system with a dormant server	75
	5.1	Introduction	75
	5.2	Model description	76
	5.3	The cycle time	77
	5.4	The waiting time	81
	5.5	The queue length	86
	5.A	Proof of Lemma 5.3.1	89
	5.B	Proof of Lemma 5.4.1	91
6	Opt	imization of k-limited service strategies	95
	6.1	Introduction	95
	6.2	Model description and preliminaries	97
	6.3	The constrained optimization problem	98
	6.4	Numerical results for the constrained problem	103
	6.5	The unconstrained optimization problem	107
	6.6	Numerical results for the unconstrained problem	112
	6.7	Concluding remarks and suggestions for further research	117
7	Opt	imization of fixed time polling schemes	119
	7.1	Introduction	119
	7.2	Model description	120
	7.3	Constructing an efficient ftp scheme I	121
	7.4	Constructing an efficient ftp scheme II	126
	7.5	Numerical results	132
	7.A	Proof of Lemma 7.3.1	136
	7.B	The Golden Ratio procedure	138
	7.C	A procedure based on extremal splittings	138
	7.D	Proof of Lemma 7.4.1	140
8	Optimal allocation of customer types to servers		
	8.1	Introduction	141
	8.2	Model description	143
	8.3	Finding an optimal random splitting	145
	8.4	The case of ordered customer types	149
	8.5	Finding an optimal source partitioning	154
	8.6	Concluding remarks and suggestions for further research	157
	8.A	Proof of Lemma 8.3.1	160
	8.B	Proof of Lemma 8.3.2	161
	8.C	Proof of Lemma 8.3.3	162
	8.D	Proof of Lemma 8.3.4	164
	8.E	A method for determining an optimal allocation	165

iv

9 Poll 9.1 9.2 9.3 9.4 9.5 9.6 9.A 9.B	ling systems with multiple coupled serversIntroductionAn $M/M/m$ queue with coupled servers and service interruptionModel descriptionThe joint queue length distribution IThe joint queue length distribution IIConcluding remarks and suggestions for further researchProof of Lemma 9.2.1Proof of Lemma 9.5.1	169 169 18171 177 182 188 190 192
10 Wai tem	ting-time approximations for multiple-server polling sys	s- 195
10.1	Introduction	195
10.2	Model description	190
10.0	The waiting time	199
10.5	Approximating the weighted sum $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$	202
10.6	Approximating the probabilities q_i^{i-1}	205
10.7	Numerical results	209
10.8	Concluding remarks and suggestions for further research	217
Bibliog	graphy	219

v

vi · · ·

CHAPTER 1

INTRODUCTION

1.1 BACKGROUND AND MOTIVATION

Queueing phenomena may be observed in several real-life situations, when service facilities (counters, elevators, telephone lines, traffic lights) cannot immediately render the amount or the kind of service required by their users. Also, at the byte level in modern data-handling technologies (communication systems, computer networks), queueing phenomena may be encountered which are typically less visible, but the effects of which at the user level are usually not less serious. Quite often, such congestion effects may be adequately studied by mathematical methods from queueing theory. Adopting the abstract terminology from queueing theory, the main entity in a queueing model is a *queue* or station, where customers arrive, which require some amount of service from a server. Typically, queueing models are of a stochastic nature, in the sense that the duration of the interarrival and service times of the successive customers is not exactly specified, but described in terms of probability distributions. The stochastic nature of queueing models reflects the fact that in most of the applications, it is intrinsically random or uncertain at what time demand occurs for what amount of service.

The classical model in queueing theory consists of a *single* queue attended by a *single* server. Single-server single-queue models have been studied extensively in the literature, cf. Cohen [73] for a rigorous treatment of the main analytical results. In several situations, the traditional single-server single-queue models have proven to be very successful in accurately predicting waiting times, queue lengths, and buffer overflow probabilities. However, in most of the recent applications, the parallel or distributed character of the service facilities involves queueing models with multiple servers, multiple queues, or both.

This monograph is primarily devoted to queueing models with *multiple* queues attended by a single server, visiting the queues one at a time, cf. Figure 1.1. Moving from one queue to another, the server typically incurs a non-negligible switch-over time. Such single-server multiple-queue models are commonly referred to as *polling* models. The term 'polling' originates from the polling data link control scheme, in which a central computer cyclically polls the terminals on a communication link to inquire whether they have any data to transmit. When a terminal completes the transmission of data, the data link may be used for some system overhead, and then the central computer polls the next terminal. In the associated polling model, the server represents the central computer, the queues correspond to the terminals, the customers represent the messages, and the switch-over time corresponds to the system overhead. In a broader perspective, polling models may arise in situations, in which there are multiple customer classes sharing a common resource, which is available to only one customer class at a time. In those situations, changing from one customer class to another usually involves a non-negligible overhead.

Stimulated by a wide variety of applications, polling models have been extensively studied in the literature, cf. Takagi [174], [175], [176] for a series of comprehensive surveys. In this monograph, we provide a generalization and unification of the main exact distributional results available for polling models, present a detailed analysis of various extensions, and discuss several optimization issues. One of the main extensions concerns *multiple-server* polling models, which are of considerable practical relevance. So far, however, they have received remarkably little attention in the literature, perhaps because of the combined mathematical difficulties arising in multiple-queue and multipleserver models.

The remainder of the chapter is organized as follows. In Section 1.2, we describe the main applications of polling models in communication systems, computer networks, and traditional fields of engineering like maintenance, manufacturing, and transportation. The wide diversity in applications is reflected in the numerous variants of polling models considered throughout the past decades, mostly focusing on the technologies emerging in the respective periods of time. It is however not in the scope of the monograph to present an encyclopedic categorization of the plethora of polling models considered in the literature. Instead, in Section 1.3, we provide rather a global classification, by identifying some fundamentally distinguishing features in the spectrum of polling models. In Section 1.4, we survey the state of the art in the analysis of polling systems. Rather than covering all technical details, we intend to illuminate the main concepts in the analysis of polling systems, which contribute to putting the monograph in the right perspective. In Section 1.5, we review the state of the art in the optimization of polling systems. Again, we seek to identify the main developments in the optimization of polling systems, rather than exhaustively address all the topics raised. In Section 1.6, we give an overview of the main results presented in the remainder of the monograph.



FIGURE 1.1. The basic polling system.

1.2 Applications of polling models

In this section, we describe the main applications of polling models in communication systems, computer networks, and traditional fields of engineering like maintenance, manufacturing, and transportation. For extensive surveys on the applications of polling models, we refer to Levy & Sidi [141], Grillo [112], and Takagi [173] (the latter two surveys focusing on computer-communication systems).

Computer-communication systems

For reasons of flexibility and efficiency, modern computer systems mostly have a distributed and parallel structure. Consider e.g. a local area network (LAN), consisting of a number of computers or stations interconnected by a common communication medium for exchanging packetized messages. For controlling the medium access in a LAN, one either needs a mechanism to resolve conflicts arising when more than one station starts to transmit simultaneously, or one needs a protocol to avoid such conflicts, by giving only one station at a time permission to transmit data packets. The performance evaluation of the latter category, the so-called conflict-free medium access mechanisms, has greatly stimulated the research in the area of polling models. Adopting the polling terminology, the server represents the right of transmission, the queues correspond to the stations, and the customers represent the data packets. In practice, several versions of conflict-free medium access protocols are known.

One variant is the *token* ring, i.e., there is an explicit or implicit token circulating on the communication ring, representing the right of transmission. When a station receives the token, it may start transmitting packets. As soon as the station finishes transmitting, it passes the token to the next station. So holding the token corresponds to utilizing the server.

Another variant is the *slotted* ring, i.e., the communication ring is subdivided into time slots of the size of a single packet, circulating at constant speed. When a station sees an empty slot pass by, it may put a packet in it. In case of destination release the receiving station subsequently removes the packet from the slot, while in case of source release the transmitting station empties the slot again. So occupying a slot corresponds to utilizing a server.

A slotted ring may be viewed as a multiple-server polling system (unless there is only a single slot). A token ring is in fact a single-server polling system, but the stations may happen to be interconnected by multiple token rings rather than only a single token ring.

Maintenance, manufacturing, transportation

In the first polling study that appeared in the open literature, Mack, Murphy, & Webb [148] considered a situation in which a patrolling repairman cyclically inspects a number of machines, checks whether or not a failure occurred, if so repairs the machine, and then moves to the next machine. In the associated polling model, the server represents the repairman, the queues correspond to the machines, and the customers represent the possible breakdowns. Königsberg & Mamer [130] studied a similar model, in which an operator at a fixed position serves a number of storage locations on a rotating carousel conveyor. Models with several independent rotating carousels have also been considered, cf. Kim & Königsberg [127], Bunday & El-Badri [61].

There are also various applications in manufacturing environments. Consider e.g. a flexible manufacturing system, in which a machine periodically changes over from performing one type of operations to another. Here the server represents the machine and the queues correspond to the various types of operations. A similar application is multi-product economic lot scheduling, cf. Sarkar & Zangwill [161].

Furthermore, there are applications in transportation networks. Consider e.g. a material handling system, in which a vehicle transfers loads from one machining center to another, cf. Bozer [55]. Here the server represents the vehicle,

1.3 Model description

the queues correspond to the machining centers, and the customers represent the loads. Similar applications are public transport systems, mail delivery, and elevator facilities, cf. Gamse & Newell [106], [107].

A last application that is worth mentioning is the control of traffic lights. In polling terms, the stream that is being given green light corresponds to the queue receiving service.

1.3 MODEL DESCRIPTION

As described in the previous section, polling models find a variety of applications in communication systems, computer networks, and fields like maintenance, manufacturing, and transportation. The wide diversity in applications is reflected in the numerous variants of polling models considered throughout the past decades, mostly focusing on the technologies emerging in the respective periods of time. However, as mentioned earlier, it is not in the scope of the monograph to present an exhaustive taxonomy of the abundance of polling models considered in the literature. Instead, we provide in this section rather a global classification, by identifying some fundamentally distinguishing features in the spectrum of polling models. For a series of comprehensive surveys of the overwhelming variety of polling models considered in the literature, we refer to Takagi [174], [175], [176].

The basic model

A polling model basically consists of multiple queues, Q_1, \ldots, Q_n , attended by a single server S. Customers arriving at Q_i are also referred to as type-*i* customers, $i = 1, \ldots, n$.

As usual in the recent polling literature, in the sequel the queues are always assumed to have infinite buffer capacity. In some applications (manufacturing, transportation), the inherent finiteness of the buffer capacity may play a major role in the operation of the system. However, in many applications, the finiteness of the buffer capacity only tends to have a minor influence on the performance of the system. In those situations, the assumption of infinite buffer capacity is quite often a reasonable idealizing approximation, which facilitates the analysis considerably.

In addition to the description of the physical layout of the system, a model description essentially includes two main facets. First, the specification of the input to the system, i.e., the rules governing the duration of the interarrival, service, and switch-over times. Second, the description of how the input is handled by the system, i.e., the rules controlling the server action.

We first consider the arrival, service, and switch-over processes. We focus on continuous-time models, i.e., the interarrival, service, and switch-over times are assumed to be continuous-valued stochastic variables. Although occasionally some subtleties may be involved, most of the results for continuous-time models carry over to discrete-time models.

The arrival process

Type-*i* customers arrive at generally distributed interarrival times \mathbf{A}_i , having distribution $A_i(\cdot)$ with Laplace-Stieltjes Transform (LST) $\alpha_i(\cdot)$, first moment α_i , and second moment $\alpha_i^{(2)}$, i = 1, ..., n. Denote by $\lambda_i := 1/\alpha_i$ the arrival rate at Q_i , i = 1, ..., n. The total arrival rate is $\lambda := \sum_{i=1}^n \lambda_i$.

As usual in the polling literature, in the sequel customers are always assumed to arrive according to Poisson processes, unless specified otherwise. In the absence of detailed information on the characteristics of the arrival process, the assumption of Poisson arrival processes is quite often a reasonable idealizing approximation, which facilitates the analysis considerably.

We focus here on models with single arrivals, i.e., customers are assumed to arrive one by one. Most of the results may however be generalized to models with batch arrivals.

The service process

Type-*i* customers require generally distributed service times \mathbf{B}_i , having distribution $B_i(\cdot)$ with LST $\beta_i(\cdot)$, first moment β_i , and second moment $\beta_i^{(2)}$, $i = 1, \ldots, n$. Define $\rho_i := \lambda_i \beta_i$ as the traffic intensity at Q_i , $i = 1, \ldots, n$. The total traffic intensity is $\rho := \sum_{i=1}^n \rho_i$.

The switch-over process

Moving from Q_i to Q_j , the server incurs a generally distributed switch-over time \mathbf{S}_{ij} , having distribution $S_{ij}(\cdot)$ with LST $\sigma_{ij}(\cdot)$, first moment s_{ij} , and second moment $s_{ij}^{(2)}$, $i, j = 1, \ldots, n$. As usual in the polling literature, in the sequel, switch-over times are always assumed to depend only on the previous queue visited or the next queue to be visited, i.e., $\mathbf{S}_{ij} = \mathbf{S}_i$ or $\mathbf{S}_{ij} = \mathbf{S}_j$, $i, j = 1, \ldots, n$, respectively. Thus, the distribution of the total switch-over time incurred during a tour along the queues has LST $\sigma(\cdot) := \prod_{i=1}^{n} \sigma_i(\cdot)$, first moment $s := \sum_{i=1}^{n} s_i$, and second moment $s^{(2)} := \sum_{i=1}^{n} \sum_{j=1}^{n} s_i s_j + \sum_{i=1}^{n} (s_i^{(2)} - s_i^2)$. To avoid ambiguity, in the sequel, \mathbf{S}_i always corresponds to the switch-over time incurred when swapping out of Q_i , unless specified otherwise.

Remark 1.3.1

The successive interarrival, service, and switch-over times are implicitly assumed to be independent. In addition, the arrival, service, and switch-over processes are assumed to be mutually independent. In some of the polling applications, however, neither of these assumptions is very realistic. Not only

1.3 Model description

bursty traffic due to packetizing of messages, or due to alternating on/off phases of sources, but also collection or reservation mechanisms for transmission of messages may cause dependence in the arrival and service processes, e.g., dependence between consecutive interarrival times, between consecutive service times, or between the interarrival and service time of a customer. As Combé [76] demonstrates, quite often such dependence structures may be adequately modeled by a batch Markovian arrival process (BMAP), which is a direct generalization of the batch Poisson arrival process. In the BMAP, the arrival process is governed by an underlying Markov chain, which in the case of an ordinary Poisson process has only a single state, cf. Lucantoni [147]. The BMAP/G/1 queue may be numerically analyzed by the matrix-geometric method, cf. Lucantoni [146], Neuts [154]. To the best of the author's knowledge, the $BMAP_i/G_i/1$ polling model has not yet been studied.

In the polling literature, it is almost exclusively assumed that the arrival, service, and switch-over processes are also independent of the state of the system. As a rare exception, Boxma & Kelbert [46] consider a polling system in which customers arrive at Q_i according to a Poisson process of rate λ_{ij} when the server is at Q_j . Bozer & Srinivasan [55] analyze a model in which the switch-over time depends on the state of the *previous* queue visited; Ferguson [92], [91] studies a model in which the switch-over time may depend on the state of the *next* queue to be visited. Models with state-dependent service times are also conceivable.

Remark 1.3.2

With regard to the customer behavior, it is almost exclusively assumed in the polling literature, that customers from some external infinite source arrive at some queue, wait for some time, receive some amount of service, and then leave the system. As a rare exception, Sidi & Levy [166] and Sidi, Levy, & Fuhrmann [167] study an open polling network in which customers, after receiving service at Q_i , either move to Q_j with probability r_{ij} , or leave the system with probability $1 - \sum_{j=1}^{n} r_{ij}$. In a manufacturing setting, customer routing may arise when parts successively undergo service in a number of stages, e.g., drilling holes of different type, painting in different colors. In the context of communication networks, customer routing may occur when a station that receives a faulty message, sends a negative acknowledgement to the station that transmitted the message to indicate that the message has to be retransmitted. Altman & Yechiali [9] analyze a *closed* polling network in which customers (belonging to a permanent population), after receiving service at Q_i , move to Q_j with probability r_{ij} (with $\sum_{j=1}^{n} r_{ij} = 1$). As a generalization of customer routing, Levy & Sidi [141] describe a model with customer branching, in which departures may trigger concurrent arrivals to the system.

7

Remark 1.3.3

As usual in the polling literature, we focus here on a model with a *discrete* structure, i.e., a finite number of distinct queues. Letting the number of queues tend to infinity, with the traffic intensity ρ fixed, we obtain a *continuous* polling model. Several models have been analyzed in which the server travels around a circle, on which customers arrive according to a uniform Poisson process, cf. Bisdikian & Merakos [20], Coffman & Gilbert [69], Fuhrmann & Cooper [102], Kroese & Schmidt [132]. Recently, various models have been considered in which the server traverses a graph or a region of higher dimension, or in which customers do not necessarily arrive according to uniform Poisson processes, cf. Altman & Foss [6], Bertsimas & Ryzin [17], Coffman & Stolyar [72], Kroese & Schmidt [133].

Remark 1.3.4

As usual in the polling literature, for now, we focus on a model with a *single* server. In the last chapters of the monograph, we consider polling models with *multiple* servers.

We now consider the rules controlling the server action. A scheduling strategy is a collection of decision instructions for determining the server action at any given time. Occasionally, a scheduling strategy will also be referred to as a scheduling discipline, a polling strategy, or a polling policy. A scheduling strategy prescribes whether the server S should serve (which customer), switch (to which queue), or idle. Those decisions are made based on some partial knowledge of the state of the system (queue lengths, past arrival patterns), and on past decisions.

Although a scheduling strategy in principle may be arbitrarily involved, it mostly decomposes into three separate control mechanisms, viz.:

i. the routing policy: in which order should S serve the queues;

ii. the service policy: while at a queue, which number of customers should S serve;

iii. the service order: while at a queue, in which order should S serve customers.

We now successively describe the main variants of these three control mechanisms.

The routing policy

The routing policy prescribes in which order S should visit the queues. In the traditional cyclic polling model, the server visits the queues in a strictly cyclic order, i.e., $Q_1, \ldots, Q_n, Q_1, \ldots, Q_n, \ldots$

One obvious generalization of strictly cyclic polling is *periodic* polling, introduced in Kruskal [134] and revisited in Eisenberg [84], Baker & Rubin [15], and Boxma, Groenendijk, & Weststrate [45]. In periodic polling, the server

1.3 Model description

visits the queues in a fixed order, listed in a *polling table* of some size m, i.e., a vector of length m with components in $\{1, \ldots, n\}$. An important special case of periodic polling is *scan* polling or *elevator* polling, in which the server visits the queues in the order $Q_1, \ldots, Q_n, Q_n, \ldots, Q_1, \ldots$. Another special case of periodic polling is *star* polling, in which the server visits the queues in the order $Q_1, \ldots, Q_{n-1}, Q_1, \ldots$.

Another natural generalization of strictly cyclic polling is *Markovian* polling, introduced in Boxma & Weststrate [53]. In Markovian polling, the server visits the queues according to a discrete-time Markov chain with state space $\{1, \ldots, n\}$, i.e., the server is routed from Q_i to Q_j with probability p_{ij} , $i, j = 1, \ldots, n$. A special case of Markovian polling is *random* polling, i.e., $p_{ij} = p_j$, $i, j = 1, \ldots, n$, analyzed in Kleinrock & Levy [128]. Mixtures of periodic polling and Markovian polling are also conceivable.

All above policies are static, in the sense that the routing decisions are made independently of the state of the system, so that the sequence of the queues visited is also independent of the input sequence to the system. In dynamic policies, the routing decisions are made based on some partial knowledge of the state of the system (queue lengths, past arrival patterns), and on past decisions, e.g., the server may be instructed to serve the longest queue. Evidently, in principle the performance of the system may improve substantially by using such information in making the routing decisions. However, gathering such information and implementing a sophisticated routing policy may involve a considerable communication overhead, and complicate the operation of the system significantly. Therefore, in practice, dynamic policies are not necessarily preferable to static policies.

The service policy

While at a queue, the service policy (or strategy, or discipline) prescribes which number of customers S should serve. There are four classical service disciplines. I. Exhaustive service.

Under exhaustive service, the server continues to work until the queue becomes empty. Customers that arrive during the course of the visit, are served in the current visit.

II. Gated service.

Under gated service, S serves only the customers that were present at the start of the visit. Customers that arrive during the course of the visit, are served in the next visit.

III. Limited service.

Under k-limited service, the server continues to work until either a prespecified number of k customers have been served, or the queue becomes empty, whichever occurs first. There are two versions of limited service: gated-limited or exhaustive-limited service, depending on whether or not S only serves the customers that were present at the start of the visit.

IV. Decrementing service.

Under k-decrementing service, the server continues to work until either there

are a prespecified number of k customers less present than at the start of the visit, or the queue becomes empty, whichever occurs first. 1-Decrementing service is commonly referred to as semi-exhaustive service.

There are also numerous probabilistic hybrids of the four classical service disciplines. To give a systematic overview, we now define a family of service disciplines which operate as follows. If there are m_i customers present at the start of the visit to Q_i , then a (random) number $\mathbf{L}_i(m_i)$ of them qualify for service. Customers arriving during the visit to Q_i qualify for service with probability p_i . The server continues to work until either a (random) number of $\mathbf{K}_i(m_i)$ customers have been served, or there are no customers left that qualify for service, whichever occurs first.

Service disciplines with $p_i = 1$ and $p_i = 0$ are frequently referred to as exhaustive-type and gated-type policies, respectively, cf. Boxma [39], Levy & Sidi [141], Levy, Sidi, & Boxma [142]. Disciplines with $\mathbf{K}_i(m_i) < \infty$ (with positive probability) are frequently referred to as limited-type policies. Similarly, disciplines with $\mathbf{L}_i(m_i) < \infty$ (with positive probability) may be viewed as decrementing-type policies.

In binomial-type policies, $\mathbf{K}_i(m_i) = \infty$ and $\mathbf{L}_i(m_i)$ is binomially distributed with mean $m_i q_i$, $0 \le q_i \le 1$, cf. Levy [138], [139]. In Bernoulli-type policies, $\mathbf{L}_i(m_i) = m_i$ and $\mathbf{K}_i(m_i)$ is the sum of m_i independent identically geometrically distributed random variables each with mean $1/(1-q_i)$, $0 \le q_i \le 1$, cf. Resing [159]. In case $\mathbf{K}_i(m_i)$ is just a single geometrically distributed random variable, we obtain ordinary Bernoulli service, introduced in Keilson & Servi [124], [164]. Ordinary Bernoulli service may be used as an emulation of k_i -limited service, under which S serves at most k_i customers at Q_i (taking $q_i = 1 - 1/k_i$). Note that Bernoulli service and k_i -limited service coincide for $q_i = 0$ as well as $q_i = 1$. In its turn, k_i -limited service is widely used as an approximation of time-limited service, under which the server stays at most for a time T_i at Q_i (taking $k_i \approx T_i/\beta_i$, the exact value depending on whether or not service is preempted when the timer expires). For deterministic service times, k_i -limited service and T_i -limited service even coincide. A similar discipline is fixed time service, under which the server stays at a queue for a fixed time, regardless of whether the queue becomes empty in the meanwhile or not. A service discipline under which S always serves a *fixed number* of customers at a queue does not really make sense, since (for static routing policies) it inherently causes the system to be unstable.

All above policies are local, in the sense that the service decisions are made at each of the queues in isolation after the start of the visit. Recently, Boxma, Levy, & Yechiali [50] proposed the globally gated service discipline as a modeling approach to reservation mechanisms like in the cyclic-reservation multipleaccess (CRMA) protocol. Under globally gated service, during a visit to Q_i , S serves only the customers present at the start of the most recent visit to Q_1 . As a generalization of ordinary gated and globally gated service, Khamisy, Altman, & Sidi [126] analyzed the synchronized gated service discipline. Under synchronized gated service, during a visit to Q_i , S serves only the customers

1.3 Model description

present at the start of the most recent visit to a 'master' queue $Q_{\pi(i)}$ with $\pi(i) \in \{1, \ldots, n\}$. Synchronized versions of other service disciplines than gated, or service disciplines with other gating epochs than at the start of the most recent visit to $Q_{\pi(i)}$, e.g. at the *completion* of the most recent visit to $Q_{\pi(i)}$, are also conceivable, cf. Bertsekas & Gallager [16], and Lee & Sengupta [136], [137]. Although global in nature, even in the latter policies the service decisions depend on the state of the system through the marginal queue length only. Policies in which the service decisions may be based on the joint queue length, have been considered in Hofri & Ross [119], Koole [131], and Liu, Nain, & Towsley [145]. With regard to the pro's and con's of such sophisticated adaptive service policies similar remarks hold as with regard to dynamic versus static routing policies.

The order of service

While at a queue, the order of service prescribes in which order S should serve customers. While the routing policy and the service policy together dictate the global priorities, the service order determines the local priorities. In the sequel, the order of service is always assumed to be First Come First Served (FCFS), i.e., customers are assumed to be served in order of arrival. In fact, the service order does neither matter for the queue length distribution nor, by Little's law, for the mean waiting times, as long as customers enter service in an order independent of their service times. Of course, the service order *does* matter for the waiting-time *distribution*. Polling models with local priority rules within queues have been considered in Fournier & Rosberg [94], and Shimogawa & Takahashi [165].

The stability condition

Finally, we briefly discuss the conditions for stability. Recently Fricker & Jaïbi [97] rigorously proved that for a system with periodic polling a necessary and sufficient condition for stability reads

$$\rho + \max_{i=1,\dots,n} \lambda_i R / M_i < 1, \tag{1.1}$$

where R is the mean total switch-over time incurred during a cycle, i.e., incurred when passing through the polling table once, while M_i is the maximum mean number of type-*i* customers served during a cycle, i.e., the mean number of type-*i* customers that would be served during a cycle, if there were an infinite number of type-*i* customers present at the start of the cycle. Here the system is said to be stable, if it admits a stationary regime with integrable cycle time. A simple traffic balance argument shows that if the system is stable, then the server is working a fraction ρ of the time, so that the mean cycle time is given by $EC = R/(1-\rho)$. So if the system is stable (1.1) may be rewritten as

$$\frac{\lambda_i R}{1-\rho} < M_i, \qquad i = 1, \dots, n, \tag{1.2}$$

saying that the mean number of type-*i* customers arriving during a cycle is smaller than the maximum mean number of type-*i* customers served during a cycle. As the server is working a fraction ρ_i of the time at Q_i , the mean total visit time at Q_i in a cycle is given by $\mathbf{EV}_i = \rho_i \mathbf{EC} = \rho_i R/(1-\rho)$, $i = 1, \ldots, n$. The mean total intervisit time at Q_i in a cycle follows from $\mathbf{EI}_i = \mathbf{EC} - \mathbf{EV}_i = (1-\rho_i)R/(1-\rho), i = 1, \ldots, n$.

Note that if the system is stable, then all the queues are stable. However, even if a system is unstable a subset of the queues may still be stable. Assuming that the queues are indexed such that $i \leq j \iff \lambda_i/M_i \leq \lambda_j/M_j$, Fricker & Jaïbi show that the queues Q_1, \ldots, Q_κ are stable while the queues $Q_{\kappa+1}, \ldots, Q_n$ are unstable with κ being defined as

$$\kappa := \max\{i : \sum_{j=1}^{i} \rho_j + \lambda_i (R + \sum_{j=i+1}^{n} \beta_j M_j) / M_i < 1\}.$$

In other words, whether or not the individual queues are stable depends on the ratio λ_i/M_i . The mean cycle time is given by $\text{EC} = (R + \sum_{j=\kappa+1}^n \beta_j M_j)/(1 - \sum_{j=$

$$\sum_{j=1}^{n} \rho_j$$
). Note that (1.1) implies $\kappa = n$.

The quantity M_i is determined by the number of visits to Q_i as specified in the polling table and the maximum mean number of customers served during a visit to Q_i as specified by the service discipline (possibly different policies at different visits). For ease of presentation, we now focus on the case of strictly cyclic polling, so that R = s, and the quantity M_i is determined by the service discipline at Q_i only. (The sufficient stability conditions for the case of strictly cyclic polling were independently established by Altman, Konstantopoulos, & Liu [8] and Georgiadis & Szpankowski [108], using different techniques. The assumptions in [8] and [108] on the service disciplines are however somewhat restrictive compared to [97].) For service disciplines like exhaustive and gated that do not impose any (probabilistic) restriction on the maximum mean number of customers served, $M_i = \infty$, so that the stability condition (1.1) reduces to $\rho < 1$, which has long been stated without formal proof, cf. Eisenberg [84]. For both the exhaustive and gated version of k_i -limited service, $M_i = k_i$, so that (1.2) reduces to $\lambda_i s/(1-\rho) < k_i$, which also has long been stated without formal proof, cf. Kühn [135]. For k_i -decrementing service, $M_i = k_i/(1-\rho_i)$, so that (1.2) reduces to $\lambda_i s(1-\rho_i)/(1-\rho) < k_i$, i = 1, ..., n, saying that the mean increase in the number of type-*i* customers during the intervisit time is smaller than the net decrease during the visit time.

In [98] Fricker & Jaïbi establish the stability condition for models with Markovian polling; cf. also Borovkov & Schassberger [25]. For dynamic scheduling strategies, there are hardly any results known on the conditions for stability; cf. Schassberger [162] for the case of gated-limited service.

Throughout the monograph, the conditions for stability are assumed to hold. We further always assume the system under consideration to be in steady state.

1.4 ANALYSIS OF POLLING SYSTEMS

In this section, we survey the state of the art in the analysis of polling systems. Rather than covering all technical details, we intend to illuminate the main concepts in the analysis of polling systems, which contribute to putting the monograph in the right perspective. We refer to Takagi [172] for a thorough monograph on the analysis of polling systems, containing a detailed enumeration of the main results.

In one of the first polling studies, Avi-Itzhak, Maxwell, & Miller [14] study a two-queue model with zero switch-over times and alternating priority (i.e. exhaustive service at both queues). They obtain the sojourn time distribution by focusing on the system busy period. Takács [171] derives the waiting-time distribution in the same model by studying the Markov chain formed by the state of the system embedded at service completion epochs. Using similar techniques, Eisenberg [83] obtains the waiting-time distribution in a two-queue model with non-zero switch-over times, and either alternating priority or strict priority, in which the server stops switching when the system is empty.

Cooper & Murray [77] study a model with an arbitrary number of queues, zero switch-over times, strictly cyclic polling, and exhaustive service at each of the queues. They derive the cycle time distribution by analyzing the Markov chain formed by the state of the system embedded at visit completion epochs. Cooper [78] obtains the waiting-time distribution for the model, by viewing the queues in isolation as vacation queues, the intervisit periods constituting the vacations. The solution method may also be used for a similar model with gated service at each of the queues.

Eisenberg [84] studies a model with an arbitrary number of queues, non-zero switch-over times, periodic polling, and exhaustive service at each of the queues, in which the server keeps switching when the system is empty. Eisenberg derives the waiting-time distribution, the marginal queue length distribution, and the joint queue length distribution at polling epochs by cleverly exploiting four Markov chains, embedded at service and visit beginnings and endings. The solution method may also be used for a similar model with gated service at each of the queues. In a recent study [86], Eisenberg shows how an adapted version of the method may be applied in case the server stops switching when the system is empty.

Over the years, several methods have been developed for computing the mean waiting times at the various queues in strictly cyclic polling systems with either exhaustive or gated service. To be specific, denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer, i.e., the time elapsing from its arrival to the start of its service, $i = 1, \ldots, n$.

One method for computing the mean waiting times is the *buffer occupancy* method, as used by Cooper & Murray [77], Cooper [78], and Eisenberg [84]. As the name suggests, in the buffer occupancy method, the mean waiting times

are computed starting from the *buffer occupancy* variables \mathbf{X}_{ij} , denoting the queue length at Q_j at the start of a visit to Q_i , i, j = 1, ..., n.

Define $F_i(z_1, \ldots, z_n) := \mathbb{E}(z_1^{\mathbf{X}_{i1}} \ldots z_n^{\mathbf{X}_{in}})$ to be the probability generating function (pgf) of the joint queue length distribution at the start of a visit to Q_i , $|z_j| \leq 1, j = 1, \ldots, n$. The buffer occupancy method starts with deriving n functional equations, expressing $F_{i+1}(\cdot)$ into $F_i(\cdot), i = 1, \ldots, n$.

The waiting times are related to the buffer occupancy variables as follows, cf. Watson [187]. For exhaustive service, writing \mathbf{X}_i for \mathbf{X}_{ii} , for Re $\omega \geq 0$,

$$E(e^{-\omega \mathbf{W}_i}) = \frac{(1 - \lambda_i \beta_i)\omega}{\omega - \lambda_i (1 - \beta_i(\omega))} \frac{1 - E((1 - \omega/\lambda_i)^{\mathbf{X}_i})}{\omega E \mathbf{X}_i/\lambda_i},$$
(1.3)

the first term on the right-hand side representing the waiting-time Laplace-Stieltjes Transform (LST) in the corresponding isolated M/G/1 queue of Q_i with arrival rate λ_i and service time LST $\beta_i(\cdot)$. For gated service, for Re $\omega \geq 0$,

$$E(e^{-\omega \mathbf{W}_{i}}) = \frac{(1 - \lambda_{i}\beta_{i})\omega}{\omega - \lambda_{i}(1 - \beta_{i}(\omega))} \frac{E((\beta_{i}(\omega)^{\mathbf{X}_{i}}) - E((1 - \omega/\lambda_{i})^{\mathbf{X}_{i}}))}{(1 - \lambda_{i}\beta_{i})\omega E\mathbf{X}_{i}/\lambda_{i}}, \quad (1.4)$$

the first term on the right-hand side standing again for the waiting-time LST in the corresponding isolated M/G/1 queue of Q_i . In Chapter 2, we will discuss the occurrence of that term in greater detail.

The above relationships yield expressions for \mathbf{EW}_i involving $f'_i := \mathbf{EX}_i$ and $f''_i := \mathbf{E}(\mathbf{X}_i(\mathbf{X}_i - 1)) = \mathbf{E}(\mathbf{X}_i^2) - \mathbf{EX}_i$ as unknowns. There are explicit expressions available for the first moments \mathbf{EX}_i ; for exhaustive service, $\mathbf{EX}_i = \lambda_i \mathbf{EI}_i = \lambda_i (1 - \rho_i) s / (1 - \rho)$; for gated service, $\mathbf{EX}_i = \lambda_i \mathbf{EC} = \lambda_i s / (1 - \rho)$. For the second moments $\mathbf{E}(\mathbf{X}_i^2)$ there are no explicit expressions available. However, the functional equations involving $F_i(\cdot)$, $i = 1, \ldots, n$, render a set of n^3 linear equations with n^3 unknowns $\mathbf{E}(\mathbf{X}_{ij}\mathbf{X}_{ik})$. The latter set of equations can best be solved in an iterative manner, which requires $O(n^3\log_{\rho}\epsilon)$ elementary operations (additions, multiplications), with ϵ denoting the level of accuracy required.

Another method for computing the mean waiting times is the station time method, as used in Ferguson & Aminetzah [93]. As the name reflects, in the station time method the mean waiting times are computed starting from the station time variables U_j , denoting the length of the station time at Q_j , $j = 1, \ldots, n$. For exhaustive service, the station time consists of the visit time plus the *preceding* switch-over time. For gated service, the station time is composed of the visit time plus the *following* switch-over time.

Define $\Theta_i(\omega_1, \ldots, \omega_n) := \mathrm{E}(\mathrm{e}^{-\omega_1 \mathbf{U}_{i-1}-\ldots-\omega_n \mathbf{U}_{i-n}})$ (all indices mod n) to be the Laplace-Stieltjes Transform (LST) of the joint station time distribution of the last n visits at the start of a visit to Q_i , $\mathrm{Re}\,\omega_j \geq 0, j = 1, \ldots, n$. The station time method starts with establishing n functional equations expressing $\Theta_{i+1}(\cdot)$ into $\Theta_i(\cdot), i = 1, \ldots, n$.

1.4 Analysis of polling systems

The waiting times are related to the station time variables as follows. For exhaustive service, for $\operatorname{Re} \omega \geq 0$,

$$E(e^{-\omega \mathbf{W}_i}) = \frac{(1 - \lambda_i \beta_i)\omega}{\omega - \lambda_i (1 - \beta_i(\omega))} \frac{1 - E(e^{-\omega \mathbf{I}_i})}{\omega E \mathbf{I}_i},$$
(1.5)

with \mathbf{I}_i denoting the intervisit time at Q_i , i.e., $\mathbf{I}_i = \mathbf{S}_{i-1} + \mathbf{U}_{i-1} + \ldots + \mathbf{U}_{i-n+1}$. Note that (1.3) and (1.5) are equivalent by the fact that for exhaustive service \mathbf{X}_i equals the number of arrivals at Q_i during \mathbf{I}_i , i.e., $\mathbf{E}(z^{\mathbf{X}_i}) = \mathbf{E}(e^{-\lambda_i(1-z)\mathbf{I}_i})$. For gated service, for $\operatorname{Re} \omega \geq 0$,

$$E(e^{-\omega \mathbf{W}_{i}}) = \frac{(1-\lambda_{i}\beta_{i})\omega}{\omega - \lambda_{i}(1-\beta_{i}(\omega))} \frac{E(e^{-\lambda_{i}(1-\beta_{i}(\omega))\mathbf{C}_{i}}) - E(e^{-\omega \mathbf{C}_{i}})}{(1-\lambda_{i}\beta_{i})\omega E\mathbf{C}_{i}}, \qquad (1.6)$$

with C_i denoting the cycle time at Q_i , i.e., $C_i = U_{i-1} + \ldots + U_{i-n}$. The equivalence of (1.4) and (1.6) follows from the fact that for gated service X_i equals the number of arrivals at Q_i during \mathbf{C}_i , i.e., $\mathbf{E}(z^{\mathbf{X}_i}) = \mathbf{E}(e^{-\lambda_i(1-z)\mathbf{C}_i})$. The above relationships yield expressions for EW_i involving $g_i = EU_i$ and $g_{ij} = E(\mathbf{U}_i \mathbf{U}_j)$ (Q_i being visited before Q_j) as unknowns. There are explicit expressions available for the means g_i ; for exhaustive service, $g_i = s_{i-1} + EV_i =$ $s_{i-1} + \rho_i s/(1-\rho)$; for gated service, $g_i = EV_i + s_i = \rho_i s/(1-\rho) + s_i$. For the covariances g_{ij} there are no explicit expressions available. However, the functional equations involving $\Theta_i(\cdot)$, i = 1, ..., n, induce a set of n^2 linear equations with n^2 unknowns g_{ij} . The latter set of equations can be solved in an iterative manner, which requires $O(n^2 \log_{\rho} \epsilon)$ elementary operations, with ϵ denoting the level of accuracy demanded. A further advantage of the station time method in comparison with the buffer occupancy method is that the structure of the set of linear equations involved is somewhat simpler. Without explicitly solving it, Ferguson & Aminetzah observe from the structure of the set of linear equations that the intensity-weighted sum $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$ yields a relatively simple expression in comparison with the extremely complicated expressions for the individual mean waiting times themselves, cf. also Watson [187]. For exhaustive service,

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \rho \frac{s^{(2)}}{2s} + \frac{s}{2(1-\rho)} \left[\rho^2 - \sum_{i=1}^{n} \rho_i^2 \right].$$
(1.7)

For gated service,

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \rho \frac{s^{(2)}}{2s} + \frac{s}{2(1-\rho)} \left[\rho^2 + \sum_{i=1}^{n} \rho_i^2 \right].$$
(1.8)

These relationships for the mean waiting times are commonly referred to as *pseudo-conservation laws*. In Chapter 2, we will discuss the existence of these pseudo-conservation laws in greater detail. A disadvantage of the station time

method is that, unlike the buffer occupancy method, it only appears to be applicable to polling systems with either exhaustive or gated service.

Sarkar & Zangwill [160] describe a refinement of the station time method. They express the n^2 unknowns g_{ij} into the n unknowns g_{ii} , and derive a set of n linear equations for the latter coefficients, which however appears to be less sparse. Konheim & Levy [129] describe a modification of the buffer occupancy method. They propose to calculate $E(\mathbf{X}_i^2)$ by the so-called descendant set approach, which allows the computation of the mean waiting time at a single queue in only $O(n\log_{\alpha}\epsilon)$ elementary operations, with ϵ the level of accuracy desired.

Concluding, although efficient numerical evaluation of the mean waiting times is non-trivial, polling systems with exhaustive or gated service do allow an exact analysis for generally distributed service times, generally distributed switchover times, and an arbitrary number of queues. Polling systems with limited or decrementing service, however, do not allow an exact analysis, apart from some special cases like two-queue cases and completely symmetric cases. Eisenberg [85] studies a two-queue model with zero switch-over times and alternating service (i.e. 1-limited service at both queues), transforming the problem of finding the joint queue length distribution into the problem of solving a singular Fredholm integral equation. Cohen & Boxma [74] analyze the same model, translating the problem into a Riemann-Hilbert boundary value problem. Using similar techniques, Boxma [37] studies a symmetric two-queue model with non-zero switch-over times and 1-limited service at both queues. Boxma & Groenendijk [43] analyze an asymmetric two-queue model with non-zero switch-over times and 1-limited service at both queues by formulating a Riemann boundary value problem. Cohen [75] considers a two-queue model with zero switch-over times and 1-decrementing (semi-exhaustive) service at both queues. The solution of the specific boundary value problem as formulated in each of the latter studies typically requires an arsenal of most advanced techniques from complex function theory, usually rendering contour-integral expressions for the mean waiting times. For polling systems with k-limited or k-decrementing service and n > 2queues, only approximative results are available, apart from some mean-value results for global performance measures, like the cycle times, or for the waiting times in a completely symmetric system (cf. Fuhrmann [100] for the special case of 1-limited service).

Summarizing, we observe a striking difference in complexity between on the one hand service disciplines, like exhaustive and gated, that can be analyzed exactly in a general setting by standard methods, and on the other hand service disciplines, like limited and decrementing, that can only be analyzed exactly in special cases by most ingenious techniques.

The existence of such a sharp distinction is illuminated in Resing [159] and independently explained in Fuhrmann [99]. Both Resing and Fuhrmann consider the class of service disciplines that satisfy the following property:

1.4 Analysis of polling systems

Property 1.4.1

If there are k_i customers present at Q_i at the start of a visit, then during the course of the visit, each of these k_i customers will effectively be replaced in an i.i.d. manner by a random population having probability generating function (pgf) $h_i(z_1, \ldots, z_n)$, which may be any *n*-dimensional pgf.

By using a multi-type branching process approach, both Resing and Fuhrmann show that the class of service disciplines that satisfy the above property allows an exact analysis. The results of Resing and Fuhrmann suggest that service disciplines that violate Property 1.4.1 defy an exact analysis, except for some special cases, like two-queue cases and completely symmetric cases.

The key element in their exposition is that if the service disciplines in a polling system satisfy Property 1.4.1, it is possible to relate the pgf $G_i(z_1, \ldots, z_n) := E(z_1^{\mathbf{Y}_{i_1}} \ldots z_n^{\mathbf{Y}_{i_n}})$ of the joint queue length distribution at the *end* of a visit to Q_i to the pgf $F_i(z_1, \ldots, z_n) := E(z_1^{\mathbf{X}_{i_1}} \ldots z_n^{\mathbf{X}_{i_n}})$ of the joint queue length distribution at the *beginning* of a visit to Q_i by

$$G_i(z_1, \dots, z_n) = F_i(z_1, \dots, z_{i-1}, h_i(z_1, \dots, z_n), z_{i+1}, \dots, z_n).$$
(1.9)

Moreover, it is possible to relate $F_{i+1}(z_1, \ldots, z_n)$ to $G_i(z_1, \ldots, z_n)$ by

$$F_{i+1}(z_1, \dots, z_n) = G_i(z_1, \dots, z_n) \sigma_i(\sum_{j=1}^n \lambda_j (1-z_j)),$$
(1.10)

irrespective of the service disciplines (ignoring here some subtleties in case the total switch-over time in a cycle is zero, cf. Chapter 3). Thus, we obtain 2n equations for 2n functions, which may be combined to obtain a functional equation for one of the functions $F_i(\cdot)$ or $G_i(\cdot)$, which may then be solved by a standard iterative procedure. As we will show in Section 2.1, most of the relevant performance measures like marginal queue lengths at an arbitrary epoch and waiting times may directly be derived from $F_i(\cdot)$ and $G_i(\cdot)$. Note that the approach of Resing and Fuhrmann is closely related to the buffer occupancy method, which was outlined earlier in the present section. In fact, the station time method, which was also sketched there, only appears to be applicable to a very restricted subclass of the service disciplines satisfying Property 1.4.1. Assuming the service disciplines to satisfy Property 1.4.1, Resing shows that

In case the total switch-over time in a cycle is non-zero, the joint queue length process at the polling epochs of a fixed but arbitrary queue constitutes a multitype branching process with immigration in each state. The particle types in the branching process correspond to the customer types in the polling model, the offspring in the branching process represents the customers arriving during the service times in the polling model, and the immigration in the branching process corresponds to the customers arriving during the switch-over times in the polling model. In case the total switch-over time in a cycle is zero, Resing shows that the joint queue length process at the polling epochs of a fixed queue constitutes a multi-type branching process with immigration in state zero only. The immigration in the branching process then corresponds to the customers arriving to an empty system in the polling model. So the models with zero and non-zero switch-over times are closely related through a common offspring generation, the only difference originating from the immigration. In Chapter 3, we expose the relationship between models with zero and non-zero switch-over times in greater technical detail.

The exhaustive service discipline satisfies Property 1.4.1 with $h_i(z_1, \ldots, z_n) = \eta_i(\sum_{j \neq i} \lambda_j(1-z_j))$. Here $\eta_i(\cdot)$ is the LST of the busy-period distribution in an

ordinary isolated M/G/1 queue with arrival rate λ_i and service time distribution $B_i(\cdot)$, satisfying the functional equation $\eta_i(\omega) = \beta_i(\omega + \lambda_i(1 - \eta_i(\omega)))$, cf. [73] p. 250. The gated service discipline satisfies Property 1.4.1 with $h_i(z_1, \ldots,$

 $z_n) = \beta_i (\sum_{j=1}^n \lambda_j (1-z_j))$. Limited and decrementing service disciplines violate Property 1.4.1, and have indeed not yielded an exact analysis, except for some

Property 1.4.1, and have indeed not yielded an exact analysis, except for some special cases, like two-queue cases and completely symmetric cases.

Models with server-position dependent arrival rates and customer branching (as the word suggests), cf. Remark 1.3.1 and Remark 1.3.2, satisfy Property 1.4.1, and may thus be analyzed exactly by using a multi-type branching process approach. There are also some service disciplines, like synchronized gated, that strictly speaking do not satisfy Property 1.4.1, but still allow an exact analysis. Most of these service disciplines, however, satisfy the following generalization of Property 1.4.1:

Property 1.4.2

If there are k_i customers present at Q_i at the beginning (or the end) of a visit to $Q_{\pi(i)}$, with $\pi(i) \in \{1, \ldots, n\}$, then during the course of the visit to Q_i , each of these k_i customers will effectively be replaced in an i.i.d. manner by a random population having pgf $h_i(z_1, \ldots, z_n)$, which may be any *n*-dimensional pgf.

In Chapter 9, we will introduce another generalization of Property 1.4.1 to explore the class of service disciplines that allow an exact analysis in the case of multiple servers.

Here we outlined the analysis for a continuous-time model with strictly cyclic polling and single Poisson arrivals. Without seriously complicating the analysis, continuous-time may be replaced by discrete-time, strictly cyclic polling may be generalized to periodic polling or Markovian polling, and single Poisson arrivals may be generalized to batch Poisson arrivals.

1.5 **Optimization** of polling systems

In this section, we review the state of the art in the optimization of polling systems. As mentioned earlier, we seek to identify the main issues in the op-

1.5 Optimization of polling systems

timization of polling systems, rather than exhaustively discuss all the achievements made. For extensive surveys on optimization of polling systems, we refer to Boxma [39] (static optimization) and Yechiali [191] (semi-dynamic optimization).

In most of the polling applications, some degree of freedom exists in the design or control of the system in choosing the parameters of the scheduling discipline (visit order, visit lengths, service order). A major objective in studying polling systems is to develop sufficient understanding of how these parameters influence the operation of the system, and how these parameters should thus be chosen so as to improve the performance of the system. Nevertheless, compared to the well-trodden area of the analysis of polling systems, the field of optimization of polling systems still remains relatively unexplored. Although for example many waiting-time approximations have been proposed for limited-type service policies, the problem of determining appropriate values for the service limits has hardly been addressed.

In the optimization of polling systems, the problem formulation is typically to optimize some measure of the system performance over some class of feasible scheduling disciplines. So there are two factors that play a role, first, what is the performance measure to be optimized, second, what is the class of feasible scheduling disciplines.

Concerning the first factor, there is probably no generic comprehensive measure to evaluate the system performance. Efficiency and fairness are commonly viewed as important aspects of the system performance. Although it is somewhat unclear exactly how efficiency and fairness should be defined, it is widely believed that there is some trade-off between them. On the one hand, exhaustive service is considered to be efficient but not very fair, as a heavily-loaded queue may dominate the complete system. On the other hand, 1-limited service is considered to be fair, as each of the queues receives at most one service per visit, but not very efficient. Ideally, a performance measure should relate all important aspects of the system performance to measurable quantities like waiting times, queue lengths, and excess probabilities, and should indicate how all those aspects weigh against each other. In view of these considerations,

 $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i \text{ (by Little's law equivalent to } \sum_{i=1}^{n} c_i E \mathbf{L}_i \text{, with } \mathbf{L}_i \text{ denoting the number of waiting type-} i \text{ customers at an arbitrary epoch} \text{ is widely accepted as a}$

reasonable measure of the system performance.

Concerning the second factor, usually the class of feasible scheduling disciplines consists of a family of strategies of a similar structure that differ by some (vectorial) parameter. We now successively discuss some optimization studies that focus on optimization of a routing vector (routing probabilities, or polling table), a service vector (service probabilities, or service limits), and a routing vector and a service vector simultaneously. In addition, we mention some optimization studies that analyze less structured dynamic polling policies.

Optimization of the routing policy for a given service policy

A considerable amount of research effort has been devoted to static optimization, i.e., optimization of static routing policies, in which the routing decisions are made independently of the state of the system. Boxma, Levy, & Weststrate [47] (cf. also [188]) consider a system with random polling, and at each of the queues either exhaustive or gated service. They address the problem of finding the routing probabilities (p_1, \ldots, p_n) that minimize $\sum_{i=1}^n \rho_i \mathbf{EW}_i$, the latter quantity being explicitly known from the pseudo-conservation law for random polling, cf. [53]. Boxma, Levy, & Weststrate [39] (cf. also [188]) consider a system with periodic polling and at each of the queues either exhaustive, gated, or 1-limited service. They study the problem of determining a polling table that minimizes $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$. The proposed approach is to use the optimal visit ratios in random polling (i.e. the routing probabilities (p_1, \ldots, p_n)) as indication for the optimal visit ratios in periodic polling (i.e. the occurrence ratios of the queues in the polling table), and then to use the Golden Ratio procedure as a heuristic for spacing the visits within the polling table. Boxma, Levy, & Weststrate [49] (cf. also [188]) address the generalized problem of determining a polling table that minimizes $\sum_{i=1}^{n} c_i \lambda_i \mathbf{EW}_i$, the latter quantity now being approximated in terms of the occurrence ratios of the queues in the polling table. Kruskal [134] studies a similar problem with deterministic arrival, service, and switch-over processes. In all cases, the optimal visit ratios are given by surprisingly simple square-root formulae. Also, a considerable amount of research effort has been put in semi-dynamic optimization, i.e., optimization of semi-dynamic routing policies, in which periodically the visit order for some future period is determined, based on some partial knowledge of the state of the system. Browne & Yechiali [59] consider a system with either exhaustive or gated service at each of the queues. They address the problem of finding the visit order, at the start of each cycle, that

address the problem of finding the visit order, at the start of each cycle, that minimizes the expected duration of the new cycle, based on knowledge of the queue lengths. The optimal visit order is given by a remarkably simple indextype rule. However, they do not explicitly indicate how minimization of the expected duration of each new cycle is supposed to contribute to optimizing the system performance, in particular minimizing the mean waiting times. Actually, the mean cycle time remains $s/(1-\rho)$, cf. [8], in other words, the server is doing nothing but deferring the service of customers to future cycles. Results of Fabian & Levy [90] suggest that among all semi-dynamic visit orders, the order that minimizes (maximizes) the expected duration of each new cycle in fact yields the largest (smallest) mean waiting times. Purely dynamic optimization of the routing policy (for a given service policy) has hardly received any attention so far.

1.5 Optimization of polling systems

Optimization of the service policy for a given routing policy

Borst, Boxma, & Levy [35] consider a system with a k-limited service strategy at each of the queues. They address the problem of determining the vector of service limits (k_1, \ldots, k_n) that minimizes $\sum_{i=1}^n c_i \lambda_i E W_i$. Chapter 6 of the present monograph is based on the results obtained in [35]. Blanc & Van der Mei [23] study a similar optimization problem in a system with a Bernoulli service strategy at each of the queues. Purely dynamic optimization of the service policy (for a given routing policy) has hardly received any attention so far.

Simultaneous optimization of routing policy and service policy

Borst, Boxma, Harink, & Huitema [34] consider a system operated with a fixed time polling (ftp) scheme. An ftp scheme specifies which queue should be visited at what time, i.e., it specifies not only the *order* of the visits, but also the *starting times* of the visits. They address the problem of constructing an ftp scheme that minimizes $\sum_{i=1}^{n} c_i \lambda_i EW_i$. Chapter 7 of the present monograph is based on the results obtained in [34].

Liu, Nain, & Towsley [145] consider a system with a dynamic polling policy, i.e., a collection of instructions for making decisions on whether the server S should serve (which customer), switch (to which queue), or idle, based on some partial knowledge of the state of the system. They attempt to identify policies that stochastically minimize the total amount of work and the total number of customers present in the system at an arbitrary epoch. They show that optimal policies are exhaustive and greedy, i.e., the server should neither idle nor switch when a queue is non-empty. In addition, they show that in a symmetric system, optimal policies are patient, and belong to the class of Stochastically-Largest-Queue policies, i.e., the server should never switch to a queue known to be stochastically smaller than another queue.

For a model with zero switch-over times, the optimal (non-preemptive) polling policy is known to be given by the $c\mu$ -rule, cf. Meilijson & Yechiali [151], and Buyukkoc, Varaiya, & Walrand [63]. For a symmetric two-queue model with non-zero switch-over times, Hofri & Ross [119] show that the policy that minimizes the sum of discounted switch-over times and the holding cost, is exhaustive service in a nonempty system, and is of threshold type for switching from an empty queue to another. For an asymmetric two-queue model with switch-over *costs* rather than switch-over *times*, Koole [131] shows that the policy that minimizes the sum of discounted switching cost and holding cost, is *not* a threshold policy, but that the best threshold policy approaches the optimal policy very well.

Monotonicity issues

Closely related to optimization issues are questions of stochastic ordering or monotonicity of the various performance measures with regard to the inducing stochastic processes (arrival process, service process, switch-over process), or with regard to the scheduling discipline (routing policy, service policy, order of service).

For a fixed but arbitrary routing policy, Levy, Sidi, & Boxma [142] establish a hierarchy of dominance relations among work-conserving, non-idling service policies with respect to the total amount of work in the system at any time. Under fairly mild assumptions, they show that the closer a service policy approaches the standard exhaustive service policy, the higher the service policy reaches in the hierarchy, so that in particular the standard exhaustive service policy figures at the top of the hierarchy.

Altman, Konstantopoulos, & Liu [8] consider a system with strictly cyclic polling, and at each of the queues either exhaustive-type or gated-type service. They show that the queue lengths at polling epochs, the visit times, the intervisit times, and the cycle times are stochastically increasing in the arrival rates, the service times, and the switch-over times.

Note that the above ordering results refer to global performance measures or performance measures that are defined at polling epochs. For detailed performance measures like waiting times or queue lengths at arbitrary epochs there are hardly any ordering results known. In view of [8], it might be conjectured that also the waiting time and the queue length at Q_i are stochastically increasing with regard to the arrival rates, the service times, and the switch-over times as well as the 'limitedness' of the service at Q_i , but most of the statements of such tendency have either been disproved by simple counterexamples or have lacked proof so far. As one of the scarce results, we establish in Chapter 5 a stochastic ordering relation for the waiting times in a globally gated polling system.

1.6 Overview of the monograph

We now give an overview of the main results presented in the remainder of the monograph.

In Chapter 2, we elaborate on the use of decomposition properties to analyze polling models, in particular discussing the existence of so-called pseudoconservation laws for the mean waiting times, cf. (1.7), (1.8). Broadening the scope somewhat, we also demonstrate the use of such decomposition properties to study a related model, namely, a queueing system with a customer collection mechanism.

In Chapter 3, we consider two different single-server polling systems: (i) a model with *zero* switch-over times, and (ii) a model with *non-zero* switch-over times, in which the server keeps cycling when the system is empty. For both

1.6 Overview of the monograph

models we relate the steady-state queue length distribution at a queue to the queue length distribution at visit beginning and visit completion instants at that queue. As a by-product, we obtain a shorter proof of the Fuhrmann-Cooper decomposition, cf. [103]. For the important class of service disciplines with a branching structure satisfying Property 1.4.1, we expose a strong relationship between both the queue length and the waiting-time distribution in the two models. We also show how the latter relationship can be exploited to reduce the computational complexity of numerical moment calculations.

In Chapters 4 and 5, we study a polling system with a *dormant* server, i.e., a polling system in which the server may be allowed to make a halt at a queue when there are no customers present in the system. In the polling literature, the server is usually assumed never to idle, in other words, to be switching when not working. In particular, the server is assumed to be switching when there are no customers present in the system. However, quite often there are very sound reasons for letting the server stop switching when there are no customers present in the system. However, needlessly circle around. In Chapter 4, we therefore derive a pseudo-conservation law for a general model, permitting a variety of service disciplines, in which the server may be allowed to make a halt at an arbitrary subset of the queues. We use the pseudo-conservation law to compare the dormant and the non-dormant server case. Furthermore, we address the question at which queues the server should make a halt to minimize the mean total amount of work in the system.

The option of idling especially goes hand in hand quite naturally with the globally gated service discipline, introduced in Boxma, Levy, & Yechiali [50]. Under globally gated service, during a tour along the queues only those customers are served that were already present at the start of the tour, while the service of customers that meanwhile arrive to the system is deferred until the next tour, cf. Section 1.3. Thus, as suggested in Boxma, Weststrate, & Yechiali [54], it does not make sense to start a tour along the queues when there are no customers present in the system. In Chapter 5, we therefore focus on a globally gated polling system in which the server makes a halt at its home base when there are no customers present in the system. We derive an explicit expression for the LST of the cycle time distribution, for the LST of the waiting-time distribution at each of the queues, and for the pgf of the joint queue length distribution at polling epochs. As a justification of the dormant server policy, the waiting time at each of the queues is shown to be smaller (in the increasingconvex-ordering sense) than in the ordinary non-dormant server case.

In Chapters 6 and 7, we discuss several optimization issues in polling systems. In Chapter 6, we consider a polling system with a k-limited service strategy. Under k-limited service, when visiting a queue, the server works until either a prespecified number of customers have been served, or the queue becomes empty, whichever occurs first, cf. Section 1.3. We are interested in the problem of determining appropriate values for the service limits that contribute to an efficient operation of the system. It appears that if we do not impose any constraint on the k_i 's, then at least one of the optimal k_i 's is always infinite. To accomplish a bound on the cycle time, we therefore also study a version of the problem with a constraint of the form $\sum_{i=1}^{n} \gamma_i k_i \leq K$. We propose four different approaches to the constrained optimization problem, based on four different approximations for the mean waiting times, which are extensively investigated by numerical experiments. Next, we discuss some properties of polling systems with k-limited service, and establish a (partially conjectured) $c\mu$ -like rule for the unconstrained optimization problem. We then propose an approximative approach to this problem, which is also elaborately examined by numerical experiments.

In Chapter 7, we consider a polling system operated with a fixed time polling (ftp) scheme. An ftp scheme specifies which queue should be visited at what time, i.e., it specifies not only the *order* of the visits, but also the *starting times* of the visits. We are interested in the problem of constructing ftp schemes that contribute to an efficient operation of the system. Starting from rather simple approximations, we formulate the problem as a mathematical program. In view of its NP-hardness, we develop a heuristic method for solving the mathematical program. The method is tested by numerical experiments.

In Chapters 8, 9, and 10, we consider multiple-queue systems with multiple servers. In Chapter 8, we consider a system consisting of several customer types attended by several parallel non-identical servers. Customers are allocated to the servers in a probabilistic manner; upon arrival customers are sent to one of the servers according to a matrix of routing probabilities. We consider the problem of finding an allocation that minimizes a weighted sum of the mean waiting times. We expose the structure of an optimal allocation, and describe in detail for some special cases how the structure may be exploited in actually determining an optimal allocation. Furthermore, we consider the problem of finding an optimal allocation, i.e., an optimal allocation that involves a 0-1 matrix of routing probabilities. We show the problem to be NP-hard and indicate how the structure of an optimal non-deterministic allocation may be used as a heuristic guideline in searching for an optimal deterministic allocation.

In Chapters 9 and 10, we consider polling systems with multiple servers, in which the cooperation, unlike in the situation in Chapter 8, also results in actual interaction of the servers. So far, there are hardly any exact results known for such multiple-server polling systems, apart from some mean-value results for global performance measures like cycle times. In Chapter 9, we consider systems in which the servers are assumed to be *coupled*, i.e., the servers always visit the queues together. Guided by Property 1.4.1, we explore the class of systems that allow an exact analysis. For these systems, we present distributional results for the waiting time, the marginal queue length, and the joint queue length at polling epochs. The class in question includes several single-queue systems with a varying number of servers, two-queue two-server systems with exhaustive service and exponential service times, as well as infinite-server systems with an arbitrary number of queues, exhaustive or gated service, and

1.6 Overview of the monograph

deterministic service times.

In Chapter 10, we consider systems in which the servers are assumed to be *independent*, i.e., each of the servers visits the queues according to its own cyclic schedule. These systems appear to completely defy the derivation of exact analytical waiting-time results, which motivates the search for accurate approximations. We derive such waiting-time approximations for systems with the exhaustive and gated service discipline. The approximations are tested for a wide range of parameter combinations.

Throughout the monograph, stochastic variables are denoted by capitals and printed bold. References to the literature are presented as the name(s) of the author(s), followed by the corresponding index in the reference list or, in case of repeated occurrence, as the index in the reference list only, omitting the name(s) of the author(s). The chapters are each divided in a number of sections. Formulas are numbered per chapter, e.g., formula (2.6) is the sixth formula in Chapter 2. Assumptions, corollaries, examples, figures, lemma's, properties, remarks, tables, theorems, etc. are numbered per section, e.g., Theorem 4.3.1 is the first theorem in Section 3 of Chapter 4.
CHAPTER 2

DECOMPOSITION PROPERTIES AND PSEUDO-CONSERVATION LAWS IN POLLING MODELS

2.1 INTRODUCTION

Overviewing the recent polling literature, we see a prominent role played by so-called pseudo-conservation laws, which provide exact expressions for a specific weighted sum of the mean waiting times, mostly $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$, cf. Ferguson & Aminetzah [93], Watson [187]. Although the individual mean waiting times usually involve extremely complicated expressions, a pseudo-conservation law typically provides a relatively simple explicit expression, which e.g. depends on the switch-over times only through the first two moments of their sum. There are even service disciplines, like Bernoulli service, for which the individual mean waiting times are completely unknown, but for which a pseudo-conservation law is still explicitly known. Thus, pseudo-conservation laws provide a useful measure of the overall system performance. In addition, pseudo-conservation laws prove to be a valuable instrument for constructing and validating waiting-time approximations, and for determining the mean waiting times in a completely symmetric system in a simple manner. Pseudo-conservation-law-based waitingtime approximations for models with 1-limited service and an arbitrary number of queues are developed e.g. in Boxma & Meister [51], [52], Groenendijk [113], Srinivasan [168]. Waiting-time approximations for similar models with k-limited service are presented e.g. in Chang & Sandhu [66], Everitt [87], [89], Fuhrmann & Wang [104].

The existence of pseudo-conservation laws is interpreted in Boxma & Groenendijk [42], and further clarified in Boxma [38]. The framework presented by Boxma & Groenendijk unified and generalized the pseudo-conservation laws existing until then, and also explained why they existed. In addition, the approach allows a relatively simple derivation of pseudo-conservation laws, and a probabilistic interpretation of the various terms occurring in pseudo-conservation laws. Until then, pseudo-conservation laws were obtained by cumbersome ad hoc methods, which did not really explain why they existed, and what the various terms occurring represented.

The key element in the framework presented by Boxma & Groenendijk is the property of work decomposition, which builds on the fundamental property of work conservation, and is closely related to the concept of queue length decomposition, as described in Fuhrmann & Cooper [103]. To illuminate these concepts, we consider in the present chapter a single-server queueing system, not necessarily a polling model, with a Poisson arrival process of rate λ , a service time distribution $B(\cdot)$ with LST $\beta(\cdot)$ and mean β , and service interruptions. The service interruptions are assumed to result from some kind of interfering process that from time to time may keep the server from working, even when there are customers present. A period during which the server is not working, because of a service interruption, or because there are no customers present, will be referred to as a non-serving interval.

The service interruptions may be intervoven with the arrival and service process in an arbitrarily complex manner, but may not anticipate on the arrival and service times of future customers. In particular, the durations of successive service interruptions are allowed to be dependent.

For now, we abstract from what kind of interfering process causes the service interruptions. In a performability setting, a service interruption typically represents a down-period of the system. In the context of polling models, a service interruption usually corresponds to a switch-over time, or to an intervisit period with regard to a specific queue, depending upon whether the polling system in totality is viewed as a system with service interruptions, or just a specific queue in isolation. Broadening the scope, the service interruptions may also represent set-up times, shut-down times, reconfiguration times, periods during which the server performs some secondary tasks, or they may correspond to some collection or reservation mechanism for customers, on which we will focus in Section 2.4.

The remainder of the chapter is organized as follows. In Section 2.2, we describe the property of queue length decomposition for the model under consideration. Focusing on a specific queue in a polling system in isolation, we indicate how the queue length decomposition also translates into a decomposition of the waiting time, as already recognized in Section 1.4, cf. (1.3), (1.4), (1.5), (1.6). In Section 2.3, we elaborate on the related property of work decomposition. Applying it to a polling system in totality, we sketch how the work decomposition property leads to pseudo-conservation laws for the mean waiting times, as already observed in Section 1.4, cf. (1.7), (1.8). In Section 2.4, we demonstrate how the queue length and work decomposition properties may be exploited in a related model, namely, a queueing system with a customer collection mechanism.

2.2 Queue length decomposition

Using concepts from the theory of branching processes, under rather mild assumptions, Fuhrmann & Cooper [103] prove the following *queue length decomposition* property for the model under consideration:

$$\mathbf{N} \stackrel{a}{=} \mathbf{N}_{M/G/1} + \mathbf{N}_I,\tag{2.1}$$

with $\stackrel{d}{=}$ denoting equality in distribution;

 $\mathbf{N} :=$ the queue length at an arbitrary epoch;

 $\mathbf{N}_{M/G/1}$:= the queue length at an arbitrary epoch in the 'corresponding' M/G/1 system;

 $\mathbf{N}_I :=$ the queue length at an arbitrary epoch in a non-serving interval; $\mathbf{N}_{M/G/1}$ and \mathbf{N}_I being independent.

The corresponding M/G/1 system is an ordinary M/G/1 queue with similar traffic characteristics, but without any service interruptions. To find the distribution of N, it thus suffices to find the distribution of N_I, as the distribution of N_{M/G/1} is simply known from the Pollaczek-Khintchine formula, cf. [73] p. 238. From a methodological point of view, however, it is usually preferable to analyze the queue length at the beginning and the end of a non-serving interval rather than to study N_I, the queue length at an arbitrary epoch in a non-serving interval. Therefore, we now relate the distribution of N_I to the queue length distribution at such embedded epochs. Denote by N^(k)_{begin} and N^(k)_{end} the queue length at the beginning and the end of the k-th non-serving interval. Denote by N_{begin} and N_{end} a pair of stochastic variables with as joint distribution the stationary joint distribution of N^(k)_{begin} and N^(k)_{end}.

Lemma 2.2.1

$$\Pr\{\mathbf{N}_{I} = l\} = \frac{\Pr\{\mathbf{N}_{\text{begin}} \le l\} - \Pr\{\mathbf{N}_{\text{end}} \le l\}}{\mathrm{EN}_{\text{end}} - \mathrm{EN}_{\text{begin}}}.$$
(2.2)

Written in terms of pgf's,

$$\mathbf{E}(z^{\mathbf{N}_{I}}) = \frac{\mathbf{E}(z^{\mathbf{N}_{\text{begin}}}) - \mathbf{E}(z^{\mathbf{N}_{\text{end}}})}{(1-z)(\mathbf{E}\mathbf{N}_{\text{end}} - \mathbf{E}\mathbf{N}_{\text{begin}})}, \qquad |z| \le 1.$$
(2.3)

Proof

Because of the PASTA property, N_I has the same distribution as the number of customers seen by an arbitrary customer arriving in a non-serving interval. In the first K non-serving intervals, the fraction of customers that see l customers upon arrival is

$$\frac{\sum\limits_{k=1}^{K} \mathrm{I}_{\{\mathbf{N}_{\mathrm{begin}}^{(k)} \leq l \leq \mathbf{N}_{\mathrm{end}}^{(k)} - 1\}}}{\sum\limits_{k=1}^{K} (\mathbf{N}_{\mathrm{end}}^{(k)} - \mathbf{N}_{\mathrm{begin}}^{(k)})}$$

with $I_{\{E\}}$ denoting the indicator function of the event E. So, using the law of large numbers,

$$\begin{aligned} \Pr\{\mathbf{N}_{I} = l\} &= \lim_{K \to \infty} \frac{\sum\limits_{k=1}^{K} I_{\{\mathbf{N}_{begin}^{(k)} \leq l \leq \mathbf{N}_{end}^{(k)} - 1\}} / K}{\sum\limits_{k=1}^{K} (\mathbf{N}_{end}^{(k)} - \mathbf{N}_{begin}^{(k)}) / K} \\ &= \frac{\Pr\{\mathbf{N}_{begin} \leq l \leq \mathbf{N}_{end} - 1\}}{\sum_{k=1}^{K} (\mathbf{N}_{end}^{(k)} - \mathbf{N}_{begin}^{(k)})} \\ &= \frac{\Pr\{\mathbf{N}_{begin} \leq l \geq \mathbf{N}_{end} - \mathbf{E} \mathbf{N}_{begin}}{\sum_{k=1}^{K} (\mathbf{N}_{begin}^{(k)} \leq l) - \Pr\{\mathbf{N}_{end} \leq l\}}, \end{aligned}$$

as $\Pr{\{\mathbf{N}_{\text{begin}} \leq \mathbf{N}_{\text{end}}\}} = 1.$

30

The queue length decomposition property (2.1) only holds if the order of service is independent of the service times. Consequently, it typically does *not* hold for a polling system in totality, but it *does* hold for each of the queues in isolation, assuming that the order of service at each of the queues is independent of the service times. Therefore, we now focus on the queue length at a specific queue in a polling system in isolation, let us say Q_i . For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3. (In Section 2.4, we use the queue length decomposition property to analyze a system where the service interruptions arise from a customer collection procedure.) The queue length N_{begin} and N_{end} at the *beginning* and the *end* of a *non-serving interval* then correspond to the queue length Y_i at the *end* and X_i at the *beginning* of a *visit* to Q_i , respectively. Applying (2.1) and (2.3) to Q_i , we then obtain

$$\mathbf{N}_i \stackrel{a}{=} \mathbf{N}_{i|M/G/1} + \mathbf{N}_{i|I},\tag{2.4}$$

$$E(z^{\mathbf{N}_{i|I}}) = \frac{E(z^{\mathbf{Y}_{i}}) - E(z^{\mathbf{X}_{i}})}{(1-z)(E\mathbf{X}_{i} - E\mathbf{Y}_{i})}, \qquad |z| \le 1,$$
(2.5)

with

 $\mathbf{N}_i :=$ the queue length at Q_i at an arbitrary epoch;

 $\mathbf{N}_{i|M/G/1}$:= the queue length at an arbitrary epoch in the 'corresponding' M/G/1 queue of Q_i in isolation;

 $\mathbf{N}_{i|I} :=$ the queue length at Q_i at an arbitrary epoch in an intervisit period; $\mathbf{N}_{i|M/G/1}$ and $\mathbf{N}_{i|I}$ being independent.

By (2.4) and (2.5), to find the queue length distribution at Q_i at an arbitrary epoch it suffices to find the queue length distribution at Q_i at the beginning and the end of a visit, respectively. Note that for the class of service disciplines that satisfy Property 1.4.1, $E(z^{\mathbf{X}_i}) = F_i(1, \ldots, 1, z, 1, \ldots, 1)$ and $E(z^{\mathbf{Y}_i}) = G_i(1, \ldots, 1, z, 1, \ldots, 1)$, with z as *i*-th argument, are actually known.

2.3 Work decomposition

We now abstract from the polling context again. If the order of service in the system with service interruptions is FCFS, then the sojourn time \mathbf{R} of an arbitrary customer, i.e., the time from its arrival to the completion of its service, is related to the queue length \mathbf{N} (the total number of customers present, including a customer possibly in service) at an arbitrary epoch by the distributional form of Little's law, cf. Keilson & Servi [125]:

$$E(z^{N}) = E(e^{-\lambda(1-z)R}), \qquad |z| \le 1,$$
 (2.6)

provided the sojourn time of customers is independent of arrivals after their own arrival.

Similarly, the waiting time **W** of an arbitrary customer is related to the number **L** of waiting customers (excluding a customer possibly in service):

$$\mathbf{E}(z^{\mathbf{L}}) = \mathbf{E}(\mathbf{e}^{-\lambda(1-z)\mathbf{W}}), \qquad |z| \le 1.$$
(2.7)

Taking expectations in either (2.6) or (2.7) yields the original form of Little's law. Combining (2.1) and (2.6), taking $\lambda(1-z) = \omega$, we obtain the following sojourn time decomposition:

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{R}}) = \mathbf{E}(\mathbf{e}^{-\omega\mathbf{R}_{M/G/1}})\mathbf{E}((1-\omega/\lambda)^{\mathbf{N}_I}), \qquad \operatorname{Re}\omega \ge 0,$$
(2.8)

with $\mathbf{R}_{M/G/1}$ denoting the sojourn time of an arbitrary customer in the corresponding M/G/1 system.

Noting that $E(e^{-\omega \mathbf{R}}) = E(e^{-\omega \mathbf{W}})\beta(\omega)$, we obtain from (2.8) the following waiting-time decomposition:

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}}) = \mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}_{M/G/1}})\mathbf{E}((1-\omega/\lambda)^{\mathbf{N}_I}), \qquad \operatorname{Re}\omega \ge 0,$$
(2.9)

with $\mathbf{W}_{M/G/1}$ denoting the waiting time of an arbitrary customer in the corresponding M/G/1 system.

We now return to the polling setting again. Since the order of service is required to be FCFS here, the properties (2.6)-(2.9) do not hold for a polling system in totality, but they do hold for a specific queue in isolation with FCFS order of service. Applying (2.9) to Q_i , using (2.5) and the Pollaczek-Khintchine formula, cf. [73] p. 255, we obtain

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}_{i}}) = \frac{(1-\lambda_{i}\beta_{i})\omega}{\omega-\lambda_{i}(1-\beta_{i}(\omega))} \frac{\mathbf{E}((1-\omega/\lambda_{i})^{\mathbf{Y}_{i}}) - \mathbf{E}((1-\omega/\lambda_{i})^{\mathbf{X}_{i}})}{(\mathbf{E}\mathbf{X}_{i} - \mathbf{E}\mathbf{Y}_{i})\omega/\lambda_{i}}.$$

Taking for exhaustive service $\mathbf{Y}_i \equiv 0$ and $\mathbf{E}(z^{\mathbf{X}_i}) = \mathbf{E}(e^{-\lambda_i(1-z)\mathbf{I}_i})$ leads to (1.3) and (1.5). Noting that for gated service $\mathbf{E}(z^{\mathbf{Y}_i}) = \mathbf{E}(\beta_i(\lambda_i(1-z))^{\mathbf{X}_i})$ and $\mathbf{E}(z^{\mathbf{X}_i}) = \mathbf{E}(e^{-\lambda_i(1-z)\mathbf{C}_i})$ gives (1.4) and (1.6).

2.3 Work decomposition

In the previous section, we described the property of *queue length decomposition*, and indicated how it also translates into a decomposition of the waiting times at the individual queues in a polling system. In this section, we focus on the related property of *work decomposition*, and sketch how it leads to pseudo-conservation laws for the mean waiting times in a polling system, as shown by Boxma & Groenendijk [42].

Adapting the arguments of Fuhrmann & Cooper, under even milder assumptions, Boxma & Groenendijk prove the following *work decomposition* property for the model under consideration:

$$\mathbf{V} \stackrel{d}{=} \mathbf{V}_{M/G/1} + \mathbf{V}_I, \tag{2.10}$$

 \mathbf{with}

 $\mathbf{V} :=$ the amount of work in the system at an arbitrary epoch;

 $\mathbf{V}_{M/G/1}$:= the amount of work at an arbitrary epoch in the corresponding M/G/1 system;

 $\mathbf{V}_I :=$ the amount of work in the system at an arbitrary epoch in a non-serving interval;

 $\mathbf{V}_{M/G/1}$ and \mathbf{V}_I being independent.

When the amount of work in a non-serving interval were always zero, i.e., $\mathbf{V}_I \equiv 0$, (2.10) would reduce to the fundamental property of *work conserva*tion, which in fact holds even in sample-path sense. Note that in a polling system it is generally not the case that $\mathbf{V}_I \equiv 0$, since the server may be switching when there are customers present.

The work decomposition property still holds if the order of service is not independent of the service times, reflecting that in a sense the amount of work in the system is a less sensitive quantity than the queue length. Consequently, it holds in particular for the total amount of work in a polling system. Therefore, we now focus on the total amount of work in a polling system, and show how the work decomposition property leads to a pseudo-conservation law for the mean waiting times. For the specification of the arrival, service, and switchover processes, we refer to the description of the 'basic model' in Section 1.3. (In Section 2.4, we use the work decomposition property to analyze a system where the service interruptions originate from a customer collection procedure.) Applying Brumelle's formula [60],

$$E\mathbf{V} = \sum_{i=1}^{n} \rho_i E\mathbf{W}_i + \frac{1}{2} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)}.$$
 (2.11)

From the Pollaczek-Khintchine formula, cf. [73] p. 255,

n

$$\mathbf{E}\mathbf{V}_{M/G/1} = \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)}.$$
(2.12)

Taking expectations in (2.10), substituting (2.11), (2.12), we obtain the following relationship for the mean waiting times:

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \mathbf{E} \mathbf{V}_I.$$
(2.13)

2.3 Work decomposition

When the amount of work in a non-serving interval were always zero, i.e., $\mathbf{EV}_I = 0$, (2.13) would reduce to the property that $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$ does not depend on the scheduling discipline as long as $\mathbf{EV}_I = 0$, which is commonly referred to as a *conservation law*.

For strictly cyclic polling, Boxma & Groenendijk [42] show that EV_I may be determined as follows:

$$\mathbf{EV}_{I} = \rho \frac{s^{(2)}}{2s} + \frac{s}{2(1-\rho)} \left[\rho^{2} - \sum_{i=1}^{n} \rho_{i}^{2} \right] + \sum_{i=1}^{n} \mathbf{EZ}_{ii}, \qquad (2.14)$$

with \mathbf{Z}_{ii} denoting the amount of work left behind by the server at Q_i at the completion of a visit.

Substituting (2.14) into (2.13), we obtain the following relationship for the mean waiting times:

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \rho \frac{s^{(2)}}{2s} + \frac{s}{2(1-\rho)} \left[\rho^2 - \sum_{i=1}^{n} \rho_i^2 \right] + \sum_{i=1}^{n} \mathbf{E} \mathbf{Z}_{ii}, (2.15)$$

which is commonly referred to as a *pseudo-conservation law*, since $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$ now *does* depend on the scheduling discipline through the terms \mathbf{EZ}_{ii} . As a pleasing circumstance however, \mathbf{EZ}_{ii} is determined by the service discipline at Q_i only, i.e., not by the service discipline at Q_j , $j \neq i$, e.g. [42]: I. Exhaustive service:

$$\mathbf{E}\mathbf{Z}_{ii}=0.$$

II. Gated service:

$$\mathbf{E}\mathbf{Z}_{ii} = \frac{\rho_i^2 s}{1-\rho}.$$

III. 1-Limited service:

$$\mathbf{E}\mathbf{Z}_{ii} = \frac{\rho_i^2 s}{1-\rho} + \rho_i \frac{\lambda_i s}{1-\rho} \mathbf{E}\mathbf{W}_i.$$

IV. 1-Decrementing service:

$$\mathbf{E}\mathbf{Z}_{ii} = -\frac{\rho_i^2 \lambda_i^2 \beta_i^{(2)} s}{2(1-\rho)} + \rho_i \frac{\lambda_i (1-\rho_i) s}{1-\rho} \mathbf{E}\mathbf{W}_i.$$

Substituting the above expressions into (2.15) yields the pseudo-conservation laws, which before were only known to hold in cases with the same service disciplines at each of the queues.

Here we sketched the derivation of a pseudo-conservation law for a continuoustime model with strictly cyclic polling and single Poisson arrivals. Without seriously complicating the derivation, continuous-time may be replaced by discrete-time, cf. [44], strictly cyclic polling may be generalized to periodic polling, cf. [45], or Markovian polling, cf. [53], and single Poisson arrivals may be generalized to batch Poisson arrivals, cf. [38]. In Chapter 4, we show how a pseudo-conservation law is derived in case the server may be allowed to make a halt at a queue when there are no customers present in the system.

2.4 A QUEUEING SYSTEM WITH A CUSTOMER COL-LECTION MECHANISM

In the two previous sections, we sketched how decomposition properties may be exploited in the setting of polling models. In this section, we describe how such decomposition properties may be applied to a related model, namely, a queueing system with a customer collection mechanism. For a detailed analysis of an aggregated version of the model, we refer to Borst, Boxma, & Combé [32], [33]. For a generalization and unification of the results, we refer to Boxma & Combé [40], Combé [76].

The model under consideration consists of n queues, Q_1, \ldots, Q_n , attended by a single server S. For the specification of the arrival and service processes at the queues, we refer to the description of the 'basic model' in Section 1.3. Unlike in a polling model, where the server periodically visits the queues to serve (some of) the customers present, here a *collector* from time to time visits the queues, picks up (some of) the customers present, and delivers them to the server where they are served. Collectors are assumed to be sent out according to a Poisson process of rate γ , independent of the arrival and service processes at the queues. A collector visits the queues in strictly cyclic order, Q_1, \ldots, Q_n . Moving from Q_i to Q_{i+1} requires a constant travel time σ_i , $i = 1, \ldots, n$. Here σ_n is to be understood as the travel time from Q_n to the server. (Note that the travel time from the origin of the collectors to Q_1 is irrelevant here.)

Upon arrival at Q_i , the collector instantaneously picks up all the type-*i* customers that have already been present for at least a constant time τ_i (e.g. $\tau_i = 0$), i = 1, ..., n. The collectors are assumed to have infinite capacity. Upon arrival at the server, the collector instantaneously delivers all the customers that were picked up. The order of service is assumed not to discriminate between the various customer types; within the various customer classes the order of service is assumed to be FCFS.

Conceptually, a batch of collected customers may also be viewed as a single 'super' customer. The service time of the super customer comprises the total service time of the corresponding batch. So a super customer has a zero service time in case a batch happens to be empty. The arrival epoch of the super customer coincides with the arrival epoch of the corresponding collector at the server. Since the collectors have a constant travel time, the super customers arrive at the server according to a Poisson process of rate γ . Focusing on the

super customers, the model under consideration may thus be viewed as an ordinary M/G/1 queue, essentially differing however from the standard model description due to the intrinsic *dependence* between the interarrival and service times. Namely, the larger the interarrival time of a super customer, the larger the corresponding intercollector time, the larger the total service time of the corresponding batch, and the larger the service time of the super customer. To be specific, denote by (\mathbf{A}, \mathbf{B}) a pair of stochastic variables with as joint distribution the joint distribution of the interarrival and service time of an arbitrary super customer. From [32], [33],

$$E(e^{-\zeta \mathbf{A} - \omega \mathbf{B}}) = \frac{\gamma}{\gamma + \zeta + \lambda(1 - \beta(\omega))}, \qquad \operatorname{Re} \zeta \ge 0, \operatorname{Re} \omega \ge 0, \qquad (2.16)$$

with $\beta(\omega) := \sum_{i=1}^{n} \frac{\lambda_i}{\lambda} \beta_i(\omega)$, so that $\operatorname{Cov}(\mathbf{A}, \mathbf{B}) = \lambda \beta / \gamma^2$ with $\beta = \sum_{i=1}^{n} \frac{\lambda_i}{\lambda} \beta_i = \rho / \lambda$. In the sequel, we will not really highlight the effects of the dependence between the interpretion and service times, but the impact is studied in detail

between the interarrival and service times, but the impact is studied in detail in [32, 33, 40, 76].

For future convenience, we first observe that the collection procedure under consideration may be equivalently defined as follows. Arriving type-*i* customers are first 'retained' for a while, and 'released' after a constant time $\tau_i + \sum_{j=i}^n \sigma_j$. Upon arrival at Q_i , the collector instantaneously picks up only the released

type-*i* customers. Moving from Q_i to Q_{i+1} requires no travel time.

Taking the perspective of the last paragraph, we immediately see that at every epoch the population of released customers is independent of the population of customers being retained, implying

$$\mathbf{V} \stackrel{a}{=} \mathbf{V}^* + \mathbf{V}^\#,\tag{2.17}$$

with

 $\mathbf{V} :=$ the amount of work in the system at an arbitrary epoch;

 $\mathbf{V}^* :=$ the amount of work in the system at an arbitrary epoch corresponding to *released* customers;

 $V^{\#}$:= the remaining amount of work in the system at an arbitrary epoch corresponding to customers being retained;

 \mathbf{V}^* and $\mathbf{V}^{\#}$ being independent.

The distribution of $\mathbf{V}^{\#}$ is given by

$$\mathbf{E}(\mathbf{e}^{-\omega \mathbf{V}^{\#}}) = \exp\left[-\sum_{i=1}^{n} \lambda_i (1 - \beta_i(\omega))(\tau_i + \sum_{j=i}^{n} \sigma_j)\right], \qquad \operatorname{Re} \omega \ge 0. \quad (2.18)$$

Denote by \mathbf{V}^0 the amount of work at an arbitrary epoch in the 'corresponding' system with a zero-delay collection procedure, i.e., a system with similar traffic characteristics, but with $\sigma_i = 0, \tau_i = 0, i = 1, \ldots, n$.

Observing that the population of released customers in the original system

evolves similarly as the total population of customers in the corresponding zero-delay system,

$$\mathbf{V}^* \stackrel{d}{=} \mathbf{V}^0. \tag{2.19}$$

As described in Section 2.1, the following decomposition property holds for \mathbf{V}^0 , cf. [38]:

$$\mathbf{V}^0 \stackrel{d}{=} \mathbf{V}_{M/G/1} + \mathbf{V}_I^0, \tag{2.20}$$

with

 $\mathbf{V}_{M/G/1}$:= the amount of work at an arbitrary epoch in the 'corresponding' M/G/1 system without customer collection mechanism, i.e., a system with similar traffic characteristics, where the customers have immediate access to the server;

 $\mathbf{V}_{I}^{0}:=$ the amount of work in the zero-delay system at an arbitrary epoch in a non-serving interval;

 $\mathbf{V}_{M/G/1}$ and \mathbf{V}_I^0 being independent.

For future convenience, we first introduce some further terminology. A 'basic' non-serving interval is a non-serving interval beginning at the departure of a super customer that leaves no super customers behind, and ending at the arrival of the next super customer (which sees no super customers present). Since super customers arrive according to a Poisson process of rate γ , the length of a basic non-serving interval is exponentially distributed with mean $1/\gamma$. In case the next arriving super customer (which sees no super customers present) happens to have zero service time, a number of consecutive basic non-serving intervals may occur, together constituting a larger non-serving interval. A 'maximal' non-serving interval is a non-serving interval that is not strictly contained in any other larger non-serving interval. Note that the length of a maximal non-serving interval is no longer exponentially distributed.

The distribution of $\mathbf{V}_{M/G/1}$ in (2.20) follows from the Pollaczek-Khintchine formula, cf. [73] p. 255,

$$E(e^{-\omega \mathbf{V}_{M/G/1}}) = \frac{(1 - \lambda\beta)\omega}{\omega - \lambda(1 - \beta(\omega))}, \qquad \text{Re}\,\omega \ge 0.$$
(2.21)

The quantity \mathbf{V}_{I}^{0} in (2.20) may be equivalently defined as the amount of work in the zero-delay system at an arbitrary epoch in a basic non-serving interval. Let us say D is the super customer at whose departure the basic non-serving interval in question started.

The quantity \mathbf{V}_{I}^{0} then consists of two independent components, viz.,

$$\mathbf{V}_I^0 \stackrel{d}{=} \mathbf{V}_I' + \mathbf{V}_I'',\tag{2.22}$$

with

 $\mathbf{V}'_I :=$ the amount of work that arrived since the departure of D; $\mathbf{V}''_I :=$ the amount of work that arrived during the sojourn time of D;

2.4 A queueing system with a customer collection mechanism

\mathbf{V}'_{I} and \mathbf{V}''_{I} being independent.

For any non-negative stochastic variable \mathbf{T} , denote by $\mathbf{V}(\mathbf{T})$ the amount of work arriving to the system during a period of length \mathbf{T} , i.e.,

$$\mathbf{E}(\mathbf{e}^{-\omega \mathbf{V}(\mathbf{T})}) = \mathbf{E}(\mathbf{e}^{-\lambda(1-\beta(\omega))\mathbf{T}}), \qquad \operatorname{Re} \omega \ge 0.$$
(2.23)

The quantity \mathbf{V}'_I is the amount of work arriving during the past of the basic non-serving interval in question at the arrival of D. Since a basic non-serving interval is exponentially distributed, the past of a basic non-serving interval is also exponentially distributed. So \mathbf{V}'_I is distributed as the amount of work arriving during an arbitrary interarrival time \mathbf{A} , i.e., distributed as the service time \mathbf{B} of an arbitrary super customer,

$$\mathbf{V}_{I}^{\prime} \stackrel{d}{=} \mathbf{V}(\mathbf{A}) \stackrel{d}{=} \mathbf{B},\tag{2.24}$$

with, taking $\zeta = 0$ in (2.16),

$$E(e^{-\omega \mathbf{B}}) = \frac{\gamma}{\gamma + \lambda(1 - \beta(\omega))}, \qquad \text{Re}\,\omega \ge 0.$$
(2.25)

Denote by $\tilde{\mathbf{R}}$ the sojourn time of D.

$$\mathbf{V}_{I}^{\prime\prime} \stackrel{a}{=} \mathbf{V}(\tilde{\mathbf{R}}). \tag{2.26}$$

Substituting (2.19), (2.20), (2.22), (2.24), (2.26) into (2.17), we obtain the following detailed form of the work decomposition property:

$$\mathbf{V} \stackrel{d}{=} \mathbf{V}_{M/G/1} + \mathbf{V}(\mathbf{A}) + \mathbf{V}(\tilde{\mathbf{R}}) + \mathbf{V}^{\#}.$$
(2.27)

We now show how a functional equation may be derived for the LST $r(\omega) := E(e^{-\omega \mathbf{R}})$, Re $\omega \ge 0$, of the sojourn time distribution of an arbitrary super customer.

Because of the PASTA property,

,

$$\mathbf{V}^* \stackrel{d}{=} \mathbf{R}.\tag{2.28}$$

From (2.19), (2.20), (2.22), (2.24), (2.26), (2.28), we obtain the following decomposition of the sojourn time of an arbitrary super customer:

$$\mathbf{R} \stackrel{a}{=} \mathbf{V}_{M/G/1} + \mathbf{B} + \mathbf{V}(\tilde{\mathbf{R}}), \tag{2.29}$$

all three terms in the right-hand side being independent. Note that the waiting time **W** and the service time **B** of a super customer are however *not* independent, due to the dependence between the interarrival and service time. Namely, the larger the service time, the larger the interarrival time, and the smaller the waiting time. Hence, although the sojourn time is composed of the waiting time and the service time, (2.29) does *not* imply that $\mathbf{W} \stackrel{d}{=} \mathbf{V}_{M/G/1} + \mathbf{V}(\tilde{\mathbf{R}})$. In [40] it is shown how the joint distribution of (\mathbf{W}, \mathbf{B}) may be determined in a direct manner.

The sojourn time $\tilde{\mathbf{R}}$ of D may be thought of as that of an arbitrary super customer whose sojourn time is known to be smaller than an exponentially distributed interarrival time \mathbf{A} ,

$$E(e^{-\omega\tilde{\mathbf{R}}}) = \frac{r(\gamma + \omega)}{r(\gamma)}, \qquad \text{Re}\,\omega \ge 0.$$
(2.30)

Note that $r(\gamma)$ equals the probability that an arbitrary super customer finds the server idle upon arrival. Because of the PASTA property, the latter probability equals the fraction of time that the server is idle, $1 - \lambda\beta$.

So we obtain from (2.21), (2.23), (2.25), (2.29), and (2.30) the following functional equation for $r(\cdot)$:

$$r(\omega) = \frac{\gamma\omega}{\omega - \lambda(1 - \beta(\omega))} \frac{r(\gamma + \lambda(1 - \beta(\omega)))}{\gamma + \lambda(1 - \beta(\omega))}, \qquad \text{Re}\,\omega \ge 0.$$
(2.31)

We now solve the above functional equation. Denote

$$egin{aligned} f(\omega) &:= rac{\gamma \omega}{\omega - \lambda (1 - eta(\omega))}, & ext{Re}\, \omega \geq 0, \ g(\omega) &:= \gamma + \lambda (1 - eta(\omega)), & ext{Re}\, \omega \geq 0. \end{aligned}$$

Then (2.31) may be rewritten as

$$r(\omega) = \frac{f(\omega)}{g(\omega)} r(g(\omega)), \qquad \text{Re}\,\omega \ge 0.$$
 (2.32)

Define

$$g^{(0)}(\omega) := \omega, \quad \operatorname{Re} \omega \ge 0,$$

 $g^{(k)}(\omega) := g(g^{(k-1)}(\omega)), \quad \operatorname{Re} \omega \ge 0, k = 1, 2, \dots.$

Iterating (2.32) K times, we find

$$r(\omega) = r(g^{(K+1)}(\omega)) \prod_{k=0}^{K} \frac{f(g^{(k)}(\omega))}{g^{(k+1)}(\omega)}, \qquad \text{Re}\,\omega \ge 0.$$
(2.33)

Letting $K \to \infty$ in (2.33),

$$r(\omega) = r(\omega^*) \prod_{k=0}^{\infty} \frac{f(g^{(k)}(\omega))}{g^{(k+1)}(\omega)}, \qquad \text{Re}\,\omega \ge 0,$$
(2.34)

with $\omega^* := \lim_{K \to \infty} g^{(K)}(\omega)$. For a proof of the convergence for $\rho < 1$, we refer to [33], [40].

Putting $\omega = 0$ in (2.34),

$$r(\omega^*) = 1 \left/ \prod_{k=0}^{\infty} \frac{f(g^{(k)}(0))}{g^{(k+1)}(0)} \right|.$$
(2.35)

Substituting (2.35) back into (2.34),

$$r(\omega) = \prod_{k=0}^{\infty} \left[\frac{f(g^{(k)}(\omega))}{g^{(k+1)}(\omega)} \middle/ \frac{f(g^{(k)}(0))}{g^{(k+1)}(0)} \right], \qquad \text{Re}\,\omega \ge 0.$$
(2.36)

Above we showed how the work decomposition property could be exploited to derive a functional equation for the LST of the sojourn time distribution of an arbitrary super customer. We now show how similarly the queue length decomposition property can be applied to find an expression for the pgf of the joint distribution of the numbers of customers present at an arbitrary epoch. Similar to (2.17),

$$(\mathbf{N}_1,\ldots,\mathbf{N}_n) \stackrel{d}{=} (\mathbf{N}_1^*,\ldots,\mathbf{N}_n^*) + (\mathbf{N}_1^\#,\ldots,\mathbf{N}_n^\#), \tag{2.37}$$

with

 $\mathbf{N}_i :=$ the number of type-*i* customers present at an arbitrary epoch;

 $\mathbf{N}_{i}^{*} :=$ the number of released type-*i* customers present at an arbitrary epoch; $\mathbf{N}_{i}^{\#} :=$ the number of remaining type-*i* customers being retained at an arbitrary epoch;

 $(\mathbf{N}_1^*, \dots, \mathbf{N}_n^*)$ and $(\mathbf{N}_1^\#, \dots, \mathbf{N}_n^\#)$ being independent.

The joint distribution of $(\mathbf{N}_1^{\#}, \ldots, \mathbf{N}_n^{\#})$ is given by

$$E(z_1^{\mathbf{N}_1^{\#}} \dots z_n^{\mathbf{N}_n^{\#}}) = \exp[-\sum_{i=1}^n \lambda_i (1-z_i)(\tau_i + \sum_{j=i}^n \sigma_j)], \qquad (2.38)$$

for $|z_i| \leq 1, i = 1, ..., n$.

Denote by \mathbf{N}^* the total number of released customers present at an arbitrary epoch. Denote by \mathbf{N}^0 the total number of customers present at an arbitrary epoch in the corresponding system with a zero-delay collection procedure (i.e. with $\sigma_i = 0, \tau_i = 0, i = 1, ..., n$). Similar to (2.19),

$$\mathbf{N}^* \stackrel{d}{=} \mathbf{N}^0. \tag{2.39}$$

As described in Section 2.1, the following queue length decomposition property holds for N^0 , cf. [103]:

$$\mathbf{N}^0 \stackrel{a}{=} \mathbf{N}_{M/G/1} + \mathbf{N}_I^0, \tag{2.40}$$

with

 $\mathbf{N}_{M/G/1}$:= the total number of customers present at an arbitrary epoch in the corresponding M/G/1 system without a customer collection mechanism (i.e. where the customers have immediate access to the server);

 \mathbf{N}_{I}^{0} := the total number of customers present in the zero-delay system at an arbitrary epoch in a non-serving interval;

 $N_{M/G/1}$ and N_I^0 being independent.

The distribution of $\mathbf{N}_{M/G/1}$ in (2.40) follows from the Pollaczek-Khintchine formula, cf. [73] p. 238,

$$E(z^{N_{M/G/1}}) = \frac{(1 - \lambda\beta)(1 - z)\beta(\lambda(1 - z))}{\beta(\lambda(1 - z)) - z}, \qquad |z| \le 1.$$
(2.41)

For any non-negative stochastic variable \mathbf{T} , denote by $\mathbf{N}(\mathbf{T})$ the total number of customers arriving to the system during a period of length T, i.e.,

$$E(z^{N(T)}) = E(e^{-\lambda(1-z)T}), \qquad |z| \le 1.$$
 (2.42)

Substituting (2.40) into (2.39), after decomposing the quantity N_I^0 similarly to the quantity \mathbf{V}_{I}^{0} in (2.22), we obtain the following detailed form of the queue length decomposition property:

$$\mathbf{N}^* \stackrel{d}{=} \mathbf{N}_{M/G/1} + \mathbf{N}(\mathbf{A}) + \mathbf{N}(\tilde{\mathbf{R}}).$$
(2.43)

Since the order of service is assumed not to discriminate between the various customer types,

$$E(z_1^{\mathbf{N}_1^*} \dots z_n^{\mathbf{N}_n^*}) = E(\pi(z)^{\mathbf{N}^*}), \qquad |z_i| \le 1, \ i = 1, \dots, n,$$
(2.44)

with $\pi(z) := \sum_{i=1}^{n} \lambda_i z_i / \lambda$. From (2.37), (2.38), (2.44), (2.43), we obtain

$$\mathbf{E}(z_1^{\mathbf{N}_1} \dots z_n^{\mathbf{N}_n}) = \mathbf{E}(\pi(z)^{\mathbf{N}_{M/G/1}}) \mathbf{E}(\pi(z)^{\mathbf{N}(\mathbf{A})}) \mathbf{E}(\pi(z)^{\mathbf{N}(\mathbf{\bar{R}})})$$

$$\times \exp[-\sum_{i=1}^n \lambda_i (1-z_i)(\tau_i + \sum_{j=i}^n \sigma_j)],$$
(2.45)

for $|z_i| \le 1, i = 1, ..., n$. From (2.42),

$$\mathbf{E}(z^{\mathbf{N}(\mathbf{A})}) = \frac{\gamma}{\gamma + \lambda(1-z)}, \qquad |z| \le 1.$$
(2.46)

From (2.30), (2.42),

$$\mathbf{E}(z^{\mathbf{N}(\tilde{\mathbf{R}})}) = \frac{r(\gamma + \lambda(1-z))}{r(\gamma)}, \qquad |z| \le 1,$$
(2.47)

with $r(\cdot)$ as in (2.36). Thus, $E(z_1^{N_1} \dots z_n^{N_n})$ is completely specified through (2.41), (2.45)-(2.47) and (2.36). Taking $z = (1, \dots, 1, y, 1, \dots, 1)$ in (2.45) with y as *i*-th argument,

$$E(y^{\mathbf{N}_{i}}) = E(\pi_{i}(y)^{\mathbf{N}_{M/G/1}})E(\pi_{i}(y)^{\mathbf{N}(\mathbf{A})})E(\pi_{i}(y)^{\mathbf{N}(\mathbf{\bar{R}})})$$

$$\times \exp[-\lambda_{i}(1-y)(\tau_{i}+\sum_{j=i}^{n}\sigma_{j})],$$
(2.48)

2.4 A queueing system with a customer collection mechanism

with $\pi_i(y) := 1 - \lambda_i(1-y)/\lambda$.

Taking $\lambda_i(1-y) = \omega$ in (2.48), we obtain the following decomposition of the sojourn time \mathbf{R}_i of an arbitrary type-*i* customer:

$$E(e^{-\omega \mathbf{R}_{i}}) = E(e^{-\omega \mathbf{R}_{M/G/1}})E(e^{-\omega \mathbf{\bar{R}}})E(e^{-\omega \mathbf{\bar{R}}})$$

$$\times \exp[-\omega(\tau_{i} + \sum_{j=i}^{n} \sigma_{j})].$$
(2.49)

(Recall that within the various customer classes the order of service is assumed to be FCFS.)

Noting that $E(e^{-\omega \mathbf{R}_i}) = E(e^{-\omega \mathbf{W}_i})\beta_i(\omega)$, we obtain from (2.49) the following decomposition of the waiting time \mathbf{W}_i of an arbitrary type-*i* customer:

$$E(e^{-\omega \mathbf{W}_{i}}) = \frac{\beta(\omega)}{\beta_{i}(\omega)} E(e^{-\omega \mathbf{W}_{M/G/1}}) E(e^{-\omega \mathbf{A}}) E(e^{-\omega \mathbf{\tilde{R}}})$$
(2.50)

$$\times \exp[-\omega(\tau_{i} + \sum_{j=i}^{n} \sigma_{j})].$$



CHAPTER 3

Polling systems with zero and non-zero switch-over times

3.1 INTRODUCTION

In the present chapter, we consider two different single-server polling systems: (i) a model with *zero* switch-over times, and (ii) a model with *non-zero* switchover times, in which the server keeps cycling when the system is empty. For both models, we relate the steady-state queue length distribution at a queue to the queue length distribution at visit beginning and visit completion instants at that queue. As a by-product, we obtain a shorter proof of the Fuhrmann-Cooper decomposition, discussed earlier in Section 2.2. For the important class of service disciplines with a branching structure satisfying Property 1.4.1, we expose a strong relationship between both the queue length and the waitingtime distribution in the two models. We also show how the latter relationship can be exploited to reduce the computational complexity of numerical moment calculations.

As described in Section 2.1, polling systems may be viewed as queueing systems with service interruptions. Focusing on a specific queue in isolation, the service interruptions correspond to the intervisit times of the server with regard to that queue. Accordingly, the concept of queue length decomposition for queues with service interruptions (cf. Equation (2.1), Fuhrmann & Cooper [103]) has proven to be very fruitful for the analysis of polling models. It has also led to the concept of work decomposition in polling models (cf. Equation (2.10), Boxma [38]), which relates the amount of work in a system with switch-over times to the amount of work in a system with similar traffic characteristics but without switch-over times. Note that in the latter case, the switch-over times and models without switch-over times had usually been treated separately, often via different approaches; the problem with simply letting the switch-over times is that the number of polling model with non-zero switch-over times is that the number of polling epochs in an idle period tends to infinity, leading to de-

generate distributions at such epochs, cf. [140], [86]. The relationship between the two models has further been exposed in some recent papers of Cooper, Niu, & Srinivasan [79], Fuhrmann [101], and Srinivasan, Niu, & Cooper [169]; these authors consider waiting times and queue lengths instead of workloads. In the present chapter, we unify and generalize some of their results.

Firstly, in Section 3.3, we use a beautiful relation of Eisenberg [84] (see also [86]), which has received too little attention in the literature, to relate the probability generating functions (pgf's) of queue lengths at various instants in the polling system (visit beginnings and endings, service beginnings and endings). We observe that this relation, which was presented by Eisenberg [84] for the case of non-zero switch-over times, also holds for the case of zero switch-over times, and we show how it almost instantaneously gives a simple proof of the above-mentioned Fuhrmann-Cooper decomposition for the queue lengths at the various queues of a polling system.

Eisenberg's relation leads to an expression for the joint queue length pgf at service completion instants at some queue into the joint queue length pgf's at the beginning and the end of a visit to that queue. The latter pgf's can be easily related, and determined, for the important class of polling models in which the service discipline at each queue satisfies Property 1.4.1, which we restate here for the sake of completeness:

Property 3.1.1

If there are k_i customers present at Q_i at the start of a visit, then during the course of the visit, each of these k_i customers will effectively be replaced in an i.i.d. manner by a random population having pgf $h_i(z_1, \ldots, z_n)$, which may be any *n*-dimensional pgf.

Resing [159] (see also Fuhrmann [99]) has studied polling systems that satisfy this property; this includes the case of exhaustive or gated service at all queues, but it excludes the case of 1-limited service at any queue. As described in Section 1.4, for this class of polling systems, the joint queue length process at visit instants of a fixed queue is a so-called *multi-type branching process* with immigration. The theory of multi-type branching processes (cf. Athreya & Ney [13], Resing [158]) thus leads to an expression for the pgf of the joint queue length process at visit beginning (polling) instants. In Section 3.4, for models that satisfy Property 3.1.1, we use a slightly adapted version of the results of Resing [159] to relate the joint queue length pgf's at visit beginning and visit ending instants, and then to obtain those pgf's. The results expose a close similarity between the cases with and without switch-over times. In Section 3.5, we determine the steady-state marginal queue length pgf at Q_i , both for the model with and the model without switch-over times, and we relate the pgf's for those two cases; similarly for the waiting-time Laplace-Stieltjes Transform (LST) at Q_i . In Section 3.6, we describe how the results can be exploited for a very efficient numerical calculation of the waiting-time moments under different switch-over time scenarios.

3.2 MODEL DESCRIPTION

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by a single server S. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3.

The server visits the queues in strictly cyclic order, Q_1, \ldots, Q_n . We consider two versions of the model. In the first variant, all the switch-over times are zero, i.e., $\sigma(\infty) = 1$, s = 0, $s^{(2)} = 0$. In the other variant, at least one of the switch-over times is non-zero with some positive probability, i.e., $\sigma(\infty) < 1$, s > 0, $s^{(2)} > 0$.

In the model with *non-zero* switch-over times, the server keeps switching when the system becomes empty. In the model with *zero* switch-over times, when the system becomes empty, the server makes a full cycle, i.e., passes all the queues once, and subsequently stops right before Q_1 . All this requires zero time. When the first new customer arrives, the server cycles along the queues to that customer. The choice of Q_1 is arbitrary, but for the application of the theory of multi-type branching processes in Section 3.4, it will be necessary to fix one position. There we shall discuss this issue in more detail.

We assume the service disciplines to be non-idling, i.e., the server is not allowed to idle when there are customers present. For now, we do not specify the service disciplines any further.

3.3 The joint queue length distribution at various epochs

Eisenberg [84] studies the model under consideration for the case of non-zero switch-over times and the exhaustive service discipline at all queues (while briefly discussing the case of gated service at all queues). He considers the following four quantities, with N denoting a vector of numbers of customers at Q_1, \ldots, Q_n and N a realization:

 $L_i(t, N) :=$ number of service beginnings at Q_i in (0, t) for which $\mathbf{N} = N$; $M_i(t, N) :=$ number of service completions at Q_i in (0, t) for which $\mathbf{N} = N$; $F_i(t, N) :=$ number of visit beginnings at Q_i in (0, t) for which $\mathbf{N} = N$; $G_i(t, N) :=$ number of visit completions at Q_i in (0, t) for which $\mathbf{N} = N$;

In the case of a service or visit completion, the state is defined as what exists immediately after the departure of the customer.

Eisenberg [84] now makes the crucial observation that each time a visit beginning or a service completion occurs, this coincides with either a service beginning or a visit completion. Hence,

$$F_i(t,N) + M_i(t,N) = L_i(t,N) + G_i(t,N).$$
(3.1)

We observe that (3.1) not only holds for the case of non-zero switch-over times and exhaustive or gated service, but for any service discipline, and also for the case of zero switch-over times. Define the following equilibrium state probabilities for this polling model:

 $L_i(N) := \Pr(\mathbf{N} = N, S \text{ is at } Q_i \mid \text{service beginning instant});$ $M_i(N) := \Pr(\mathbf{N} = N, S \text{ is at } Q_i \mid \text{service completion instant});$ $F_i(N) := \Pr(\mathbf{N} = N \mid \text{visit beginning at } Q_i);$ $G_i(N) := \Pr(\mathbf{N} = N \mid \text{visit completion at } Q_i).$

Eisenberg [84] divides all four terms in (3.1) by the total number of service completions at all queues in (0, t), and takes the limit for $t \to \infty$. He thus relates those four equilibrium state probabilities:

$$\gamma_i F_i(N) + M_i(N) = L_i(N) + \gamma_i G_i(N).$$

Here γ_i is the long-term ratio of the number of visit completions at Q_i to the number of customers that are handled by the system; in this cyclic polling model $\gamma_i \equiv \gamma, i = 1, ..., n$. Written in terms of pgf's,

$$\gamma F_i(z) + M_i(z) = L_i(z) + \gamma G_i(z), \qquad (3.2)$$

for $z = (z_1, \ldots, z_n)$, $|z_j| \le 1$, $j = 1, \ldots, n$; here $F_i(z)$ and $G_i(z)$ denote the pgf of the joint queue length distribution at visit beginnings and visit completions of Q_i , respectively, while $L_i(z)$ and $M_i(z)$ denote the pgf of the joint distribution of queue length vector and server position at service beginnings and service completions, respectively.

Now Eisenberg observes that $M_i(z)$ and $L_i(z)$ are related via

$$M_{i}(z) = L_{i}(z)\beta_{i}(\sum_{j=1}^{n} \lambda_{j}(1-z_{j}))/z_{i},$$
(3.3)

for $|z_j| \leq 1, j = 1, ..., n$. It follows from (3.2) and (3.3) that

$$M_{i}(z) = \frac{\gamma \beta_{i}(\sum_{j=1}^{n} \lambda_{j}(1-z_{j}))}{z_{i} - \beta_{i}(\sum_{j=1}^{n} \lambda_{j}(1-z_{j}))} [F_{i}(z) - G_{i}(z)].$$
(3.4)

Eisenberg, considering the variant with switch-over times and exhaustive service, subsequently expresses $F_i(z)$ into $G_{i-1}(z)$. For the moment we refrain from that (see Section 3.4), but we observe that formula (3.4) is generally valid for the class of polling systems described in Section 3.2 (with and without switch-over times).

Taking z = (1, ..., 1, y, 1, ..., 1) in (3.4), with y as *i*-th argument, and dividing by the probability λ_i/λ that an arbitrary service completion is at Q_i , gives the queue length pgf at Q_i at a service completion instant at Q_i . A standard up-

3.3 The joint queue length distribution at various epochs

and down-crossing argument, combined with PASTA, shows that the queue length distribution at Q_i at its service completion instants, at its customer arrival instants, and in steady-state, are all the same. Hence, with N_i the steady-state queue length at Q_i and with X_i and Y_i the steady-state queue lengths at Q_i at the beginning and end of a visit at that queue: for $|y| \leq 1$,

$$\mathbf{E}(y^{\mathbf{N}_{i}}) = \frac{\lambda}{\lambda_{i}} \frac{\gamma \beta_{i}(\lambda_{i}(1-y))}{y - \beta_{i}(\lambda_{i}(1-y))} [\mathbf{E}(y^{\mathbf{X}_{i}}) - \mathbf{E}(y^{\mathbf{Y}_{i}})].$$
(3.5)

Note that $1/\gamma$ equals the mean number of customers served per cycle, hence also the mean number of customers that arrive per cycle: $1/\gamma = \lambda EC$, with EC the mean length of one cycle of S along the queues (a cycle w.r.t. Q_i is defined as the period between the start of two successive visits to Q_i ; it is easily seen that the mean cycle time is the same for all *i*). Because S spends on the average a fraction ρ_i of a cycle at Q_i , we can write:

$$\mathbf{E}\mathbf{X}_{i} - \mathbf{E}\mathbf{Y}_{i} = \lambda_{i}(1-\rho_{i})\mathbf{E}\mathbf{C} = \frac{\lambda_{i}(1-\rho_{i})}{\lambda\gamma}.$$
(3.6)

From (3.5) and (3.6), for $|y| \le 1$:

$$E(y^{\mathbf{N}_{i}}) = \frac{(1-\rho_{i})(1-y)\beta_{i}(\lambda_{i}(1-y))}{\beta_{i}(\lambda_{i}(1-y)) - y} \frac{E(y^{\mathbf{Y}_{i}}) - E(y^{\mathbf{X}_{i}})}{(1-y)(E\mathbf{X}_{i} - E\mathbf{Y}_{i})}.$$
(3.7)

The first term in the right-hand side is the pgf $E(y^{N_{i|M/G/1}})$ of the queue length distribution in a 'corresponding' isolated M/G/1 queue of Q_i with arrival rate λ_i and service time distribution LST $\beta_i(\cdot)$. Now consider the second term. Observe that \mathbf{Y}_i not only denotes the queue length at Q_i at the end of a visit to that queue, but also the queue length at Q_i at the beginning of an intervisit period for that queue; while \mathbf{X}_i denotes the queue length at Q_i at the end of such an intervisit period. Introducing $\mathbf{N}_{i|I}$, a stochastic variable with distribution the queue length distribution at an arbitrary instant in an intervisit period of Q_i , we have from Lemma 2.2.1:

$$\mathbf{E}(y^{\mathbf{N}_{i|I}}) = \frac{\mathbf{E}(y^{\mathbf{Y}_i}) - \mathbf{E}(y^{\mathbf{X}_i})}{(1-y)(\mathbf{E}\mathbf{X}_i - \mathbf{E}\mathbf{Y}_i)}.$$
(3.8)

This relation appears in the polling literature for various special cases (e.g., for exhaustive vacation models, where $\mathbf{Y}_i \equiv 0$). It holds also for non-Poisson arrivals, when $\mathbf{N}_{i|I}$ is defined as the queue length that is observed by an arbitrary customer that arrives at Q_i during an intervisit period.

Together, formulae (3.7) and (3.8) yield the well-known Fuhrmann-Cooper queue length decomposition [103], applied to a queue in a polling model with or without switch-over times: for $|y| \leq 1$,

$$\mathbf{E}(y^{\mathbf{N}_i}) = \mathbf{E}(y^{\mathbf{N}_{i|\mathcal{M}/\mathcal{G}/1}})\mathbf{E}(y^{\mathbf{N}_{i|\mathcal{I}}}).$$
(3.9)

Remark 3.3.1

Fuhrmann and Cooper [103] state five conditions under which their decomposition holds:

(i). Customers arrive at Q_i according to a Poisson process of rate λ_i .

(ii). All customers arriving at Q_i are eventually served.

(iii). Customers enter service in an order that is independent of their service times.

(iv). Service is non-preemptive.

(v). The rules that govern when the server begins and ends visit periods to Q_i , do not anticipate future jumps of the Poisson arrival process at Q_i .

These assumptions indeed hold in our polling model, and are implicitly used in the derivation of (3.9). The above proof, with as key steps (3.1), (3.3) and (3.8), in fact also holds for vacation models without a polling context. Note that the relations $1/\gamma = \lambda EC$ and (3.6) hold generally for queues with some vacation (intervisit) mechanism. We refer to Keilson & Servi [125] for another short proof of the Fuhrmann-Cooper decomposition.

The waiting-time LST at Q_i immediately follows from (3.9), when we assume that within each of the queues customers are served in order of arrival. Denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer. Denote by $\mathbf{W}_{i|M/G/1}$ the waiting time of an arbitrary customer in the 'corresponding' isolated M/G/1 queue of Q_i . By the distributional form of Little's law, cf. Keilson & Servi [125], similar to equation (2.9) in Section 2.2,

$$\mathbf{E}(\mathbf{e}^{-\omega \mathbf{W}_i}) = \mathbf{E}(\mathbf{e}^{-\omega \mathbf{W}_i|M/G/1}) \mathbf{E}((1 - \omega/\lambda_i)^{\mathbf{N}_i|I}).$$
(3.10)

In Section 3.5, we shall return to this relation, for the case of polling models that satisfy Property 3.1.1.

3.4 The joint queue length distribution at polling epochs

In the previous section, we have seen that Eisenberg's results [84] yield simple relations between the pgf $M_i(z)$ of the joint queue length vector at service completion epochs (or $L_i(z)$, at service beginning epochs) and the pgf's $F_i(z)$ and $G_i(z)$ of the joint queue length vector at visit beginning and visit completion epochs. We now restrict ourselves to polling models for which the service discipline at each queue satisfies Property 3.1.1. Property 3.1.1 prescribes how each of the customers present at Q_i at the visit beginning is replaced by independent families of customers at its visit completion. This enables one to express $G_i(\cdot)$ nicely into $F_i(\cdot)$, and to finally determine each of the functions $F_i(\cdot)$ (after which the pgf's $G_i(\cdot)$, $M_i(\cdot)$ and $L_i(\cdot)$ follow). In our analysis we

3.4 The joint queue length distribution at polling epochs

follow Resing [159].

First some words on the ergodicity conditions. In the sequel, we assume that $\rho < 1$ and $s_i < \infty$ for all *i*. Resing [159] proves that for the subclass of socalled Bernoulli-type service disciplines, including exhaustive and gated service, cf. Section 1.3, these conditions together constitute sufficient ergodicity conditions. His proof is based on the observation that for these Bernoulli-type service disciplines the derivatives of $h_i(z_1, \ldots, z_n)$ take the form $\frac{\partial h_i(z)}{\partial z_i}|_{z=1} = \lambda_j \alpha_i \frac{\beta_i}{1-\rho_i}$, $i \neq j$, $\frac{\partial h_i(z)}{\partial z_i}|_{z=1} = 1 - \alpha_i$, with α_i some coefficient in (0, 1] determined by the parameters of Q_i . It may be easily verified, however, that the latter form of the derivatives applies for any non-idling service discipline that satisfies Property 3.1.1 with $h_i(z_1, \ldots, z_n) \neq z_i$. Since the proof in Resing [159] further does not rely on the specific form of α_i , we may conclude that $\rho < 1$ and $s_i < \infty$ for all *i* together constitute sufficient ergodicity conditions for any non-idling service discipline that satisfies Property 3.1.1 with $h_i(z_1, \ldots, z_n) \neq z_i$. Property 3.1.1 implies that

$$G_i(z) = F_i(z_1, \dots, z_{i-1}, h_i(z), z_{i+1}, \dots, z_n).$$
(3.11)

In the case of gated service, $h_i(z)$ is simply the pgf of the joint distribution of the numbers of arrivals at all queues during one service time at Q_i : $h_i(z) = \sum_{i=1}^{n} \frac{1}{2} \sum_{i=1}^{n} \frac{1}{$

$$\beta_i(\sum_{j=1}^n \lambda_j(1-z_j)).$$

In the case of exhaustive service, $h_i(z) = \eta_i(\sum_{j \neq i} \lambda_j(1-z_j))$, with $\eta_i(\cdot)$ the LST of the length of the bury period in a 'componenting' isolated M/C/1 groups

of the length of the busy period in a 'corresponding' isolated M/G/1 queue of Q_i .

Next we relate $F_i(z)$ to $G_{i-1}(z)$.

In the case of non-zero switch-over times:

$$F_i(z) = G_{i-1}(z)\sigma_{i-1}(\sum_{j=1}^n \lambda_j(1-z_j)).$$
(3.12)

In the case of zero switch-over times (in the sequel we add a superscript 0 for that case, to distinguish its quantities from those for non-zero switch-over times):

$$F_i^0(z) = G_{i-1}^0(z), (3.13)$$

for i = 2, ..., n. The relation between $F_1^0(z)$ and $G_n^0(z)$ deserves special attention, because of our convention, mentioned in Section 3.2, concerning the behavior of the server when the system is empty. When all queues in the model with zero switch-over times become empty, S makes a full cycle, and subsequently stops right before Q_1 (all this requires zero time). When the first new customer arrives, S cycles along the queues to that customer. The consequence of this is that when the system is empty at the start of a visit to Q_1 , then the next visit to Q_1 does not take place until a customer has arrived. We can write

$$F_1^0(z) = G_n^0(z) - F_1^0(0)[1 - g^0(z)], (3.14)$$

$$g^0(z):=\sum_{i=1}^nrac{\lambda_i}{\lambda}z_i$$

The function $g^0(\cdot)$ represents the 'immigration process' of the multi-type branching process: it is the pgf of the arrival process of customers during periods in which the system is empty.

Remark 3.4.1

Although we shall sometimes find it convenient to concentrate on Q_1 , it should be noted that our convention for the position of S in an empty system does not affect the waiting-time and queue length distributions.

In fact, our convention slightly differs from that of Resing [159], who assumes that when the system is empty, S immediately stops right behind Q_1 , and hence

takes $g^0(z) = \sum_{i=1}^n \frac{\lambda_i}{\lambda} f_i(z)$. Our convention enables us to simultaneously apply

the theory of multi-type branching processes and Eisenberg's approach.

Substituting (3.11) into (3.12), respectively (3.13) and (3.14), we can relate $F_i(\cdot)$ to $F_{i-1}(\cdot)$. We distinguish between the two cases of zero and non-zero switch-over times. In both cases, the following branching functions play a crucial role, thus establishing the link between both cases. Define

$$f(z) := (f_1(z), \dots, f_n(z)), \tag{3.15}$$

with

$$f_i(z) := h_i(z_1, \dots, z_i, f_{i+1}(z), \dots, f_n(z))$$
(3.16)

for $|z_j| \leq 1, j = 1, ..., n$. This is the offspring pgf, the pgf of the joint distribution of the numbers of customers at the end of a cycle w.r.t. Q_1 that are descendants of a type-*i* customer. In this branching process setting, a descendant of some customer K is a customer that has arrived during the service time of K or of one of its descendants. Define

$$f^{(0)}(z) := z,$$

 $f^{(k)}(z) := f(f^{(k-1)}(z)), \qquad k \ge 1$

for $|z_j| \le 1, j = 1, ..., n$.

Case I: Zero switch-over times Substituting (3.11) into (3.13),

$$F_i^0(z) = F_{i-1}^0(z_1, \dots, z_{i-2}, h_{i-1}(z), z_i, \dots, z_n)$$
(3.17)

50

with

for i = 2, ..., n. Starting from (3.14) and (3.11) for i = n, and subsequently using (3.17) for i = n, n - 1, ..., 2, one obtains

$$F_1^0(z) = F_1^0(f(z)) - F_1^0(0)[1 - g^0(z)].$$
(3.18)

Iterating (3.18) yields

$$F_1^0(z) = 1 - F_1^0(0) \sum_{k=0}^{\infty} [1 - g^0(f^{(k)}(z))], \qquad (3.19)$$

with

$$F_1^0(0) = \left[1 + \sum_{k=0}^{\infty} [1 - g^0(f^{(k)}(0))]\right]^{-1},$$

the infinite sum being convergent when the ergodicity conditions are fulfilled.

Introducing, for $|z_j| \leq 1, j = 1, \ldots, n$,

$$H(z) := \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_i (1 - f_i^{(k)}(z)), \qquad (3.20)$$

we can write

$$F_1^0(z) = 1 - F_1^0(0) \sum_{k=0}^{\infty} \sum_{i=1}^n \frac{\lambda_i}{\lambda} (1 - f_i^{(k)}(z))$$

= $1 - F_1^0(0) H(z) / \lambda,$ (3.21)

with

$$F_1^0(0) = [1 + H(0)/\lambda]^{-1}.$$

Case II: Non-zero switch-over times Substituting (3.11) into (3.12),

$$F_{i}(z) = F_{i-1}(z_{1}, \dots, z_{i-2}, h_{i-1}(z), z_{i}, \dots, z_{n})\sigma_{i-1}(\sum_{j=1}^{n} \lambda_{j}(1-z_{j})). \quad (3.22)$$

Applying (3.22) n times (which corresponds to following the server during one full cycle w.r.t. Q_1),

$$F_1(z) = F_1(f(z))g(z), (3.23)$$

with

$$g(z) = \prod_{i=1}^{n} \sigma_i (\sum_{j=1}^{i} \lambda_j (1-z_j) + \sum_{j=i+1}^{n} \lambda_j (1-f_j(z))).$$

The function $g(\cdot)$ represents the 'immigration process' of this multi-type branching process: it is the pgf of the vector of all customers that either have arrived in the switch-over periods of the past cycle (measured w.r.t. Q_1), or are descendants of such customers.

Iterating (3.23) yields

$$F_{1}(z) = \prod_{k=0}^{\infty} g(f^{(k)}(z))$$

$$= \prod_{k=0}^{\infty} \prod_{i=1}^{n} \sigma_{i}(\sum_{j=1}^{i} \lambda_{j}(1 - f_{j}^{(k)}(z)) + \sum_{j=i+1}^{n} \lambda_{j}(1 - f_{j}^{(k+1)}(z))),$$
(3.24)

the infinite product being convergent when the ergodicity conditions are fulfilled.

From (3.21) and (3.24) we see that $F_1(z)$ as well as $F_1^0(z)$ is determined by $\sum_{j=1}^n \lambda_j (1 - f_j^{(k)}(z))$. For constant switch-over times, the connection becomes even closer, as (3.24) in that case reduces to a simpler expression:

$$F_1(z) =$$
 (3.25)

$$\exp\left[-\sum_{k=0}^{\infty}\sum_{i=1}^{n}s_{i}\left\{\sum_{j=1}^{i}\lambda_{j}(1-f_{j}^{(k)}(z))+\sum_{j=i+1}^{n}\lambda_{j}(1-f_{j}^{(k+1)}(z))\right\}\right].$$

Using (3.20), we can rewrite (3.25) into

$$F_1(z) = \exp[-sH(z) + \sum_{j=1}^n \lambda_j (1-z_j) \sum_{i=1}^{j-1} s_i].$$
(3.26)

3.5 MARGINAL QUEUE LENGTHS AND WAITING TIMES

In the previous section, the queue length pgf's $F_i(z)$ and $F_i^0(z)$ at visit beginning instants have been determined for the class of cyclic polling models in which Property 3.1.1 holds for all service disciplines. In Section 3.3, we already obtained a decomposition for the pgf of the marginal queue length distribution at Q_i , and for the waiting-time LST at Q_i , into a corresponding M/G/1 term and a term involving $E(y^{\mathbf{X}_i})$ and $E(y^{\mathbf{Y}_i})$ (via the pgf $E(y^{\mathbf{N}_{i|I}})$). In particular, denoting

$$\begin{split} \tilde{h}_i(y) &:= h_i(1, \dots, 1, y, 1, \dots, 1); \\ \tilde{F}_i(y) &:= F_i(1, \dots, 1, y, 1, \dots, 1); \\ \tilde{F}_i^0(y) &:= F_i^0(1, \dots, 1, y, 1, \dots, 1), \end{split}$$

3.5 Marginal queue lengths and waiting times

with y as *i*-th argument, it follows from (3.8) and (3.11) for the case of non-zero switch-over times that

$$\mathbf{E}(y^{\mathbf{N}_{i|I}}) = \frac{\tilde{F}_i(\tilde{h}_i(y)) - \tilde{F}_i(y)}{(1-y)\tilde{F}_i'(1)(1-\tilde{h}_i'(1))};$$
(3.27)

the same result holds for the case of zero switch-over times, replacing $\tilde{F}_i(\cdot)$ by $\tilde{F}_i^0(\cdot)$ in (3.27). Similarly indicating queue lengths, and waiting times, by a superscript 0 in the case of zero times, we find from (3.9) and (3.10):

$$E(y^{\mathbf{N}_{i}}) = E(y^{\mathbf{N}_{i}^{0}}) \frac{[\tilde{F}_{i}(\tilde{h}_{i}(y)) - \tilde{F}_{i}(y)]\tilde{F}_{i}^{0\prime}(1)}{[\tilde{F}_{i}^{0}(\tilde{h}_{i}(y)) - \tilde{F}_{i}^{0}(y)]\tilde{F}_{i}^{\prime}(1)},$$
(3.28)

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}_{i}}) = \mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}_{i}}) \frac{[\tilde{F}_{i}(\tilde{h}_{i}(1-\omega/\lambda_{i})) - \tilde{F}_{i}(1-\omega/\lambda_{i})]\tilde{F}_{i}^{0\prime}(1)}{[\tilde{F}_{i}^{0}(\tilde{h}_{i}(1-\omega/\lambda_{i})) - \tilde{F}_{i}^{0}(1-\omega/\lambda_{i})]\tilde{F}_{i}^{\prime}(1)}.$$
 (3.29)

For exhaustive service, $\tilde{h}_i(\cdot) \equiv 1$; for gated service, $\tilde{h}_i(y) = \beta_i(\lambda_i(1-y))$. Differentiating (3.10) and (3.29) once, putting $\omega = 0$,

$$\mathbf{E}\mathbf{W}_{i} = \frac{\lambda_{i}\beta_{i}^{(2)}}{2(1-\lambda_{i}\beta_{i})} + \frac{\tilde{F}_{i}^{\prime\prime}(1)}{2\tilde{F}_{i}^{\prime}(1)\lambda_{i}}(1+\tilde{h}_{i}^{\prime}(1)) - \frac{\tilde{h}_{i}^{\prime\prime}(1)}{2(1-\tilde{h}_{i}^{\prime}(1))\lambda_{i}},$$
(3.30)

and

$$\mathbf{EW}_{i} - \mathbf{EW}_{i}^{0} = \left[\frac{\tilde{F}_{i}^{\prime\prime}(1)}{2\tilde{F}_{i}^{\prime}(1)\lambda_{i}} - \frac{\tilde{F}_{i}^{0\prime\prime}(1)}{2\tilde{F}_{i}^{0\prime}(1)\lambda_{i}}\right](1 + \tilde{h}_{i}^{\prime}(1)).$$
(3.31)

Let us now (without loss of generality) concentrate on \mathbf{W}_1 and \mathbf{W}_1^0 . Denoting $\tilde{f}_i^{(k)}(y) := f_i^{(k)}(y, 1, \dots, 1)$

(the k-th generation type-1 customer offspring of a type-i customer), it follows from (3.21) that

$$\tilde{F}_1^0(y) = 1 - F_1^0(0)\tilde{H}(y)/\lambda, \qquad (3.32)$$

with

$$\tilde{H}(y) := \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_i (1 - \tilde{f}_i^{(k)}(y)).$$
(3.33)

In particular, $\tilde{F}_1^{0\prime}(1) = -F_1^0(0)\tilde{H}'(1)/\lambda$. It follows from (3.24) and (3.33) that $\tilde{F}_1'(1) = -s\tilde{H}'(1)$, and hence

$$\tilde{F}_1'(1) = \tilde{F}_1^{0'}(1)s\lambda/F_1^0(0).$$
(3.34)

From (3.29), (3.32), (3.34),

Corollary 3.5.1

$$E(e^{-\omega \mathbf{W}_{1}}) = E(e^{-\omega \mathbf{W}_{1}^{0}}) \frac{\tilde{F}_{1}(\tilde{h}_{1}(1-\omega/\lambda_{1})) - \tilde{F}_{1}(1-\omega/\lambda_{1})}{s[\tilde{H}(\tilde{h}_{1}(1-\omega/\lambda_{1})) - \tilde{H}(1-\omega/\lambda_{1})]},$$
(3.35)

which for exhaustive service $(\tilde{h}_1(\cdot) \equiv 1)$ and gated service $(\tilde{h}_1(1 - \omega/\lambda_1) = \beta_1(\omega))$, corresponds to Theorems 2 and 5 in Srinivasan et al. [169], respectively.

For constant switch-over times, it follows from (3.26) that

$$\tilde{F}_1(y) = \exp[-s\tilde{H}(y)], \qquad (3.36)$$

and hence we have from (3.35)

Corollary 3.5.2

$$E(e^{-\omega \mathbf{W}_{1}}) =$$

$$E(e^{-\omega \mathbf{W}_{1}}) \frac{\exp[-s\tilde{H}(\tilde{h}_{1}(1-\omega/\lambda_{1}))] - \exp[-s\tilde{H}(1-\omega/\lambda_{1})]}{s[\tilde{H}(\tilde{h}_{1}(1-\omega/\lambda_{1})) - \tilde{H}(1-\omega/\lambda_{1})]},$$
(3.37)

which for exhaustive service $(\tilde{h}_1(\cdot) \equiv 1)$ and gated service $(\tilde{h}_1(1 - \omega/\lambda_1) = \beta_1(\omega))$, reduces to Theorems 3 and 6 in Srinivasan et al. [169], respectively.

Remark 3.5.1 Substituting $\sigma_i(\omega) = 1 - s_i\omega + o(s_i)$ for $s_i \to 0, i = 1, ..., n$, in (3.24) yields that

$$\begin{split} \tilde{F}_1(y) &= \\ 1 - \sum_{i=1}^n s_i \sum_{k=0}^\infty \left[\sum_{j=1}^i \lambda_j (1 - \tilde{f}_j^{(k)}(y)) + \sum_{j=i+1}^n \lambda_j (1 - \tilde{f}_j^{(k+1)}(y)) \right] + o(s) = \\ 1 - s \tilde{H}(y) + o(s). \end{split}$$

Hence, for $s \to 0$ in (3.35), \mathbf{W}_1 approaches \mathbf{W}_1^0 in distribution. Using (3.21) and (3.29), the same statement follows for other service disciplines satisfying Property 3.1.1.

3.6 Computational aspects

In this last section, we describe how the waiting-time results obtained can be exploited for a very efficient numerical calculation of the various moments under different switch-over time scenarios. For ease of presentation, we focus on the first moment, but similar observations hold for the higher moments.

To determine \mathbf{EW}_1 and \mathbf{EW}_1^0 according to formulae (3.30) and (3.31), we need to compute the quantities $\tilde{F}_1'(1)$, $\tilde{F}_1''(1)$, $\tilde{F}_1^{0'}(1)$, and $\tilde{F}_1^{0''}(1)$. The quantities $\tilde{h}_1'(1)$ and $\tilde{h}_1''(1)$ occurring in (3.30) and (3.31) may simply be determined from the service discipline at Q_1 .

We first introduce some notation. Denote $\phi_i^{(k)} := \frac{d}{dy} \tilde{f}_i^{(k)}(y)|_{y=1}, \ \psi_i^{(k)} := \frac{d^2}{dy^2} \tilde{f}_i^{(k)}(y)|_{y=1}, \ i = 1, \dots, n, \ k = 0, \dots, \infty.$ Define $\Phi_i := \sum_{k=0}^{\infty} \phi_i^{(k)}, \ \Psi_i := \sum_{k=0}^{\infty} \psi_i^{(k)}, \ i = 1, \dots, n, \ \Phi := \sum_{i=1}^n \lambda_i \Phi_i, \ \Psi := \sum_{i=1}^n \lambda_i \Psi_i.$ Differentiating (3.32),

$$\tilde{F}_{1}^{0'}(1) = \frac{F_{1}^{0}(0)}{\lambda} \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_{i} \phi_{i}^{(k)} = \frac{F_{1}^{0}(0)}{\lambda} \Phi; \qquad (3.38)$$

$$\tilde{F}_{1}^{0''}(1) = \frac{F_{1}^{0}(0)}{\lambda} \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_{i} \psi_{i}^{(k)} = \frac{F_{1}^{0}(0)}{\lambda} \Psi.$$
(3.39)

Taking z = (y, 1, ..., 1) in (3.24), differentiating w.r.t. to y,

$$\tilde{F}'_{1}(1) = s \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_{i} \phi_{i}^{(k)} = s \Phi; \qquad (3.40)$$

$$\tilde{F}_{1}^{\prime\prime}(1) = \left(\tilde{F}_{1}^{\prime}(1)\right)^{2} + s \sum_{k=0}^{\infty} \sum_{i=1}^{n} \lambda_{i} \psi_{i}^{(k)}$$
(3.41)

$$+ \sum_{k=0}^{\infty} \sum_{i=1}^{n} \left(s_{i}^{(2)} - s_{i}^{2} \right) \left(\sum_{j=1}^{i} \lambda_{j} \phi_{j}^{(k)} + \sum_{j=i+1}^{n} \lambda_{j} \phi_{j}^{(k+1)} \right)^{2}$$

= $\left(\tilde{F}_{1}^{\prime}(1) \right)^{2} + s \Psi + \sum_{i=1}^{n} \left(s_{i}^{(2)} - s_{i}^{2} \right) \chi_{i},$

with $\chi_i = \sum_{k=0}^{\infty} \left(\sum_{j=1}^{i} \lambda_j \phi_j^{(k)} + \sum_{j=i+1}^{n} \lambda_j \phi_j^{(k+1)} \right)^{-1}$. From (3.30), (3.31), (3.38)-(3.41), we obtain

$$\mathbf{EW}_{1}^{0} = \frac{\lambda_{1}\beta_{1}^{(2)}}{2(1-\lambda_{1}\beta_{1})} + \frac{2\Psi}{\Phi\lambda_{1}}(1+\tilde{h}_{1}'(1)) - \frac{\tilde{h}''(1)}{2(1-\tilde{h}_{1}'(1))\lambda_{1}},$$
(3.42)

$$\mathbf{EW}_{1} - \mathbf{EW}_{1}^{0} = \frac{(s\Phi)^{2} + \sum_{i=1}^{\infty} (s_{i}^{(2)} - s_{i}^{2})\chi_{i}}{s\Phi\lambda_{1}} (1 + \tilde{h}_{1}'(1)).$$
(3.43)

Note that the coefficients Φ , Ψ , and χ_i , $i = 1, \ldots, n$, are completely *independent* of the switch-over times. Apparently, the presence of switch-over times *always* increases the mean waiting times. That might not be very surprising, but it should be noted that the (perhaps also intuitively appealing) claim that the mean waiting times are increasing in the mean switch-over times is known to be false (cf. Sarkar & Zangwill [160]). We see from (3.43) that, for given s, the mean waiting times are increasing in the *variance* of each of the individual switch-over times, cf. Sarkar & Zangwill [160], in particular being minimal for constant switch-over times. In the latter case, (3.43) reduces to

$$\mathbf{EW}_{1} - \mathbf{EW}_{1}^{0} = \frac{s\Phi}{\lambda_{1}} (1 + \tilde{h}_{1}'(1)), \qquad (3.44)$$

which illustrates once more that the relationship between the models with and without switch-over times takes a remarkably simple form for constant switchover times, cf. (3.37). The mean waiting times then increase linearly with s, and for given s, they are insensitive to the individual s_i 's, as observed earlier by Fuhrmann [101] and Cooper et al. [79] for exhaustive and gated service. In this respect, it is interesting to mention some results for a two-queue model with exhaustive service at Q_1 and 1-limited service at Q_2 ; (cf. Groenendijk [114]; note that the branching property is now violated): for constant switch-over times, the waiting-time distributions are still insensitive to the individual s_i 's, given s; but the linearity as displayed in (3.44) no longer holds.

Finally, we observe from (3.43) that, once the coefficients Φ , Ψ and χ_i , $i = 1, \ldots, n$ have been calculated, the mean waiting times may be determined for different switch-over time scenarios with very little additional effort, cf. also Srinivasan et al. [169]. Those quantities Φ , Ψ , and χ_i , $i = 1, \ldots, n$, may be computed as follows. Taking $z = (y, 1, \ldots, 1)$ in (3.16),

$$\tilde{f}_i^{(k+1)}(y) = h_i(\tilde{f}_1^{(k)}(y), \dots, \tilde{f}_i^{(k)}(y), \tilde{f}_{i+1}^{(k+1)}(y), \dots, \tilde{f}_n^{(k+1)}(y)).$$
(3.45)

Define \mathbf{T}_i as the visit time at Q_i generated by an arbitrary type-*i* customer present at the start of a visit to Q_i . Then $\frac{\partial}{\partial z_j} h_i(z)|_{z=1} = \lambda_j \mathbf{ET}_i$, $\frac{\partial^2}{\partial z_j \partial z_i} h_i(z)|_{z=1}$ $= \lambda_j \lambda_l \mathbf{E}(\mathbf{T}_i^2), \ j \neq i, \ l \neq i$. Define \mathbf{V}_i as the total visit time at Q_i in a cycle. From $\mathbf{EV}_i = \mathbf{EX}_i \mathbf{ET}_i$, $\mathbf{EX}_i = \mathbf{EY}_i + \lambda_i (1 - \rho_i) \mathbf{EC}$, $\mathbf{EV}_i = \rho_i \mathbf{EC}$, $\mathbf{EY}_i =$ $\mathbf{EX}_i \frac{\partial}{\partial z_i} h_i(z)|_{z=1}$, it immediately follows that $\frac{\partial}{\partial z_i} h_i(z)|_{z=1} = (\lambda_i - 1/\beta_i) \mathbf{ET}_i + 1$. Thus, differentiating (3.45),

$$\phi_{i}^{(k+1)} = \sum_{j=1}^{i} \frac{\partial}{\partial z_{j}} h_{i}(z)_{|z=1} \phi_{j}^{(k)} + \sum_{j=i+1}^{n} \frac{\partial}{\partial z_{j}} h_{i}(z)_{|z=1} \phi_{j}^{(k+1)} \quad (3.46)$$

$$= \mathbf{ET}_{i} \left[\sum_{j=1}^{i} \lambda_{j} \phi_{j}^{(k)} + \phi_{i}^{(k)} / \mathbf{ET}_{i} - \phi_{i}^{(k)} / \beta_{i} + \sum_{j=i+1}^{n} \lambda_{j} \phi_{j}^{(k+1)} \right];$$

$$\psi_{i}^{(k+1)} = \sum_{j=1}^{i} \frac{\partial}{\partial z_{j}} h_{i}(z)_{|z=1} \psi_{j}^{(k)} + \sum_{j=i+1}^{n} \frac{\partial}{\partial z_{j}} h_{i}(z)_{|z=1} \psi_{j}^{(k+1)} \quad (3.47)$$

3.6 Computational aspects

$$\begin{split} + & \sum_{j=1}^{i} \sum_{l=1}^{i} \frac{\partial^{2}}{\partial z_{j} \partial z_{l}} h_{i}(z)|_{z=1} \phi_{j}^{(k)} \phi_{l}^{(k)} \\ + & 2 \sum_{j=1}^{i} \sum_{l=i+1}^{n} \frac{\partial^{2}}{\partial z_{j} \partial z_{l}} h_{i}(z)|_{z=1} \phi_{j}^{(k)} \phi_{l}^{(k+1)} \\ + & \sum_{j=i+1}^{n} \sum_{l=i+1}^{n} \frac{\partial^{2}}{\partial z_{j} \partial z_{l}} h_{i}(z)|_{z=1} \phi_{j}^{(k+1)} \phi_{l}^{(k+1)} \\ = & \operatorname{ET}_{i} \left[\sum_{j=1}^{i} \lambda_{j} \psi_{j}^{(k)} + \frac{\psi_{i}^{(k)}}{\operatorname{ET}_{i}} - \frac{\psi_{i}^{(k)}}{\beta_{i}} + \sum_{j=i+1}^{n} \lambda_{j} \psi_{j}^{(k+1)} \right] \\ + & \operatorname{E}(\operatorname{T}_{i}^{2}) \left(\sum_{j=1}^{i} \lambda_{j} \phi_{j}^{(k)} + \sum_{j=i+1}^{n} \lambda_{j} \phi_{j}^{(k+1)} \right)^{2} \\ + & \phi_{i}^{(k)} \left[2 \sum_{j=1}^{i-1} \tau_{ij} \phi_{j}^{(k)} + \tau_{ii} \phi_{i}^{(k)} + 2 \sum_{j=i+1}^{n} \tau_{ij} \phi_{j}^{(k+1)} \right], \end{split}$$

with $\tau_{ij} := \frac{\partial^2}{\partial z_i z_j} h_i(z)_{|z=1} - \lambda_i \lambda_j E(\mathbf{T}_i^2)$. Summing (3.46) and (3.47) over $k = 0, \dots, \infty$, we obtain

$$\frac{\Phi_i}{\beta_i} - \frac{\phi_i^{(0)}}{\mathbf{ET}_i} + \sum_{j=i+1}^n \lambda_j \phi_j^{(0)} = \Phi,$$
(3.48)

and

$$\frac{\Psi_i}{\beta_i} - \frac{\psi_i^{(0)} + \mathcal{E}(\mathbf{T}_i^2)\chi_i + \xi_i}{\mathbf{ET}_i} + \sum_{j=i+1}^n \lambda_j \psi_j^{(0)} = \Psi, \qquad (3.49)$$

with $\xi_i = \sum_{k=0}^{\infty} \phi_i^{(k)} \left[2 \sum_{j=1}^{i-1} \tau_{ij} \phi_j^{(k)} + \tau_{ii} \phi_i^{(k)} + 2 \sum_{j=i+1}^n \tau_{ij} \phi_j^{(k+1)} \right].$ Multiplying (3.48) and (3.49) by ρ_i , summing over $i = 1, \dots, n$, using $\phi_1^{(0)} = 1$, $\phi_i^{(0)} = 0, i = 2, \dots, n, \ \psi_i^{(0)} = 0, i = 1, \dots, n$, we obtain

$$\Phi = \frac{\rho_1 / \mathrm{ET}_1}{1 - \rho},\tag{3.50}$$

and

$$\Psi = \frac{\sum_{i=1}^{n} \rho_i \left[\mathbf{E}(\mathbf{T}_i^2) \chi_i + \xi_i \right] / \mathbf{E} \mathbf{T}_i}{1 - \rho}.$$
(3.51)

For exhaustive service, i.e., $\mathbf{ET}_i = \beta_i/(1-\rho_i)$, $\mathbf{E}(\mathbf{T}_i^2) = \beta_i^{(2)}/(1-\rho_i)^3$, $\tau_{ij} = -\lambda_i\lambda_j\mathbf{E}(\mathbf{T}_i^2)$, the expressions for χ_i and ξ_i , i = 1, ..., n, reduce to those obtained by Srinivasan et al. [169].

For constant switch-over times, we find from (3.43) and (3.50) that $\mathbf{EW}_1 - \mathbf{EW}_1^0 = (1 + \tilde{h}'_1(1))\mathbf{EC}\beta_1/(2\mathbf{ET}_1)$, which shows that the difference then only depends on the parameters of the other queues through $\mathbf{EC} = s/(1 - \rho)$. In particular, for exhaustive and gated service, $\mathbf{EW}_1 - \mathbf{EW}_1^0 = (1 - \rho_1)\mathbf{EC}/2$ and $\mathbf{EW}_1 - \mathbf{EW}_1^0 = (1 + \rho_1)\mathbf{EC}/2$, respectively, as obtained in [101] and [79] by using different techniques.

Note that the bulk of the computational effort is involved with the calculation of the quantities χ_i and ξ_i , i = 1, ..., n. The coefficients $\phi_i^{(k)}$, i = 1, ..., n, $k = 0, ..., \infty$, needed for these calculations, may be computed recursively from (3.46), supplemented with $\phi_1^{(0)} = 1$, $\phi_i^{(0)} = 0$, i = 2, ..., n. The number of elementary operations (additions, multiplications) involved is $O(n\log_{\rho}(\epsilon))$ with ϵ the level of accuracy desired. Once the quantities χ_i and ξ_i , i = 1, ..., n, have been calculated, EW₁ may be computed for any value of the switch-over times in O(n) elementary operations.

CHAPTER 4

A PSEUDO-CONSERVATION LAW FOR A POLLING SYSTEM WITH A DORMANT SERVER

4.1 INTRODUCTION

In the present chapter, we consider a polling system with a *dormant* server, i.e., a polling system in which the server may be allowed to make a halt at a queue when there are no customers present in the system. In the polling literature, the server is usually assumed never to idle, in other words, to be switching when not working. In particular, the server is assumed to be switching when there are no customers present in the system. As a rare exception, Eisenberg [83] considers a two-queue model with either alternating priority (the exhaustive service discipline at both queues) or strict priority, in which the server remains idling at a queue when there are no customers present in the system. Eisenberg [84] studies a model with an arbitrary number of queues and the exhaustive service discipline at all queues, in which the server does not idle. In a recent study [86], Eisenberg shows however how an adapted version of the solution method in [84] may be used to analyze a model in which the server makes a halt at some of the queues when the system is empty. The outline of the solution method in [86] may also be used to treat a similar model with the gated service discipline. Gersht & Marbukh [109] consider a two-queue model with two types of disciplines for switching from one queue to another. For both types of disciplines, they show that for some region of the system parameters the discipline that minimizes the mean waiting cost inserts forced idle periods. Liu, Nain, & Towsley [145] identify polling policies, allowing idling as a possible action, that stochastically minimize the total amount of work in the system at an arbitrary epoch. They show that optimal policies are exhaustive and greedy, i.e., the server should neither switch nor idle when at a non-empty queue. In addition, they prove that in symmetric polling systems patient policies are optimal, i.e., when the entire system is empty, the server should remain idling at the last visited queue. Gupta & Srinivasan [115] derive explicit expressions for the waiting-time distribution in a similar model as in [86] by using an approach based on the concept of "descendant sets". They show that, while a patient server policy is generally better in the sense of a reduction of the amount of work in the system, there do exist cases where a roving server strategy is better. Blanc & Van der Mei [24] use the power-series algorithm to analyze the performance of a system in which the server may be allowed to make a halt at a queue when the entire system is empty. They find that the performance may improve considerably by allowing the server to make a halt at a queue, especially in light traffic.

One reason why usually in the polling literature the server is nevertheless assumed never to idle, may be that the option of idling in general slightly complicates the operation of the system. If at all technically feasible, some mechanism is needed to control the server and to keep track of the customers present in the system. Consequently, the option of idling in general also slightly complicates the mathematical analysis of the system. Another reason may be that the option of idling will have the biggest impact in light traffic, when the performance will be satisfactory anyhow.

However, quite often there are very sound reasons for letting the server stop switching when there are no customers present in the system. In many situations, some mechanism to control the server and to keep track of the customers present in the system is needed anyhow. The option of idling then arises quite naturally. In manufacturing and maintenance environments, e.g., one usually requires already some kind of supporting system to schedule the jobs. In such situations, it makes sense to let the server make a halt at a queue when the entire system is empty, rather than to let the server needlessly circle around. One option is then allowing the server to make a halt at all of the queues, i.e., to stop switching as soon as the entire system is empty. Another option is allowing the server only to make a halt at some of the queues (thus possibly forcing the server to keep switching for a while), e.g. at a queue which functions as home base, or at the queue where a new customer is most likely to arrive. The latter option may be recognized in the dynamic control of traffic lights. When there are no vehicles waiting, typically the main stream is given passage, until a waiting vehicle of a crossing stream is detected.

In many situations, there are moreover significant cost involved in switching. In manufacturing and maintenance applications, the switch-over usually represents the change-over from one type of jobs to another, which may involve labor cost, material cost, or transportation cost. In such situations, a potential saving in switching cost is an additional reason for letting the server stop switching when there are no customers present in the system.

Apart from the practical relevance, it is theoretically interesting to gain some insight into the effect of idling. In the present chapter, we therefore derive a pseudo-conservation law for a general model, permitting a wide variety of service disciplines, in which the server may be allowed to make a halt at an

4.2 Model description

arbitrary subset of the queues. A pseudo-conservation law provides a relatively simple expression for a specific weighted sum of the mean waiting times, cf. Section 2.3. By its comparative simplicity, a pseudo-conservation law is likely to provide some insight into the effect of idling, whereas the individual mean waiting times themselves involve expressions far too complicated to do so. Linked up with this, the determination of the individual mean waiting times, if at all possible, requires an intricate analysis, and relies substantially on the features of the model under consideration, whereas the derivation of a pseudo-conservation law in fact only demands the calculation of mean working/idling times, and only marginally leans on the characteristics of the model under consideration.

The remainder of the chapter is organized as follows. In Section 4.2, we present a detailed model description. In Section 4.3, we derive a pseudo-conservation law for the model under consideration. We use the pseudo-conservation law in Section 4.4 to compare the dormant and the non-dormant server case. Furthermore, we address the question at which queues the server should make a halt to minimize the mean total amount of work in the system.

4.2 MODEL DESCRIPTION

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by a single server S. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3.

The server visits the queues in strictly cyclic order, Q_1, \ldots, Q_n . As soon as the server arrives at Q_i , it starts serving type-*i* customers (possibly none), as prescribed by the service discipline. For now, we do not specify the service disciplines at the various queues. We only demand the service disciplines at the various queues to be non-preemptive and work-conserving, i.e., no work is created or destroyed, cf. Boxma [38]. As soon as the server has finished serving type-*i* customers (possibly none), as prescribed by the service discipline, it departs from Q_i . However, we put in the proviso that the server may be allowed to make a halt at Q_i when there are no customers present in the system. The server then remains idling at Q_i , awaiting a new customer to arrive at one of the queues. For now, we do not specify the additional criteria for deciding when to make a halt. In case of the 1-limited service discipline e.g., it seems to make sense only to make a halt at a queue when the server has not served any customer yet. As soon as a new customer arrives, the server resumes its activities. If the new customer arrives at Q_i , and if the service discipline permits to do so, then the server immediately starts serving the newly arrived customer. Otherwise, the server immediately starts switching to Q_{i+1} . The newly arrived customer then does not need to be served first, as, on its way to the newly arrived customer, the server may encounter other customers at other queues.

The case in which the server makes a halt at none of the queues when there are no customers present in the system, will also be referred to as the non-dormant server case. The case in which the server makes a halt at all of the queues when there are no customers present in the system, will also be referred to as the purely dormant server case.

As a particular variant of a polling system with a dormant server, we focus in the next chapter on a globally gated polling system, in which the server only makes a halt at its home base. In Eisenberg [86], the latter stopping convention is referred to as the "Continue-Cycle-To-Home-Base" rule, as opposed to the "Jump-Directly-To-Home-Base" rule, where the server, on emptying the system, executes a single change-over that takes it directly to the home base. In the purely dormant server case, Eisenberg [86] speaks of the "Stop Immediately" rule. Analogously, the starting convention that we adopted here is referred to as the "Resume-Cycle" rule, as opposed to the "Jump-Directly-To-New-Arrival" rule, where the server executes a single change-over that takes it directly to the queue receiving the new arrival.

Finally some words on the stability conditions. We claim – without formal proof – that the opportunity to idle when the system is empty does not affect the stability conditions as discussed for the ordinary non-dormant server case in Section 1.3. The reason is that if the stability conditions were violated, then the system would be empty with probability 0, so that also the opportunity to idle would arise only with probability 0. Throughout the chapter, the stability conditions are assumed to hold.

4.3 A PSEUDO-CONSERVATION LAW

In this section, we derive a pseudo-conservation law for the model under consideration. In the next section, we will use the pseudo-conservation law to compare the dormant and the non-dormant server case.

We first introduce some notation. Denote by π_i the probability that at an arbitrary epoch the server is idling at Q_i , i = 1, ..., n. If the server is not allowed to make a halt at Q_i , then of course $\pi_i \equiv 0$. In general, the probabilities π_i are not simply known. For the exhaustive and gated service discipline, they can be determined along the lines of [86]. For the Bernoulli service discipline (comprising the exhaustive and 1-limited service discipline as extreme cases), they can be determined numerically using the power-series algorithm, cf. [24]. The total probability that at an arbitrary epoch the server is idling is $\pi := \sum_{i=1}^{n} \pi_i$. Denote by \mathbf{C}_i the cycle time with respect to Q_i , i.e., the time between two successive polling epochs at Q_i , $i = 1, \ldots, n$. Although the distribution of \mathbf{C}_i in general depends on i, EC_i obviously does not. Noting that the server is

62
4.3 A pseudo-conservation law

working a fraction ρ of the time and idling a fraction π of the time,

$$EC_i = \frac{s}{1 - \rho - \pi}, \qquad i = 1, \dots, n.$$
 (4.1)

Denote by \mathbf{V}_i the total time that the server is working at Q_i during a cycle, $i = 1, \ldots, n$. As ρ_i is the fraction of time that the server is working at Q_i , using (4.1),

$$\mathbf{EV}_i = \frac{\rho_i s}{1 - \rho - \pi}, \qquad i = 1, \dots, n.$$
(4.2)

Denote by I_i the total time that the server is idling at Q_i during a cycle, i = 1, ..., n. As π_i is the fraction of time that the server is idling at Q_i , using (4.1),

$$EI_i = \frac{\pi_i s}{1 - \rho - \pi}, \qquad i = 1, \dots, n.$$
 (4.3)

We now derive the pseudo-conservation law for the model under consideration. The approach is similar to the approach in Boxma & Groenendijk [42] for the ordinary non-dormant server case. It is easily verified that the model under consideration satisfies the assumptions mentioned in Boxma [38]. Hence, as described in Section 2.3, the mean waiting times in the model under consideration satisfy the following relationship:

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \mathbf{E} \mathbf{Y},$$
(4.4)

with **Y** denoting the amount of work in the system at an arbitrary epoch in a non-serving interval, i.e., a switching interval or an idling interval. We now determine **EY**, by distinguishing whether the server is switching or idling and conditioning on the type of switching interval. Denote by \mathbf{Y}_i the total amount of work in the system at an arbitrary epoch in a switching interval from Q_i to Q_{i+1} , $i = 1, \ldots, n$. By definition, there are no customers present in the system when the server is idling, in other words, the total amount of work in the system at an arbitrary epoch in an idling interval is zero. Hence, with $\mathbf{EI} := \sum_{i=1}^{n} \mathbf{EI}_i$,

$$\mathbf{E}\mathbf{Y} = \sum_{i=1}^{n} \frac{s_i}{s + \mathbf{E}\mathbf{I}} \mathbf{E}\mathbf{Y}_i.$$
(4.5)

 \mathbf{Y}_i is composed of two terms, viz.:

i. \mathbf{Z}_i , the total amount of work in the system at the beginning of the switchover from Q_i to Q_{i+1} ;

ii. the total amount of work that arrives in the system between the beginning

of the switch-over from Q_i to Q_{i+1} and the epoch under consideration; so

$$\mathbf{E}\mathbf{Y}_{i} = \mathbf{E}\mathbf{Z}_{i} + \rho \frac{s_{i}^{(2)}}{2s_{i}}, \qquad i = 1, \dots, n.$$

$$(4.6)$$

We now fragment \mathbf{Z}_i further into work that accumulated at different queues during different periods. Denote by \mathbf{Z}_{ij} the amount of work at Q_j at the beginning of a switch-over from Q_i to Q_{i+1} , $i = 1, \ldots, n$, $j = 1, \ldots, n$. Then

$$\mathbf{E}\mathbf{Z}_{i} = \sum_{j=1}^{n} \mathbf{E}\mathbf{Z}_{ij}, \qquad i = 1, \dots, n.$$
(4.7)

 \mathbf{Z}_{ij} is composed of two terms, viz.:

i. \mathbf{Z}_{jj} , the amount of work at Q_j at the beginning of the switch-over from Q_j to Q_{j+1} ;

ii. the amount of work that arrives at Q_j between the beginning of the switchover from Q_j to Q_{j+1} and the beginning of the switch-over from Q_i to Q_{i+1} ; so, using (4.2) and (4.3),

$$\mathbf{E}\mathbf{Z}_{ij} = \mathbf{E}\mathbf{Z}_{jj} + \rho_j \sum_{k=j+1}^{i} \left(s_{k-1} + \frac{(\rho_k + \pi_k)s}{1 - \rho - \pi} \right), \qquad i \neq j, \tag{4.8}$$

where the summation is to be interpreted cyclically.

Substituting (4.6), (4.7), and (4.8) into (4.5), noting that $\frac{s}{s + \text{EI}} = \frac{1 - \rho - \pi}{1 - \rho}$,

$$EY = \frac{s}{1-\rho} \sum_{i=1}^{n} \sum_{j=1}^{i-1} \rho_i \rho_j + \frac{\rho(1-\rho-\pi)}{(1-\rho)s} \left[\sum_{i=1}^{n} \sum_{j=1}^{i-1} s_i s_j + \frac{1}{2} \sum_{i=1}^{n} s_i^{(2)} \right] \\ + \frac{1}{1-\rho} \sum_{i=1}^{n} \sum_{j\neq i} \sum_{k=j+1}^{i} s_i \rho_j \pi_k + \frac{1-\rho-\pi}{1-\rho} \sum_{i=1}^{n} EZ_{ii} \\ = \frac{s}{2(1-\rho)} \left[\rho^2 - \sum_{i=1}^{n} \rho_i^2 \right] + \frac{1-\rho-\pi}{1-\rho} \rho \frac{s^{(2)}}{2s}$$

$$+ \frac{1}{1-\rho} \sum_{i=1}^{n} \sum_{j\neq i} \sum_{k=j+1}^{i} s_i \rho_j \pi_k + \frac{1-\rho-\pi}{1-\rho} \sum_{i=1}^{n} EZ_{ii}.$$
(4.9)

Substituting (4.9) into (4.4), we obtain the following relationship:

4.3 A pseudo-conservation law

Theorem 4.3.1

The mean waiting times in the model under consideration satisfy the following relationship:

$$\sum_{i=1}^{n} \rho_{i} \mathbf{E} \mathbf{W}_{i} = \rho \frac{\sum_{i=1}^{n} \lambda_{i} \beta_{i}^{(2)}}{2(1-\rho)} + \frac{s}{2(1-\rho)} \left[\rho^{2} - \sum_{i=1}^{n} \rho_{i}^{2} \right] +$$

$$\frac{1-\rho-\pi}{1-\rho} \rho \frac{s^{(2)}}{2s} + \frac{1}{1-\rho} \sum_{i=1}^{n} \sum_{j\neq i} \sum_{k=j+1}^{i} s_{i} \rho_{j} \pi_{k} + \frac{1-\rho-\pi}{1-\rho} \sum_{i=1}^{n} \mathbf{E} \mathbf{Z}_{ii},$$
(4.10)

with $E\mathbf{Z}_{ii}$ the mean amount of work that is left behind by the server at Q_i .

We may interpret the terms in (4.10) as follows. The term in the left-hand side is the mean amount of waiting work in the system at an arbitrary epoch. The first term in the right-hand side is the mean amount of waiting work in the corresponding system without switch-over times at an arbitrary epoch. The remaining terms in the right-hand side reflect the influence of the switch-over times. Together these terms constitute the mean total amount of work in the system at an arbitrary epoch in a non-serving interval, i.e., a switching interval or an idling interval. The last term in the right-hand side represents the mean amount of work at an arbitrary epoch in a non-serving interval that is left behind by the server at the various queues. Together the second, third, and fourth term in the right-hand side constitute the mean total amount of work at an arbitrary epoch in a non-serving interval that arrived at the various queues since the server left those queues. Separately, the second, third, and fourth term represent the mean amount of work at an arbitrary epoch in a non-serving interval that arrived at the various queues during the working, switching, and idling intervals since the server left those queues, respectively. These terms do not depend on the service disciplines at the various queues, at least as far as their global structure is concerned; the probabilities π_i that occur in these terms probably do depend on the service disciplines at the various queues. Apparently, at least as far as the global structure of (4.10) is concerned, the last term in the right-hand side completely captures the influence of the service disciplines at the various queues.

Remember that we determined the terms in the right-hand side of (4.10), except the first term, by distinguishing whether the server is switching or idling and conditioning on the type of switching interval. The second, third, and fifth term may in fact also be determined without conditioning on the type of switching interval, the fourth term does not allow such an alternative determination.

The last term in the right-hand side of (4.10), representing the mean amount of work that is left behind by the server at the various queues, still remains to be specified. Obviously, $\mathbf{E}\mathbf{Z}_{ii}$ is influenced by the service discipline at Q_i . However, as a pleasing circumstance, $\mathbf{E}\mathbf{Z}_{ii}$ is in fact influenced by the service discipline at Q_i only, i.e., not by the service discipline at Q_j , $j \neq i$. On the basis of the service discipline at Q_i only, we can split $E\mathbf{Z}_{ii}$ into work that arrived during working, switching, and idling intervals, whose means do not depend on the service disciplines at the various queues, cf. (4.2) and (4.3).

We now determine \mathbf{EZ}_{ii} for the exhaustive (I), gated (II), 1-limited (III), and globally gated (IV) service discipline. We need to distinguish whether the server, when idling at Q_i , has already served a customer during its visit to Q_i or not. Denote by π'_i the probability that at an arbitrary epoch the server is idling at Q_i and has not served any customer yet, $i = 1, \ldots, n$. Denote by π''_i the probability that at an arbitrary epoch the server is idling at Q_i and has already served a customer, $i = 1, \ldots, n$. Neither π'_i nor π''_i are simply known, but of course $\pi_i \equiv \pi'_i + \pi''_i$. Let E, G, L, and GG represent the index set of the queues where the exhaustive, gated, 1-limited, and globally gated service discipline is used, respectively.

I. Exhaustive: S serves type-i customers until Q_i is empty. So,

$$\mathbf{E}\mathbf{Z}_{ii} = 0, \quad i \in E. \tag{4.11}$$

II. Gated: S serves exactly those type-*i* customers present at its arrival at Q_i . A customer that arrives when S is idling at Q_i and has not served any customer yet, is also served. So, using (4.2) and (4.3),

$$\mathbf{E}\mathbf{Z}_{ii} = \frac{\rho_i(\rho_i + \pi_i'')s}{1 - \rho - \pi}, \quad i \in G.$$

$$(4.12)$$

III. 1-Limited: S serves one type-*i* customer, provided there is a customer present at its arrival at Q_i . A customer that arrives when S is idling at Q_i and has not served any customer yet, is also served. Thus, S leaves behind the amount of work that arrives during the waiting time of the customer that is possibly served and during its visit to Q_i , but not during the possible idling period before S serves a customer. It follows from (4.1) that on average $\frac{\lambda_i s}{1-\rho-\pi}$ customers are served during a visit to Q_i . In particular, under the 1-limited service discipline, with probability $\frac{\lambda_i s}{1-\rho-\pi}$ a customer is served during a visit to Q_i . So, using (4.2) and (4.3),

$$\mathbf{E}\mathbf{Z}_{ii} = \rho_i \left(\frac{\lambda_i s}{1 - \rho - \pi} \mathbf{E}\mathbf{W}_i + \frac{(\rho_i + \pi_i'')s}{1 - \rho - \pi}\right), \quad i \in L.$$
(4.13)

IV. Globally gated: S serves exactly those type-*i* customers present at its most recent arrival at Q_1 . A customer that arrives when S is idling at Q_1 and has not served any customer yet, is also served. So, using (4.2) and (4.3),

$$\mathbf{E}\mathbf{Z}_{ii} = \rho_i \left(\frac{(\rho_1 + \pi_1'')s}{1 - \rho - \pi} + \sum_{j=2}^i \left(s_{j-1} + \frac{(\rho_j + \pi_j)s}{1 - \rho - \pi} \right) \right), \quad i \in GG.$$
(4.14)

4.3 A pseudo-conservation law

We now assume that at each queue either the exhaustive, gated, 1-limited, or globally gated service discipline is used, i.e., $E \cup G \cup L \cup GG = \{1, \ldots, n\}$. Substituting (4.11), (4.12), (4.13), and (4.14) into (4.10), we obtain the following pseudo-conservation law:

Theorem 4.3.2

The mean waiting times in the model under consideration satisfy the following relationship:

$$\sum_{i \in E, G, GG} \rho_i \mathbf{E} \mathbf{W}_i + \sum_{i \in L} \rho_i \left(1 - \frac{\lambda_i s}{1 - \rho} \right) \mathbf{E} \mathbf{W}_i =$$

$$\rho \frac{\sum_{i=1}^n \lambda_i \beta_i^{(2)}}{2(1 - \rho)} + \frac{s}{2(1 - \rho)} \left[\rho^2 - \sum_{i \in E} \rho_i^2 + \sum_{i \in G, L, GG} \rho_i^2 + 2 \sum_{i \in GG} \sum_{j=1}^{i-1} \rho_i \rho_j \right] +$$

$$\frac{1 - \rho - \pi}{1 - \rho} \left[\rho \frac{s^{(2)}}{2s} + \sum_{i \in GG} \sum_{j=1}^{i-1} \rho_i s_j \right] +$$

$$\frac{1}{1 - \rho} \left[\sum_{i=1}^n \sum_{j \neq i} \sum_{k=j+1}^i s_i \rho_j \pi_k + s \sum_{i \in G, L} \rho_i \pi_i'' + s \sum_{i \in GG} \rho_i (\pi_1'' + \sum_{j=2}^i \pi_j) \right].$$
(4.15)

Elaborating on the explanation of the terms in (4.10), we may interpret the terms in (4.15) as follows. The terms in the left-hand side, excluding the term $\sum_{i \in L} \rho_i \frac{\lambda_i s}{1-\rho} E \mathbf{W}_i$, together constitute the mean amount of waiting work in the

system at an arbitrary epoch. The first term in the right-hand side is the mean amount of waiting work in the corresponding system without switch-over times at an arbitrary epoch. The remaining terms in the right-hand side, and the term $\sum_{i \in L} \rho_i \frac{\lambda_i s}{1-\rho} \mathbf{EW}_i$, together constitute the mean total amount of work in the

system at an arbitrary epoch in a non-serving interval, i.e., a switching interval or an idling interval. Separately, the second, third, and fourth term represent the mean amount of work at an arbitrary epoch in a non-serving interval that arrived at the various queues during working, switching, and idling intervals, respectively, excluding the work that arrived at 1-limited queues before the server polled those queues for the last time.

Remark 4.3.1

It is easily verified that the mean waiting times in the model of Eisenberg [83] with the exhaustive service discipline at both queues indeed satisfy (4.15) with $n = 2, E = \{1, 2\}.$

For $\pi_i = 0, i = 1, ..., n$, (4.15) reduces to the pseudo-conservation law for the

ordinary non-dormant server case, cf. Boxma & Groenendijk [42].

For $E = \{1, \ldots, n\}$, (4.15) reduces to the pseudo-conservation law obtained by Gupta & Srinivasan [115] for a similar model with the exhaustive service discipline at all queues. In the latter study, the pseudo-conservation law is obtained as a by-product from the determination of the individual mean waiting times rather than by using a work decomposition approach.

Remark 4.3.2

Notice that the pseudo-conservation law is not as explicit in the dormant server case as in the ordinary non-dormant server case. In the dormant server case, to obtain the pseudo-conservation law explicitly, we are committed to an intricate analysis to determine the probabilities π_i , if at all possible. For the exhaustive and gated service discipline, they can be determined along the lines of [86]. For the globally gated service discipline, mathematically a most tractable service discipline, there is no obstacle to determining the probabilities π_i either. For the 1-limited service discipline, there is no method available for determining the probabilities π_i analytically. However, they can be determined numerically using the power-series algorithm, cf. [24].

Remark 4.3.3

Like in the ordinary non-dormant server case, cf. Groenendijk [113], a pseudoconservation law is useful for supporting approximations for the mean waiting times and for finding the exact mean waiting times in a symmetric system in a simple manner. Various approximations for the mean waiting times are conceivable, but we do not pursue this matter here any further.

Remark 4.3.4

In this section, we obtained a pseudo-conservation law for a model with cyclic polling and single Poisson arrivals. Without seriously complicating the above analysis, cyclic polling may be generalized to polling guided by a table, cf. [45], or Markovian polling, i.e., the visit order is governed by a Markov chain with state space $\{1, \ldots, n\}$, cf. [53], and single Poisson arrivals may be generalized to batch Poisson arrivals, cf. Boxma [38].

68

4.4 A COMPARISON BETWEEN THE DORMANT AND THE NON-DORMANT SERVER CASE

In the previous section we obtained a pseudo-conservation law for the model under consideration. In this section, we use the pseudo-conservation law to compare the dormant and the non-dormant server case. Specifically, we compare the mean waiting times in a symmetric system in the purely dormant and the non-dormant server case. Furthermore, we address the question at which queues the server should make a halt to minimize the mean total amount of work in the system.

Let us label the waiting times in the dormant and the non-dormant server case with a hat and a tilde respectively. From (4.15),

$$\begin{split} &\sum_{i \in E,G,GG} \rho_i(\mathbf{E}\hat{\mathbf{W}}_i - \mathbf{E}\tilde{\mathbf{W}}_i) + \sum_{i \in L} \rho_i \left(1 - \frac{\lambda_i s}{1 - \rho}\right) (\mathbf{E}\hat{\mathbf{W}}_i - \mathbf{E}\tilde{\mathbf{W}}_i) = (4.16) \\ &- \frac{\pi}{1 - \rho} \left[\rho \frac{s^{(2)}}{2s} + \sum_{i \in GG} \sum_{j=1}^{i-1} \rho_i s_j \right] + \\ &\frac{1}{1 - \rho} \left[\sum_{i=1}^n \sum_{j \neq i} \sum_{k=j+1}^i s_i \rho_j \pi_k + s \sum_{i \in G,L} \rho_i \pi_i'' + s \sum_{i \in GG} \rho_i (\pi_1'' + \sum_{j=2}^i \pi_j) \right] = \\ &- \frac{\rho \pi}{1 - \rho} \left\{ \left[\frac{s^{(2)}}{2s} + \sum_{i \in GG} \sum_{j=1}^{i-1} \frac{\rho_i}{\rho} s_j \right] - \right. \\ &\left[\sum_{i=1}^n \sum_{j \neq i} \sum_{k=j+1}^i s_i \frac{\rho_j}{\rho} \frac{\pi_k}{\pi} + s \sum_{i \in G,L} \frac{\rho_i}{\rho} \frac{\pi_i''}{\pi} + s \sum_{i \in GG} \frac{\rho_i}{\rho} (\frac{\pi_1''}{\pi} + \sum_{j=2}^i \frac{\pi_j}{\pi}) \right] \right\}, \end{split}$$

where π of course refers to the dormant server case. The interpretation of the terms in the left-hand side and in the first form of the right-hand side of (4.16) follows immediately from the interpretation of the terms in (4.15). The terms in the left-hand side, excluding the term $\sum_{i \in L} \rho_i \frac{\lambda_i s}{1-\rho} (\mathbf{E} \hat{\mathbf{W}}_i - \mathbf{E} \tilde{\mathbf{W}}_i)$, together constitute the difference in the mean total amount of work in the system at an

arbitrary epoch between the dormant and the non-dormant server case.

The first term in the first form of the right-hand side represents the difference in the mean amount of work at an arbitrary epoch in a non-serving interval that arrived during *switching* intervals (excluding the work that arrived at 1limited queues during switching intervals before the server polled those queues for the last time). The first term itself is the product of two terms. The term inside the square brackets represents the mean amount of work at an arbitrary epoch in a switching interval that arrived during switching intervals (excluding the work that arrived at 1-limited queues during switching intervals before the server polled those queues for the last time). Notice that these quantities are the same in the dormant and the non-dormant server case. The term in front is the probability that in the dormant server case an arbitrary epoch in a nonserving interval concerns an idling epoch rather than a switching epoch. The first term is the product of these two terms, since the amount of work at an arbitrary idling epoch in the dormant server case is by definition zero.

The second term in the first form of the right-hand side represents the difference in the mean amount of work at an arbitrary epoch in a non-serving interval that arrived during *idling* intervals. The second term is of course just the mean amount of work at an arbitrary epoch in a non-serving interval that arrived during idling intervals in the dormant server case, since in the non-dormant server case there are by definition no idling intervals.

The interpretation of the terms in the second form of the right-hand side of (4.16) requires a somewhat different point of view. Define f(i, j, k), g(i, j, k), and h(i, j, k) as the number of times the server needs to switch from Q_i to Q_{i+1} when currently in an empty system switching from Q_k to Q_{k+1} , idling at Q_k not having served any customer yet, and idling at Q_k having already served a customer, respectively, before it can perform work currently arriving at Q_j , taking into account the service discipline at Q_j . Then the first and the second term in the first form of the right-hand side may be written as

$$\frac{\pi}{1-\rho} \sum_{i=1}^{n} \frac{s_i}{s} \sum_{j=1}^{n} \rho_j \left(\frac{s_i^{(2)}}{2s_i} + \sum_{k=1}^{n} s_k f(i,j,k) \right) = \frac{\pi\rho}{1-\rho} \sum_{i=1}^{n} \frac{s_i}{s} \sum_{j=1}^{n} \frac{\rho_j}{\rho} \left(\frac{s_i^{(2)}}{2s_i} + \sum_{k=1}^{n} s_k f(k,j,i) \right),$$

and

$$\frac{1-\rho-\pi}{1-\rho}\sum_{i=1}^{n}\frac{s_{i}}{s}\sum_{j=1}^{n}\rho_{j}\sum_{k=1}^{n}\left(\frac{\pi_{k}'s}{1-\rho-\pi}g(i,j,k)+\frac{\pi_{k}''s}{1-\rho-\pi}h(i,j,k)\right) = \frac{\pi\rho}{1-\rho}\left(\sum_{i=1}^{n}\frac{\pi_{i}'}{\pi}\sum_{j=1}^{n}\frac{\rho_{j}}{\rho}\sum_{k=1}^{n}s_{k}g(k,j,i)+\sum_{i=1}^{n}\frac{\pi_{i}''}{\pi}\sum_{j=1}^{n}\frac{\rho_{j}}{\rho}\sum_{k=1}^{n}s_{k}h(k,j,i)\right),$$

respectively. As the amount of work arriving at Q_j constitutes a fraction $\frac{\rho_j}{\rho}$ of the total amount of work arriving to the system, we may interpret

$$\sum_{i=1}^{n} \frac{s_i}{s} \sum_{j=1}^{n} \frac{\rho_j}{\rho} \left(\frac{s_i^{(2)}}{2s_i} + \sum_{k=1}^{n} s_k f(k, j, i) \right),$$

$$\sum_{i=1}^{n} \frac{\pi'_i}{\pi} \sum_{j=1}^{n} \frac{\rho_j}{\rho} \sum_{k=1}^{n} s_k g(k, j, i) + \sum_{i=1}^{n} \frac{\pi''_i}{\pi} \sum_{j=1}^{n} \frac{\rho_j}{\rho} \sum_{k=1}^{n} s_k h(k, j, i)$$

as the mean total switch-over time to be incurred by the server when currently in an empty system switching and idling, respectively, before it can perform work currently arriving to the the system (at an arbitrary place), taking into account the various service disciplines. For brevity, let us refer to these quantities as the mean access time of a switching and an idling server, respectively. Concluding, the difference in the mean total amount of work between the dormant and the non-dormant server case is just the difference in the mean access time between a switching and an idling server, preceded by the multiplier $\frac{\pi\rho}{1-\rho}$.

We now compare the mean waiting times in a symmetric system in the purely dormant and the non-dormant server case for the exhaustive (I), gated (II), and 1-limited (III) service discipline. To do so, we take in (4.16) $\rho_i = \rho/n$, $s_i = s/n$, $\pi_i = \pi/n$, $E\hat{\mathbf{W}}_i = E\hat{\mathbf{W}}$, $E\tilde{\mathbf{W}}_i = E\tilde{\mathbf{W}}$, and for the 1-limited service discipline in addition $\lambda_i = \lambda/n$, $\pi_i'' = \pi''/n$.

We cannot compare the mean waiting times in a symmetric system for the globally gated service discipline. Even when the Q_i are all identical, the asymmetric nature of the globally gated service discipline will preclude that the mean waiting times \mathbf{EW}_i and the probabilities π_i are all identical. I. Exhaustive:

$$E\hat{\mathbf{W}} - E\tilde{\mathbf{W}} = \frac{\pi s}{2(1-\rho)} \left[\frac{n-1}{n} - \frac{s^{(2)}}{s^2} \right]$$

$$\leq -\frac{\pi s}{2n(1-\rho)} \leq 0.$$
(4.17)

II. Gated:

III. 1-Limited:

$$\mathbb{E}\hat{\mathbf{W}} - \mathbb{E}\tilde{\mathbf{W}} = \frac{s}{2(1-\rho-\frac{\lambda s}{n})} \left[\pi \left(\frac{n-1}{n} - \frac{s^{(2)}}{s^2}\right) + \frac{2\pi''}{n} \right] \qquad (4.19)$$

$$\leq \frac{\pi s}{2(n(1-\rho)-\lambda s)} \leq \frac{s}{2n}.$$

For the exhaustive service discipline, always $E\hat{W} \leq E\tilde{W}$, which agrees with the result of Liu, Nain, & Towsley [145] that in a symmetric system, the server

should remain idling at a queue when the entire system is empty to minimize the total amount of work. Also $E\hat{\mathbf{W}} \leq E\tilde{\mathbf{W}}$ for the gated service discipline when the server is only allowed to make a halt at a queue when it has not served any customer yet, i.e., when $\pi'' = 0$. When the server is also allowed to make a halt at a queue when it has already served a customer, i.e., when $\pi'' = \pi$, $E\hat{\mathbf{W}} \leq E\tilde{\mathbf{W}}$ ($E\hat{\mathbf{W}} \geq E\tilde{\mathbf{W}}$) for the gated service discipline, iff the coefficient of variation of the total switch-over time is larger (smaller) than 1 + 1/n. In particular, $E\hat{\mathbf{W}} = E\tilde{\mathbf{W}}$ when the total switch-over time is Erlang-*n* distributed, so when the individual switch-over times are exponentially distributed. Because of the memoryless property of the exponential distribution, customers then indeed do not observe any difference between the dormant and the non-dormant server case. Similar remarks hold for the 1-limited service discipline.

We finally address the question at which queues the server should make a halt to minimize the mean total amount of work in the system. Liu, Nain, & Towsley [145] identify polling policies, allowing idling as a possible action, that stochastically minimize the total amount of work in the system at an arbitrary epoch. They show that optimal policies are exhaustive and greedy, i.e., the server should neither switch nor idle when at a non-empty queue. In addition, they prove that in symmetric polling systems, patient policies are optimal, i.e., when the entire system is empty the server should remain idling at the last visited queue. Since a non-exhaustive policy cannot minimize the total amount of work, we assume in the sequel the service discipline to be exhaustive.

In asymmetric systems, the total amount of work is not always minimal in the purely dormant server case. In some asymmetric systems the total amount of work is smaller even in the non-dormant server case. Consider e.g. a system with n = 2, $\lambda_1 = \infty$, $\beta_1 = 0$, $s_1^{(2)} = s_1^2$, $s_2 = 0$. As far as the amount of work is concerned, such a system corresponds in the purely dormant and the non-dormant server case to an ordinary M/G/1 queue with set-up times and multiple vacations of length s_1 , respectively. In the latter case, the amount of work is of course smaller. However, in not a single system the total amount of work is minimal in the non-dormant server case, as we will prove now. If the server makes a halt only at Q_h , then (4.16) reduces to

$$\sum_{i=1}^{n} \rho_i (\mathbf{E}\hat{\mathbf{W}}_i - \mathbf{E}\tilde{\mathbf{W}}_i) = -\frac{\pi_h}{1-\rho} \left[\rho \frac{s^{(2)}}{2s} - \sum_{i \neq h-1} \sum_{j=i+1}^{h-1} s_i \rho_j \right].$$

So it suffices to show that there is at least one Q_h such that $\sum_{i \neq h-1} \sum_{j=i+1}^{h-1} s_i \rho_j \leq \sum_{i \neq h-1}^{h-1} s_i \rho_i$

$$\rho \frac{s^{(2)}}{2s}. \text{ Now } \sum_{h=1}^{n} \sum_{i \neq h-1} \sum_{j=i+1}^{n-1} s_i \rho_j = \frac{1}{2} n \rho s - \sum_{i=1}^{n} s_i \rho_i \text{ implies } \min_{1 \leq h \leq n} \sum_{i \neq h-1} \sum_{j=i+1}^{n-1} s_i \rho_j$$
$$\leq \frac{1}{2} \rho s \leq \rho \frac{s^{(2)}}{2s}. \text{ So there is indeed at least one such } Q_h.$$

We now know that making a halt only at Q_h is beneficial, iff $\sum_{i \neq h-1} \sum_{j=i+1}^{h-1} s_i \rho_j$

 $\leq \rho \frac{s^{(2)}}{2s}$. In addition, we now assume that Q_h belongs to the set of queues at which the server should make a halt, when making a halt only at Q_h is beneficial. In other words, we propose to let the server make a halt at Q_h , iff

 $\sum_{i \neq h-1} \sum_{j=i+1}^{h-1} s_i \rho_j \leq \rho \frac{s^{(2)}}{2s}.$ In accordance with the interpretation of the second

form of (4.16), the criterion is seen to select queues Q_h , such that the mean access time of the server is smaller when idling at Q_h than when switching.

Written as $\sum_{i \neq h-1} \sum_{j=i+1}^{h-1} \frac{s_i}{s} \frac{\rho_j}{\rho} \leq \frac{s^{(2)}}{2s^2}$, the criterion is seen to suggest idling at

queues Q_h preceded by queues with relatively light traffic and large switch-over times and followed by queues with relatively heavy traffic and small switch-over times, and is also seen to suggest idling more strongly accordingly as the variability of the total switch-over time is larger. Notice that in a symmetric system

the inequality reduces to $1 - 1/n \leq \frac{s^{(2)}}{s^2}$, which always holds, cf. (4.17). Blanc & Van der Mei [24] obtain a similar rule from light-traffic considerations. Numerical experiments in [24] suggest that the rule performs very well.

Remark 4.4.1

The problem at which queues the server should make a halt to minimize the mean total amount of work in the system, may be formulated as a semi-Markov decision problem, cf. Tijms [184] p. 200. The decision epochs are the epochs of a visit completion; the possible decisions (actions) are either switching or idling when the visit completion leaves the entire system empty and only switching otherwise. The states are (i, l_1, \ldots, l_n) , where *i* is the queue at which the visit completion occurs and l_j is the number of waiting customers at $Q_j, j = 1, \ldots, n$. The crucial observation is that the system under consideration satisfies the following Markovian property: given the state at some decision epoch and the decision (action) chosen, the evolution after that decision epoch does not depend on the evolution before that decision epoch.

A semi-Markov decision problem formulates the problem of finding a strategy that minimizes the mean total cost per unit of time. A strategy prescribes here at which queues the server should make a halt. The mean total cost per unit of time may be related here to the mean total amount of work in the system, when we define the cost appropriately. If c_i represents the waiting cost per unit of time of an arbitrary type-*i* customer, $i = 1, \ldots, n$, then the mean total cost per unit of time equal $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i$. When we take $c_i = \beta_i, i = 1, \ldots, n$,

the mean total cost per unit of time equal $\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i$, the mean amount of waiting work in the system. Minimizing the mean amount of waiting work and

minimizing the mean total amount of work are equivalent, since the difference, the mean amount of work in service, always equals $\frac{1}{2}\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}$.

It is straightforward, when action a is chosen in the current state h, to calculate p(a, h, k), the probability that at the next decision epoch the state will be k, t(a, h), the expected time until the next decision epoch, and c(a, h), the expected cost incurred until the next decision epoch. The resulting semi-Markov decision problem may be solved numerically by truncating the state space.

Remark 4.4.2

In the present chapter, we allowed the server only to make a halt at a queue when there are no customers present in the system. In fact, we may allow the server also to make a halt at a queue in other cases when there are customers present in the system. A first option might be to maintain the service disciplines at the various queues, but to decide at the completion of *each* visit whether to switch or to idle, and not only at the completion of a visit that leaves the entire system empty. A second option might be also to drop the service disciplines at the various queues, and to decide at the completion of *each service* whether to serve another customer if present, to switch, or to idle, like Liu, Nain, & Towsley [145].

Once having enlarged the freedom of decisions in the operation of the system, it is quite natural to consider the problem of finding a strategy that optimizes the performance of the system. As the enlarged freedom of decisions will complicate the analysis even further, there is little hope to solve the problem analytically. Remember that the pseudo-conservation law did not even hold enough information to solve the more specific problem at which queues the server should make a halt to minimize the mean total amount of work in the system. However, like the latter more specific problem, the problem of finding a strategy that minimizes $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i$ may still be handled as a semi-Markov decision problem.

CHAPTER 5

A GLOBALLY GATED POLLING SYSTEM WITH A DORMANT SERVER

5.1 INTRODUCTION

In the previous chapter we considered a polling system with a dormant server. We obtained a pseudo-conservation law for a general model, permitting a wide variety of service disciplines, in which the server may be allowed to make a halt at an arbitrary subset of the queues. In the present chapter, we focus on a particular model from that broad class, in which the option of idling arises quite naturally: a system with the globally gated service discipline, in which the server only makes a halt at its home base. The globally gated service discipline, introduced in Boxma, Levy, & Yechiali [50], operates as follows. Suppose the server arrives at its home base. Then all the customers in the system are marked instantaneously, and the server immediately starts a tour along the queues. During this tour only the marked customers are served. The service of customers that meanwhile arrive to the system is deferred until the next tour along the queues. Boxma, Weststrate, & Yechiali [41] propose the globally gated service discipline to be used by a repair crew, in charge of the maintenance activities at several installations. As indicated in [41], under the globally gated service discipline it does not make sense to start a tour along the queues when there are no customers present in the system. In the present chapter, we therefore consider a globally gated polling system in which the server makes a halt at its home base when there are no customers present in the system. The globally gated service discipline then operates as follows. Suppose again the server arrives at its home base. If there are customers present in the system, they are all marked instantaneously, and the server starts a tour along the queues, acting as described earlier. If there are no customers present in the system, the server remains idling at its home base, awaiting a new customer to arrive at one of the queues. As soon as a new customer arrives, it is marked instantaneously, and the server starts a tour along the queues. During this tour only the newly arrived customer is served. The service of customers that meanwhile arrive to the system is again deferred until the next tour along the queues.

The remainder of the chapter is organized as follows. In Section 5.2, we present a detailed model description. In Section 5.3, we derive an explicit expression for the Laplace-Stieltjes Transform (LST) of the cycle time distribution. We obtain the LST of the waiting time-distribution at each of the queues in Section 5.4. As a justification of the dormant server policy, we show the waiting time at each of the queues to be smaller (in the increasing-convex-ordering sense) than in the ordinary non-dormant server case. In Section 5.5, we derive the probability generating function (pgf) of the joint queue length distribution at polling epochs.

5.2 MODEL DESCRIPTION

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by a single server S. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3.

The server visits the queues in strictly cyclic order, Q_1, \ldots, Q_n . Suppose the server is just about to visit Q_1 . If there are customers present in the system, they are all marked instantaneously, and the server immediately starts visiting Q_1, \ldots, Q_n . During the coming cycle only the marked customers are served. At each queue, customers are served in order of arrival. The service of customers that meanwhile arrive to the system is deferred until the next cycle. If there are no customers present in the system, the server remains idling at Q_1 , awaiting a new customer to arrive at one of the queues. As soon as a new customer arrives, it is marked instantaneously, and the server starts visiting Q_1, \ldots, Q_n . During the coming cycle only the newly arrived customer is served. Again, customers that meanwhile arrive to the system are served during the next cycle. During the cycle the server is not allowed to make a halt at a queue when the completion of a service leaves the system empty. In other words, the server is only allowed to make a halt when the system is empty at the beginning of a visit to Q_1 .

Remark 5.2.1

For n = 1 the model under consideration reduces to a gated vacation model with *single* vacations, while in the non-dormant server case the model corresponds to a gated vacation model with *multiple* vacations, cf. Takagi [174] pp. 205-213. The latter model is analyzed in detail in Takine & Hasegawa [177].

5.3 The cycle time

In this section, we relate the cycle time distribution to the joint queue length distribution at the beginning of a cycle and at the beginning of a subsequent cycle. The approach is similar to the approach in Boxma, Levy, & Yechiali [50] for the ordinary non-dormant server case. Assuming an equilibrium distribution, we obtain a functional equation for the pgf of the joint queue length distribution at the beginning of a cycle and for the LST of the cycle time distribution. The latter functional equation is solved explicitly. The cycle time distribution will play a crucial role in the analysis of the waiting-time distribution and the joint queue length distribution at polling epochs in Sections 5.4 and 5.5, respectively.

We first introduce some notation. Denote by $\mathbf{C}^{(m)}$ the length of the *m*-th cycle, i.e., the time between the start of the *m*-th visit to Q_1 and the start of the (m+1)-th visit to $Q_1, m = 1, 2, \ldots$ Denote by $\mathbf{I}^{(m)}$ the length of the *m*-th idling period, i.e., the *m*-th idling time at Q_1 (possibly zero), $m = 1, 2, \ldots$ Denote by $\overline{\mathbf{C}}^{(m)}$ the length of the *m*-th *restricted* cycle, i.e., the *m*-th cycle time minus the *m*-th idling time, $m = 1, 2, \ldots$. Let $\alpha_m(\zeta, \omega) := \mathbf{E}(\mathbf{e}^{-\zeta\mathbf{I}^{(m)}} - \omega \overline{\mathbf{C}}^{(m)})$ for Re $\zeta \geq 0$, Re $\omega \geq 0$, $m = 1, 2, \ldots$. Let $\gamma_m(\omega) := \mathbf{E}(\mathbf{e}^{-\omega \overline{\mathbf{C}}^{(m)}})$ for Re $\omega \geq 0$, $m = 1, 2, \ldots$ Denote by $\mathbf{Y}_i^{(m)}$ the number of customers present at queue *i* at the beginning of the *m*-th cycle, $i = 1, \ldots, n, m = 1, 2, \ldots$. Let $\xi_m(z_1, \ldots, z_n) := \mathbf{E}(z_1^{\mathbf{Y}_1^{(m)}} \ldots z_n^{\mathbf{Y}_n^{(m)}})$ for $|z_i| \leq 1, i = 1, \ldots, n, m = 1, 2, \ldots$. By the nature of the globally gated service discipline, $\mathbf{I}^{(m)}$, $\overline{\mathbf{C}}^{(m)}$, $\mathbf{Y}_i^{(m)}$, and $\mathbf{Y}_i^{(m+1)}$, $i = 1, \ldots, n, m = 1, 2, \ldots$, are related as follows. On the one hand, $\mathbf{Y}_i^{(m)}$ equals the number of customers that are served at Q_i during the *m*-th restricted cycle, unless $(\mathbf{Y}_1^{(m)}, \ldots, \mathbf{Y}_n^{(m)}) = (0, \ldots, 0)$, i.e., there are no customers present at the beginning of the *m*-th cycle. In that case, the server remains idling at Q_1 for a period which is negative exponentially distributed with parameter λ , until a customer arrives at one of the queues; such an arrival occurs at Q_i with probability λ_i/λ . So

$$E(e^{-\zeta \mathbf{I}^{(m)} - \omega \bar{\mathbf{C}}^{(m)}} | \mathbf{Y}_{1}^{(m)}, \dots, \mathbf{Y}_{n}^{(m)}) =$$

$$\sigma(\omega) \left[\prod_{i=1}^{n} \beta_{i}(\omega)^{\mathbf{Y}_{i}^{(m)}} - \left(1 - \frac{\lambda}{\lambda + \zeta} \sum_{i=1}^{n} \frac{\lambda_{i}}{\lambda} \beta_{i}(\omega) \right) \mathbf{I}_{\{\mathbf{Y}^{(m)} = 0\}} \right],$$
(5.1)

with $I_{\{\mathbf{Y}^{(m)}=0\}}$ denoting the indicator function of the event $(\mathbf{Y}_1^{(m)}, \ldots, \mathbf{Y}_n^{(m)}) = (0, \ldots, 0)$. Unconditioning (5.1),

$$\alpha_m(\zeta,\omega) = (5.2)$$

$$\sigma(\omega) \left[\xi_m(\beta_1(\omega),\ldots,\beta_n(\omega)) - \left(1 - \frac{\lambda}{\lambda+\zeta} \sum_{i=1}^n \frac{\lambda_i}{\lambda} \beta_i(\omega) \right) \xi_m(0,\ldots,0) \right].$$

On the other hand, $\mathbf{Y}_{i}^{(m+1)}$ equals the number of customers that arrive at Q_{i} during the *m*-th restricted cycle. So

$$E(z_1^{\mathbf{Y}_1^{(m+1)}} \dots z_n^{\mathbf{Y}_n^{(m+1)}} \mid \bar{\mathbf{C}}^{(m)} = t) = e^{-\sum_{i=1}^n \lambda_i (1-z_i)t}.$$
(5.3)

Define $\epsilon(z_1, \ldots, z_n) := \sum_{i=1}^n \lambda_i (1 - z_i)$ for $|z_i| \le 1, i = 1, \ldots, n$. Unconditioning (5.3),

$$\xi_{m+1}(z_1,\ldots,z_n) = \gamma_m(\epsilon(z_1,\ldots,z_n)).$$
(5.4)

Denote by C, I, \overline{C} , and \mathbf{Y}_i stochastic variables with the limiting distribution for $m \to \infty$ of $\mathbf{C}^{(m)}$, $\mathbf{I}^{(m)}$, $\overline{\mathbf{C}}^{(m)}$, and $\mathbf{Y}_i^{(m)}$, $i = 1, \ldots, n$, respectively. Let $\alpha(\zeta, \omega) := \mathbf{E}(\mathrm{e}^{-\zeta \mathbf{I} - \omega \overline{\mathbf{C}}})$ for $\operatorname{Re} \zeta \geq 0$, $\operatorname{Re} \omega \geq 0$. Let $\gamma(\omega) := \mathbf{E}(\mathrm{e}^{-\omega \overline{\mathbf{C}}})$ for $\operatorname{Re} \omega \geq 0$. Let $\xi(z_1, \ldots, z_n) := \mathbf{E}(z_1^{\overline{\mathbf{Y}}_1}, \ldots, z_n^{\mathbf{Y}_n})$ for $|z_i| \leq 1, i = 1, \ldots, n$. From (5.2) and (5.4),

$$\alpha(\zeta,\omega) =$$

$$\sigma(\omega) \left[\xi(\beta_1(\omega),\ldots,\beta_n(\omega)) - \left(1 - \frac{\lambda}{\lambda+\zeta} \sum_{i=1}^n \frac{\lambda_i}{\lambda} \beta_i(\omega)\right) \xi(0,\ldots,0) \right],$$
(5.5)

and

$$\xi(z_1,\ldots,z_n) = \gamma(\epsilon(z_1,\ldots,z_n)).$$
(5.6)

Combining (5.5) and (5.6),

$$\alpha(\zeta,\omega) =$$

$$\sigma(\omega) \left[\gamma(\sum_{i=1}^{n} \lambda_i (1 - \beta_i(\omega))) - \gamma(\lambda) \left(1 - \frac{\lambda}{\lambda + \zeta} \sum_{i=1}^{n} \frac{\lambda_i}{\lambda} \beta_i(\omega) \right) \right],$$
(5.7)

and

$$\xi(z_1, \dots, z_n) =$$

$$\sigma(\epsilon(z)) \left[\xi(\beta_1(\epsilon(z)), \dots, \beta_n(\epsilon(z))) - \left(1 - \sum_{i=1}^n \frac{\lambda_i}{\lambda} \beta_i(\epsilon(z))\right) \xi(0, \dots, 0) \right].$$
(5.8)

Remark 5.3.1

As mentioned in Section 1.4, strictly speaking the globally gated service discipline does not satisfy Property 1.4.1, which provides a global characterization of the class of service disciplines that allow an exact analysis in a multi-type branching process framework. Still, the joint queue length process at the beginning of a cycle, $\{(\mathbf{Y}_1^{(m)},\ldots,\mathbf{Y}_n^{(m)}), m = 1,2,\ldots\}$, constitutes a multi-type branching process with state-dependent immigration. The crucial observation

5.3 The cycle time

is that the globally gated service discipline satisfies the following property: if there are k_i customers present at Q_i at the beginning of a cycle, then each of these k_i customers will be 'effectively replaced' in an i.i.d. manner by a random population having pgf $\beta_i(\epsilon(z))$. Adopting the terminology of the theory of multi-type branching processes, the offspring generating functions are given by $f_i(z) = \beta_i(\epsilon(z)), i = 1, \ldots, n$, the immigration generating function for the non-zero states is given by $g(z) = \sigma(\epsilon(z))$, and the immigration generating function for the zero state is given by g(z)h(z) with $h(z) = \sum_{i=1}^n \frac{\lambda_i}{\lambda} f_i(z)$. From the theory of multi-type branching processes we have

$$\xi(z_1,\ldots,z_n) = g(z)[\xi(f_1(z),\ldots,f_n(z)) - (1-h(z))\xi(0,\ldots,0)],$$

which is identical to (5.8).

Below we solve the functional equation (5.7). We first derive some preliminary results from (5.7). Noting that $E(e^{-\zeta I}) = \alpha(\zeta, 0)$ for $\operatorname{Re} \zeta \ge 0$ and $E(e^{-\omega \tilde{C}}) = \alpha(0, \omega)$ for $\operatorname{Re} \omega \ge 0$,

$$EC = \frac{s + \frac{\gamma(\lambda)}{\lambda}}{1 - \rho};$$
(5.9)

$$\mathbf{EI} = \frac{\gamma(\lambda)}{\lambda};\tag{5.10}$$

$$\mathbf{E}\bar{\mathbf{C}} = \frac{s + \rho \frac{\gamma(\lambda)}{\lambda}}{1 - \rho}; \tag{5.11}$$

$$\mathbf{E}\bar{\mathbf{C}}^{2} = \frac{s^{(2)} + (2\rho s + \sum_{i=1}^{n} \lambda_{i}\beta_{i}^{(2)})\mathbf{E}\mathbf{C}}{1 - \rho^{2}}.$$
(5.12)

Remark 5.3.2

We may also obtain (5.10) directly by observing

 $-\alpha$

$$\mathrm{E}(\mathbf{I} \mid \mathbf{I} > 0) = \frac{1}{\lambda},$$

while

$$\Pr{\{\mathbf{I} > 0\}} = \Pr{\{(\mathbf{Y}_1, \dots, \mathbf{Y}_n) = (0, \dots, 0)\}} = \int_{t=0}^{\infty} e^{-\lambda t} d\Pr{\{\bar{\mathbf{C}} < t\}} = \gamma(\lambda).$$

We may also obtain (5.9) directly from (4.1) and (5.10) by observing that $\pi = \frac{EI}{EC}$.

We now solve the functional equation (5.7). Obviously, it suffices to find an expression for $\gamma(\omega)$, as substituting such an expression into (5.7) yields an expression for $\alpha(\zeta, \omega)$. Define $\delta(\omega) := \sum_{i=1}^{n} \lambda_i (1 - \beta_i(\omega))$ for $\operatorname{Re} \omega \ge 0$. Putting $\zeta = 0$ in (5.7),

$$\gamma(\omega) = \sigma(\omega) \left[\gamma(\delta(\omega)) - \frac{\gamma(\lambda)}{\lambda} \delta(\omega) \right], \quad \text{Re } \omega \ge 0.$$
 (5.13)

Define recursively

$$\begin{split} \delta^{(0)}(\omega) &:= \omega, & \operatorname{Re} \omega \geq 0; \\ \delta^{(k)}(\omega) &:= \delta(\delta^{(k-1)}(\omega)), & \operatorname{Re} \omega \geq 0, k = 1, 2, \dots \end{split}$$

Iterating (5.13) K times,

$$\gamma(\omega) = \prod_{k=0}^{K} \sigma(\delta^{(k)}(\omega)) \gamma(\delta^{(K+1)}(\omega)) -$$

$$\frac{\gamma(\lambda)}{\lambda} \sum_{k=0}^{K} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega)), \operatorname{Re} \omega \ge 0, K = 1, 2, \dots$$
(5.14)

Lemma 5.3.1

Lemma 5.3.1 i. $\lim_{K \to \infty} \delta^{(K)}(\omega) = 0 \text{ for all } \omega \text{ with } \operatorname{Re} \omega \ge 0.$ ii. $\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega)) \text{ converges for all } \omega \text{ with } \operatorname{Re} \omega \ge 0.$ iii. $\sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega)) \text{ converges for all } \omega \text{ with } \operatorname{Re} \omega \ge 0.$

Proof

See Appendix 5.A.

Lemma 5.3.1 implies, letting $K \to \infty$ in (5.14),

$$\gamma(\omega) = \prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega)) - \frac{\gamma(\lambda)}{\lambda} \sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega)).$$
(5.15)

Putting $\omega = \lambda$ in (5.15),

$$\gamma(\lambda) = \frac{\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\lambda))}{1 + \frac{1}{\lambda} \sum_{k=0}^{\infty} \delta^{(k+1)}(\lambda) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\lambda))}.$$
(5.16)

5.4 The waiting time

Substituting (5.16) back into (5.15),

$$\gamma(\omega) = \prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega)) -$$

$$\frac{\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\lambda))}{\lambda + \sum_{k=0}^{\infty} \delta^{(k+1)}(\lambda) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\lambda))} \sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega)).$$
(5.17)

Substituting (5.17) into (5.7) yields an expression for $\alpha(\zeta, \omega)$. Finally some words on the joint past and residual lifetime distribution of a restricted cycle. This distribution will play a crucial role in the analysis of the waiting-time distribution in the next section. Denote by $\bar{\mathbf{C}}_P$ and $\bar{\mathbf{C}}_R$ stochastic variables with the distribution of the past and residual lifetime of a restricted cycle, respectively. From Cohen [73] p. 113 we have

$$E(e^{-\omega_{P}\bar{\mathbf{C}}_{P}-\omega_{R}\bar{\mathbf{C}}_{R}}) = \frac{1}{E\bar{\mathbf{C}}} \frac{\gamma(\omega_{R})-\gamma(\omega_{P})}{\omega_{P}-\omega_{R}}, \quad \operatorname{Re}\omega_{P} \ge 0, \quad \operatorname{Re}\omega_{R} \ge 0. \quad (5.18)$$

In particular,

$$E(e^{-\omega\bar{\mathbf{C}}_{P}}) = E(e^{-\omega\bar{\mathbf{C}}_{R}}) = \frac{1-\gamma(\omega)}{\omega E\bar{\mathbf{C}}}, \qquad \text{Re}\,\omega \ge 0.$$
(5.19)

From (5.19),

$$\mathbf{E}\bar{\mathbf{C}}_{P} = \mathbf{E}\bar{\mathbf{C}}_{R} = \frac{\mathbf{E}\bar{\mathbf{C}}^{2}}{2\mathbf{E}\bar{\mathbf{C}}}.$$
(5.20)

5.4 The waiting time

In formula (5.18) in the previous section, we expressed the LST of the joint past and residual lifetime distribution of a restricted cycle into the LST of the restricted cycle time distribution. In this section, we relate the waiting-time distribution to the joint past and residual lifetime distribution of a restricted cycle. Thus, we obtain an expression for the LST of the waiting-time distribution in terms of the LST of the restricted cycle time distribution.

We first introduce some notation. Denote by $\mathbf{V}_i(\mathbf{N})$ the total service time of \mathbf{N} type-*i* customers, i = 1, ..., n, for any non-negative integer-valued stochastic variable \mathbf{N} . So $\mathbf{E}(\mathbf{e}^{-\omega \mathbf{V}_i(\mathbf{N})}) = \mathbf{E}(\beta_i(\omega)^{\mathbf{N}})$, Re $\omega \ge 0$, i = 1, ..., n. Denote by $\mathbf{A}_i(\mathbf{T})$ the number of type-*i* customers arriving during a period of length \mathbf{T} , i = 1, ..., n, for any non-negative real-valued stochastic variable \mathbf{T} . So $\mathbf{E}(z_i^{\mathbf{A}_i(\mathbf{T})}) = \mathbf{E}(\mathbf{e}^{-\lambda_i(1-z_i)\mathbf{T}})$, $|z_i| \le 1$, i = 1, ..., n. Denote by \mathbf{B}_i and \mathbf{S}_i stochastic variables having distribution $B_i(\cdot)$ and $S_i(\cdot)$, i = 1, ..., n, respectively.

We now analyze the waiting-time distribution of an arbitrary type-*i* customer, by distinguishing whether the customer arrives during a restricted cycle or during an idling period (thus terminating the idling period immediately by initiating a new restricted cycle), in other words, whether the customer sees the server working/switching or idling upon arrival. The waiting time $\mathbf{W}_i^{(B)}$ of an arbitrary type-*i* customer that arrives during a restricted cycle, is composed of i. the residual lifetime of the restricted cycle in which it arrives;

ii. the total service time of all customers that arrive at Q_1, \ldots, Q_{i-1} during the same restricted cycle;

iii. the total service time of all customers that arrive at Q_i during the past lifetime of the restricted cycle in which it arrives;

iv. the total switch-over time incurred by the server when moving from Q_1 to Q_i ;

i.e.,

$$\mathbf{W}_i^{(\mathrm{B})} \stackrel{d}{=} \bar{\mathbf{C}}_R + \sum_{j=1}^{i-1} \mathbf{V}_j (\mathbf{A}_j (\bar{\mathbf{C}}_P + \bar{\mathbf{C}}_R)) + \mathbf{V}_i (\mathbf{A}_i (\bar{\mathbf{C}}_P)) + \sum_{j=1}^{i-1} \mathbf{S}_j.$$

So, using (5.18),

$$E(e^{-\omega \mathbf{W}_{i}^{(B)}}) = \prod_{j=1}^{i-1} \sigma_{j}(\omega) \times$$

$$\int_{t_{P}=0}^{\infty} \int_{t_{R}=0}^{\infty} e^{-\omega t_{R}} \prod_{j=1}^{i-1} \left\{ e^{-\lambda_{j}(1-\beta_{j}(\omega))(t_{P}+t_{R})} \right\} e^{-\lambda_{i}(1-\beta_{i}(\omega))t_{P}} \times$$

$$d_{t_{P},t_{R}} \Pr\{\bar{\mathbf{C}}_{P} < t_{P}, \bar{\mathbf{C}}_{R} < t_{R}\} =$$
(5.21)

$$\prod_{j=1}^{i-1} \sigma_j(\omega) \frac{1}{\mathrm{E}\bar{\mathbf{C}}} \frac{\gamma(\sum_{j=1}^i \lambda_j(1-\beta_j(\omega))) - \gamma(\sum_{j=1}^{i-1} \lambda_j(1-\beta_j(\omega)) + \omega)}{\omega - \lambda_i(1-\beta_i(\omega))}.$$

The waiting time $\mathbf{W}_{i}^{(I)}$ of an arbitrary type-*i* customer that arrives during an idling period, is composed solely of the total switch-over time incurred by the server when moving from Q_{1} to Q_{i} , i.e.,

$$\mathbf{W}_i^{(\mathrm{I})} \stackrel{d}{=} \sum_{j=1}^{i-1} \mathbf{S}_j.$$

So,

$$\mathbf{E}(\mathbf{e}^{-\omega \mathbf{W}_{i}^{(1)}}) = \prod_{j=1}^{i-1} \sigma_{j}(\omega).$$
(5.22)

Combining (5.21) and (5.22), noting that an arbitrary customer, irrespective of which type, arrives during a restricted cycle and an idling period with probability $E\bar{C}/EC$ and EI/EC, respectively,

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}_{i}}) =$$

$$\prod_{j=1}^{i-1} \sigma_{j}(\omega) \frac{1}{\mathbf{EC}} \left[\mathbf{EI} + \frac{\gamma(\sum_{j=1}^{i} \lambda_{j}(1-\beta_{j}(\omega))) - \gamma(\sum_{j=1}^{i-1} \lambda_{j}(1-\beta_{j}(\omega)) + \omega)}{\omega - \lambda_{i}(1-\beta_{i}(\omega))} \right].$$
(5.23)

Remark 5.4.1

For n = 1, using (5.9), (5.10) and (5.13), (5.23) reduces to

$$\mathbf{E}(\mathbf{e}^{-\omega\mathbf{W}}) = \frac{(1-\rho)\omega}{\omega - \lambda(1-\beta(\omega))} \left[\frac{\mathbf{E}\mathbf{I}}{s+\mathbf{E}\mathbf{I}} + \frac{s}{s+\mathbf{E}\mathbf{I}} \frac{1-\sigma(\omega)}{s\omega} \frac{\gamma(\omega)}{\sigma(\omega)} \right],$$

in which we recognize the well-known waiting-time decomposition property of M/G/1 vacation models.

From (5.23), using (5.9) and (5.12),

$$\mathbf{EW}_{i} = \left[1 + 2\sum_{j=1}^{i-1} \rho_{j} + \rho_{i} \right] \frac{\mathbf{E}\bar{\mathbf{C}}^{2}}{2\mathbf{E}\mathbf{C}} + \sum_{j=1}^{i-1} s_{j} =$$
(5.24)
$$\left[1 + 2\sum_{j=1}^{i-1} \rho_{j} + \rho_{i} \right] \frac{1}{1+\rho} \left[\frac{\sum_{j=1}^{n} \lambda_{j} \beta_{j}^{(2)}}{2(1-\rho)} + \frac{\rho s}{1-\rho} + \frac{s^{(2)}}{2(s+\frac{\gamma(\lambda)}{\lambda})} \right] + \sum_{j=1}^{i-1} s_{j}.$$

Remark 5.4.2

For n = 1 (5.24) reduces to

$$\mathbf{EW} = [1+\rho]\frac{\mathbf{E}\bar{\mathbf{C}}^2}{2\mathbf{E}\mathbf{C}} = \frac{\lambda\beta^{(2)}}{2(1-\rho)} + \frac{\rho s}{1-\rho} + \frac{s^{(2)}}{2(s+\frac{\gamma(\lambda)}{\lambda})},$$

which agrees with formula (5.40b) in Takagi [174] p. 213.

Remark 5.4.3

Noting that $\sum_{i=1}^{n} \rho_i [1 + 2 \sum_{j=1}^{i-1} \rho_j + \rho_i] = \rho(1+\rho)$, we may obtain from (5.24) the pseudo-conservation law for the model under consideration,

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \rho (1+\rho) \frac{\mathbf{E} \bar{\mathbf{C}}^2}{2\mathbf{E} \mathbf{C}} + \sum_{i=1}^{n} \rho_i \sum_{j=1}^{i-1} s_j =$$

$$\rho \frac{\sum_{j=1}^{n} \lambda_j \beta_j^{(2)}}{2(1-\rho)} + \frac{\rho^2 s}{1-\rho} + \rho \frac{s^{(2)}}{2(s+\frac{\gamma(\lambda)}{\lambda})} + \sum_{i=1}^{n} \rho_i \sum_{j=1}^{i-1} s_j.$$

We may also obtain the pseudo-conservation law, without knowledge of the individual mean waiting times, using (5.9) and (5.10), from (4.15) with GG =

$$\{1,\ldots,n\}, \pi'_1 = \frac{\mathbf{EI}}{\mathbf{EC}} = \frac{\gamma(\lambda)}{\lambda} \frac{1-\rho}{s+\frac{\gamma(\lambda)}{\lambda}}, \pi''_1 = 0, \pi_i = 0, i \neq 1.$$

Remark 5.4.4

As seen from (5.24), the ordering of the queues involves an ordering of the mean waiting times,

$$EW_i - EW_{i+1} = -[\rho_i + \rho_{i+1}] \frac{E\bar{C}^2}{2EC} - s_i \le 0, \qquad i = 1, \dots, n-1.$$

The ordering of the queues even involves a stochastic ordering of the waiting times, as seen from the derivation of (5.23).

On the one hand, the ordering of the waiting times may be argued to be unfair. Elevator polling (or scan polling) to some extent meets this objection to the globally gated service discipline. In a globally gated system with elevator polling, the server alternatingly passes through 'up' cycles, visiting the queues in the order Q_1, \ldots, Q_n , and 'down' cycles, visiting the queues in the order Q_n, \ldots, Q_1 , cf. Section 1.3. Thus, as an immediate advantage, elevator polling saves the switch-over time from Q_n to Q_1 . At the start of each cycle, up or down, a similar centralized gating procedure is executed as described earlier. So alternatingly Q_1 and Q_n function as home base. Assuming the switchover time from Q_{i+1} to Q_i to have the same mean as the switch-over time from Q_i to Q_{i+1} , Altman, Khamisy, & Yechiali [7] show that in a globally gated system with elevator polling, the mean waiting times at all queues are equal, irrespective of the traffic characteristics! It is not too surprising that a similar observation holds in the dormant server case. Obviously, the waitingtime distributions at all queues are not equal. E.g., the variance of the waiting times is likely to be larger at the queues visited in the beginning or the end of a cycle than at the queues visited in the middle of a cycle.

On the other hand, the ordering of the waiting times may be exploited to effectuate some kind of priorization. Following the line of this idea, similarly to Boxma, Levy, & Yechiali [50], simple index rules may be established for both static and dynamic optimization of the system performance, measured in terms of the mean waiting times.

5.4 The waiting time

As a justification of the dormant server policy, we now show the waiting time at each of the queues to be smaller (in the increasing-convex-ordering sense) than in the ordinary non-dormant server case. Let us label the stochastic variables and the associated LST's and pgf's in the dormant and non-dormant server case with a hat and a tilde, respectively. From [50] we have

$$\mathbf{E}\tilde{\mathbf{W}}_{i} = \left[1 + 2\sum_{j=1}^{i-1} \rho_{j} + \rho_{i}\right] \frac{\mathbf{E}\tilde{\mathbf{C}}^{2}}{2\mathbf{E}\tilde{\mathbf{C}}} + \sum_{j=1}^{i-1} s_{j} =$$

$$\left[1 + 2\sum_{j=1}^{i-1} \rho_{j} + \rho_{i}\right] \frac{1}{1+\rho} \left[\frac{\sum_{j=1}^{n} \lambda_{j} \beta_{j}^{(2)}}{2(1-\rho)} + \frac{\rho s}{1-\rho} + \frac{s^{(2)}}{2s}\right] + \sum_{j=1}^{i-1} s_{j}.$$
(5.25)

Subtracting (5.25) from (5.24),

$$\mathbf{E}\hat{\mathbf{W}}_{i} - \mathbf{E}\tilde{\mathbf{W}}_{i} = -\left[1 + 2\sum_{j=1}^{i-1}\rho_{j} + \rho_{i}\right]\frac{1}{1+\rho}\frac{\hat{\gamma}(\lambda)}{\lambda}\frac{\lambda}{s+\frac{\hat{\gamma}(\lambda)}{\lambda}}\frac{s^{(2)}}{2s} \le 0.$$

Proceeding by differentiating the LST of the waiting-time distribution not just once but several times, we may prove that in fact not only the mean waiting times are smaller, but also each of the higher-order moments, i.e., $E(\hat{W}_i^k) \leq E(\tilde{W}_i^k)$ for any $k \geq 1$. By using coupling techniques, we may however prove that the waiting times are in fact even smaller in the increasing-convex-ordering sense, i.e., $Ef(\hat{W}_i) \leq Ef(\hat{W}_i)$ for any non-decreasing convex function $f(\cdot)$.

i.e., $\mathrm{E}f(\hat{\mathbf{W}}_i) \leq \mathrm{E}f(\tilde{\mathbf{W}}_i), i = 1, \ldots, n$, for any non-decreasing convex function $f(\cdot)$.

Proof

See Appendix 5.B.

The ordering relation stated in the above lemma adds to the modest collection of stochastic ordering results that are known for polling systems so far. The scarce results that are known exclusively refer to stochastic monotonicity properties of *global* performance measures, like the total amount of work in the system or the cycle time, or refer to monotonicity of quantities like the joint queue length at polling epochs with regard to the parameters of the service discipline, or with regard to the underlying stochastic processes. Levy et al.

[142] showed that the total amount of work in the system is decreasing in the degree of exhaustiveness of the service discipline. Altman et al. [8] proved that the cycle time and the joint queue length at polling epochs are stochastically increasing in the arrival rates, service times, and switch-over times. To the best of the author's knowledge, there are however no ordering results known at all for the individual waiting times of the nature of the ordering relation stated in the above lemma. One might be inclined to conjecture that also the individual waiting times are stochastically decreasing in the degree of exhaustiveness of the service discipline or increasing in the arrival rates, service times, and switch-over times, but such statements have either been disproved by simple counterexamples or have lacked proof so far.

5.5 The queue length

In Section 5.3, we expressed the pgf of the joint queue length distribution at the beginning of a cycle into the LST of the restricted cycle time distribution. In this section, we relate the joint queue length distribution at polling epochs to the joint queue length distribution at the beginning of a cycle. Thus, we obtain an expression for the pgf of the joint queue length distribution at polling epochs in terms of the LST of the restricted cycle time distribution. We analyze both the queue lengths at Q_1, \ldots, Q_n at a polling epoch at Q_i , and the queue lengths at Q_1, \ldots, Q_n . As described in Section 2.2, the marginal queue length distribution at Q_i at an arbitrary epoch may be derived from the waiting-time distribution at Q_i as obtained in (5.23).

We first introduce some notation. Denote by \mathbf{X}_{ij} the number of customers at Q_j at a polling epoch at Q_i , i.e., at the start of a visit to Q_i , i = 1, ..., n, j = 1, ..., n. Denote by \mathbf{D}_i the indicator function of the event that an arbitrary customer (that arrives to an empty system) arrives at Q_i , i.e., a stochastic variable which is 1 with probability λ_i/λ and 0 with probability $1 - \lambda_i/\lambda$, i = 1, ..., n.

By the nature of the globally gated service discipline, X_{ij} and Y_j , i = 1, ..., n, j = 1, ..., n, are related as follows.

$$\mathbf{X}_{ij} \stackrel{d}{=} \mathbf{Y}_j (i \le j) + \mathbf{A}_j (\sum_{h=1}^{i-1} (\mathbf{V}(\mathbf{Y}_h) + \mathbf{S}_h)) + \left(\mathbf{D}_j (i \le j) + \mathbf{A}_j (\sum_{h=1}^{i-1} \mathbf{D}_h \mathbf{B}_h) \right) \mathbf{I}_{\{\mathbf{Y}=0\}},$$
(5.26)

with $I_{\{Y=0\}}$ denoting the indicator function of the event $(Y_1, \ldots, Y_n) = (0, \ldots, 0)$, and $(i \leq j)$ is equal to 1 if $i \leq j$, and 0 if i > j. The notation was further introduced in Section 5.4.

5.5 The queue length

We first study the joint distribution of $\mathbf{X}_{i1}, \ldots, \mathbf{X}_{in}$, the queue lengths at Q_1, \ldots, Q_n at a polling epoch at $Q_i, i = 1, \ldots, n$. From (5.26),

$$E(z_1^{\mathbf{X}_{i1}} \dots z_n^{\mathbf{X}_{in}} | \mathbf{Y}_1, \dots, \mathbf{Y}_n) =$$

$$\prod_{h=1}^{i-1} \sigma_h(\epsilon(z)) \times \left[\prod_{h=1}^{i-1} \beta_h(\epsilon(z))^{\mathbf{Y}_h} \prod_{h=i}^n z_h^{\mathbf{Y}_h} - \left(1 - \sum_{h=1}^{i-1} \frac{\lambda_h}{\lambda} \beta_h(\epsilon(z)) - \sum_{h=i}^n \frac{\lambda_h}{\lambda} z_h \right) \mathbf{I}_{\{\mathbf{Y}=0\}} \right],$$
(5.27)

with $\epsilon(z_1,\ldots,z_n) = \sum_{j=1}^n \lambda_j(1-z_j), |z_j| \le 1, j = 1,\ldots,n.$ Unconditioning (5.27), using (5.6),

$$E(z_1^{\mathbf{X}_{i_1}} \dots z_n^{\mathbf{X}_{i_n}}) =$$

$$\prod_{h=1}^{i-1} \sigma_h(\epsilon(z)) \times \left[\gamma(\sum_{h=1}^{i-1} \lambda_h (1 - \beta_h(\epsilon(z)) + \sum_{h=i}^n \lambda_h (1 - z_h)) - \frac{\gamma(\lambda)}{\lambda} \left(\sum_{h=1}^{i-1} \lambda_h (1 - \beta_h(\epsilon(z))) + \sum_{h=i}^n \lambda_h (1 - z_h) \right) \right].$$
(5.28)

For i = n + 1, using (5.6) and interpreting $X_{n+1,j}$ as Y_j , j = 1, ..., n, (5.28) reduces to (5.8). From (5.28),

$$\operatorname{Cov}(\mathbf{X}_{ik}, \mathbf{X}_{il}) = \lambda_k \lambda_l \times \left[\operatorname{Var}(\sum_{h=1}^{i-1} \mathbf{S}_h) + \operatorname{EC} \sum_{h=1}^{i-1} \lambda_h \beta_h^{(2)} + (5.29) \right] \\ (\operatorname{E}\bar{\mathbf{C}}^2 - (\operatorname{EC})^2) \left(\sum_{h=1}^{i-1} \rho_h + (i \le k)\right) \left(\sum_{h=1}^{i-1} \rho_h + (i \le l)\right) \right].$$

For i = n + 1, using (5.12) and interpreting $\mathbf{X}_{n+1,j}$ as \mathbf{Y}_j , j = 1, ..., n, (5.29) reduces to

$$\operatorname{Cov}(\mathbf{Y}_k, \mathbf{Y}_l) = \lambda_k \lambda_l (\mathrm{E}\bar{\mathbf{C}}^2 - (\mathrm{E}\bar{\mathbf{C}})^2),$$

which may also be obtained from (5.6).

We finally study the joint distribution of $\mathbf{X}_{11}, \ldots, \mathbf{X}_{nn}$, the queue lengths at Q_1, \ldots, Q_n seen by the server at successive polling epochs at Q_1, \ldots, Q_n . From (5.26),

$$\mathbf{E}(z_1^{\mathbf{X}_{11}}\dots z_n^{\mathbf{X}_{nn}} \mid \mathbf{Y}_1,\dots,\mathbf{Y}_n) =$$
(5.30)

$$\prod_{i=1}^{n} \sigma_i \left(\sum_{h=i+1}^{n} \lambda_h (1-z_h)\right) \times \left[\prod_{i=1}^{n} \beta_i \left(\sum_{h=i+1}^{n} \lambda_h (1-z_h)\right)^{\mathbf{Y}_i} \prod_{i=1}^{n} z_i^{\mathbf{Y}_i} - \left(1 - \sum_{i=1}^{n} \frac{\lambda_i}{\lambda} z_i \beta_i \left(\sum_{h=i+1}^{n} \lambda_h (1-z_h)\right)\right) \mathbf{I}_{\{\mathbf{Y}=0\}}\right]$$

Unconditioning (5.30), using (5.6),

$$\mathbf{E}(z_1^{\mathbf{X}_{11}} \dots z_n^{\mathbf{X}_{nn}}) =$$

$$\prod_{i=1}^n \sigma_i \left(\sum_{h=i+1}^n \lambda_h (1-z_h)\right) \times \left[\gamma\left(\sum_{i=1}^n \lambda_i (1-z_i\beta_i(\sum_{h=i+1}^n \lambda_h (1-z_h)))\right) - \frac{\gamma(\lambda)}{\lambda} \left(\sum_{i=1}^n \lambda_i (1-z_i\beta_i(\sum_{h=i+1}^n \lambda_h (1-z_h)))\right)\right].$$
(5.31)

From (5.31),

$$\operatorname{Cov}(\mathbf{X}_{kk}, \mathbf{X}_{ll}) = \lambda_k \lambda_l \times \left[\operatorname{Cov}(\sum_{i=1}^{k-1} \mathbf{S}_i, \sum_{i=1}^{l-1} \mathbf{S}_i) + (5.32) \right]$$
$$\operatorname{EC}\left(q(k, l) + \sum_{i=1}^{\min(k, l) - 1} \lambda_i \beta_i^{(2)} \right) + (\operatorname{E}\bar{\mathbf{C}}^2 - (\operatorname{EC})^2) (\sum_{i=1}^{k-1} \rho_i + 1) (\sum_{i=1}^{l-1} \rho_i + 1) \right],$$

where q(k,l) is equal to β_k if k < l, β_l if k > l, and 0 if k = l.

Remark 5.5.1

Unlike in the ordinary non-dormant server case, cf. Boxma, Weststrate, & Yechiali [41], neither the queue lengths at a polling epoch, nor the queue lengths seen by the server at successive polling epochs need to be positively correlated. E.g., if all service times and switch-over times are zero, then (5.29) and (5.32) reduce to

$$\operatorname{Cov}(\mathbf{X}_{ik}, \mathbf{X}_{il}) = -\frac{\lambda_k \lambda_l}{\lambda^2} (i \le k) (i \le l),$$

 \mathbf{and}

$$\operatorname{Cov}(\mathbf{X}_{kk},\mathbf{X}_{ll})=-rac{\lambda_k\lambda_l}{\lambda^2},$$

respectively. This may also be seen immediately, as during every cycle exactly one customer is served, which occurs at Q_i with probability λ_i/λ , i = 1, ..., n.

5.A Proof of Lemma 5.3.1

APPENDICES

5.A Proof of Lemma 5.3.1

Lemma 5.3.1 *i.* $\lim_{K \to \infty} \delta^{(K)}(\omega) = 0$ for all ω with $\operatorname{Re} \omega \ge 0$. *ii.* $\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega))$ converges for all ω with $\operatorname{Re} \omega \ge 0$. *iii.* $\sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega))$ converges for all ω with $\operatorname{Re} \omega \ge 0$.

Proof

For any non-negative valued stochastic variable **H**, having distribution $H(\cdot)$ with LST $\eta(\cdot)$, holds

$$|1 - \eta(\omega)| = |\int_{t=0}^{\infty} (1 - e^{-\omega t}) dH(t)|$$

$$\leq \int_{t=0}^{\infty} |1 - e^{-\omega t}| dH(t)$$

$$\leq \int_{t=0}^{\infty} |\omega t| dH(t)$$

$$= \mathbf{EH} |\omega|, \qquad (5.33)$$

for all ω with $\operatorname{Re} \omega \geq 0$, since $|1 - e^{-z}| \leq |z|$ for all z with $\operatorname{Re} z \geq 0$. *Proof of i.*

Re
$$\delta^{(k)}(\omega) \ge 0, \qquad k = 0, 1, \dots,$$
 (5.34)

for all ω with Re $\omega \geq 0$. Further, using (5.33),

$$| \delta(\omega) | = | \sum_{i=1}^{n} \lambda_i (1 - \beta_i(\omega)) |$$

$$\leq \sum_{i=1}^{n} \lambda_i | 1 - \beta_i(\omega) |$$

$$\leq \sum_{i=1}^{n} \lambda_i \beta_i | \omega |$$

$$= \rho | \omega |,$$

for all ω with $\operatorname{Re} \omega \geq 0$. Hence, by induction, using (5.34),

$$|\delta^{(k)}(\omega)| \le \rho^k |\omega|, \qquad k = 0, 1, \dots,$$
 (5.35)

for all ω with $\operatorname{Re} \omega \geq 0$.

As $\rho < 1$ is assumed to hold, we conclude that $\lim_{K \to \infty} \delta^{(K)}(\omega) = 0$ for all ω with $\operatorname{Re}\omega\geq 0.$

Proof of ii.

From the theory of infinite products, cf. Titchmarsh [185] p. 18, we have that $\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega)) \text{ converges iff } \sum_{k=0}^{\infty} [1 - \sigma(\delta^{(k)}(\omega))] \text{ converges.}$ Using (5.33), (5.35), (5.34), and the assumption that $\rho < 1$,

$$|\sum_{k=0}^{\infty} [1 - \sigma(\delta^{(k)}(\omega))]| \leq \sum_{k=0}^{\infty} |1 - \sigma(\delta^{(k)}(\omega))|$$
$$\leq \sum_{k=0}^{\infty} s |\delta^{(k)}(\omega)|$$
$$\leq s \sum_{k=0}^{\infty} \rho^{k} |\omega|$$
$$= \frac{s}{1 - \rho} |\omega|$$
$$< \infty,$$

for all ω with $\operatorname{Re} \omega \geq 0$. So $\prod_{k=0}^{\infty} \sigma(\delta^{(k)}(\omega))$ converges for all ω with $\operatorname{Re} \omega \geq 0$. Proof of iii.

Because of (5.34),

$$|\sigma(\delta^{(k)}(\omega))| \le 1, \qquad k = 0, 1, \dots,$$
 (5.36)

for all ω with Re $\omega \ge 0$. Using (5.35), (5.36), and the assumption that $\rho < 1$,

$$\begin{aligned} |\sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega))| &\leq \sum_{k=0}^{\infty} |\delta^{(k+1)}(\omega)| \prod_{l=0}^{k} |\sigma(\delta^{(l)}(\omega))| \\ &\leq \sum_{k=0}^{\infty} \rho^{k+1} |\omega| \\ &= \frac{\rho}{1-\rho} |\omega| \\ &< \infty, \end{aligned}$$

for all ω with $\operatorname{Re} \omega \geq 0$. So $\sum_{k=0}^{\infty} \delta^{(k+1)}(\omega) \prod_{l=0}^{k} \sigma(\delta^{(l)}(\omega))$ converges for all ω with $\operatorname{Re} \omega \geq 0$.

90

5.B Proof of Lemma 5.4.1

Lemma 5.4.1

 $\hat{\mathbf{W}}_i \leq_{\mathrm{icx}} \hat{\mathbf{W}}_i, \qquad i=1,\ldots,n,$

i.e., $\mathrm{E}f(\hat{\mathbf{W}}_i) \leq \mathrm{E}f(\tilde{\mathbf{W}}_i), i = 1, \ldots, n$, for any non-decreasing convex function $f(\cdot)$.

Proof

We sketch the intuitive idea of the proof. For a detailed technical proof, we refer to appendix B of Borst [26]. We assume the arrival, service, and switch-over processes in the dormant and non-dormant server case to be coupled as follows. In both cases, the server experiences exactly the same switch-over times, but - since the dormant and non-dormant server case evolve according to different operational rules - the same switch-over time is not necessarily experienced at the same point in time, i.e., the switch-over times may be shifted in time. Moreover, in the dormant server case, when the server is actually idling, we assume that the server is experiencing a switch-over time, which is however immediately interrupted as soon as a new customer arrives, just as if the server would have been idling, awaiting a new customer to arrive. The remainder of the switch-over time is then resumed as soon as the server starts idling again. During one and the same switch-over time, the arrival processes in both cases proceed synchronously, i.e., the same customer arrives at the same relative time (with regard to the switch-over time in question), requiring the same service time. Thus, the server also provides exactly the same service times in both cases, but – since the dormant and non-dormant server case evolve according to different operational rules – the same service time is not necessarily provided at the same point in time. Also, during one and the same service time, the arrival processes in both cases proceed synchronously. So the arrivals in both cases may be shifted in time, however, congruently to the service or switch-over times in which they fall, so that the same customer arrives at the same relative time with regard to the service time or switch-over time in question. By the memoryless property of the Poisson process, the coupling does not affect the stochastic properties of the arrival process. Neither does the coupling affect the stochastic properties of the service and switch-over processes. Thus, we obtain coupled but still marginally unbiased induced stochastic processes (like waiting times and queue lengths) in the dormant and non-dormant server case. Suppose that at time $t = T_0$ in both cases the system is empty, and the server is at its home base Q_1 , just back from switching. The server then starts switching for a time of length S_0 . S_0 , S_1 , S_2 , ... are independent stochastic variables with common distribution $S(\cdot)$. During the switch-over time S_0 , a number of **K** customers arrive, let us say C_1, \ldots, C_K , at (relative) time $t = A_1$, $t = A_1 + A_2, \dots, t = A_1 + \dots + A_K$, requiring service times of length B_1 , $\mathbf{B}_2, \ldots, \mathbf{B}_{\mathbf{K}}, \mathbf{A}_1, \mathbf{A}_2, \ldots$ are independent exponentially distributed stochastic variables with mean $1/\lambda$. **B**₁, **B**₂,... are independent stochastic variables with common distribution $\sum_{i=1}^{n} \frac{\lambda_i}{\lambda} B_i(\cdot)$.

In the dormant server case, at time $t = T_0 + A_1$, the server interrupts switching, suspending the remainder $S_0 - A_1$ of the switch-over time S_0 , and starts a cycle along the queues to serve the newly arrived customer C_1 , just as if the server would have remained idling at Q_1 from time $t = T_0$ on, awaiting the new customer to arrive. At time $t = T_1$, after L_1 cycles, the system is empty again, and the server is back again at its home base Q_1 (these events occurring simultaneously for the first time). The server then starts switching, resuming the switch-over time S_0 .

At time $t = \mathbf{T}_1 + \mathbf{A}_2$, the server again interrupts switching, suspending the remainder $\mathbf{S}_0 - \mathbf{A}_1 - \mathbf{A}_2$ of the switch-over time \mathbf{S}_0 , and starts a cycle along the queues to serve the newly arrived customer C_2 , again just as if the server would have remained idling at Q_1 from time $t = \mathbf{T}_1$ on, awaiting the new customer to arrive.

In the non-dormant server case, at time $t = \mathbf{T}_0 + \mathbf{A}_1$ the server just continues switching, disregarding the newly arrived customer C_1 . At time $t = \mathbf{T}_0 + \mathbf{S}_0$, the server finishes switching, and starts a cycle along the queues to serve the newly arrived customers $C_1, \ldots, C_{\mathbf{K}}$.

In the dormant server case, at time $t = \mathbf{T}_{\mathbf{K}}$, after $\mathbf{L}_1 + \ldots + \mathbf{L}_{\mathbf{K}}$ cycles, the system is empty, and the server is back at its home base Q_1 (these events occurring simultaneously for the K-th time). The server then starts switching, resuming the switch-over time \mathbf{S}_0 . At time $t = \mathbf{U}_0$, the server finishes switching, $\mathbf{U}_0 = \mathbf{T}_{\mathbf{K}} + \mathbf{D}$, $\mathbf{D} = \mathbf{S}_0 - \mathbf{A}_1 - \ldots - \mathbf{A}_{\mathbf{K}}$. At time $t = \mathbf{U}_0$, also in the non-dormant server case the system is empty, and the server just finishes switching. In both cases, the server has then experienced exactly the same switch-over times, viz., $\mathbf{S}_0, \mathbf{S}_1, \ldots, \mathbf{S}_{\mathbf{L}_1 + \ldots + \mathbf{L}_{\mathbf{K}}}$, and has provided exactly the same service times, viz., the service times of the customers arriving during $\mathbf{S}_0, \mathbf{S}_1, \ldots, \mathbf{S}_{\mathbf{L}_1 + \ldots + \mathbf{L}_{\mathbf{K}}}$, and of their descendants. Let us say the total number of type-*i* customers among them is \mathbf{M}_i . (Here the descendants of a customer are recursively defined as the customers arriving during its service time or during the service time of one of its descendants.) Concluding, at time $t = \mathbf{U}_0$, in *both* cases, the system is empty, and the server is at its home base Q_1 , just back from switching.

Let $R_i^{(h)}$ be the *h*-th type-*i* customer served from time $t = \mathbf{T}_0$ on in the dormant server case, $h = 1, 2, \ldots$ Denote by $\hat{\mathbf{W}}_i^{(h)}$ and $\tilde{\mathbf{W}}_i^{(h)}$ the waiting time of $R_i^{(h)}$ in the dormant and non-dormant server case, respectively, $h = 1, 2, \ldots$ As the stochastic processes $\{\hat{\mathbf{W}}_i^{(h)}, h = 1, 2, \ldots\}$ and $\{\tilde{\mathbf{W}}_i^{(h)}, h = 1, 2, \ldots\}$ are regenerative with regard to h = 1 and $h = \mathbf{M}_i + 1$,

$$\mathbf{E}f(\hat{\mathbf{W}}_i) = \frac{1}{\mathbf{E}\mathbf{M}_i} \mathbf{E}(\sum_{h=1}^{\mathbf{M}_i} f(\hat{\mathbf{W}}_i^{(h)})), \tag{5.37}$$

and



FIGURE 5.1. A dormant server.

FIGURE 5.2. A non-dormant server.

Consider now Figure 5.1 and Figure 5.2, representing the customer offspring process in the dormant and non-dormant server case, respectively. In both cases, dots at the same (horizontal) level correspond to customers served in the same cycle. An arc indicates that the customer at the head arrives during the service time of the customer at the tail. A dot without any incoming arc represents a customer which arrives during a switch-over time, or, in the dormant server case, during an idling period. A dot without any outgoing arc represents a customer requiring a service time during which no single customer arrives.

To prove that (5.38) majorizes (5.37), we need to introduce some additional terminology. In the dormant server case, the interval from time $t = \mathbf{T}_{k-1}$ to time $t = \mathbf{T}_k$, comprising the cycles $\mathbf{L}_1 + \ldots + \mathbf{L}_{k-1} + 1$ to $\mathbf{L}_1 + \ldots + \mathbf{L}_k$, is referred to as the k-th busy interval, $k = 1, \ldots, \mathbf{K}$. Customers arriving during \mathbf{S}_0 (thus interrupting \mathbf{S}_0 in the dormant server case), together with their descendants, are called *primary* customers (corresponding to the light-grey dots in Figure 5.1 and Figure 5.2). The remaining customers, i.e., customers arriving during during $\mathbf{S}_1, \ldots, \mathbf{S}_{\mathbf{L}_{\mathbf{K}}}$, together with their descendants, are called *secondary* customers (the dark-grey dots).

With the busy intervals as background, the dormant and non-dormant server case differ in the service of primary customers, but not in the service of secondary customers. In the dormant server case, the service of primary customers is balanced over the **K** busy intervals. In the non-dormant server case, the service of primary customers is concentrated in the first busy interval. The service of secondary customers occurs in both cases in corresponding cycles. So, in the non-dormant server case, the primary customers all bother one another, and all bother the same secondary customers. To make the latter intuitive idea precise, observe that the waiting time of every customer is composed of switch-over time, service time of primary customers, and service time of secondary customers. For any waiting time $\mathbf{W}_i^{(h)}$, denote by $\mathbf{V}_{i,k}^{(h)}$ the share constituted by service time of primary customers that are descendants of C_k , $k = 1, \ldots, \mathbf{K}$. Denote by $\mathbf{V}_i^{(h)}$ the remaining share in $\mathbf{W}_i^{(h)}$, i.e., the share constituted by switch-over time and service time of secondary customers. For a primary customer $R_i^{(h)}$ we have

$$\hat{\mathbf{V}}_{i}^{(h)} + \mathbf{Z}_{i}^{(h)} \stackrel{d}{=} \tilde{\mathbf{V}}_{i}^{(h)}, \qquad \mathbf{Z}_{i}^{(h)} \ge 0.$$
 (5.39)

Assuming that $R_i^{(h)}$ is a descendant of C_k ,

$$\hat{\mathbf{V}}_{i,k}^{(h)} = \tilde{\mathbf{V}}_{i,k}^{(h)}, \qquad \hat{\mathbf{V}}_{i,m}^{(h)} = 0, \qquad m \neq k.$$
(5.40)

For a secondary customer $R_i^{(h)}$ we have

h

$$\hat{\mathbf{V}}_i^{(h)} = \tilde{\mathbf{V}}_i^{(h)}.\tag{5.41}$$

Let $H_{i,l}$ be the index set of the secondary type-*i* customers served in the *l*-th cycle in the dormant server case, $l = 1, \ldots, \mathbf{L}_1 + \ldots + \mathbf{L}_{\mathbf{K}}$. Let $H_{i,kl}$ be the index set of the secondary type-*i* customers served in the $(\mathbf{L}_1 + \ldots + \mathbf{L}_{k-1} + l$ -th cycle in the dormant server case if $l \leq \mathbf{L}_k$, $k = 1, \ldots, \mathbf{K}$. If $l > \mathbf{L}_k$, let $H_{i,kl}$ be the empty set. We then have

$$\sum_{k \in H_{i,kl}} \hat{\mathbf{V}}_{i,k}^{(h)} \stackrel{d}{=} \sum_{h \in H_{i,l}} \tilde{\mathbf{V}}_{i,k}^{(h)}, \qquad \sum_{h \in H_{i,kl}} \hat{\mathbf{V}}_{i,m}^{(h)} = 0, \qquad m \neq k.$$
(5.42)

The relations (5.39), (5.40), (5.41), and (5.42) constitute the key elements in proving that (5.38) majorizes (5.37). For a detailed comparison, we refer to appendix B of Borst [26].

CHAPTER 6

Optimization of k-limited service strategies

6.1 INTRODUCTION

In the present chapter, we consider a polling system with a k-limited service strategy. Under k-limited service, when visiting a queue, the server works until either a prespecified number of k customers have been served, or the queue becomes empty, whichever occurs first. We are interested in the problem of determining appropriate values for the service limits that contribute to an efficient operation of the system.

As described in Section 1.2, polling systems may be used to model communication systems with a conflict-free medium access protocol. The standard control mechanism to regulate the operation of those systems, is either to limit the *number* of transmissions, or to limit the transmission *time* granted to the queues. As a major benefit, by setting such service limits, one may accomplish a bound on the cycle time, i.e., the time until a queue acquires the right of transmission again. Moreover, one may effectuate some kind of priorization, by assigning different service limits to different queues, according to their relative importance.

Although the use of service limits is very standard, it is still not very well understood how the limits should be set so as to optimize the performance of such communication systems. Driven by these considerations, we address in the present chapter the problem how in a polling system with k_i -limited service at Q_i , the limits should be set so as to minimize the waiting cost as performance measure. It appears that if we do not impose any constraint on the k_i 's, then at least one of the optimal k_i 's is always infinite. To accomplish a bound on the cycle time, we therefore also study a version of the problem with a constraint

of the form $\sum_{i=1}^{n} \gamma_i k_i \leq K$.

Although similar in their operation, limits on the visit time are analytically even harder to handle than limits on the number of services during a visit. Therefore, k_i -limited service is frequently used as an approximation of T_i -limited service, with $k_i \approx T_i/\beta_i$, the exact value depending on whether or not service is preempted when the timer expires. Consequently, the optimal k_i 's are likely to provide a reasonable indication for the optimal T_i 's. Note that for deterministic service times, corresponding to the practically relevant case of transmission of packets of fixed length, k_i -limited service and T_i -limited service even coincide. Blanc & Van der Mei [23] consider a similar optimization problem in a polling system with Bernoulli service. Under Bernoulli service with parameter q_i , when visiting Q_i , S serves at least one customer, and at every service completion Sserves yet another customer with probability q_i , and leaves Q_i with probability $1 - q_i$; S also leaves Q_i when it becomes empty. The number of potential services being geometrically distributed with mean $1/(1-q_i)$, Bernoulli service may be used as an emulation of k_i -limited service, with $q_i = 1 - 1/k_i$. Note that Bernoulli service and k_i -limited service even coincide for $q_i = 0$ as well as $q_i = 1$.

Polling systems with limited-type service strategies are extremely hard to analyze, not to mention optimize. As noted in Section 1.4, limited-type service policies do not satisfy Property 1.4.1, which provides a global characterization of the class of service disciplines that are amenable to an exact analysis. Indeed, polling systems with limited-type service strategies have not allowed an exact analysis, apart from some special cases, like two-queue cases and completely symmetric cases. Eisenberg [85] studies a two-queue model with zero switch-over times and 1-limited service at both queues, transforming the problem of finding the joint queue length distribution into the problem of solving a singular Fredholm integral equation. Cohen & Boxma [74] analyze the same model, translating the problem into a Riemann-Hilbert boundary value problem. Using similar techniques, Boxma [37] studies a symmetric two-queue model with non-zero switch-over times and 1-limited service at both queues. Boxma & Groenendijk [43] analyze an asymmetric two-queue model with nonzero switch-over times and 1-limited service at both queues by formulating a Riemann boundary value problem. Fuhrmann [100] derives the mean waiting time in a completely symmetric system with 1-limited service and an arbitrary number of queues. The two-queue model with 1-limited service at one queue and *exhaustive* service at the other has turned out to be relatively simple to analyze, cf. Groenendijk [114] Section 6.3. The two-queue model with 1-limited service at one queue and gated service at the other, however, appears to defy an exact analysis. Polling systems with time-limited service have not yielded to an exact analysis either. See Coffman et al. [71] for an interesting analysis of a two-queue model with exponentially distributed time limits.

The fact that limited-type service policies are extremely hard to analyze, has considerably added to the importance of pseudo-conservation laws, which provide relatively simple exact expressions for a specific weighted sum of the mean waiting times, cf. Section 2.3. Thus, pseudo-conservation laws are an important instrument in constructing and validating waiting-time approximations. For the case of 1-limited service at each of the queues, the pseudo-conservation law was originally derived in Watson [187]. As described in Section 2.3, Boxma & Groenendijk [42] developed an approach to obtain a pseudo-conservation law in a considerably more general framework, covering 1-limited service at each of the queues. Boxma & Meister [52] use the pseudo-conservation law to derive waiting-time approximations for 1-limited service. Groenendijk [114] presents a more refined procedure to compute such approximations. For the general case of k-limited service, the pseudo-conservation law still contains an unknown term. Fuhrmann & Wang [104] obtain waiting-time approximations for k-limited service by bounding that term. Everitt [87], [89] derives such approximations by approximating that term. Chang & Sandhu [66] present a more refined procedure to calculate waiting-time approximations for k-limited service.

The remainder of the chapter is organized as follows. In Section 6.2, we present a detailed model description, and formulate a pseudo-conservation law for the mean waiting times. In Section 6.3, we propose four different approximative approaches to the constrained optimization problem, based on four different approximations for the mean waiting times. These approaches are numerically investigated in Section 6.4. In Section 6.5, we discuss some properties of polling systems with k-limited service, and establish a (partially conjectured) $c\mu$ -like rule for the unconstrained optimization problem. We then also propose an approximative approach to this problem. This approach is numerically examined in Section 6.6. In Section 6.7, we conclude with some remarks and suggestions for further research.

6.2 MODEL DESCRIPTION AND PRELIMINARIES

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by a single server S. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3.

The server visits the queues in strictly cyclic order, Q_1, \ldots, Q_n . When visiting Q_i , S works until either k_i customers have been served or the queue becomes empty, whichever occurs first. Note that the case $k_i = \infty$ amounts to ordinary exhaustive service. Fuhrmann & Wang [104] call this discipline exhaustive-limited service, as opposed to gated-limited service in case of which S, when meeting l_i customers at Q_i upon arrival, only serves min $\{l_i, k_i\}$ of them. In case of G-limited service, $k_i = \infty$ amounts to ordinary gated service.

As mentioned in Section 1.3, a necessary and sufficient condition for stability is $\lambda_i s/(1-\rho) < k_i, i = 1, ..., n$. Throughout the chapter, the stability condition is assumed to hold.

Denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer, i = 1, ..., n. Let c_i represent the waiting cost per unit of time of a type-*i* customer, i = $1, \ldots, n$. The mean total waiting cost per unit of time amounts to $\sum_{i=1}^{n} c_i \lambda_i \mathbf{EW}_i$. In this chapter, we are interested in the problem of finding the values for k_1, \ldots, k_n that minimize that quantity, both for the constrained case with $\sum_{i=1}^{n} \gamma_i k_i \leq K$, and for the unconstrained case. Note that taking $\gamma_i \equiv 1$ puts a limit on the number of services in a cycle. Choosing $\gamma_i \equiv \beta_i$ yields, for deterministic service and switch-over times, a bound on the cycle time.

Everitt [87] has derived the following pseudo-conservation law for the mean waiting times:

$$\sum_{i=1}^{n} \rho_i \left(1 - \frac{\lambda_i s}{k_i (1-\rho)} \right) \mathbf{E} \mathbf{W}_i = D + \frac{s}{1-\rho} \sum_{i=1}^{n} \frac{\rho_i^2}{k_i} - \sum_{i=1}^{n} \frac{\rho_i (1-\rho_i) g_i^{(2)}}{2\lambda_i k_i}, (6.1)$$

with

$$D = \rho \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\rho)} + \rho \frac{s^{(2)}}{2s} + \frac{s}{2(1-\rho)} \left[\rho^2 - \sum_{i=1}^{n} \rho_i^2 \right],$$
(6.2)

and $g_i^{(2)} = E\mathbf{M}_i(\mathbf{M}_i - 1)$, with \mathbf{M}_i denoting the number of customers served during a visit to Q_i . So for $k_i = 1$, $g_i^{(2)} = 0$, but for $k_i \neq 1$, $g_i^{(2)}$ is not known exactly.

6.3 The constrained optimization problem

In this section, we study the problem of finding the service limits k_1, \ldots, k_n , constrained to $\sum_{i=1}^n \gamma_i k_i \leq K$, that minimize the mean total waiting cost $\sum_{i=1}^n c_i \lambda_i E \mathbf{W}_i$. We successively consider the following four waiting-time approximations:

- I. an approximation based on a 1-limited polling table;
- II. a simple k-limited approximation;

accurate indication for $\bar{k}_1^*, \ldots, \bar{k}_n^*$.

III. a Fuhrmann & Wang-like k-limited approximation;

IV. the original Fuhrmann & Wang k-limited approximation.

In fact, the approximations ignore the integrality of k_1, \ldots, k_n . The numbers k_1, \ldots, k_n , not necessarily integers, that minimize the resulting approximations for $\sum_{i=1}^{n} c_i \lambda_i \mathbf{E} \mathbf{W}_i$, are denoted by k_1^*, \ldots, k_n^* . The integers k_1, \ldots, k_n that minimize the resulting approximations for $\sum_{i=1}^{n} c_i \lambda_i \mathbf{E} \mathbf{W}_i$, are denoted by $\bar{k}_1^*, \ldots, \bar{k}_n^*$. As the approximations are rather smooth, k_1^*, \ldots, k_n^* are likely to provide an
6.3 The constrained optimization problem

I. An approximation based on a 1-limited polling table.

A generalization of the cyclic visit order considered so far is a fixed, generally non-cyclic, visit order. Such a visit order may be described in a *polling table*, which may contain $m_i \ge 1$ visits to Q_i . Our approximation idea is the following. There is some resemblance between adopting the k_i -limited service discipline at Q_i , visiting Q_i once, and adopting the 1-limited service discipline at Q_i , visiting Q_i k_i times; in either case, the server is allowed to serve at most k_i customers in one 'cycle'. So the optimal visit numbers m_1, \ldots, m_n for the 1-limited service discipline may provide an indication for k_1^*, \ldots, k_n^* .

Boxma, Levy, & Weststrate [48] study the problem of finding those polling table visit numbers m_1, \ldots, m_n , that minimize $\sum_{i=1}^n c_i \lambda_i E \mathbf{W}_i$. They propose the following waiting-time approximation, under the assumption that the m_i visits to Q_i are spaced as evenly as possible:

$$\mathbf{EW}_{i} \approx A \frac{1 - \rho + \rho_{i}}{1 - \rho - \lambda_{i} \frac{\sum m_{j} s_{j}}{m_{i}}} \frac{\mathbf{EC}}{m_{i}}, \qquad i = 1, \dots, n,$$
(6.3)

with $EC = \sum_{j=1}^{n} m_j s_j / (1 - \rho)$ the mean cycle time, and A some unknown constant. The constant A could be determined using the pseudo-conservation law

for polling tables, cf. [45], but its value is not relevant for the determination of the optimal values of m_i , which follow easily from (6.3):

$$m_{i}^{*} = \lambda_{i}R + (1 - \rho - \sum_{j=1}^{n} \lambda_{j}s_{j})R \frac{\sqrt{c_{i}\lambda_{i}(1 - \rho + \rho_{i})/s_{i}}}{\sum_{j=1}^{n} s_{j}\sqrt{c_{j}\lambda_{j}(1 - \rho + \rho_{j})/s_{j}}}.$$
 (6.4)

Here R represents an arbitrary scaling factor, reflecting the homogeneity of the objective function in m_1, \ldots, m_n .

As remarked earlier, the optimal visit numbers m_1, \ldots, m_n for the 1-limited service discipline may provide an indication for k_1^*, \ldots, k_n^* . However, visiting Q_i k_i times in a cycle differs from visiting Q_i only once, in the respect of the switch-over time incurred. In the former (latter) case, the switch-over time corresponding to Q_i is incurred k_i times (once) per cycle. So the optimal visit numbers m_1, \ldots, m_n may be better candidates to provide an indication for k_1^*, \ldots, k_n^* , when the mean switch-over times in (6.3) are scaled by a factor $1/m_i$; this gives

$$\mathbf{E}\mathbf{W}_{i} \approx A \frac{1-\rho+\rho_{i}}{1-\rho-\lambda_{i}} \frac{\mathbf{E}\mathbf{C}}{m_{i}}, \qquad i=1,\dots,n,$$
(6.5)

with $EC = s/(1 - \rho)$. This leads to

$$m_{i}^{*} = \frac{\lambda_{i}s}{1-\rho} + (K - \sum_{j=1}^{n} \gamma_{j} \frac{\lambda_{j}s}{1-\rho}) \frac{\sqrt{c_{i}\lambda_{i}(1-\rho+\rho_{i})/\gamma_{i}}}{\sum_{j=1}^{n} \gamma_{j} \sqrt{c_{j}\lambda_{j}(1-\rho+\rho_{j})/\gamma_{j}}}.$$
 (6.6)

One may interpret (6.6) as follows. The server should be allowed to serve at least $\frac{\lambda_i s}{1-\rho}$ customers during a visit to Q_i , to satisfy the stability condition. The

remaining service capacity, $K - \sum_{j=1}^{n} \gamma_j \frac{\lambda_j s}{1-\rho}$, should be assigned proportionally

to $\sqrt{c_i \lambda_i (1 - \rho + \rho_i)/\gamma_i}$. Some reflection indicates that a queue with relatively high c_i , λ_i , ρ_i , or $1/\gamma_i$ should indeed be assigned a relatively high capacity.

Rule (6.6) is just as simple as (6.4) but yields better results. Still, the numerical results in Section 6.4 reveal that it does not always perform well. Below we investigate a quite different idea.

II. A simple k-limited approximation.

We now imitate the derivation of the waiting-time approximation for cyclic polling systems with 1-limited service [52] and for polling tables with 1-limited service ([48], leading to (6.3)). The waiting time of a (tagged) type-*i* customer is composed of:

i. the time from its arrival to the start of the subsequent visit of the server to Q_i , i.e., a residual cycle time \mathbf{RC}_i with regard to Q_i ;

ii. the time from the start of the latter visit to its service, i.e., approximately \mathbf{X}_i/k_i cycle times \mathbf{C}_i^+ with regard to Q_i (atypical cycles, as each contains k_i services at Q_i), when the (tagged) customer finds \mathbf{X}_i waiting type-*i* customers upon arrival.

Applying a traffic balance argument, $\mathrm{EC}_i^+ \approx k_i \beta_i + s + (\rho - \rho_i) \mathrm{EC}_i^+$. Noting that $\mathrm{EX}_i = \lambda_i \mathrm{EW}_i$, we thus obtain

$$\mathbf{E}\mathbf{W}_{i} \approx \frac{1-\rho+\rho_{i}}{1-\rho-\frac{\lambda_{i}}{k_{i}}s} \mathbf{E}\mathbf{R}\mathbf{C}_{i}, \qquad i=1,\dots,n.$$
(6.7)

For $k_i = 1$, (6.7) reduces to $\mathrm{EW}_i \approx \frac{1-\rho+\rho_i}{1-\rho-\lambda_i s} \mathrm{ERC}_i$, the known approximation [52] for the 1-limited service discipline. However, for $k_i = \infty$, (6.7) reduces to $\mathrm{EW}_i \approx \frac{1-\rho+\rho_i}{1-\rho} \mathrm{ERC}_i$, rather than $\mathrm{EW}_i = (1-\rho_i)\mathrm{ERC}_i$, the known exact result for the exhaustive service discipline (defining a cycle with regard to Q_i as the interval between two successive departures of S from Q_i). The reason for this discrepancy is that the derivation of (6.7) ignores the possibility that an arriving type-*i* customer finds S visiting Q_i , and can still be served during that visit.

Starting from (6.7), assuming $\text{ERC}_i \approx \text{ERC} = B \text{ EC} = Bs/(1-\rho)$, with B some unknown constant,

$$k_{i}^{*} = \frac{\lambda_{i}s}{1-\rho} + (K - \sum_{j=1}^{n} \gamma_{j} \frac{\lambda_{j}s}{1-\rho}) \frac{\lambda_{i} \sqrt{c_{i}(1-\rho+\rho_{i})/\gamma_{i}}}{\sum_{j=1}^{n} \gamma_{j} \lambda_{j} \sqrt{c_{j}(1-\rho+\rho_{j})/\gamma_{j}}}.$$
 (6.8)

6.3 The constrained optimization problem

One may interpret (6.8) similarly to (6.6).

Note that (6.8) slightly differs from (6.6) in the proportional assignment of the remaining service capacity, $K - \sum_{j=1}^{n} \gamma_j \frac{\lambda_j s}{1-\rho}$, which may be explained as follows. Visiting $Q_i k_i$ times differs from visiting Q_i only once, not only in the

respect of the switch-over time incurred, as remarked earlier, but also differs in the respect of the residual time until visiting Q_i . In the former case, the residual subcycle time is assumed to behave inversely proportional to k_i , whereas in the latter case the residual cycle time is assumed to behave constantly.

III. A Fuhrmann & Wang-like k-limited approximation.

To remedy the weakness of (6.7) indicated above, a natural heuristic approach is to take a weighted sum of the 1-limited waiting-time approximation $\mathrm{EW}_i \approx \frac{1-\rho+\rho_i}{1-\rho-\lambda_i s}\mathrm{ERC}_i$ and the exhaustive mean waiting-time result $\mathrm{EW}_i = (1-\rho_i)\mathrm{ERC}_i$, with weight factors $u_i(k_i)$ and $1-u_i(k_i)$, respectively. The choice $u_i(k_i) = \frac{1-\rho-\lambda_i s}{k_i(1-\rho)-\lambda_i s}$ has the desirable properties that $u_i(1) = 1, u_i(\infty) = 0$, and $\mathrm{EW}_i \to \infty$ for $k_i \to \frac{\lambda_i s}{1-\rho}$.

This in facts yields the approximation (30) of Fuhrmann & Wang [104]:

$$\mathbf{EW}_{i} \approx \frac{(1-\rho_{i})(1-\rho) + \frac{\rho_{i}}{k_{i}}(2-\rho)}{1-\rho - \frac{\lambda_{i}s}{k_{i}}} \mathbf{ERC}_{i}, \qquad i = 1, \dots, n.$$
(6.9)

Starting from (6.9), assuming $\text{ERC}_i \approx \text{ERC} = B \text{ EC}$, with B some unknown constant,

$$k_i^* = \frac{\lambda_i s}{1 - \rho} + \left(K - \sum_{j=1}^n \gamma_j \frac{\lambda_j s}{1 - \rho}\right) \frac{\sqrt{c_i \lambda_i \delta_i / \gamma_i}}{\sum_{j=1}^n \gamma_j \sqrt{c_j \lambda_j \delta_j / \gamma_j}},\tag{6.10}$$

with $\delta_i = \rho_i(2-\rho) + \lambda_i s(1-\rho_i)$. One may interpret (6.10) similarly to (6.6).

IV. The original Fuhrmann & Wang k-limited approximation.

Fuhrmann & Wang [104] also assume $\text{ERC}_i \approx \text{ERC}$, but they do not assume ERC = B EC. Instead, they approximate ERC by substituting (6.9) into (6.1), taking $g_i^{(2)} = 0$,

$$\mathbf{ERC} \approx \frac{D + \frac{s}{1 - \rho} \sum_{j=1}^{n} \frac{\rho_j^2}{k_j}}{\sum_{j=1}^{n} [\rho_j (1 - \rho_j) + \frac{\rho_j^2}{k_j} \frac{2 - \rho}{1 - \rho}]}.$$
(6.11)

Chapter 6 Optimization of k-limited service strategies

2

Remember that $g_i^{(2)} = \mathrm{E}\mathbf{M}_i(\mathbf{M}_i - 1) \ge (\mathrm{E}\mathbf{M}_i)^2 - \mathrm{E}\mathbf{M}_i = \left(\frac{\lambda_i s}{1-\rho}\right)^2 - \frac{\lambda_i s}{1-\rho}$. Taking $g_i^{(2)} = \max\{0, \left(\frac{\lambda_i s}{1-\rho}\right)^2 - \frac{\lambda_i s}{1-\rho}\}$ in (6.1) would probably improve the accuracy of (6.11). We did however not consider this option, as the numerical results in Section 6.4 show that the rule based on (6.11) performs already very well.

Substituting (6.11) back into (6.9),

$$\mathrm{E}\mathbf{W}_{i} \approx \frac{(1-\rho_{i})(1-\rho) + \frac{\rho_{i}}{k_{i}}(2-\rho)}{1-\rho - \frac{\lambda_{i}s}{k_{i}}} \frac{D + \frac{s}{1-\rho} \sum_{j=1}^{n} \frac{\rho_{j}^{2}}{k_{j}}}{\sum_{j=1}^{n} [\rho_{j}(1-\rho_{j}) + \frac{\rho_{j}^{2}}{k_{j}} \frac{2-\rho}{1-\rho}]}.$$
 (6.12)

The optimal service limits based on (6.12) cannot be solved analytically, but can easily be determined numerically.

In the next section, we test the simple rules (6.6), (6.8), (6.10), and the rule based on (6.12).

Remark 6.3.1

Fuhrmann & Wang [104] concentrate on k-limited service under a gated regime at all queues. They use the reasoning leading to our approximation II, observe the discrepancy for $k_i = \infty$ (the reason for which is explained above (6.8)), and then modify their approximation in a way that amounts to our taking a weighted sum. Tedijanto [181] considers cyclic polling systems with a Bernoulli service policy. He proposes a waiting-time approximation which coincides with (6.9) when one replaces the Bernoulli parameters q_i by $1 - 1/k_i$. His approximation is used by Blanc & Van der Mei [23] to find those q_i that minimize a weighted sum of the mean waiting times, cf. Section 6.1.

Remark 6.3.2

Setting $\gamma_i = \beta_i$, K = L - s, i.e., imposing a limit L on the mean cycle time at periods of overload (namely, when all queues are loaded), (6.10) reduces to

$$k_i^* \beta_i = \frac{\rho_i s}{1 - \rho} + \left(L - \frac{s}{1 - \rho}\right) \frac{\sqrt{c_i \rho_i \delta_i}}{\sum\limits_{j=1}^n \sqrt{c_j \rho_j \delta_j}},\tag{6.13}$$

with $\delta_i = \rho_i(2-\rho) + \lambda_i s(1-\rho_i)$. One may interpret (6.13) as follows. The server should be allowed to visit Q_i at least for a time $\frac{\rho_i s}{1-\rho}$, to satisfy the stability condition. The remaining non-switch-over time, $L - \frac{s}{1-\rho}$, should

be assigned proportionally to $\sqrt{c_i \rho_i \delta_i}$. This suggests a rule for the optimal setting of time-limits in polling models with a time-limited service discipline. Note that in the case of constant service times, the k-limited and time-limited service disciplines coincide.

6.4 NUMERICAL RESULTS FOR THE CONSTRAINED PROBLEM

In this section, we give an overview of the numerical results that we gathered to test the rules (6.6), (6.8), (6.10), and the rule based on (6.12) proposed in the previous section. For a wide variety of cases, we compared the optimal values of the service limits and the waiting cost with the values achieved by the proposed rules.

To evaluate the mean waiting times, we used the power-series algorithm (PSA). The PSA allows an accurate numerical determination of the mean waiting times in polling models for which the joint queue length process has the structure of a multi-dimensional quasi birth-death process, cf. [21], [22].

The main drawback from which the PSA suffers, is that the time and memory requirements grow exponentially with the number of queues. We therefore confined ourselves to cases with only a few queues. We have confidence, however, that the various approaches will perform at least as good for a larger number of queues. In Section 6.5, we shall discuss the case of large n in more detail.

A further drawback from which the PSA suffers is that the time and memory requirements grow rapidly with the number of stages of the service and switch-over time distributions. For this reason, most of the numerical tests are conducted for cases with exponential service and switch-over times. The following arguments support our belief that the results for other service and switch-over time distributions will be similar in general. It should be noted that the k_i -values prescribed by the rules (6.6), (6.8), and (6.10) are insensitive to the form of the service time and switch-over time distributions. Indeed, the form of the pseudo-conservation law, and of the waiting-time approximations based upon it, suggests that the optimal policy and optimal cost will depend

on higher service time moments mainly through $\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}$, the influence of in-

dividual service time distributions being marginal, and that the influence of the switch-over time distributions on the optimal policy will be very minor. Numerical experiments of Blanc & Van der Mei [23] for cyclic polling with a Bernoulli service policy (cf. Section 6.1) support this view completely. Our tests with an Erlang-2 service time distribution (cf. Table 6.6 of Section 6.6) also point in the direction of a robustness of the optimal policy w.r.t. service time distributions; a robustness that was also observed in designing optimal polling tables, cf. [48] pp. 152, 153, and 161. Finally, it should be observed that Fuhrmann & Wang [104] p. 50 test their approximation only for exponentially distributed service times, but state that 'limited experience indicates that the accuracy for other service time distributions seems to be similar in general'.

The numerical results are presented in Tables 6.1-6.3. Table 6.1 contains 13 two-queue cases, Table 6.2 contains 2 three-queue cases, and Table 6.3 presents a five-queue case. Most of the parameter combinations are taken from [48]. In the two-queue cases we imposed the constraint $k_1 + k_2 \leq 12$, in the three-queue cases $k_1 + k_2 + k_3 \leq 12$, and in the five-queue case $k_1 + k_2 + \ldots + k_5 \leq 20$. The constraint may be interpreted as a limit on the maximum number of services in a cycle. The displayed cost figures are the 'exact' waiting-cost figures obtained from the PSA.

$\lambda_1 = \lambda_2 =$	$= 0.75; \beta_1 =$	$=\beta_2=0$.1; $s_1 = s_2$	$= 0.1; \rho$	= 0.15.			
	optimum		(6.8	3)	(6.10) (6.1		2)	
(c_1, c_2)	(k_1,k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1,k_2)	cost
(1, 0.1)	(10, 2)	0.145	(9, 3)	0.146	(9, 3)	0.146	(10, 2)	0.145
(1, 0.2)	(8, 4)	0.159	(8, 4)	0.159	(8, 4)	0.159	(9, 3)	0.159
(1, 0.5)	(6, 6)	0.199	(7, 5)	0.199	(7, 5)	0.199	(7, 5)	0.199
(1, 1)	(6, 6)	0.265	(6, 6)	0.265	(6, 6)	0.265	(6, 6)	0.265
(1, 2)	(6, 6)	0.397	(5, 7)	0.397	(5, 7)	0.397	(5, 7)	0.397
(1, 5)	(4, 8)	0.794	(4, 8)	0.794	(4, 8)	0.794	(3, 9)	0.794
(1, 10)	(2, 10)	1.455	(3, 9)	1.455	(3, 9)	1.455	(2, 10)	1.455

$\lambda_1 = \lambda_2 =$	0.75; $\beta_1 =$	0.9; $\beta_2 =$	$0.1; s_1 = s$	$\rho_2 = 0.1; \rho$	= 0.75.			
	optimum		(6.	(6.8)		(6.10)		.2)
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost
(0.9, 0.1)	(8, 4)	2.169	(10, 2)	2.307	(9, 3)	2.209	(8, 4)	2.169
(1, 0.1)	(8, 4)	2.355	(10, 2)	2.476	(9, 3)	2.386	(9, 3)	2.386
(1, 0.2)	(6, 6)	2.676	(9, 3)	3.006	(8, 4)	2.842	(7, 5)	2.738
(1, 0.5)	(4, 8)	3.176	(8, 4)	4.301	(7, 5)	3.819	(4, 8)	3.176
(1, 2)	(3, 9)	4.972	(6, 6)	6.996	(6, 6)	6.996	(2, 10)	5.033
(1, 5)	(2, 10)	7.693	(5, 7)	10.99	(4, 8)	9.391	(2, 10)	7.693
(1, 10)	(2, 10)	12.13	(4, 8)	16.30	(4, 8)	16.30	(1, 11)	12.80

$\lambda_1 = \lambda_2 = 0.5; \ \beta_1 = 0.9; \ \beta_2 = 0.1; \ s_1 = s_2 = 0.1; \ \rho = 0.5.$										
	optimum		(6.8)	3) (6.10)		(6.1)	2)			
(c_1, c_2)	(k_1,k_2)	cost	(k_1, k_2)	cost	(k_1,k_2)	cost	(k_1, k_2)	cost		
(0.9, 0.1)	(8, 4)	0.502	(9, 3)	0.502	(9, 3)	0.502	(9, 3)	0.502		
(1, 0.1)	(9, 3)	0.548	(10, 2)	0.550	(10, 2)	0.550	(9, 3)	0.548		
(1, 0.2)	(6, 6)	0.620	(9, 3)	0.632	(9, 3)	0.632	(7, 5)	0.623		
(1, 0.5)	(3, 9)	0.780	(8, 4)	0.861	(8, 4)	0.861	(3, 9)	0.780		
(1, 2)	(1, 11)	1.370	(6, 6)	1.804	(6, 6)	1.804	(1, 11)	1.370		
(1, 5)	(1, 11)	2.292	(4, 8)	3.229	(4, 8)	3.229	(1, 11)	2.292		
(1, 10)	(1, 11)	3.827	(4, 8)	5.942	(4, 8)	5.942	(1, 11)	3.827		

$\lambda_1 = \lambda_2$	$\lambda_1 = \lambda_2 = 0.4; \ \beta_1 = 0.9; \ \beta_2 = 0.1; \ s_1 = s_2 = 1.5; \ \rho = 0.4.$										
$-\frac{n_1-n_2}{2}$	optimum		(6.8)		(6.1)	(6.10)		.2)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost			
(1, 50)	(3, 9)	68.77	(4, 8)	71.78	(4, 8)	71.78	(3, 9)	68.77			
(1, 10)	(4, 8)	15.92	(5, 7)	16.74	(5, 7)	16.74	(4, 8)	15.92			
(1, 3)	(5, 7)	6.126	(6, 6)	6.305	(6, 6)	6.305	(5, 7)	6.126			
(1, 1)	(6, 6)	3.037	(7, 5)	3.156	(7, 5)	3.156	(6, 6)	3.037			
(3, 1)	(7, 5)	5.770	(8, 4)	5.968	(8, 4)	5.968	(7, 5)	5.770			
(10, 1)	(8, 4)	14.66	(9, 3)	15.54	(9, 3)	15.54	(8, 4)	14.66			
(50, 1)	(9, 3)	63.64	(9, 3)	63.64	(9, 3)	63.64	(9, 3)	63.64			

6.4 Numerical results for the constrained problem

$\lambda_1 = \lambda_2 = 0.8; \ \beta_1 = 0.9; \ \beta_2 = 0.1; \ s_1 = s_2 = 0.4; \ \rho = 0.8.$										
	optimum		(6.	8)	(6.10) (6		(6.1	5.12)		
(c_1, c_2)	(k_1, k_2)	cost								
(1, 50)	(4, 8)	120.9	(5, 7)	132.5	(4, 8)	120.9	(4, 8)	120.9		
(1, 10)	(5, 7)	34.19	(6, 6)	76.59	(5, 7)	34.19	(4, 8)	38.96		
(1, 3)	(5, 7)	16.98	(7, 5)	41.70	(6, 6)	27.90	(5, 7)	16.98		
(1, 1)	(5, 7)	12.06	(7, 5)	17.69	(7, 5)	17.69	(6, 6)	13.99		
(3, 1)	(6, 6)	28.04	(8, 4)	39.79	(7, 5)	29.04	(7, 5)	29.04		
(10, 1)	(7, 5)	68.78	(8, 4)	70.43	(8, 4)	70.43	(7, 5)	68.78		
(50, 1)	(8, 4)	245.6	(8, 4)	245.6	(8, 4)	245.6	(8, 4)	245.6		

$\lambda_1 = 0.76$	35; $\lambda_2 = 0.0$	$085; \beta_1 =$	$\beta_2 = 0.1;$	$s_1 = s_2 =$	$= 0.1; \rho = 0$	0.085.		
	optimum		(6.8)		(6.10)		(6.12)	
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost
(1, 0.1)	(10, 2)	0.124	(11, 1)	0.124	(11, 1)	0.124	(11, 1)	0.124
(1, 0.2)	(9, 3)	0.125	(10, 2)	0.125	(11, 1)	0.125	(11, 1)	0.125
(1, 0.5)	(9, 3)	0.130	(10, 2)	0.130	(11, 1)	0.130	(11, 1)	0.130
(1, 1)	(9, 3)	0.137	(9, 3)	0.137	(11, 1)	0.137	(11, 1)	0.137
(1, 2)	(8, 4)	0.151	(8, 4)	0.151	(10, 2)	0.151	(10, 2)	0.151
(1, 5)	(8, 4)	0.194	(7, 5)	0.194	(10, 2)	0.194	(9, 3)	0.194
(1, 10)	(8, 4)	0.266	(6, 6)	0.266	(9, 3)	0.266	(8, 4)	0.266

$\lambda_1 = 0.70$	$\lambda_1 = 0.765; \ \lambda_2 = 0.085; \ \beta_1 = \beta_2 = 0.5; \ s_1 = s_2 = 0.1; \ \rho = 0.425.$									
	optimum		(6.8)		(6.10)		(6.12)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost		
(1, 0.1)	(11, 1)	0.393	(11, 1)	0.393	(11, 1)	0.393	(11, 1)	0.393		
(1, 0.2)	(11, 1)	0.400	(11, 1)	0.400	(11, 1)	0.400	(11, 1)	0.400		
(1, 0.5)	(11, 1)	0.422	(11, 1)	0.422	(11, 1)	0.422	(11, 1)	0.422		
(1, 1)	(10, 2)	0.456	(9, 3)	0.456	(11, 1)	0.458	(11, 1)	0.458		
(1, 2)	(8, 4)	0.519	(9, 3)	0.519	(10, 2)	0.520	(9, 3)	0.519		
(1, 5)	(4, 8)	0.684	(8, 4)	0.701	(9, 3)	0.706	(6, 6)	0.692		
(1, 10)	(2, 10)	0.918	(8, 4)	1.006	(9, 3)	1.018	(2, 10)	0.918		

1 - 0.70	$\lambda_1 = 0.765; \ \lambda_2 = 0.085; \ \beta_1 = \beta_2 = 1; \ s_1 = s_2 = 0.1; \ \rho = 0.85.$									
$x_1 = 0.10$	$\frac{10}{10}, \frac{1}{12} = 0.0$	000, p1 -	$-p_2 - 1, s_1$		0.1, p = 0.	00.				
	optimum		(6.8	.8) (6.10)		0)	(6.12)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost		
(1, 0.1)	(11, 1)	4.333	(11, 1)	4.333	(11, 1)	4.333	(11, 1)	4.333		
(1, 0.2)	(11, 1)	4.466	(11, 1)	4.466	(11, 1)	4.466	(11, 1)	4.466		
(1, 0.5)	(11, 1)	4.865	(11, 1)	4.865	(11, 1)	4.865	(11, 1)	4.865		
(1, 1)	(11, 1)	5.529	(10, 2)	5.565	(11, 1)	5.529	(10, 2)	5.565		
(1, 2)	(9, 3)	6.013	(10, 2)	6.090	(10, 2)	6.090	(9, 3)	6.013		
(1, 5)	(7, 5)	7.047	(9, 3)	7.154	(9, 3)	7.154	(6, 6)	7.114		
(1, 10)	(6, 6)	8.428	(8, 4)	8.692	(9, 3)	9.055	(4, 8)	8.630		

$\lambda_1 = 0.765; \ \lambda_2 = 0.085; \ \beta_1 = \beta_2 = 0.5; \ s_1 = s_2 = 0.4; \ \rho = 0.425.$										
	optimum		(6.8)		(6.10)		(6.12)			
(c_1, c_2)	(k_1, k_2)	cost								
(1, 50)	(3, 9)	5.899	(5, 7)	6.174	(7, 5)	6.441	(3, 9)	5.899		
(1, 10)	(6, 6)	1.932	(7, 5)	1.934	(9, 3)	1.950	(8, 4)	1.941		
(1, 3)	(9, 3)	1.127	(9, 3)	1.127	(10, 2)	1.133	(10, 2)	1.133		
(1, 1)	(10, 2)	0.888	(10, 2)	0.888	(11, 1)	0.905	(11, 1)	0.905		
(3, 1)	(11, 1)	2.413	(11, 1)	2.413	(11, 1)	2.413	(11, 1)	2.413		
(10, 1)	(11, 1)	7.690	(11, 1)	7.690	(11, 1)	7.690	(11, 1)	7.690		
(50, 1)	(11, 1)	37.85	(11, 1)	37.85	(11, 1)	37.85	(11, 1)	37.85		

$\lambda_1 = 0.7$	$\lambda_1 = 0.765; \ \lambda_2 = 0.085; \ \beta_1 = \beta_2 = 1; \ s_1 = s_2 = 0.4; \ \rho = 0.85.$										
	optimum		(6.	(6.8)		(6.10)		.2)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost			
(1, 50)	(8, 4)	28.52	(8, 4)	28.52	(8, 4)	28.52	(7, 5)	30.33			
(1, 10)	(10, 2)	12.65	(9, 3)	12.81	(10, 2)	12.65	(9, 3)	12.81			
(1, 3)	(10, 2)	9.20	(10, 2)	9.20	(11, 1)	9.61	(10, 2)	9.20			
(1, 1)	(11, 1)	4.92	(11, 1)	4.92	(11, 1)	4.92	(11, 1)	4.92			
(3, 1)	(11, 1)	10.08	(11, 1)	10.08	(11, 1)	10.08	(11, 1)	10.08			
(10, 1)	(11, 1)	28.11	(11, 1)	28.11	(11, 1)	28.11	(11, 1)	28.11			
(50, 1)	(11, 1)	131.2	(11, 1)	131.2	(11, 1)	131.2	(11, 1)	131.2			

0.7	-0.765 , -0.095 , $a_{1} - a_{2} - 1$, $a_{2} - 0.1$, $a_{3} - 0.75$, $a_{4} - 0.95$									
$\lambda_1 = 0.7$	$\lambda_1 = 0.703; \ \lambda_2 = 0.083; \ \mu_1 = \mu_2 = 1; \ s_1 = 0.1; \ s_2 = 0.03.$									
	optimum		(6	(6.8) (6.10)		10)	(6.12)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost		
(1, 50)	(8, 4)	29.60	(8, 4)	29.60	(8, 4)	29.60	(7, 5)	30.33		
(1, 10)	(10, 2)	13.00	(9, 3)	13.14	(10, 2)	13.00	(9, 3)	13.14		
(1, 3)	(10, 2)	9.413	(10, 2)	9.413	(11, 1)	9.857	(11, 1)	9.857		
(1, 1)	(11, 1)	5.060	(11, 1)	5.060	(11, 1)	5.060	(11, 1)	5.060		
(3, 1)	(11, 1)	10.38	(11, 1)	10.38	(11, 1)	10.38	(11, 1)	10.38		
(10, 1)	(11, 1)	29.01	(11, 1)	29.01	(11, 1)	29.01	(11, 1)	29.01		
1 (50 1)	1 (11 1)	135.5	1(11,1)	135.5	(11.1)	135.5	(11.1)	135.5		

$\lambda_1 = 0.5;$	$\lambda_1 = 0.5; \ \lambda_2 = 0.25; \ \beta_1 = \beta_2 = 1; \ s_1 = 0.1; \ s_2 = 0.2; \ \rho = 0.75.$										
	optimum		(6.	8) (6.10		10)	(6.1	.2)			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost	(k_1, k_2)	cost			
(1, 18.4)	(2, 10)	10.66	(4, 8)	12.54	(4, 8)	12.54	(1, 11)	12.30			
(1, 7.6)	(3, 9)	6.278	(5, 7)	6.885	(6, 6)	7.434	(2, 10)	6.293			
(1, 6)	(3, 9)	5.515	(6, 6)	6.286	(6, 6)	6.286	(2, 10)	5.646			
(1, 3.95)	(3, 9)	4.537	(6, 6)	4.813	(6, 6)	4.813	(3, 9)	4.537			
(1, 2.75)	(5, 7)	3.863	(7, 5)	4.198	(7, 5)	4.198	(4, 8)	3.882			
(1, 1.5)	(6, 6)	3.036	(8, 4)	3.280	(8, 4)	3.280	(6, 6)	3.036			
(1, 0.7)	(10, 2)	2.265	(9.3)	2.294	(9.3)	2.294	(9, 3)	2.294			

$\lambda_1 = 0.5$	$\lambda_1 = 0.5; \ \lambda_2 = 1; \ \beta_1 = 1; \ \beta_2 = 0.3; \ s_1 = 0.2; \ s_2 = 0.6; \ \rho = 0.8.$								
	optin	num	(6.	8)	(6.1	.0)	(6.1	.2)	
(c_1, c_2)	(k_1, k_2)	cost	(k_1,k_2)	cost	(k_1,k_2)	cost	(k_1, k_2)	cost	
(1, 50)	(3, 9)	129.7	(3, 9)	129.7	(3, 9)	129.7	(3, 9)	129.7	
(1, 10)	(3, 9)	32.09	(4, 8)	69.06	(4, 8)	69.06	(3, 9)	32.09	
(1, 3)	(3, 9)	15.01	(4, 8)	23.77	(4, 8)	23.77	(3, 9)	15.01	
(1, 1)	(3, 9)	10.13	(5, 7)	12.18	(5, 7)	12.18	(4, 8)	10.83	
(3, 1)	(5, 7)	17.53	(6, 6)	19.61	(6, 6)	19.61	(5, 7)	17.53	
(10, 1)	(6, 6)	32.79	(7, 5)	41.42	(7, 5)	41.42	(6, 6)	32.79	
(50, 1)	(7, 5)	99.50	(7, 5)	99.50	(7, 5)	99.50	(7, 5)	99.50	

TABLE 6.1. The constrained case; two-queue models.

$\lambda_1 = \lambda_2 = \lambda_3 = 0.7; \ \beta_1 = 0.8; \ \beta_2 = \beta_3 = 0.1; \ s_1 = s_2 = s_3 = 0.05; \ \rho = 0.7.$								
	optim	um	(6.8)		(6.10))	(6.12)
$(c_1, c_{2,3})$	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost
(3.5, 1)	(3, 4, 5)	7.00	(7, 2, 3)	8.55	(6, 3, 3)	7.72	(4, 4, 4)	7.20
(14.4, 1)	(9, 2, 1)	18.09	(8, 2, 2)	18.20	(8, 2, 2)	18.20	(8, 2, 2)	18.20
(51.8, 1)	(9, 2, 1)	50.50	(10, 1, 1)	51.80	(9, 1, 2)	52.07	(10, 1, 1)	51.80

$\lambda_1 = 0.531; \ \lambda_2 = 0.212; \ \lambda_3 = 0.106; \ \beta_1 = \beta_2 = \beta_3 = 0.9; \ s_1 = s_2 = s_3 = 0.3; \ \rho = 0.7641.$								
	optim	um	(6.8)		(6.10)		(6.12)	
(c_1,c_2,c_3)	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost	$(k_{1,2,3})$	cost
(1, 0.1, 1)	(9, 1, 2)	2.660	(8, 2, 2)	2.668	(8, 2, 2)	2.668	(8, 2, 2)	2.668
(1, 5, 1)	(5, 5, 2)	7.796	(6, 5, 1)	8.541	(6, 5, 1)	8.541	(5, 6, 1)	8.055
(1, 1, 5)	(6, 3, 3)	6.252	(6, 3, 3)	6.252	(6, 3, 3)	6.252	(6, 3, 3)	6.252

TABLE 6.2. The constrained case; three-queue models.

$\lambda_1 = 0.35; \ \lambda_2 = \ldots = \lambda_5 = 0.1; \ \beta_1 = 1; \ \beta_2 = \ldots = \beta_5 = 1; \ s_1 = 0.1; \ s_2 = \ldots = s_5 = 0.05; \ \rho = 0.75.$								
	optimi	ım	(6.8))	(6.10)	(6.12)
(c_1, c_{2-5})	(k_1, k_{2-5})	cost						
(1, 0.1)	(16, 1)	0.883	(16, 1)	0.883	(16, 1)	0.883	(16, 1)	0.883
(1, 0.5)	(16, 1)	1.804	(8, 3)	1.987	(12, 2)	1.807	(16, 1)	1.804
(1, 1)	(4, 4)	2.420	(8, 3)	2.934	(8, 3)	2.934	(8, 3)	2.934
(1, 2)	(4, 4)	3.631	(4, 4)	3.631	(8, 3)	4.827	(4, 4)	3.631

TABLE 6.3. The constrained case; a five-queue model.

6.5 The unconstrained optimization problem

Discussion of the numerical results.

The various rules perform reasonably well; in the majority of the examples, the waiting cost achieved is less than 10% larger than the minimum waiting cost. It is however interesting to compare how the various rules perform. For compactness, we did not display the results for the rule (6.6), but on average the rules (6.6) and (6.8) perform similarly. Sometimes (6.6) performs better, sometimes (6.8). The rule (6.10) performs slightly better than the rule (6.8). The underlying approximation of (6.10) is theoretically indeed better than the underlying approximation of (6.8). The former shows the correct exact behavior when $k_i \to \infty$, the latter does not. As the imposed constraint prevents that $k_i \to \infty$, the difference in performance is however minor.

The rule based on the approximation (6.12) performs by far the best; only 8 out of the 101 times the relative error exceeds 5%. The approximation (6.12) is indeed theoretically better than the underlying approximation of (6.10). The former catches the influence of k_i on EW_j , the latter does not. Thus, the difference in performance will especially be dramatic in cases where that influence plays a crucial role, as is illustrated by the numerical results. When we take a closer look at those two-queue cases where the rules (6.6), (6.8), and (6.10) perform poorly, we notice that simultaneously β_1 is larger than β_2 , and c_1 is smaller than c_2 . These rules, which completely ignore the influence of k_1 on EW_2 , then choose k_1 too large and k_2 too small. These two-queue cases are typically cases where the influence of k_1 on EW_2 plays a crucial role: for large β_1 , the influence of k_1 on EW_2 is large, and for large c_2 this influence is heavily weighted in the waiting cost.

6.5 The unconstrained optimization problem

In this section, we study the problem of finding the (unconstrained) service limits k_1, \ldots, k_n that minimize the waiting $\cot \sum_{i=1}^n c_i \lambda_i E \mathbf{W}_i$. We first derive several monotonicity properties of polling systems with k-limited service and switch-over periods. The main result is a (partially conjectured) rule, stating that for minimizing the waiting cost in such systems, the queues with the highest value of c_i/β_i should be assigned $k_i = \infty$, i.e., receive exhaustive service. This property is very similar to the well-known $c\mu$ -rule derived for systems with no switch-over periods. We subsequently propose to use the Fuhrmann & Wang approximation for the unconstrained waiting-cost minimization. We specifically investigate to what extent the Fuhrmann & Wang approximation satisfies the above-mentioned properties.

Proposition 6.5.1

In a stable polling system with cyclic visit order and k-limited service, the sum $\sum_{j=1}^{n} \rho_j \mathbf{E} \mathbf{W}_j$ is non-increasing in each of the service limits $k_i, i = 1, ..., n$.

Proof

Let \mathbf{V}_t be the total amount of work in the system at time t. Let \mathbf{V} be a stochastic variable with distribution the steady-state distribution of the total amount of work in the system. As shown in [142], \mathbf{V}_t is non-increasing in k_i (the proof in [142] is a path-wise proof). Hence, EV is non-increasing in k_i . Now, it is known that

$$\mathbf{EV} = \sum_{j=1}^{n} \rho_j \mathbf{EW}_j + \sum_{j=1}^{n} \rho_j \frac{\beta_j^{(2)}}{2\beta_j},$$

and thus $\sum_{j=1}^{n} \rho_j E \mathbf{W}_j$ is non-increasing in k_i .

Conjecture 6.5.1

In a stable polling system with cyclic visit order and k-limited service, the mean waiting time at Q_i , \mathbf{EW}_i , is decreasing in its service limit k_i , and increasing in k_j for every $j \neq i$.

While the claim made in Conjecture 6.5.1 is very appealing and intuitive, it seems difficult to prove it. A reasonable line of argument can nonetheless be provided as follows. To see the effect of k_j on EW_i , one can view the services given at Q_j as switch-over periods (whose durations are distributed as the sum of several independent random variables, whose number is the number of customers being served at Q_j). It is easy to see that, as long as the system is stable, the mean number of services given at Q_j per visit is constant $(\frac{\lambda_j s}{1-\rho})$, which does not depend on k_j). On the other hand, the second moment of the number of services does depend on k_j . Increasing k_j will increase the number of services given at Q_j when that queue is loaded (has more than k_j customers), and thus is likely to decrease the number of services given at Q_j when it has less than k_j customers. This implies that the variance of the number of services given at Q_j increases with k_j . Going back to the viewpoint of Q_i , we see that when k_j increases, the customers at Q_i observe switch-over periods of the same mean but of higher variance. Viewing Q_i as an M/G/1 queue with vacations (where the services of the other queues and the switch-over periods together constitute one large vacation), it is likely that increasing the vacation second moment, while keeping its mean the same, increases the mean waiting time at Q_i . The latter property is indeed known to hold for exhaustive and gated M/G/1 vacation queues, cf. Takagi [174]. We thus conclude that the mean waiting time at Q_i is increasing in k_j for any $j \neq i$. Combining this fact with Proposition 6.5.1 implies that the mean waiting time at Q_i must be decreasing in k_i .

6.5 The unconstrained optimization problem

Remark 6.5.1

The assumption that the system is stable plays an essential role in Conjecture 6.5.1. If Q_j is unstable, then increasing k_j will not only increase the variance of the intervisit time of Q_i , but also the mean. If the increment of the first moment is larger than the increment of the second moment, then the residual intervisit time of Q_i will *decrease*. The mean waiting time at Q_i will then also *decrease*.

Proposition 6.5.1 and Conjecture 6.5.1 lead to the main result of this section, namely, that for optimality at least one of the queues, viz., the one whose c_i/β_i attains the maximum value, must be served without limits.

Conjecture 6.5.2

If $c_i/\beta_i = \max_{j=1,\dots,n} c_j/\beta_j$, then the optimal service limit $k_i^* = \infty$.

Proof

We need to show that for all (k_1, \ldots, k_n) , the sum $\sum_{j=1}^n c_j \lambda_j E \mathbf{W}_j$ is decreasing

in k_i , i.e., $\frac{\Delta}{\Delta k_i} \sum_{j=1}^n \lambda_j c_j \mathbf{E} \mathbf{W}_j \leq 0$ (with $\frac{\Delta}{\Delta k_i}$ denoting partial difference):

$$\begin{split} \frac{\Delta}{\Delta k_i} \sum_{j=1}^n \lambda_j c_j \mathbf{E} \mathbf{W}_j &= \frac{c_i}{\beta_i} \frac{\Delta}{\Delta k_i} \rho_i \mathbf{E} \mathbf{W}_i + \sum_{j \neq i} \frac{c_j}{\beta_j} \frac{\Delta}{\Delta k_i} \rho_j \mathbf{E} \mathbf{W}_j \\ &\leq \frac{c_i}{\beta_i} \frac{\Delta}{\Delta k_i} \rho_i \mathbf{E} \mathbf{W}_i + \frac{c_i}{\beta_i} \sum_{j \neq i} \frac{\Delta}{\Delta k_i} \rho_j \mathbf{E} \mathbf{W}_j \\ &= \frac{c_i}{\beta_i} \frac{\Delta}{\Delta k_i} \sum_{j=1}^n \rho_j \mathbf{E} \mathbf{W}_j \leq 0. \end{split}$$

The inequality in the second line follows from the facts that $\frac{\Delta}{\Delta k_i} \mathbf{E} \mathbf{W}_j$ is non-negative (due to Conjecture 6.5.1), and that $c_i/\beta_i \geq c_j/\beta_j$ (condition of the theorem). The inequality in the third line follows from Proposition 6.5.1.

Remark 6.5.2

Conjecture 6.5.2 implies that if the service limit policies are of the limitedgated type (namely, serve up to k_i customers, but only of those present at the queue at the polling instant), then the queues with the maximum value of c_i/β_i should be served according to the gated policy with $k_i^* = \infty$. If the service limit policies are of the exhaustive-limited type (namely, serve up to k_i customers, but allow to include in these services customers that arrived during the service of the queue), then the queues with the maximum value of c_i/β_i should be served according to the exhaustive policy with $k_i^* = \infty$. This reminds of the $c\mu$ -rule derived for systems with no switch-over periods, and in which the server is free to move from queue to queue. According to the $c\mu$ -rule, the queues with the highest value of c_i/β_i should receive the highest priority in the system, which implies, in particular, exhaustive service at those queues.

Keeping in mind the properties discussed above, we now study the problem of finding the service limits k_1, \ldots, k_n that minimize the waiting $\cot \sum_{i=1}^n c_i \lambda_i E \mathbf{W}_i$. We propose to use the Fuhrmann & Wang approximation (6.12). As observed in the previous section, for the constrained waiting-cost minimization, the Fuhrmann & Wang approximation outperforms the simpler approximations. For the unconstrained waiting-cost minimization, the simpler approximations would be useless anyhow, as they completely ignore the influence of k_j on $E \mathbf{W}_i$, which would always lead to $k_1^* = \infty, \ldots, k_n^* = \infty$.

We now specifically investigate to what extent the Fuhrmann & Wang approximation (6.12) satisfies the properties discussed above.

Proposition 6.5.2

The approximation (6.12) of EW_i is: i) decreasing in k_i , and ii) increasing in k_j , $j \neq i$.

Proof

A straightforward computation shows that $\mathbf{EW}_i \mid_{k_i=r} \geq \mathbf{EW}_i \mid_{k_i=r+1}$, and that $\mathbf{EW}_i \mid_{k_j=r} \leq \mathbf{EW}_i \mid_{k_j=r+1}, j \neq i$. (The latter inequality holds provided $D \geq \frac{s}{2-\rho} \sum_{j=1}^n \rho_j (1-\rho_j)$, which may be shown to hold by substitution of the definition of D.)

Proposition 6.5.2 supports the use of (6.12) in trying to obtain the optimal service limit values for the actual polling system. Moreover, in the numerical experiments that will be presented in the next section, we will find that the minimum value of $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i$, where (6.12) is used to evaluate $E \mathbf{W}_i$, is always attained when the service limit of the queue with the maximum value of c_i/β_i is set to $k_i = \infty$ (exhaustive service).

In the numerical experiments for both the constrained case (Section 6.4) and the unconstrained case (Section 6.6), time and memory requirements of the PSA have forced us to confine ourselves to models with only a few queues. Let us now discuss what happens when the number of queues, n, approaches infinity, distinguishing four cases: for all j:

I. s_j fixed, $\beta_j = O(1/n)$, λ_j fixed; II. $s_j = O(1/n)$, β_j fixed, $\lambda_j = O(1/n)$; III. $s_j = O(1/n)$, $\beta_j = O(1/n)$, λ_j fixed; IV. s_j fixed, β_j fixed, $\lambda_j = O(1/n)$.

In case I, $\lambda_i s/(1-\rho) \to \infty$ and hence necessarily $k_i \to \infty$; this is not an interesting case. Case II reduces to continuous polling on a circle; cf. Fuhrmann & Cooper [102]. Each customer will be served in the cycle in which it arrives, even if the k_i values equal one; the actual choice of the k_i is irrelevant. Cases III and IV are equivalent up to a scaling of time by a factor n. Let us discuss case III in more detail. For the constrained situation, (6.7), (6.9) and (6.12) all reduce to

$$\mathbf{EW}_{i} \approx B \frac{1-\rho}{1-\rho - \frac{\lambda_{i}s}{k_{i}}},\tag{6.14}$$

with B some constant, leading to

$$k_i^* = \frac{\lambda_i s}{1 - \rho} + \left(K - \sum_{j=1}^n \gamma_j \frac{\lambda_j s}{1 - \rho}\right) \frac{\lambda_i \sqrt{c_i / \gamma_i}}{\sum\limits_{j=1}^n \gamma_j \lambda_j \sqrt{c_j / \gamma_j}}.$$
(6.15)

Note that the weakness of approximation (6.7), indicated above (6.8), disappears when $n \to \infty$. Approximation (6.14) may be expected to perform very well. For the unconstrained situation, (6.12) also reduces to (6.14). Hence, the waiting cost is minimized by taking $k_i = \infty$ for all i = 1, ..., n. Indeed, for large finite n, an increment of k_i by one reduces $c_i \lambda_i E \mathbf{W}_i$ much stronger than it increases $\sum_{j \neq i} c_j \lambda_j E \mathbf{W}_j$, as is indicated by the following rough reasoning.

To make things simple, let us assume that $k_2 = \ldots = k_n = \infty$; now increase k_1 by one. Customers in Q_1 only notice this increment at a server visit when at least $k_1 + 1$ customers are present. Suppose such an event occurs in the *m*-th cycle. This saves one Q_1 customer one cycle time $C_{1,m+1}$, which is O(1). Here $\mathbf{C}_{j,m}$ denotes the *m*-th cycle time for Q_j . What is the effect on some other queue Q_j ? First the bad effect. S reaches $Q_j \Delta_j$ later; this delay consists of a service time at Q_1 (of O(1/n)), and of extended visit times at Q_2, \ldots, Q_{j-1} ; $\Delta_j = O(1/n)$. Each of the customers at Q_j experiences this additional delay as an addition to its waiting time. There are on the average $\lambda_j EC_{j,m}$ such customers. The total mean 'loss' for Q_j is $\Delta_j \lambda_j EC_{j,m}$. Here we ignore an $O(n^{-2})$ contribution: compared to an ordinary cycle, this one lasted already Δ_j longer, during which additional period also on the average $\lambda_j \Delta_j$ customers have arrived, who each experience an extra delay Δ_i . There is also a *benefit* for Q_j . During the extra delay also customers arrive at Q_j , who are just in time to be served in *this* cycle; without the extra delay they would have arrived just after the departure of S from Q_j , and would have had to wait a full cycle. The total mean 'gain' for Q_j is $\Delta_j \lambda_j EC_{j,m+1} + O(n^{-2})$. The result on Q_j

of these two counteracting effects is $\lambda_j \Delta_j (\mathbf{EC}_{j,m} - \mathbf{EC}_{j,m+1}) + O(n^{-2})$ (the propagation of an extra service in Q_1 in later cycles should also have an $O(n^{-2})$ effect). Obviously, $\mathbf{EC}_{j,m} - \mathbf{EC}_{j,m+1} = O(1)$, but it seems in fact likely that $\mathbf{EC}_{j,m} - \mathbf{EC}_{j,m+1} = O(1/n)$. In the latter case, increasing k_1 by one has an $O(n^{-2})$ effect on Q_j , which agrees with (6.14).

6.6 NUMERICAL RESULTS FOR THE UNCONSTRAINED PROBLEM

In this section, we give an overview of the numerical results that we gathered to test the approach proposed in the previous section. For a wide variety of cases, we compared the optimal values of the service limits and the waiting cost with the values achieved by the proposed approximative approach.

We confined ourselves to cases with only a few queues, and focused on cases with an exponential service and switch-over time distribution, although we did investigate some cases with an Erlang-2 service time distribution as well. The results for an Erlang-2 service time distribution appear to be similar to the results for an exponential service time distribution.

The numerical results are presented in Tables 6.4-6.7. Table 6.4 contains the same 13 two-queue cases as Table 6.1 of Section 6.4, Table 6.5 contains 4 threequeue cases with exponential service time distributions, Table 6.6 contains the same cases but with Erlang-2 service time distributions with the same mean, and Table 6.7 contains the same five-queue case as Table 6.3 of Section 6.4. The displayed cost figures are the 'exact' waiting cost figures obtained from the PSA.

$\lambda_1 = \lambda_2$	$= 0.75; \beta_1 =$	$= \beta_2 = 0$	$.1; s_1 = s_2$	= 0.1;			
$\rho = 0.15.$	$\rho = 0.15.$						
	optim	um	approximation				
(c_1, c_2)	(k_1,k_2)	cost	(k_1, k_2)	cost	%		
(1, 0.1)	$(\infty, 2)$	0.145	$(\infty, 3)$	0.145	0.0		
(1, 0.2)	$(\infty, 5)$	0.159	(∞, ∞)	0.159	0.0		
(1, 0.5)	$(\infty, 13)$	0.199	(∞, ∞)	0.199	0.0		
(1, 1)	(∞, ∞)	0.265	(∞, ∞)	0.265	0.0		
(1, 2)	$(13, \infty)$	0.397	(∞, ∞)	0.397	0.0		
(1, 5)	$(5,\infty)$	0.794	(∞, ∞)	0.794	0.0		
(1, 10)	$(2, \infty)$	1.455	$(3, \infty)$	1.455	0.0		

)	۱1	=	λ_2	=	0	.7	5;	β_1	=	0.9;	β_2	=	0.1	;
---	----	---	-------------	---	---	----	----	-----------	---	------	-----------	---	-----	---

$s_1 = s_2 =$	$s_1 = s_2 = 0.1; \ \rho = 0.75.$						
	optin	num	appr	approximation			
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%		
(1, 0.1)	$(\infty, 17)$	2.158	(∞, ∞)	2.162	0.2		
(1, 0.2)	$(12, \infty)$	2.535	$(20, \infty)$	2.558	0.9		
(1, 0.5)	$(5, \infty)$	3.119	$(5, \infty)$	3.119	0.0		
(1, 1)	$(4,\infty)$	3.817	(3, ∞)	3.836	0.5		
(1, 2)	(3, ∞)	4.949	$(2,\infty)$	5.030	1.6		
(1, 5)	$(2,\infty)$	7.685	$(2,\infty)$	7.685	0.0		
(1, 10)	$(2, \infty)$	12.11	$(1, \infty)$	12.80	5.7		

6.6 Numerical results for the unconstrained problem

.

	05.0-	0.0.0.	- 0.1.		
$\lambda_1 = \lambda_2 =$	$= 0.5; p_1 = 0.5; p_2 = 0.5; p_3 = 0.5; p_4 = 0.5; p_$	$0.9; p_2 = 5$	= 0.1;		
$s_1 = s_2 =$	= 0.1; p = 0			wimetion	
		um	appro	ximation	07
(c_1, c_2)	(κ_1,κ_2)	COSL	(κ_1,κ_2)	COSL	70
(1, 0.1)	(∞, ∞)	0.540	(∞, ∞)	0.540	0.0
(1, 0.2)	$(7, \infty)$	0.617	(∞, ∞)	0.623	1.0
(1, 0.5)	$(3, \infty)$	0.780	$(3, \infty)$	0.780	0.0
(1, 1)	$(2, \infty)$	0.992	$(2, \infty)$	0.992	0.0
(1, 2)	$(1,\infty)$	1.370	$(1, \infty)$	1.370	0.0
(1, 5)	$(1, \infty)$	2.292	$(1, \infty)$	2.292	0.0
(1, 10)	$(1,\infty)$	3.827	$(1,\infty)$	3.827	0.0
$\lambda_1 = \lambda_2 = 0.4$	$= 0.4; \beta_1 =$	$0.9; \beta_2 =$	$= 0.1; s_1 =$	$s_2 = 1.5$;
	optin	num	app	roximati	on
(((1) (0)))	(k_1, k_2)	cost	(k_1, k_2)	Cost	1 %
(c_1, c_2)	(n_1, n_2)	1 205	(m_1, m_2)	1 205	
(1, 0.1)	(ω, ω)	1.303	(ω, ω)	1.300	
(1, 0.2)	(∞, ∞)	1.409	$\left \left(\infty, \infty \right) \right $	1.408	
(1, 0.5)	(20,∞)	1.963	$ (\infty, \infty) $	1.963	
(1, 1)	$(12, \infty)$	2.775	(∞, ∞)	2.787	0.4
(1, 2)	(8,∞)	4.336	(∞, ∞)	4.434	2.3
(1, 5)	$(5,\infty)$	8.734	$ (10, \infty)$	9.092	2 4.1
(1, 10)	$(4, \infty)$	15.70	$(5, \infty)$	15.89	1.2
$\lambda_1 = \lambda_2$	$= 0.8; \beta_1 =$	$0.9; \beta_2 =$	$= 0.1; s_1 =$	$s_2 = 0.4$:
a = 0.8	0.0, 1	,	, .1	-2	,
	ontin	m	ann	rovimati	
(62, 62)	$\frac{0}{k_1}$	cost	$\frac{app}{k_1 k_2}$	Cost	1 %
(c_1, c_2)	(κ_1,κ_2)	2 600	(~1, ~2)		
(1, 0.1)	(∞, ∞)	3.000	$ (\infty,\infty) $	3.000	0.0
(1, 0.2)	(∞, ∞)	4.549	(∞, ∞)	4.549	0.0
(1, 0.5)	$(17, \infty)$	6.104	$ (20,\infty) $	6.145	6 0.7
(1, 1)	$(11, \infty)$	8.174	$ (13,\infty) $	8.207	0.4
(1, 2)	(9, ∞)	11.58	(9,∞)	11.58	0.0
(1, 5)	$(6, \infty)$	20.17	$(6, \infty)$	20.17	0.0
(1, 10)	$(5, \infty)$	32.83	$(5,\infty)$	32.83	0.0
$\lambda_1 = 0.76$	$\lambda_2 = 0.0$	$\beta_{1} = 0.05; \beta_{1} = 0.05$	$\beta_2 = 0.1;$		
$s_1 = s_2 =$	$= 0.1; \rho \equiv 0$.085.			
	optim	um	appro	ximation	
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%
(1, 0.1)	$(\infty, 2)$	0.124	$(\infty, 1)$	0.124	0.0
(1, 0.2)	(∞, ∞)	0.125	(∞, ∞)	0.125	0.0
(1, 0.5)	(∞, ∞)	0.130	(∞, ∞)	0.130	0.0
(1, 1)	(∞, ∞)	0.137	(∞, ∞)	0.137	0.0
(1, 2)	(∞, ∞)	0.151	(∞, ∞)	0.151	0.0
(1,5)	(∞, ∞)	0.194	(∞, ∞)	0.194	0.0
(1, 10)	(∞, ∞)	0.266	(∞, ∞)	0.266	0.0
	(∞, ∞)	0.200	(∞, ∞)	0.200	0.0
$\lambda_1 = 0.76$	$35; \lambda_2 = 0.0$	$\beta_{085; \beta_1} =$	$\beta_2 = 0.5;$		
1	$= 0.1; \rho = 0$.425.			
$s_1 = s_2 =$		um	appro	ximatior	
$s_1 = s_2 =$	optim				1 % I
$s_1 = s_2 = (c_1, c_2)$	optim (k_1, k_2)	cost	(k_1, k_2)	COSt	/0
$\begin{array}{c} s_1 = s_2 = \\ \hline \\ (c_1, c_2) \\ \hline \\ (1, 0.1) \end{array}$	$\begin{array}{c} \text{optim} \\ (k_1, k_2) \\ (\infty, 1) \end{array}$	cost 0.390	$\frac{(k_1,k_2)}{(\infty,1)}$	0.390	0.0
$\begin{array}{c} s_1 = s_2 = \\ \hline (c_1, c_2) \\ \hline (1, 0.1) \\ (1, 0.2) \end{array}$	$\begin{array}{c} \text{optim} \\ \hline (k_1, k_2) \\ (\infty, 1) \\ (\infty, 1) \end{array}$	cost 0.390 0.397	$\frac{(k_1,k_2)}{(\infty,1)}$ $(\infty,1)$	0.390 0.397	0.0
$\begin{array}{c} s_1 = s_2 = \\ \hline (c_1, c_2) \\ \hline (1, 0.1) \\ (1, 0.2) \\ (1, 0.5) \end{array}$	$\begin{array}{c} \text{optim} \\ \hline (k_1, k_2) \\ \hline (\infty, 1) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, 1) \end{array}$	cost 0.390 0.397 0.419	$\frac{(k_1, k_2)}{(\infty, 1)} \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, 1)$	0.390 0.397 0.419	0.0 0.0 0.0
$\begin{array}{c} s_1 = s_2 = \\ \hline \\ (c_1, c_2) \\ \hline \\ (1, 0.1) \\ (1, 0.2) \\ (1, 0.5) \\ (1, 1) \end{array}$	$\begin{array}{c} \text{optim} \\ \hline (k_1, k_2) \\ \hline (\infty, 1) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, \infty) \end{array}$	cost 0.390 0.397 0.419 0.453	(k_1, k_2) $(\infty, 1)$ $(\infty, 1)$ $(\infty, 1)$ (∞, ∞)	0.390 0.397 0.419 0.453	0.0 0.0 0.0 0.0
$\begin{array}{c} s_1 = s_2 = \\ \hline \\ \hline \\ (c_1, c_2) \\ \hline \\ (1, 0.1) \\ (1, 0.2) \\ (1, 0.5) \\ (1, 1) \\ (1, 2) \end{array}$	$\begin{array}{c} \text{optim} \\ (k_1, k_2) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, \infty) \\ (10, \infty) \end{array}$	cost 0.390 0.397 0.419 0.453 0.518	(k_1, k_2) $(\infty, 1)$ $(\infty, 1)$ $(\infty, 1)$ (∞, ∞) (∞, ∞)	0.390 0.397 0.419 0.453 0.519	0.0 0.0 0.0 0.0 0.0 0.2
$\begin{array}{c} s_1 = s_2 = \\ \hline \\ (c_1, c_2) \\ \hline \\ (1, 0.1) \\ (1, 0.2) \\ (1, 0.5) \\ (1, 1) \\ (1, 2) \\ (1, 5) \end{array}$	$\begin{array}{c} \begin{array}{c} \text{optim} \\ \hline (k_1, k_2) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, \infty) \\ (10, \infty) \\ (4, \infty) \end{array}$	cost 0.390 0.397 0.419 0.453 0.518 0.684	$ \begin{array}{c} (k_1, k_2) \\ (\infty, 1) \\ (\infty, 1) \\ (\infty, 0) \\ (\infty, \infty) \\ (\infty, \infty) \\ (7, \infty) \end{array} $	Cost 0.390 0.397 0.419 0.453 0.519 0.697	0.0 0.0 0.0 0.0 0.0 0.2 1.9
$\begin{array}{c} s_1 = s_2 = \\ \hline \\ \hline \\ (c_1, c_2) \\ \hline \\ (1, 0.1) \\ (1, 0.2) \\ (1, 0.5) \\ (1, 1) \\ (1, 2) \\ (1, 5) \\ (1, 10) \end{array}$	$\begin{array}{c} \begin{array}{c} \text{optim} \\ \hline (k_1, k_2) \\ \hline (\infty, 1) \\ (\infty, 1) \\ (\infty, 0) \\ (10, \infty) \\ (10, \infty) \\ (2, \infty) \end{array}$	cost 0.390 0.397 0.419 0.453 0.518 0.684 0.684	(k_1, k_2) $(\infty, 1)$ $(\infty, 1)$ $(\infty, 0)$ (∞, ∞) (∞, ∞) $(7, \infty)$ $(2, \infty)$	Cost 0.390 0.397 0.419 0.453 0.519 0.697 0.018	0.0 0.0 0.0 0.0 0.2 1.9

113

Chapter 6 Optimization of k-limited service strategies

$\lambda_1 = 0.76$	65; $\lambda_2 = 0.0$	$085; \beta_1 =$	$\beta_2 = 1;$			
$s_1 = s_2 =$	$= 0.1; \rho = 0$.85.				
	optim	um	appro	ximation	1	
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%	
(1, 0.1)	$(\infty, 1)$	3.139	$(\infty, 1)$	3.139	0.0	
(1, 0.2)	$(\infty, 1)$	3.381	$(\infty, 1)$	3.381	0.0	
(1, 0.5)	$(\infty, 2)$	4.069	$(\infty, 1)$	4,106	0.9	
(1, 0.0)	$(\infty, 2)$	5.031	$(\infty, 1)$	5.031		
(1, 1)	(ω, ω)	5.031	(ω, ω)	5.031	17	
(1, 2)	$(19, \infty)$	0.740	(17, 00)	5.840	1.1	
(1, 5)	$(8,\infty)$	7.023	(i, ∞)	7.042	0.3	
(1, 10)	$(6, \infty)$	8.426	$(4, \infty)$	8.630	2.4	
$\lambda_1 = 0.76$	65; $\lambda_2 = 0.0$	$085; \beta_1 =$	$\beta_2 = 0.5;$			
$s_1 = s_2 =$	$= 0.4; \rho = 0$.425.				
	optim	um	appro	ximatior	1	
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	8	
(01,02)	(m_1, m_2)	0.752	(m_1, m_2)	0.752	00	
(1, 0.1)	(00, 1)	0.755	$(\omega, 1)$	0.755		
(1, 0.2)	$(\infty, 1)$	0.109	$(\infty, 1)$	0.109		
(1, 0.5)	$(\infty, 4)$	0.807	(∞, ∞)	0.807		
(1, 1)	(∞, ∞)	0.869	(∞, ∞)	0.869	0.0	
(1, 2)	(∞, ∞)	0.994	(∞, ∞)	0.994	0.0	
(1, 5)	$(10, \infty)$	1.361	(∞, ∞)	1.367	0.4	
(1, 10)	$(6, \infty)$	1.933	(∞, ∞)	1.990	2.9	
<u>_</u>						
1 - 0.70	$(5.)_{-0.0}$	195. A	$\beta_{-} = 1$			٦
$x_1 = 0.70$	$J_{3}, \lambda_{2} = 0.0$	$p_{00}, p_1 =$	$p_2 = 1$,			
$s_1 = s_2 =$	$= 0.4; \rho = 0$.85.				_
	optin	num	app	roximati	on	
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%	
(1, 0.1)	$(\infty, 2)$	3.685	$(\infty, 1)$	3.730) 1.2	
(1, 0.2)	$(\infty, 2)$	3.971	$(\infty, 2)$	3.971	ι 0.0	
(1, 0.5)	$(\infty, 8)$	4.698	(∞, ∞)	4.778	3 1.7	
(1, 1)	(∞, ∞)	5.673		5.673	3 0.0	
(1, 2)	(19. m)	7.313	lim mi	7.464	1 2.0	
(1, 5)	$(10, \infty)$	0 370	$(20,\infty)$	0.505	2 24	
(1, 0)	$(13, \infty)$	12 50	$(20, \infty)$	10 50		
(1, 10)	(14, 00)	12.09	$(14, \infty)$	12.09	0.0	
						_
$\lambda_1 = 0.76$	55; $\lambda_2 = 0.0$	$085; \beta_1 =$	$\beta_2 = 1;$			
$s_1 = 0.1;$	$s_2 = 0.7; \rho$	0 = 0.85.				
	optin	num	app	roximati	on	7
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	1 %	1
(1, 0, 1)	$(\infty, 2)$	3,740	$(\infty, 1)$	3 780	1 1 3	4
(1, 0.1)		4 032		1 1 100		
(1, 0.2)		4.002	(ω, z)	4.034		
(1, 0.0)	(∞, δ)	4.110	(∞, ∞)	4.858		
(1, 1)	(∞, ∞)	5.769	(∞, ∞)	5.76		
(1, 2)	$(19,\infty)$	7.543	(∞, ∞)	7.597	0.7	
(1, 5)	$(19, \infty)$	9.646	(20, ∞)	9.786	3 1.5	
(1, 10)	$ (14, \infty) $	12.86	(14, ∞)	12.86	0.0	
						_
$\lambda_1 = 0.5$	$\lambda_0 = 0.25$	$R_1 = R_2$	= 1.			
$s_1 = 0.0$	$s_0 = 0.20$	-0.75	1			
31 - 0.1,	32 = 0.2, p	- 0.10.				
<i>(</i>)	optim	uin	appro	ximation		
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%	
(1, 0.1)	$(\infty, 1)$	1.137	$(\infty, 1)$	1.137	0.0	
(1, 0.2)	$(\infty, 2)$	1.362	$(\infty, 1)$	1.389	2.0	
(1, 0.5)	$(\infty, 3)$	1.895	$(\infty, 3)$	1.895	0.0	
(1, 1)'	(∞, ∞)	2.575	(∞, ∞)	2.575	0.0	
(1, 2)	$(6,\infty)$	3,409	$(8, \infty)$	3.427	0.5	
(1, 5)	(3, ~)	5.035	$(3, \infty)$	5 035		
(1, 0)	$(2, \infty)$	7 262	$(2, \infty)$	7 262		
(1, 10)	(4,00)	1.202	(4,00)	1.202	0.0	

114

6.6 Numerical results for the unconstrained problem

$\lambda_1 = 0.5$	$\lambda_2 = 1; \beta_1$	$= 1; \beta_2 =$	= 0.3;					
$s_1 = 0.2;$	$s_1 = 0.2; \ s_2 = 0.6; \ \rho = 0.8.$							
	optin	num	appr	approximation				
(c_1, c_2)	(k_1, k_2)	cost	(k_1, k_2)	cost	%			
(1, 0.1)	$(\infty, 20)$	2.188	(∞, ∞)	2.342	7.0			
(1, 0.2)	$(\infty, 20)$	2.820	(∞, ∞)	2.911	3.2			
(1, 0.5)	$(19, \infty)$	4.475	(∞, ∞)	4.618	3.1			
(1, 1)	(8,∞)	6.611	$(15, \infty)$	6.928	4.8			
(1, 2)	$(6, \infty)$	9.981	$(7, \infty)$	10.11	1.3			
(1, 5)	$(4,\infty)$	18.49	(4,∞)	18.49	0.0			
(1, 10)	$(3, \infty)$	31.66	$(3, \infty)$	31.66	0.0			

TABLE 6.4. The unconstrained case; two-queue models.

$\lambda_1 = \lambda_2 = \lambda_3$	$\lambda_1 = \lambda_2 = \lambda_3 = 0.25; \ \beta_1 = 0.2; \ \beta_2 = 0.6; \ \beta_3 = 2.2;$							
$s_1 = s_2 = s_3$	$s_1 = s_2 = s_3 = 0.1; \ \rho = 0.75.$							
	optimu	m	approximation					
(c_1, c_2, c_3)	(k_1, k_2, k_3)	cost	(k_1, k_2, k_3)	cost	%			
(10, 10, 10)	$(\infty, \infty, 2)$	34.26	$(\infty, \infty, 2)$	34.26	0.0			
(10, 3, 10)	$(\infty, 4, 2)$	29.40	$(\infty, \infty, 2)$	29.43	0.1			
(10, 10, 3)	$(\infty, \infty, 1)$	18.22	$(\infty, \infty, 1)$	18.22	0.0			
(10, 3, 3)	$(\infty, 8, 1)$	14.71	$(\infty, \infty, 1)$	14.72	0.0			
(10, 3, 1)	$(\infty, 8, 1)$	9.28	$(\infty, \infty, 1)$	9.28	0.0			
(10, 1, 3)	$(\infty, 2, 1)$	13.64	$(\infty, 3, 1)$	13.65	0.1			
(10, 1, 1)	$(\infty, 3, 1)$	8.26	$(\infty, \infty, 1)$	8.28	0.3			

$\lambda_1 = 0.6; \ \lambda_2 = 0.2;$	$\lambda_3 = 0.05; \ \beta_1 = 0.2; \ \beta_2 = 0.6; \ \beta_3 = 2.2;$
$s_1 = s_2 = s_3 = 0.5;$	$ \rho = 0.35. $

	optimu	m	approximation			
(c_1,c_2,c_3)	(k_1, k_2, k_3)	cost	(k_1, k_2, k_3)	cost	%	
(10, 10, 10)	$(\infty, \infty, 1)$	14.78	$(\infty, \infty, 1)$	14.78	0.0	
(10, 3, 10)	$(\infty, 3, 1)$	12.35	$(\infty, \infty, 1)$	12.38	0.3	
(10, 10, 3)	$(\infty, \infty, 1)$	13.89	$(\infty, \infty, 1)$	13.89	0.0	
(10, 3, 3)	$(\infty, 3, 1)$	11.47	$(\infty, \infty, 1)$	11.49	0.2	
(10, 3, 1)	$(\infty, 3, 1)$	11.22	$(\infty, \infty, 1)$	11.24	0.2	
(10, 1, 3)	$(\infty, 2, 1)$	10.66	$(\infty, 1, 1)$	10.78	1.1	
(10, 1, 1)	$(\infty, 2, 1)$	10.41	$(\infty, 1, 1)$	10.54	1.2	

$\lambda_1 = 0.6; \ \lambda_2 = 0.2; \ \lambda_3 = 0.05; \ \beta_1 = \beta_2 = \beta_3 = 1;$								
$s_1 = s_2 = s_3 = 0.1; \ \rho = 0.85.$								
	optimur	m	approximation					
(c_1,c_2,c_3)	(k_1,k_2,k_3)	cost	(k_1,k_2,k_3)	cost	%			
(10, 10, 10)	(∞, ∞, ∞)	53.05	(∞, ∞, ∞)	53.05	0.0			
(10, 3, 10)	$(\infty, 2, \infty)$	32.78	$(\infty, 2, \infty)$	32.78	0.0			
(10, 10, 3)	$(\infty, \infty, 1)$	45.06	$(\infty, \infty, 1)$	45.06	0.0			
(10, 3, 3)	$(\infty, 3, 1)$	30.04	$(\infty, 2, 1)$	30.13	0.3			
(10, 3, 1)	$(\infty, 3, 1)$	28.53	$(\infty, 2, 1)$	28.97	1.5			
(10, 1, 3)	$(\infty, 1, 2)$	22.19	$(\infty, 1, 2)$	22.19	0.0			
(10, 1, 1)	$(\infty, 2, 1)$	21.26	$(\infty, 1, 1)$	21.35	0.4			

$\lambda_1 = 0.3; \ \lambda_2 = 0.8; \ \lambda_3 = 0.1; \ \beta_1 = 0.2; \ \beta_2 = 0.5; \ \beta_3 = 2;$								
$s_1 = 2; s_2 = 0.1; s_3 = 0.5; \rho = 0.66.$								
	optimu	m	approximation					
(c_1,c_2,c_3)	$\left(k_{1},k_{2},k_{3} ight)$	cost	(k_1,k_2,k_3)	cost	%			
(10, 10, 10)	$(\infty, \infty, 2)$	63.47	$(\infty, \infty, 4)$	65.59	3.3			
(10, 3, 10)	$(\infty, 18, 4)$	37.30	(∞, ∞, ∞)	37.76	1.2			
(10, 10, 3)	$(\infty, \infty, 2)$	55.48	$(\infty, \infty, 2)$	55.48	0.0			
(10, 3, 3)	$(\infty, \infty, 2)$	31.60	$(\infty, \infty, 2)$	31.60	0.0			
(10, 3, 1)	$(\infty, \infty, 2)$	29.32	$(\infty, \infty, 2)$	29.32	0.0			
(10, 1, 3)	$(\infty, 14, 2)$	24.55	$(\infty, \infty, 3)$	25.27	2.9			
(10, 1, 1)	$(\infty, 16, 2)$	22.46	$(\infty, \infty, 2)$	22.50	0.2			

 TABLE 6.5. The unconstrained case; three-queue models; exponential service times.

Chapter 6 Optimization of k-limited service strategies

$\lambda_1 = \lambda_2 = \lambda_3 = 0.25; \ \beta_1 = 0.2; \ \beta_2 = 0.6; \ \beta_3 = 2.2;$							
$s_1 = s_2 = s_3 = 0.1; \ \rho = 0.75.$							
	optimu	m	approx	cimation	1 07		
(c_1, c_2, c_3)	(k_1, k_2, k_3)	cost	$(\kappa_1,\kappa_2,\kappa_3)$	COSt	70		
(10, 10, 10)	$(\infty, \infty, 2)$	27.23	$(\infty, \infty, 2)$	27.23	0.0		
(10, 3, 10)	$(\infty, 4, 2)$	23.21	$(\infty, \infty, 2)$	23.23			
(10, 10, 3)	$(\infty, \infty, 1)$	14.40	$(\infty, \infty, 1)$	14.40	0.0		
(10, 3, 3)	$(\infty, 8, 1)$	11.70	$(\infty, \infty, 1)$	7 440	0.0		
(10, 3, 1)	$(\infty, 14, 1)$	10.96	$(\infty, \infty, 1)$	10.89	0.0		
(10, 1, 3)	$(\infty, 2, 1)$	10.80	$(\infty, 3, 1)$	10.00			
(10, 1, 1)	$(\infty, 3, 1)$	0.033	$(\infty, \infty, 1)$	0.052	0.3		
<u></u>		<u> </u>					
$\lambda_1 = 0.6; \ \lambda_2$	$= 0.2; \lambda_3 = 0.0$	$J_5; \beta_1 = 0$	$\beta_{2}; \beta_{2} = 0.6; \beta_{2}$	$g_3 = 2.2;$			
$s_1 = s_2 = s_3$	$= 0.5; \rho = 0.3;$	J.					
	optimu		approx	cination	1 07		
(c_1, c_2, c_3)	$(\kappa_1, \kappa_2, \kappa_3)$	COST	$(\kappa_1,\kappa_2,\kappa_3)$	Cost	1 70		
(10, 10, 10)	$(\infty, \infty, 1)$	13.79	$(\infty, \infty, 1)$	13.79	0.0		
(10, 3, 10)	$(\infty, 3, 1)$	11.04	$(\infty, \infty, 1)$	11.50			
(10, 10, 3)	$(\infty, \infty, 1)$	12.97	$(\infty, \infty, 1)$	12.97			
(10, 3, 3)	$(\infty, 3, 1)$	10.72	$(\infty, \infty, 1)$	10.74	0.1		
(10, 3, 1)	$(\infty, 4, 1)$	10.49	$(\infty, \infty, 1)$	10.50			
(10, 1, 3)	$(\infty, 2, 1)$	9.979	$(\infty, 1, 1)$	10.09			
(10, 1, 1)	$(\infty, 2, 1)$	9.751	$(\infty, 1, 1)$	9.875	1.3		
$\lambda_1 \equiv 0.0; \ \lambda_2$ $s_1 \equiv s_2 \equiv s_3$	$= 0.2; \ \lambda_3 \equiv 0.0 \\ = 0.1; \ \rho = 0.8$	5; $p_1 = p_2$	$p_2 = p_3 = 1;$				
	optimum approximation						
(c_1, c_2, c_3)	(k_1, k_2, k_3)	cost	(k_1, k_2, k_3)	cost	%		
(10, 10, 10)	(∞, ∞, ∞)	41.03	(∞, ∞, ∞)	41.03	0.0		
(10, 3, 10)	$(\infty, 2, \infty)$	25.72	$(\infty, 2, \infty)$	25.72	0.0		
(10, 10, 3)	$(\infty, \infty, 1)$	35.32	$(\infty, \infty, 1)$	35.32	0.0		
(10, 3, 3)	$(\infty, 2, 1)$	23.57	$(\infty, 2, 1)$	23.57	0.0		
(10, 3, 1)	$(\infty, 3, 1)$	22.44	$(\infty, 2, 1)$	22.65	0.9		
(10, 1, 3)	$(\infty, 1, 2)$	17.34	$(\infty, 1, 2)$	17.34	0.0		
(10, 1, 1)	$(\infty, 1, 1)$	16.67	$(\infty, 1, 1)$	16.67	0.0		
,							
$\lambda_1 = 0.3; \ \lambda_2$	$= 0.8; \lambda_3 = 0.1$	$l; \beta_1 = 0.$	2; $\beta_2 = 0.5; \beta_3$	= 2;			
$s_1 = 2; s_2 =$	$0.1; s_3 = 0.5; \mu$	o = 0.66.					
	optimu	m	approxi	imation			
(c_1, c_2, c_3)	(k_1, k_2, k_3)	cost	(k_1, k_2, k_3)	cost	%		
(10, 10, 10)	$(\infty, \infty, 3)$	55.40	$(\infty, \infty, 4)$	55.94	1.0		
(10, 3, 10)	$(\infty, 18, \infty)$	33.93	(∞, ∞, ∞)	34.36	1.2		
(10, 10, 3)	$(\infty, \infty, 2)$	48.62	$(\infty, \infty, 2)$	48.62	0.0		
(10, 3, 3)	$(\infty, \infty, 2)$	28.69	$(\infty, \infty, 2)$	28.69	0.0		
(10, 3, 1)	$(\infty, \infty, 2)$	26.67	$(\infty, \infty, 2)$	26.67	0.0		
(10, 1, 3)	$(\infty, 14, 2)$	22.39	$(\infty, \infty, 3)$	23.27	3.9		
(10, 1, 1)	$(\infty, 16, 2)$	20.48	$(\infty, \infty, 2)$	20.98	2.4		

TABLE 6.6. The unconstrained case; three-queue models; Erlang-2 service

times.

· · · · · · · · · · · · · · · · · · ·							
$\lambda_1 = 0.35; \ \lambda_2 = \ldots = \lambda_5 = 0.1; \ \beta_1 = 1; \ \beta_2 = \ldots = \beta_5 = 1;$							
$s_1 = 0.1; s_1$	$_2 = \ldots = s_5$	= 0.05;	$ \rho = 0.75. $				
	optimum approximation						
(c_1, c_{2-5})	(k_1, k_{2-5})	cost	(k_1, k_{2-5})	cost	%		
(1, 0.1)	$(\infty, 1)$	0.882	$(\infty, 1)$	0.882	0.0		
(1, 0.5)	$(\infty, 3)$	1.776	$(\infty, 4)$	1.776	0.0		
(1, 1)	(∞, ∞)	2.623	(∞, ∞)	2.623	0.0		
(1, 2)	(3, ∞)	3.898	$(4, \infty)$	3.898	0.0		

TABLE 6.7. The unconstrained case; a five-queue model.

Discussion of the numerical results.

The proposed approach performs extremely well; in the majority of the 151 examples, the achieved waiting cost is less than 1% larger than the minimum waiting cost. Only twice the achieved waiting cost is more than 5% larger than the minimum waiting cost, not once more than 10% larger.

The optimal service limits, as well as the service limits obtained from the Fuhrmann & Wang approximation, always satisfied the property stated in Conjecture 6.5.2, i.e., if $c_i/\beta_i = \max_{j=1,...,n} c_j/\beta_j$, then $k_i^* = \infty$.

The results for an Erlang-2 service time distribution are similar to the results for an exponential service time distribution. The waiting cost for an Erlang-2 service time distribution is always smaller than the waiting cost for an exponential service time distribution with the same mean. Intuitively, the waiting times are indeed likely to be smaller when the variance of the service time distribution is smaller. The *optimal service limits* for an Erlang-2 service time distribution, however, hardly differ from the optimal service limits for an exponential service time distribution with the same mean.

6.7 Concluding Remarks and suggestions for further research

We have studied the problem of finding the optimal service limits in a cyclic polling system with the k-limited service discipline. The use of the Fuhrmann & Wang approximation is shown to be very effective in finding the optimal service limits. In the numerical experiments, we have observed that the waiting cost according to the Fuhrmann & Wang approximation sometimes differs dramatically from the 'true' waiting cost obtained from the PSA, but that still the optimal service limits according to the Fuhrmann & Wang approximation agree with the 'true' optimal service limits obtained from the PSA. Even when completely misjudging the mean waiting time, the Fuhrmann & Wang approximation apparently *does* capture the major factors important for efficient operation of the system.

The waiting-time approximation for the Bernoulli service discipline that Blanc & Van der Mei [23] use to find the optimal Bernoulli parameters q_i , and the Fuhrmann & Wang approximation that we use, coincide when $q_i = 1 - 1/k_i$, cf. Remark 6.3.1. The effectiveness of both approximations suggests that, as far as optimization is concerned, the Bernoulli service discipline is a very good emulation of the k-limited service discipline. Yet, as far as evaluation of the mean waiting time is concerned, the Bernoulli service discipline is often not a very good approximation of the k-limited service discipline. Due to the stochastic nature of the Bernoulli service discipline, the mean waiting times tend to be larger than for the k-limited service discipline, cf. Tedijanto [181] Chapter 5, [182], [183].

In the present chapter, we have been concerned with optimization of the *ser*vice discipline, ranging from 1-limited to exhaustive, for a given cyclic server routing. Earlier studies mostly were concerned with optimization of the *server* routing for a given service discipline, like 1-limited, gated, or exhaustive, cf. Section 1.5. We feel that it would also be worthwhile to consider simultaneous optimization of the server routing and the service discipline. Simultaneous optimization of the number of visits and the amount of service per visit would enable more flexible priorization of the various queues.

At a few instances, we faced difficult monotonicity questions: monotonicity of EW_i in k_j , monotonicity of the mean waiting time for an M/G/1 queue with vacations in the vacation time variance. Relatively few monotonicity results for polling and vacation models have been obtained; this seems an interesting area for further research.

CHAPTER 7

OPTIMIZATION OF FIXED TIME POLLING SCHEMES

7.1 INTRODUCTION

In the present chapter, we consider a polling system operated with a fixed time polling (ftp) scheme. An ftp scheme specifies which queue should be visited at what time, i.e., it specifies not only the *order* of the visits, but also the *starting times* of the visits. We are interested in the problem of constructing ftp schemes that contribute to an efficient operation of the system.

As a major benefit, ftp schemes provide by definition a guarantee on the server return time, i.e., the time until a queue receives service again. The visit times being specified beforehand, ftp schemes are also very attractive from a scheduling point of view, especially when the polling processes in question interact with other communication or production processes. Moreover, ftp schemes are very appropriate to effectuate some kind of priorization, by scheduling more visits, longer visits, or both, to high-priority queues. The consequence, however, is that ftp schemes may force the server to idle at an empty queue, while there are customers waiting at other queues.

The similarity between ftp schemes and k-limited service, as considered in the previous chapter, lies in the limitation of the amount of service provided during a visit. The difference is that in ftp schemes, the limitation refers to the visit *time* rather than the *number* of services during a visit, as in k-limited service, and that the visit is not completed until the service limit is reached, even when the queue becomes empty.

The present chapter originates from consultancy work done by order of PTT Research, Groningen. In the course of 1992, PTT Telecom started to gradually introduce itemized telephone billing in the Netherlands. To transfer the call data involved from the telephone switches to the billing center, PTT Telecom applies a polling technique similar to the polling data link control scheme described in Section 1.1. According to a fixed time schedule, a so-called mediation system polls the telephone switches. When polled by the mediation system, the telephone switches may start to transmit files with call records to the billing center. For efficiency reasons, the delays occurring in collecting the call data should be kept within bounds.

To the best of the author's knowledge, ftp schemes have not been studied before in a strict polling context. However, in other multi-class queueing settings, various similar disciplines have been considered under a variety of names. Very similar to ftp schemes are e.g. so-called clocked schedules, studied in [1], [82], [96]. However, in these studies the multi-class queueing structure corresponds to a priority discipline in accordance with the timing requirements of tasks in real-time systems, rather than a distinction between separate queues like in a polling context. Therefore, these studies are primarily concerned with the short-term delay of high-priority tasks with extremely stringent timing requirements, or with the probability of urgent tasks not getting served in the scheduled interval, or with the long-term delay of low-priority tasks with more liberal timing requirements, rather than with achieving a minimal overall time delay. Also related to ftp schemes are so-called Time Division Multiplexing (TDM) protocols, considered e.g. in [118], [120], [156].

It is most unlikely that ftp schemes allow an exact analysis, except for deterministic arrival, service, and switch-over processes. Note that ftp schemes do not satisfy Property 1.4.1, which provides a global characterization of the class of service disciplines that are amenable to an exact analysis. We therefore feel justified in resorting to approximations for the mean waiting times. Since we are primarily concerned with minimizing a weighted sum of the mean waiting times, rather than evaluating the mean waiting times themselves, we deliberately seek very simple approximations, which do not necessarily need to be very accurate, as long as they rightly capture the *behavior* of the mean waiting times.

The remainder of the chapter is organized as follows. In Section 7.2, we present a detailed model description. In Section 7.3, we formulate the problem of constructing an efficient ftp scheme as a mathematical program. In view of its NP-hardness, we develop in Section 7.4 a heuristic method for solving the mathematical program. In Section 7.5, we give an overview of the numerical results that we gathered to validate the method.

7.2 MODEL DESCRIPTION

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by a single server S. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic' model in Section 1.3, but in the present chapter we do not assume that the customers arrive according to Poisson processes.

The server operates according to an ftp scheme, which specifies not only the order of the visits, but also the starting times of the visits. We represent an

7.3 Constructing an efficient ftp scheme I

ftp scheme by a vector pair (P,T), $P_k \in \{1,\ldots,n\}$, $T_k \ge 0$, $k = 1,\ldots,m$. The vector P contains the polling table associated with the ftp scheme, i.e., the k-th visit is to queue P_k , $k = 1,\ldots,m$. The vector T contains the extended visit times associated with the ftp scheme, i.e., T_k is the time between the start of the k-th visit and the start of the (k + 1)-th visit, $k = 1,\ldots,m$, where m + 1 is to be understood as 1.

To supplement an ftp scheme, it should be specified which customers qualify for service during a visit. In addition, it should be specified what is to be done in case the available visit time expires before all entitled customers have been served. In this chapter, we assume that service is gated, i.e., only customers that were present at the start of a visit, qualify for service during the visit. In case the available visit time expires before all entitled customers have been served, we assume that the next visit is postponed. To still fulfil the guarantee on the starting times, we demand however that the available visit times are most rarely exceeded, i.e., we demand that the available visit times are sufficiently long to meet the gated service discipline (with high probability). These assumptions were made in the light of the actual telecommunication application that motivated the present study, cf. Section 7.1.

In this chapter, we are interested in the problem of finding an ftp scheme that minimizes the mean total waiting cost per unit of time. Denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer, i = 1, ..., n. Let c_i represent the waiting cost per unit of time of a type-*i* customer. The mean total waiting cost per unit of time amounts to $\sum_{i=1}^{n} c_i \lambda_i \mathbf{EW}_i$.

7.3 Constructing an efficient FTP scheme I

As stated in the previous section, we are interested in the problem of constructing an ftp scheme that minimizes the mean total waiting cost per unit of time, under the side-constraint of most rarely exceeding the available visit times. Starting from rather simple approximations, we formulate in this section the problem under consideration as a mathematical program. In view of its NP-hardness, we describe in the next section a heuristic method for solving the mathematical program. The approach bears resemblance to the approach in Kruskal [134] to the problem of determining efficient visit orders for polling systems with deterministic arrival, service, and switch-over processes, and the approach in Boxma, Levy, & Weststrate [48] to a similar problem for polling systems with a Poisson arrival process and general service and switch-over processes.

We first introduce some notation. Denote by \mathbf{U}_k the k-th available visit time, i.e., the k-th extended visit time minus the switch-over time into queue P_k , by \mathbf{V}_k the k-th required visit time, i.e., the time the server needs to do the work during the k-th visit, and by SC_k the k-th subcycle time, i.e., the time between the start of the k-th visit and the start of the previous visit to queue P_k , k = 1, ..., m. Denote by C the cycle time, i.e., the time the server needs to pass through the ftp scheme once. By the nature of the ftp scheme, T_k , SC_k , and C are deterministic, but \mathbf{U}_k and \mathbf{V}_k are not (unless, respectively, the switch-over process and the arrival and service processes are strictly deterministic, the interarrival time in addition being a divisor of the cycle time).

Of course, T_k , \mathbf{U}_k , \mathbf{V}_k , SC_k , and C are closely related. Firstly, the k-th available visit time \mathbf{U}_k is what remains of the k-th extended visit time T_k , after the switch-over time into queue P_k ,

$$T_k = \mathbf{U}_k + \mathbf{S}_{P_k}, \qquad k = 1, \dots, m, \tag{7.1}$$

so that in fact \mathbf{U}_k may even take a negative value, when \mathbf{S}_{P_k} happens to take a value bigger than T_k . Secondly, by the nature of the gated service discipline, the k-th required visit time \mathbf{V}_k equals the amount of work that arrives at queue P_k during the k-th subcycle time SC_k , $k = 1, \ldots, m$.

The k-th subcycle time is composed of the extended visit times between the start of the k-th visit and the start of the previous visit to queue P_k ,

$$SC_k = \sum_{l=1}^{m} h_{kl} T_l, \qquad k = 1, \dots, m.$$
 (7.2)

Here the matrix $H = (h_{kl})$ is defined by

$$h_{kl} := \begin{cases} 1 & \text{if } P_{l+1}, \dots, P_{k-1} \neq P_k \\ 0 & \text{otherwise} \end{cases} \qquad k, l = 1, \dots, m,$$

i.e., h_{kl} indicates whether the *l*-th extended visit time belongs to the *k*-th subcycle time: $h_{kl} = 0$ iff the *k*-th subcycle time begins after the start of the *l*-th visit.

The cycle time is composed of all the extended visit times,

$$C = \sum_{l=1}^{m} T_l.$$
 (7.3)

The cycle time may as well be viewed as consisting of all the subcycle times corresponding to any Q_i ,

$$C = \sum_{\{k:P_k=i\}} SC_k, \qquad i = 1, \dots, n.$$
(7.4)

Note that substituting (7.2) into (7.4) indeed yields (7.3), as $\sum_{\{k:P_k=i\}} h_{kl} = 1$, $i = 1, \ldots, n, l = 1, \ldots, m$.

To be able to formulate the problem as a mathematical program, we now express the mean total waiting cost per unit of time, as well as the side-constraint of most rarely exceeding the available visit times, in terms of the ftp scheme. To start with the latter, we may represent the side-constraint by

$$\Pr{\{\mathbf{U}_k < \mathbf{V}_k\} \le \pi_{P_k}, \qquad k = 1, \dots, m,}$$
(7.5)

7.3 Constructing an efficient ftp scheme I

with π_i , i = 1, ..., n, prespecified bounds for the probabilities of exceeding the available visit times. Since (7.5) does not really fit into the framework of a mathematical program, we replace (7.5) by a constraint of the type

$$\mathrm{E}\mathbf{U}_k - Y_k \ge \mathrm{E}\mathbf{V}_k + Z_k, \qquad k = 1, \dots, m.$$

Here Y_k and Z_k are measures for the variability of respectively \mathbf{U}_k and \mathbf{V}_k such that $\Pr{\{\mathbf{U}_k < \mathbf{E}\mathbf{U}_k - Y_k\}} \le \gamma_{P_k}\pi_{P_k}$ and $\Pr{\{\mathbf{V}_k > \mathbf{E}\mathbf{V}_k + Z_k\}} \le (1 - \gamma_{P_k})\pi_{P_k}$, with $0 \le \gamma_i \le 1$, i = 1, ..., n. Thus, we slightly strengthen (7.5), since $\Pr{\{\mathbf{U}_k \ge \mathbf{V}_k\}} \ge \Pr{\{\mathbf{U}_k \ge \mathbf{E}\mathbf{U}_k - Y_k, \mathbf{V}_k \le \mathbf{E}\mathbf{V}_k + Z_k\}} = \Pr{\{\mathbf{U}_k \ge \mathbf{E}\mathbf{U}_k - Y_k\}} \Pr{\{\mathbf{V}_k \le \mathbf{E}\mathbf{V}_k + Z_k\}} = 1 - \pi_{P_k} + \gamma_{P_k}(1 - \gamma_{P_k})\pi_{P_k}^2}$. From (7.1), we have $\mathbf{E}\mathbf{U}_k = T_k - s_{P_k}, k = 1, ..., m$. Since \mathbf{V}_k equals the amount of work that arrives at Q_{P_k} during SC_k , we have $\mathbf{E}\mathbf{V}_k = \rho_{P_k}SC_k, k = 1, ..., m$. We further take $Y_k = \delta_{P_k}s_{P_k}, k = 1, ..., m$, and $Z_k = \epsilon_{P_k}\rho_{P_k}SC_k, k = 1, ..., m$. Here δ_i and ϵ_i are measures for the variability of respectively the switch-over times into Q_i and the interarrival and service times at Q_i , such that

$$\Pr\{\mathbf{S}_{i} > (1+\delta_{i})s_{i}\} \leq \gamma_{i}\pi_{i},$$

$$\sum_{h=0}^{\infty} \left(A_{i}^{h*}(\hat{SC}_{i}) - A_{i}^{(h+1)*}(\hat{SC}_{i})\right) \left(1 - B_{i}^{(h+1)*}(\rho_{i}(1+\epsilon_{i})\hat{SC}_{i})\right) \leq (1-\gamma_{i})\pi_{i},$$

with $F^{j*}(\cdot)$ denoting the *j*-fold convolution of $F(\cdot)$. Here \hat{SC}_i is a rough estimate for the subcycle times corresponding to Q_i , preferably as pessimistic as possible from the viewpoint of determining ϵ_i . Thus, we slightly strengthen (7.5) further, since

$$\Pr\{\mathbf{U}_{k} < \mathbf{E}\mathbf{U}_{k} - Y_{k}\} = \Pr\{\mathbf{S}_{P_{k}} > (1 + \delta_{P_{k}})s_{P_{k}}\},\$$
$$\Pr\{\mathbf{V}_{k} > \mathbf{E}\mathbf{V}_{k} + Z_{k}\} = \Pr\{\mathbf{V}_{k} > \rho_{P_{k}}(1 + \epsilon_{P_{k}})SC_{k}\},\$$
$$\Pr\{\mathbf{V}_{k} > t\} \le \sum_{h=0}^{\infty} \left(A_{P_{k}}^{h*}(SC_{k}) - A_{P_{k}}^{(h+1)*}(SC_{k})\right) \left(1 - B_{P_{k}}^{(h+1)*}(t)\right), \ t \ge 0,\$$

in particular for $t = \rho_{P_k}(1 + \epsilon_{P_k})SC_k$. The latter inequality holds, since $A_{P_k}^{h*}(SC_k) - A_{P_k}^{(h+1)*}(SC_k)$ is the probability that (h+1) type- P_k customers arrive during SC_k , under the pessimistic assumption that the first customer arrives at the beginning of SC_k , and $1 - B_{P_k}^{(h+1)*}(t)$ is the probability that the service of (h+1) type- P_k customers is not finished within time $t, t \ge 0$. Instead of $Z_k = \epsilon_{P_k}\rho_{P_k}SC_k$, one might take e.g. $Z_k = \epsilon_{P_k}\rho_{P_k}SC_k + (1+\zeta_{P_k})\beta_{P_k}$. Here ζ_i is a measure for the variability of the service times at Q_i , such that

Here ζ_i is a measure for the variability of the service times at Q_i , such that $1 - B_i((1 + \zeta_i)\beta_i) \leq (1 - \gamma_i)\pi_i$, $i = 1, \ldots, n$. The factor $(1 + \zeta_i)\beta_i$ would avoid that $\epsilon_i \to \infty$ in the hypothetic situation that $\hat{SC}_i \downarrow 0$. If one takes $Z_k = \epsilon_{P_k}\rho_{P_k}SC_k + (1 + \zeta_{P_k})\beta_{P_k}$ instead of $Z_k = \epsilon_{P_k}\rho_{P_k}SC_k$, then everywhere $(1 + \delta_i)s_i$ is to be replaced by $(1 + \delta_i)s_i + (1 + \zeta_i)\beta_i$.

When the distributions of the interarrival, service, and switch-over times are

specified, there are no serious complications in expressing δ_i and ϵ_i in terms of π_i . Nevertheless, in real life one may rather determine δ_i and ϵ_i empirically than make a questionable assumption about the distributions of the interarrival, service, and switch-over times, needed in expressing δ_i and ϵ_i in terms of π_i .

Concluding, we represent the side-constraint of most rarely exceeding the available visit times by

$$T_k \ge \rho_{P_k} (1 + \epsilon_{P_k}) SC_k + (1 + \delta_{P_k}) s_{P_k}, \qquad k = 1, \dots, m.$$
 (7.6)

Note that summing (7.6) with respect to k = 1, ..., m, using (7.3) and (7.4), yields $C \ge \left(\rho + \sum_{i=1}^{n} \epsilon_i \rho_i\right) C + \sum_{i=1}^{n} m_i (1 + \delta_i) s_i$. Apparently, $\rho + \sum_{i=1}^{n} \epsilon_i \rho_i < 1$ is a necessary condition for the extended visit times to be all non-negative. In the proof of Lemma 7.4.1, it will also appear to be a sufficient condition. In the sequel, the condition $\rho + \sum_{i=1}^{n} \epsilon_i \rho_i < 1$ is always assumed to hold.

We now express the mean total waiting cost per unit of time in terms of the ftp scheme. To approximate the waiting time of an arbitrary type-*i* customer, we condition on the event that the type-*i* customer in question arrives during the *k*-th subcycle time SC_k with $P_k = i$. The latter event occurs with probability SC_k/C , as the probability that a customer arrives in a specific subcycle is proportional to the length of the subcycle, irrespective of the nature of the arrival process (unless the arrival process is strictly deterministic, the interarrival time in addition being a divisor of the cycle time). We further act as if the available visit times are never exceeded. The waiting time of an arbitrary type-*i* customer that arrives during the *k*-th subcycle time SC_k , with $P_k = i$, is then composed of:

i. the time from its arrival to the start of the next visit to Q_i , i.e., the residual lifetime \mathbf{RSC}_k of SC_k at the arrival epoch of the customer;

ii. the time from the start of the next visit to Q_i to the start of its service, i.e., the time the server needs to do the work \mathbf{PV}_k that arrived at Q_i during the past lifetime \mathbf{PSC}_k of SC_k at the arrival epoch of the customer.

Since SC_k is deterministic, $ERSC_k = \frac{1}{2}SC_k$, $EPSC_k = \frac{1}{2}SC_k$. We further use the approximation $EPV_k \approx \rho_i EPSC_k$, which in fact is exact for a Poisson arrival process, cf. Boxma et al. [48]. For less variable arrival processes, the approximation will probably result in a slight overestimation, which however will tend to zero for SC_k sufficiently large compared to $1/\lambda_i$, cf. Kruskal [134]. Concluding, we approximate the mean waiting time of an arbitrary type-*i* customer by

$$\mathbf{EW}_{i} \approx \frac{1+\rho_{i}}{2C} \sum_{\{k:P_{k}=i\}} SC_{k}^{2}, \qquad i = 1, \dots, n,$$

$$(7.7)$$

7.3 Constructing an efficient ftp scheme I

which yields for the mean total waiting cost per unit of time

$$\sum_{i=1}^{n} c_i \lambda_i \mathbf{E} \mathbf{W}_i \approx \frac{1}{2C} \sum_{i=1}^{n} c_i \lambda_i (1+\rho_i) \sum_{\{k:P_k=i\}} SC_k^2$$

$$= \frac{1}{2C} \sum_{k=1}^{m} c_{P_k} \lambda_{P_k} (1+\rho_{P_k}) SC_k^2.$$
(7.8)

Having expressed the mean total waiting cost per unit of time, as well as the side-constraint of most rarely exceeding the available visit times, in terms of the ftp scheme, we are now able to formulate the problem as a mathematical program.

Problem (I).

sub

$$\min \quad \frac{1}{2C} \sum_{i=1}^{n} c_i \lambda_i (1+\rho_i) \sum_{\{k:P_k=i\}} SC_k^2 =$$

$$\frac{1}{2C} \sum_{i=1}^{m} c_{P_k} \lambda_{P_k} (1+\rho_{P_k}) SC_k^2$$
(7.9)

$$\overline{2C} \sum_{k=1}^{C_{P_k}} C_{P_k}$$

$$T_{k} \ge \rho_{P_{k}}(1 + \epsilon_{P_{k}})SC_{k} + (1 + \delta_{P_{k}})s_{P_{k}}, \qquad k = 1, \dots, m; \quad (7.10)$$

$$SC_k = \sum_{l=1}^{m} h_{kl} T_l, \qquad k = 1, \dots, m;$$
 (7.11)

$$C = \sum_{l=1}^{m} T_l;$$
 (7.12)

$$m_i = |\{k : P_k = i\}| \ge 1, \qquad i = 1, \dots, n;$$
 (7.13)

$$h_{kl} = \begin{cases} 1 & \text{if } P_{l+1}, \dots, P_{k-1} \neq P_k \\ 0 & \text{otherwise} \end{cases} \qquad k, l = 1, \dots, m; (7.14)$$

$$P_k \in \{1, \dots, n\}, \qquad k = 1, \dots, m;$$
 (7.15)

$$T_k \ge 0, \qquad k = 1, \dots, m. \tag{7.16}$$

Note that the determination of m, the length of the polling table, is part of the optimization problem.

For a specific parameter choice, problem (I) amounts to the problem of partitioning m-2 numbers into 2 sets, such that the sums of the numbers in both sets are as equal as possible, which is known to be NP-hard; cf. Lemma 7.3.1.

Lemma 7.3.1

Problem (I) is NP-hard.

Proof See Appendix 7.A. Lemma 7.3.1 suggests that there is little hope of solving problem (I) exactly in a reasonable amount of time. In the next section, we therefore describe a method for solving problem (I) approximately.

7.4 Constructing an efficient ftp scheme II

In the previous section, we formulated the problem under consideration as a mathematical program. In view of its NP-hardness, we describe in this section a heuristic method for solving the mathematical program. The idea is to divide problem (I) into three subproblems, which are somewhat easier to handle, viz.: 1. Determination of the visit numbers m_1, \ldots, m_n .

- L. Determination of the visit numbers m_1, \ldots
- 2. Determination of the visit order.
- 3. Determination of the extended visit times T_1, \ldots, T_m .

Ad 1. Determination of the visit numbers.

To simplify the determination of the visit numbers, we forget about the visit order for now. So we ignore of which extended visit times the k-th subcycle time is composed, but of course we do not ignore that the subcycle times corresponding to any Q_i together make up the cycle time, cf. (7.4). Translated to problem (I), we replace the constraints (7.11), (7.14), and (7.15) by the constraint (7.4). It is easily verified that in the resulting problem the optimal subcycle times corresponding to Q_i are all equal, i = 1, ..., n, i.e., $SC_k = C/m_{P_k}$, k = 1, ..., m. Observe that $\sum_{h=1}^{H} x_h^2$, under the constraint $\sum_{h=1}^{H} x_h = X$, is minimal for $x_h = X/H$, h = 1, ..., H. Note that $SC_k = C/m_{P_k}$, k = 1, ..., m, suggests spacing the visits to the various queues as evenly as possible, as intuitively is indeed expected to be optimal, cf. Kruskal [134], Boxma et al. [48]. As seen from (7.10) and (7.12), all the optimal extended visit times at Q_i are then equal too, i = 1, ..., n, i.e., $T_k = \rho_{P_k} (1 + \epsilon_{P_k}) C / m_{P_k} + (1 + \delta_{P_k}) s_{P_k}$, $k = 1, \ldots, m$. Denote by D_i the common value of all these optimal extended visit times at Q_i , i = 1, ..., n. As we forget about the visit order for now, the ultimate extended visit times at Q_i will probably deviate from D_i . Remember that because of the constraint (7.10), the extended visit times cannot be determined before the visit order is determined. Nevertheless, D_i will probably be a good indication for the ultimate extended visit times at Q_i , which will be useful in determining a good visit order later on. To simplify the determination of the visit numbers even further, we relax the integrality constraint (7.13) for now too. Concluding, we formulate the problem of determining the visit numbers as follows.

7.4 Constructing an efficient ftp scheme II

Problem (II).

$$\min \qquad \sum_{i=1}^{n} \frac{c_i \lambda_i (1+\rho_i) C}{2m_i} \tag{7.17}$$

sub
$$D_i = \frac{\rho_i (1 + \epsilon_i)C}{m_i} + (1 + \delta_i)s_i, \qquad i = 1, \dots, n;$$
 (7.18)

$$C = \sum_{i=1}^{n} m_i D_i; \tag{7.19}$$

$$n_i > 0, \qquad i = 1, \dots, n.$$
 (7.20)

Note that the objective function as well as the constraints are homogeneous with regard to (m_1, \ldots, m_n, C) , as is to be expected, since concatenating an ftp scheme several times does not make any difference. So we know beforehand that the optimal solution contains a positive scaling factor with regard to (m_1, \ldots, m_n, C) . Using the Lagrangean multiplier technique, we find that the optimal solution is

$$D_i^* = \frac{\rho_i (1+\epsilon_i) C^*}{m_i^*} + (1+\delta_i) s_i, \qquad i = 1, \dots, n;$$
(7.21)

$$C^{*} = \frac{R \sum_{i=1}^{n} \sqrt{c_{i} \lambda_{i} (1 + \rho_{i}) (1 + \delta_{i}) s_{i}}}{1 - \rho - \sum_{i=1}^{n} \epsilon_{i} \rho_{i}};$$
(7.22)

$$m_i^* = R_{\sqrt{\frac{c_i \lambda_i (1+\rho_i)}{(1+\delta_i)s_i}}}, \qquad i=1,\dots,n.$$
 (7.23)

Here R is the positive scaling factor mentioned above, due to which some freedom remains in determining the total number of visits m. One may e.g. choose R such that $\left|\frac{m_i^* - [m_i^*]}{m_i^*}\right| \leq \eta_i$, $i = 1, \ldots, n$, with [x] denoting the nearest integer to x, and η_i prespecified bounds for the relative deviation of the true, integer visit numbers from the desirable, generally non-integer visit numbers. Alternatively, one may choose R such that the cycle time has some desirable value, which matches e.g. a daily or hourly pattern in a specific application. For (7.21), (7.22), and (7.23), the objective function (7.17) takes the value

$$\frac{\left(\sum_{i=1}^{n} \sqrt{c_i \lambda_i (1+\rho_i)(1+\delta_i) s_i}\right)^2}{2\left(1-\rho-\sum_{i=1}^{n} \epsilon_i \rho_i\right)}.$$
(7.24)

As seen from the argumentation preceding the formulation of problem (II), formula (7.24) provides a lower bound for the value of (7.9) for the optimal solution of problem (I). As far as the determination of the visit numbers is concerned, problem (II) may thus be conceived as a relaxation of problem (I).

127

Remark 7.4.1

Apart from the slack coefficient δ_i , formula (7.23) agrees with results in Kruskal [134] for polling systems with deterministic arrival, service, and switch-over processes, and in Boxma, Levy, & Weststrate [48] for polling systems with a Poisson arrival process and general service and switch-over processes. Kruskal [134] as well as Boxma et al. [48] are mainly interested in determining the optimal visit numbers, just like we are here. In fact, the determination of the visit *times* does not play an essential role in [134] and [48], as they are completely governed by the service discipline that is used (either exhaustive or gated). Here, however, the determination of the visit times adds an extra dimension to the problem, as they should be fixed, but still should be sufficiently long (with high probability) to not interfere with the gated service discipline, leading to the side-constraint $T_k \geq \rho_{P_k}(1+\epsilon_{P_k})SC_k + (1+\delta_{P_k})s_{P_k}$. Simply taking $T_k = \rho_{P_k} SC_k + s_{P_k}$, we find ourself in the setting of Kruskal's paper [134], where the side-constraint is automatically fulfilled by the lack of any statistical fluctuation in the arrival, service, and switch-over processes. The fact that, in spite of the differences, formula (7.23) agrees with the results in [134] and [48], suggests that the optimal visit numbers are quite robust with regard to the actual length of the visit times, as well as the statistical properties of the arrival, service, and switch-over processes, which has already been argued in the study of Boxma et al. [48].

In the special case that we confine ourself beforehand to strictly cyclic polling, i.e., $m_i = 1, i = 1, ..., n$, problem (II) reduces to

$$\min \qquad \sum_{i=1}^{n} \frac{c_i \lambda_i (1+\rho_i) C}{2} \tag{7.25}$$

sub
$$D_i = \rho_i (1 + \epsilon_i) C + (1 + \delta_i) s_i, \quad i = 1, ..., n;$$
 (7.26)

$$C = \sum_{i=1}^{n} D_i.$$
 (7.27)

The only feasible and hence optimal solution is

C

$$D_i^* = \rho_i (1 + \epsilon_i) C^* + (1 + \delta_i) s_i, \qquad i = 1, \dots, n;$$
(7.28)

* =
$$\frac{\sum_{i=1}^{n} (1+\delta_i) s_i}{1-\rho - \sum_{i=1}^{n} \epsilon_i \rho_i}$$
. (7.29)

7.4 Constructing an efficient ftp scheme II

For (7.28) and (7.29), the objective function (7.25) takes the value

$$\frac{\left(\sum_{i=1}^{n} c_i \lambda_i (1+\rho_i)\right) \left(\sum_{i=1}^{n} (1+\delta_i) s_i\right)}{2 \left(1-\rho-\sum_{i=1}^{n} \epsilon_i \rho_i\right)}.$$
(7.30)

Because of the Hölder inequality, (7.30) cannot be smaller than (7.24), as is to be expected, when we confine ourself to strictly cyclic polling. In fact, the difference between (7.30) and (7.24) gives a rough estimate of the increase in the mean total waiting cost per unit of time, when we confine ourself to strictly cyclic polling.

Ad 2. Determination of the visit order.

To facilitate the determination of the visit order, we assume that the extended visit times at Q_i are all equal to D_i^* , the indication for the extended visit times at Q_i obtained in (7.21). Translated to problem (I), we replace the constraint (7.10) by the constraint $T_k = D_{P_k}^*$. As seen from the proof of Lemma 7.3.1 however, the determination of the optimal visit order for fixed m_i and fixed $T_k = D_{P_k}^*$ is still NP-hard. Nevertheless, we rather solve problem (I) for fixed m_i and fixed $T_k = D_{P_k}^*$ approximately than an even further garbled version of problem (I) exactly.

In Appendix 7.B, we describe the Golden Ratio procedure, which is an approved method for spacing the visits to the various queues as evenly as possible, cf. [118], [120], and [156]. To be specific, define $X_k := \sum_{l=1}^m h_{kl}$, i.e., X_k is the number of visits between the start of the k-th visit and the start of the previous visit to queue P_k , k = 1, ..., m. The Golden Ratio procedure aims at making the numbers X_k with $P_k = i$ as equal as possible. In fact, the numbers X_k with $P_k = i$ are guaranteed to take at most three different values. However, these three different values are not guaranteed to be all nearly equal. Moreover, the Golden Ratio procedure aims at making the numbers of visits X_k with $P_k = i$ as equal as possible, instead of the *periods* between visits SC_k with $P_k = i$, as we should, cf. the argumentation preceding the formulation of problem (II). In other words, the Golden Ratio procedure aims at solving problem (I) for $T_k = 1, k = 1, \ldots, m$, instead of $T_k = D^*_{P_k}, k = 1, \ldots, m$. Lastly, the Golden Ratio procedure does not take into account the coefficients $c_i \lambda_i (1 + \rho_i)$ in the objective function (7.9) to weigh the improvement in the spacing of the visits to one queue against the deterioration in the spacing of the visits to another queue. In Appendix 7.C, we describe a procedure based on extremal splittings, which to some extent meets these objections. Whichever of these procedures is used, it is always worthwhile to make sure that the visit order is optimal with respect to some neighborhood. One may e.g. attempt to improve the visit order by interchanging pairs of consecutive visits.

Ad 3. Determination of the extended visit times.

At first sight, it does not seem to make sense to protract a visit any longer than needed to satisfy the side-constraint of most rarely exceeding the available visit times. Remember that the server will most rarely be busy during the extra time. Still, for extreme parameter choices, it may make sense to protract a visit. Take e.g. $n = 101, \beta_i = 0, \delta_i = 0, i = 1, \dots, n$. The side-constraint (7.5) then reduces to $T_k \ge s_{P_k}$, $k = 1, \ldots, m$. Take $c_1 \lambda_1 = 10000$, $c_i \lambda_i = 1$, $i = 2, \dots, 100, c_{101}\lambda_{101} = 100, s_1 = 1, s_i = 1, i = 2, \dots, 100, s_{101} = 100.$ From (7.23) we then obtain $m_1 = 100R$, $m_i = R$, $i = 2, ..., 100, m_{101} = R$, which for R = 1 yields the polling table $P_{2i-1} = 1, i = 1, ..., 100, P_{2i-2} = i$, $i = 2, ..., 100, P_{200} = 101$. It is easily verified that (7.9) is larger for $T_k = s_{P_k}$, $k = 1, \dots, 200$, than for $T_{2i-2} = 2 > s_{P_{2i-2}}$ (protracted visit), $T_k = s_{P_k}$, $k \neq 2i - 2$ for any i = 2, ..., 100. Nevertheless, for realistic parameter choices, it will seldom really pay off to protract a visit. To facilitate the determination of the extended visit times, we therefore assume that the side-constraint of most rarely exceeding the available visit times is satisfied without any slack. Translated to problem (I), we assume that the constraint (7.10) is satisfied without any slack. Thus, determining the extended visit times amounts to solving a set of linear equations; cf. Lemma 7.4.1.

Lemma 7.4.1

The set of linear equations

$$T_{k} = \rho_{P_{k}}(1 + \epsilon_{P_{k}}) \sum_{l=1}^{m} h_{kl} T_{l} + (1 + \delta_{P_{k}}) s_{P_{k}}, \qquad k = 1, \dots, m,$$
(7.31)

has a unique solution; this solution is non-negative.

Proof

See Appendix 7.D.

Remark 7.4.2

Note that summing (7.31) with respect to $\{k : P_k = i\}$ yields for the mean total available visit time at Q_i during a cycle $\sum_{\{k:P_k=i\}} T_k - m_i s_i = \rho_i (1+\epsilon_i)C + m_i \delta_i s_i$, as $\sum_{\{k:P_k=i\}} h_{kl} = 1, i = 1, \ldots, n, l = 1, \ldots, m$. The mean total available visit time may be viewed as consisting of (i) $\rho_i C$, the time needed to satisfy the stability condition and (ii) $\epsilon_i \rho_i C + m_i \delta_i s_i$, the extra time above $\rho_i C$ needed to

stability condition and (ii) $\epsilon_i \rho_i C + m_i \delta_i s_i$, the extra time above $\rho_i C$ needed to satisfy the side-constraint of most rarely exceeding the available visit times.

7.4 Constructing an efficient ftp scheme II

We finally summarize the method for constructing an ftp scheme as follows.

1. Determination of the visit numbers.

Calculate the desirable visit frequencies

$$f_{i}^{*} = \frac{\sqrt{\frac{c_{i}\lambda_{i}(1+\rho_{i})}{(1+\delta_{i})s_{i}}}}{\sum_{j=1}^{n}\sqrt{\frac{c_{j}\lambda_{j}(1+\rho_{j})}{(1+\delta_{j})s_{j}}}}, \qquad i = 1, \dots, n.$$
(7.32)

Determine the total number of visits m^* . One may choose m^* with $m^* = \sum_{i=1}^n [m^* f_i^*], [m^* f_i^*] \ge 1$, e.g. such that

$$\frac{|f_i^* - \frac{[m^*f_i^*]}{m^*}|}{f_i^*} \le \eta_i, \qquad i = 1, \dots, n,$$
(7.33)

with [x] denoting the nearest integer to x, and η_i prespecified bounds for the relative deviation of the true visit frequencies from the desirable visit frequencies.

Alternatively, one may choose m^* such that

$$C^* = \frac{m^* \sum_{i=1}^n \sqrt{c_i \lambda_i (1+\rho_i)(1+\delta_i) s_i}}{\left(1-\rho - \sum_{i=1}^n \epsilon_i \rho_i\right) \sum_{i=1}^n \sqrt{\frac{c_i \lambda_i (1+\rho_i)}{(1+\delta_i) s_i}}} \approx C_0,$$

with C_0 some desirable value for the cycle time. Take $m_i^* = [m^* f_i^*], i = 1, ..., n$.

2. Determination of the visit order.

Construct a polling table P^* , using e.g. one of the methods described in the appendices.

Calculate

$$D_i^* = \rho_i (1+\epsilon_i) \sqrt{\frac{(1+\delta_i)s_i}{c_i\lambda_i(1+\rho_i)}} \frac{\sum_{j=1}^n \sqrt{c_j\lambda_j(1+\rho_j)(1+\delta_j)s_j}}{1-\rho - \sum_{j=1}^n \epsilon_j\rho_j} + (1+\delta_i)s_i,$$

as indication for the extended visit times at Q_i .

3. Determination of the extended visit times. Solve the set of linear equations

$$T_k^* = \rho_{P_k^*} (1 + \epsilon_{P_k^*}) \sum_{l=1}^{m^*} h_{kl}^* T_l^* + (1 + \delta_{P_k^*}) s_{P_k^*}, \qquad k = 1, \dots, m^*.$$
(7.34)

7.5 NUMERICAL RESULTS

In the previous section, we developed a method for constructing an efficient ftp scheme. In this section, we give an overview of the numerical results that we gathered to validate the method. The numerical results were obtained from a simulation tool developed at PTT Research, Groningen, cf. Harink, Cramer, & Huitema [117]. Due to the complexity of the problem, it seems not possible to identify the true optimal ftp scheme, neither in an analytic manner nor by an exhaustive search. Instead, we therefore compare the proposed ftp scheme with some other natural, but less sophisticated, ftp schemes. In addition, we compare the proposed ftp scheme with some 'neighboring' ftp schemes, partly to test the method, partly as a possible first step in improving the method itself.

Throughout the section, the service times and the switch-over times are assumed to be constant, but possibly varying from queue to queue. The interarrival times are assumed to be almost constant, but also possibly varying from queue to queue; at Q_i the interarrival times are distributed as $1/\lambda_i + \mathbf{N}_i$, with \mathbf{N}_i normally distributed with mean 0 and standard deviation $1/(10\lambda_i)$. These assumptions were made in the light of the actual telecommunication application that motivated the present study, cf. Section 7.1. The cost coefficients are assumed to be $c_i = 1/\lambda$, i.e., the goal is minimizing the overall mean waiting time. For the reason mentioned above formula (7.6), we replace the side-constraint (7.6) by $T_k \ge \rho_{P_k}(1 + \epsilon_{P_k})SC_k + (1 + \delta_{P_k})S_{P_k} + \beta_{P_k}$. Consequently, in comparison with Sections 7.3 and 7.4, s_i is everywhere replaced by $s_i + \beta_i$. We consider the following eight models.

- I. $n = 4; \lambda_i = 0.75; \beta_i = 0.25; \rho_i = 0.1875; s_i = 0.25; i = 1, ..., n.$
- II. n = 4; $\lambda_i = 0.375$; $\beta_i = 0.5$; $\rho_i = 0.1875$; $s_i = 0.25$; i = 1, ..., n.
- III. $n = 2; \lambda_1 = \lambda_2 = 0.75; \beta_1 = \beta_2 = 0.5; \rho_1 = \rho_2 = 0.375;$
 - $s_1 = 0.05; s_2 = 0.45.$
- IV. $n = 2; \lambda_1 = \lambda_2 = 0.75; \beta_1 = 0.1; \beta_2 = 0.9; \rho_1 = 0.075; \rho_2 = 0.675; s_1 = s_2 = 0.25.$
- V. $n = 2; \lambda_1 = 0.15; \lambda_2 = 1.35; \beta_1 = \beta_2 = 0.5; \rho_1 = 0.075; \rho_2 = 0.675; s_1 = s_2 = 0.25.$
- VI. n = 4; $\lambda_1 = \lambda_2 = \lambda_3 = \lambda_4 = 0.375$; $\beta_1 = \beta_2 = 0.1$; $\beta_3 = \beta_4 = 0.9$;
- $\begin{array}{l} \rho_1=\rho_2=0.0375;\,\rho_3=\rho_4=0.3375;\,s_1=s_3=0.05;\,s_2=s_4=0.45.\\ \text{VII.} \quad n=4;\,\lambda_1=\lambda_2=0.075;\,\lambda_3=\lambda_4=0.675;\,\beta_1=\beta_2=\beta_3=\beta_4=0.5; \end{array}$
- $\rho_1 = \rho_2 = 0.0375; \rho_3 = \rho_4 = 0.3375; s_1 = s_3 = 0.05; s_2 = s_4 = 0.45.$
- VIII. n = 4; $\lambda_1 = \lambda_2 = 0.075$; $\lambda_3 = \lambda_4 = 0.675$; $\beta_1 = \beta_3 = 0.1$; $\beta_2 = \beta_4 = 0.9$; $\rho_1 = 0.0075$; $\rho_2 = \rho_3 = 0.0675$; $\rho_4 = 0.6075$; $s_1 = s_2 = s_3 = s_4 = 0.25$.

Note that for each model $\rho = 0.75$. We have further taken for each model $\delta_i = 0, \eta_i = 0.05, \epsilon_i = 0.01, i = 1, \dots, n$.

The first two models are completely symmetric, so that the optimal ftp scheme should be of the form $m_i = 1, i = 1, ..., n, T_k = T, k = 1, ..., m$, for some

7.5 Numerical results

unknown constant T. Table 7.1 contains the mean waiting time for T = 2.062in model I, and for T = 3.092 in model II, as obtained from the developed method, as well as for various larger values of T. Smaller values of T cannot be chosen in view of the side-constraint $T_k \ge \rho_{P_k}(1+\epsilon_{P_k})SC_k+(1+\delta_{P_k})s_{P_k}+\beta_{P_k}$. The 95% confidence interval for the mean waiting time approximately equals the value listed in the table +/-0.5%. Note that the mean waiting time is indeed minimal for T = 2.062 in model I, and for T = 3.092 in model II.

	Ι	, in the second se	II
T = 2.062	4.777	T = 3.092	7.111
T = 2.100	4.866	T = 3.200	7.362
T = 2.400	5.577	T = 3.500	8.070
T = 2.700	6.294	T = 4.000	9.262
T = 3.000	7.003	T = 4.500	10.45

TABLE 7.1. The mean waiting times for models I and II.

The remaining six models are asymmetric, so that the influence of the various system parameters on the optimal ftp scheme can be investigated. By means of the models III, IV, and V, we can examine the effect of variations in the arrival rate, the service times, and the switch-over times. On the basis of the models VI, VII, and VIII, we can investigate the effect of pairwise variations. For these six models, we consider the following ftp schemes. The first ftp scheme (FTP 1) is the original ftp scheme, constructed by the method described in Section 7.4. In the second ftp scheme (FTP 2), the visit numbers are equal for all queues, i.e., $m_i = 1, i = 1, ..., n$ (strictly cyclic polling, which is treated as a special case in Section 7.4). The visit times are determined according to the third step of the method described in Section 7.4. In the third ftp scheme (FTP 3), the available visit times are equal for all visits, i.e., $U_k = U$, $k = 1, \ldots, m$, while the visit numbers are equal to $m_i = M \rho_i (1 + \epsilon_i)$. The extended visit times, however, may still differ, due to the switch-dependent switch-over times: $T_k = U + s_{P_k}$. The constant U should be chosen such that $D_i \ge \rho_i(1+\epsilon_i)C/m_i + s_i + \beta_i$. Some straightforward calculations show that this

implies
$$U \ge \left(\sum_{j=1}^{n} \rho_j (1+\epsilon_j) s_j + \beta_i\right) / \left(1 - \rho - \sum_{j=1}^{n} \epsilon_j \rho_j\right)$$
. The constant U

is chosen equal to the maximum over i = 1, ..., n of the right-hand side of this expression. The visit order is determined according to the second step of the method described in Section 7.4. The fourth ftp scheme (FTP 4) is similar to FTP 1, except that the polling table is constructed by the procedure based on extremal splittings instead of the Golden Ratio procedure.

In addition, we consider some 'neighboring' ftp schemes of FTP 1, in which the number of visits to one of the queues is either incremented or decremented by 1; in the ftp scheme indicated by FTP 'mi', the number of visits to Q_i is decremented by 1; in the ftp scheme denoted by FTP 'pi', the number of visits to Q_i is incremented by 1.

To be able to judge the numerical results correctly, we first compare in Table 7.2 the desired visit frequencies f_i for FTP 1, cf. (7.32), with the realized visit frequencies $r_i = m_i/m$ for the models III through VIII.

	III	IV	V			III	IV	V
f_1	0.5679	0.5922	0.2108		r_1	0.5625	0.5938	0.2188
f_2	0.4321	0.4078	0.7892		r_2	0.4375	0.4063	0.7813
				•		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
	VI	VII	VIII			VI	VII	VIII
f_1	0.4252	0.1289	0.1482		r_1	0.4375	0.1250	0.1515
f_2	0.2221	0.0981	0.0842		r_2	0.2188	0.0938	0.0808
f_3	0.1918	0.4390	0.4578		r_3	0.1875	0.4375	0.4545
f_4	0.1609	0.3340	0.3098		r_4	0.1563	0.3438	0.3131

TABLE 7.2. The desired and realized visit frequencies for models III-VIII.

Table 7.3 contains the mean overall waiting time for the models III through VIII. The lines indicated by (7.9) 1 and (7.9) 4 in Table 7.3, give the value of the mean waiting time approximation (7.9) for FTP 1 and FTP 4, respectively. Comparison of these lines with the preceding lines suggests that the approximation (7.9) tends to yield a slight structural overestimation of the mean waiting time, as indeed expected for almost constant interarrival times (in view of the assumption $EPV_k \approx \rho_i EPSC_k$ in Section 7.3). The first line in Table 7.3 gives the value of (7.24) for the various models. As remarked earlier, (7.24) provides a lower bound for the waiting-time approximation (7.9). In other words, the difference between the first line and the lines indicated by (7.9) 1 and (7.9) 4, provides an upper bound on how far FTP 1 and FTP 4 can be off in minimizing the waiting-time approximation. The difference also gives an indication how far FTP 1 and FTP 4 can at most be off in minimizing the *true* waiting time.

	III	IV	V	
(7.24)	4.1756	4.1290	3.7425	
FTP 1	4.1604	4.3018	3.6521	
(7.9) 1	4.3997	4.5379	3.8912	
FTP 2	4.0140	4.0301	4.7513	
FTP 3	4.2301	19.152	4.7315	
FTP 4	4.0719	4.2988	3.5831	
(7.9) 4	4.3101	4.5366	3.8221	
FTP p1	4.1863	4.3428	3.6295	
FTP m1	4.0435	4.2536	3.6778	
FTP p2	4.0405	4.4105	3.6490	
FTP m2	4.2148	4.3880	3.6566	
•	VI	VII	VIII	
---------	--------	--------	--------	--
(7.24)	6.7580	6.1170	5.9773	
FTP 1	7.0327	6.2255	6.4314	
(7.9) 1	7.2695	6.4654	6.6707	
FTP 2	7.1341	7.8523	7.8761	
FTP 3	41.244	9.1655	32.874	
FTP 4	7.2130	6.0879	6.2917	
(7.9) 4	7.4495	6.3243	6.5301	
FTP p1	7.0837	6.2007	6.4459	
FTP m1	6.9831	6.2815	6.4414	
FTP p2	7.0140	6.3226	6.4499	
FTP m2	7.1309	6.4903	6.5360	
FTP p3	7.1098	6.2950	6.4560	
FTP m3	7.0469	6.1906	6.4529	
FTP p4	7.1538	6.1994	6.3943	
FTP m4	7.4608	6.2802	6.4893	

TABLE 7.3. The mean waiting times for models III-VIII.

The number printed in boldface represents the optimum. For all models, FTP 1, FTP 4, and the neighboring schemes of FTP 1 give very similar results; FTP 4 tends to perform slightly better than FTP 1. Apparently, the procedure based on extremal splittings indeed tends to yield a slightly better polling table than the Golden Ratio procedure. FTP 2 is on the average slightly worse, while FTP 3 is generally bad. The low variability of the arrival, service, and switch-over processes appears to result in a relative insensitivity to the right choice of the visit numbers in the neighborhood of the desired visit numbers. Remember that a wrong choice for the visit numbers may still be compensated for in the determination of the visit times.

The fact that in the models III and IV, FTP 2 performs slightly better than FTP 1 and FTP 4, may be explained as follows. As observed in Section 7.4, the visits to the various queues should be spaced as evenly as possible. For FTP 2, the very nature of cyclic polling allows the visits to the various queues to be perfectly evenly spaced, whereas for FTP 1 and FTP 4, the desired visit frequencies in the models III and IV, cf. Table 7.2, do not even allow the visits to be reasonably evenly spaced. In the derivation of the desired visit frequencies, the visits to the queues were however assumed to be perfectly evenly spaced. The fact that nevertheless FTP 2 performs only slightly better than FTP 1 and FTP 4, actually supports the approach used. As Table 7.3 confirms, FTP 2 is likely to outperform FTP 1 and FTP 4 only when the number of queues is small, and the difference in the desired visit numbers not too large.

APPENDICES

7.A Proof of Lemma 7.3.1

Lemma 7.3.1

Problem (I) is NP-hard.

Proof

To prove that problem (I) is NP-hard, we need to show that the decision variant of problem (I) is NP-complete. A decision problem is said to be NP-complete, if (i) it belongs to the class NP, and (ii) every problem in the class NP is (polynomially) reducible to it, cf. Garey & Johnson [105]. For brevity, let us refer to the decision variant of problem (I) as the decision problem TABLE. TABLE reads as follows: given parameters λ_i , β_i , s_i , c_i , δ_i , ϵ_i , i = 1, ..., n, and an arbitrary number r, does problem (I) have a solution which is feasible and for which the value of the objective function is not larger than r?

Obviously, TABLE belongs to the class NP. As the notion of reducibility is transitive, it in fact remains to be shown that *some* known NP-hard problem is (polynomially) reducible to TABLE. Here the problem PARTITION turns out to be an appropriate choice as known NP-hard problem. PARTITION reads as follows: given a set $A = \{a_1, \ldots, a_p\}$ of p integers, does A include a subset B,

such that $\sum_{a_i \in B} a_i = \sum_{a_i \in A \setminus B} a_i = \frac{1}{2} \sum_{i=1}^{p} a_i$?

We now prove that PARTITION is (polynomially) reducible to TABLE. Given an instance a_1, \ldots, a_p for PARTITION, construct an instance λ_i , β_i , s_i , c_i , δ_i , ϵ_i , $i = 1, \ldots, n$, and r for TABLE in the following manner.

$$n := p + 1;$$

$$\beta_{i} := 0, \delta_{i} := 0, \epsilon_{i} := 0, \quad i = 1, \dots, p + 1;$$

$$s_{i} := a_{i}, c_{i}\lambda_{i} := a_{i}, \quad i = 1, \dots, p;$$

$$s_{p+1} := 1, c_{p+1}\lambda_{p+1} := 4;$$

$$r := \frac{1}{2} \left(\sum_{i=1}^{p} a_{i} + 2\right)^{2}.$$

(7.35)

We now need to prove that a_1, \ldots, a_p constitute a 'yes' instance for PARTITION, iff $\lambda_i, \beta_i, s_i, c_i, \delta_i, \epsilon_i, i = 1, \ldots, n$, and r as defined in (7.35), constitute a 'yes' instance for TABLE. We first show that a_1, \ldots, a_p constitute a 'yes' instance for PARTITION, iff there exists a feasible polling scheme (P,T), such that $m_i = 1$, $i = 1, \ldots, p, m_{p+1} = 2, SC_k = \frac{1}{2} \sum_{i=1}^p a_i + 1$ for both k with $P_k = p + 1$. $\{\Rightarrow\} \text{ The set } A = \{a_1, \dots, a_p\} \text{ includes a subset } B, \text{ such that } \sum_{a_i \in B} a_i = \sum_{\substack{a_i \in A \setminus B \\ i = 1}}^{p} a_i \text{. Let us say } B = \{a_{i_1}, \dots, a_{i_q}\}, A \setminus B = \{a_{i_{q+1}}, \dots, a_{i_p}\}.$ Take $P_1 = p + 1, P_{k+1} = i_k$ for $k = 1, \dots, q, P_{q+2} = p + 1, P_{k+2} = i_k$ for $k = q + 1, \dots, p, T_k = s_{P_k}, k = 1, \dots, p + 2$. Then $SC_1 = \sum_{k=q+2}^{p+2} T_k = \sum_{a_i \in A \setminus B} a_i + 1 = \frac{1}{2} \sum_{i=1}^{p} a_i + 1, SC_{q+2} = \sum_{k=1}^{q+1} T_k = \sum_{a_i \in B} a_i + 1 = \frac{1}{2} \sum_{i=1}^{p} a_i + 1.$ $\{ \Leftarrow \} \text{ Let us say } P_{k_1} = p + 1, P_{k_2} = p + 1, k_1 < k_2. \text{ So } SC_{k_1} = \frac{1}{2} \sum_{i=1}^{p} a_i + 1, SC_{k_2} = \frac{1}{2} \sum_{i=1}^{p} a_i + 1.$ $SC_{k_2} = \frac{1}{2} \sum_{i=1}^{p} a_i + 1. \text{ Now, using } (7.4), C = SC_{k_1} + SC_{k_2} = \sum_{i=1}^{p} a_i + 2 = \sum_{k=1}^{p+2} s_{P_k},$ while, using (7.3), $C = \sum_{k=1}^{p+2} T_k.$ Hence, $T_k \ge s_{P_k}, k = 1, \dots, p + 2$, implies $T_k = s_{P_k}, k = 1, \dots, p + 2.$ Take $B = \{a_{P_k} : k_1 < k < k_2\}.$ Then $\sum_{a_i \in B} a_i = \sum_{k=k_1+1}^{k_2-1} T_k = SC_{k_2} - T_{k_1} = \frac{1}{2} \sum_{i=1}^{p} a_i, \sum_{a_i \in A \setminus B} a_i = \sum_{k=k_2+1}^{p+2} T_k + \sum_{k=1}^{k_1-1} T_k = SC_{k_1} - T_{k_2} = \frac{1}{2} \sum_{i=1}^{p} a_i.$ We now show that there exists a feasible polling scheme (P, T), such that

We now show that there exists a feasible polling scheme (P,T), such that $m_i = 1, i = 1, \ldots, p, m_{p+1} = 2, SC_k = \frac{1}{2} \sum_{i=1}^{p} a_i + 1$ for both k with $P_k = p + 1$, iff $\lambda_i, \beta_i, s_i, c_i, \delta_i, \epsilon_i, i = 1, \ldots, n$, and r as defined in (7.35), constitute a 'yes' instance for problem TABLE.

 $\{\Rightarrow\}$ As before, $C = \sum_{i=1}^{p} a_i + 2$. So the value of the objective function for (P, T) is

$$\begin{split} &\frac{1}{2C}\sum_{i=1}^{n}c_{i}\lambda_{i}(1+\rho_{i})\sum_{\{k:P_{k}=i\}}SC_{k}^{2}=\\ &\frac{1}{2\left(\sum_{i=1}^{p}a_{i}+2\right)}\left\{\sum_{i=1}^{p}a_{i}\left(\sum_{i=1}^{p}a_{i}+2\right)^{2}+8\left(\frac{1}{2}\sum_{i=1}^{p}a_{i}+1\right)^{2}\right\}=\\ &\frac{1}{2}\left(\sum_{i=1}^{p}a_{i}+2\right)^{2}=r. \end{split}$$

 $\{\Leftarrow\}$ Problem (I) with λ_i , β_i , s_i , c_i , δ_i , ϵ_i , i = 1, ..., n, and r as defined in (7.35) has a solution which is feasible, and for which the value of the objective function is not larger than r. Using the Lagrangean multiplier technique and the argumentation preceding the formulation of problem (II), it is easily verified that the

value of the objective function is not larger than $\frac{1}{2} \left(\sum_{i=1}^{p} a_i + 2 \right)^2$ only for (P,T) with $m_i = R, i = 1, \ldots, p, m_{p+1} = 2R, SC_k = C/m_{P_k}, C = \sum_{i=1}^{p} m_i a_i + m_{p+1},$ which for R = 1 yields the result that we have in view.

7.B THE GOLDEN RATIO PROCEDURE

Calculate the numbers $g(k) = k\phi^{-1} \mod 1$ with $\phi^{-1} = \frac{1}{2}(\sqrt{5}-1) \approx 0.618034$, the so-called Golden Ratio, $k = 1, \ldots, m$. Let the numbers g(k) with $\sum_{j=1}^{i-1} m_j + 1 \le k \le \sum_{j=1}^{i} m_j$ correspond to the visits to $Q_i, i = 1, \ldots, n$. Put the numbers $g(k), k = 1, \ldots, m$, in increasing order. Let the *l*-th smallest number correspond to the *l*-th position in $P, l = 1, \ldots, m$.

Formally, $P_{\pi(k)} := i$ for k with $\sum_{j=1}^{i-1} m_j + 1 \le k \le \sum_{j=1}^{i} m_j$, π representing the permutation such that $g(k) \le g(l) \iff \pi(k) \le \pi(l), k, l = 1, \dots, m$.

7.C A PROCEDURE BASED ON EXTREMAL SPLITTINGS

Before we give a detailed description, we first sketch the main motivation behind the procedure. Recall that we want to construct a polling table P that approximately minimizes (7.9) for fixed m_i and fixed $T_k = D_{P_k}^*$.

On the one hand, as seen from the argumentation preceding the formulation of problem (II), if the visits to the various queues are perfectly evenly spaced, then the polling table is optimal. In fact, substituting $SC_k = \sum_{i=1}^n m_i D_i^*/m_{P_k}$ into (7.9) yields a lower bound for the value of (7.9) for the optimal table. On the other hand, if the visits to the various queues are perfectly evenly spaced, then the polling table obviously satisfies the following property: between any two consecutive visits to Q_i there is exactly one visit to every Q_j with $m_i = m_j$, $i \neq j$. For brevity, let us refer to this property as property (E). The reverse statement does not hold. Even if the polling table satisfies property (E), then for arbitrary parameter choices, the subcycle times may still be arbitrarily far from equal, and the value of (7.9) may still be arbitrarily far from minimal. Nevertheless, if the polling table satisfies property (E), then for reasonable parameter choices the visits are likely to be reasonably evenly spaced, and the polling table is likely to be reasonably good. We therefore use property (E) as the main guideline in constructing a polling table.

7.C A procedure based on extremal splittings

Let $M = \{m_i : i \in \{1, ..., n\}\}$ be the set of visit numbers that occur. Let $I^{(r)} = \{i \in \{1, ..., n\} : m_i = r\}$ be the set of the queues with common visit number r for $r \in M$.

Suppose that one has already constructed a subtable P of size |P| for all the visits to the queues $i \in I^{(r_1)}, \ldots, I^{(r_q)}$ with common visit numbers $r_1, \ldots, r_q \in M$. Initially, $|P| = 0, \{r_1, \ldots, r_q\} = \emptyset$.

Initially, |P| = 0, $\{r_1, \ldots, r_q\} = \emptyset$. If $M \setminus \{r_1, \ldots, r_q\} \neq \emptyset$, then select a visit number r from $M \setminus \{r_1, \ldots, r_q\}$. Construct a subtable $Q^{(r)}$ of size $|Q^{(r)}| = r \times |I^{(r)}|$ for all the visits to the queues $i \in I^{(r)}$, by just concatenating r times an arbitrary sequence of the queues $i \in I^{(r)}$. Formally, with $i_1, \ldots, i_{|I^{(r)}|}$ an arbitrary sequence of the queues $i \in I^{(r)}$, $Q_{j+(k-1) \times |I^{(r)}|}^{(r)} := i_j$ for $j = 1, \ldots, |I^{(r)}|$, $k = 1, \ldots, r$. Obviously, $Q^{(r)}$ satisfies property (E).

Construct subsequently a subtable $P^{(r)}$ of size $|P^{(r)}| = |P| + |Q^{(r)}|$, by inserting the visits from the subtable $Q^{(r)}$ in the subtable P in the following manner. Put the visits from $Q^{(r)}$ at positions in $P^{(r)}$ as evenly spaced as possible, i.e., put the visit at the k-th position in $Q^{(r)}$ at the (k+d(k))-th position in $P^{(r)}$, $k = 1, \ldots, |Q^{(r)}|$. Here $d(k) = \left[(k-1) \times \frac{|P|}{|Q^{(r)}|}\right]$, $k = 1, \ldots, |Q^{(r)}|$,

with
$$[x]$$
 denoting the nearest integer to x . Put the visits from P at the re-
maining positions in $P^{(r)}$, i.e., put the visit at the $\chi(l+l_0)$ -th position in P
at the *l*-th position in $P^{(r)}$ that is not yet occupied by a visit from $Q^{(r)}$,
 $l = 1, \ldots, |P|$. Here $\chi(k) = ((k-1) \mod |P|) + 1$. Choose l_0 from
 $\{1, \ldots, |P|\}$ such that the objective function (7.9) properly applied to $P^{(r)}$

{1,..., | P |} such that the objective function (7.9) properly applied to $P^{(r)}$ is minimal. Formally, $P_{k+d(k)}^{(r)} := Q_k^{(r)}$, $k = 1, ..., |Q^{(r)}|$, $P_{k+l}^{(r)} := P_{\chi(l+l_0)}$, $k = 1, ..., |Q^{(r)}|$, l = d(k) + 1, ..., d(k+1). Thus, in $P^{(r)}$ the number of visits from P between the k-th and (k+1)-th visit from $Q^{(r)}$ equals $d(k+1) - d(k) = \left[k \times \frac{|P|}{|Q^{(r)}|}\right] - \left[(k-1) \times \frac{|P|}{|Q^{(r)}|}\right]$, $k = 1, ..., |Q^{(r)}|$. This distancing is closely related to extremal splittings of point processes, cf. Ha-

jek [116]. Note that the internal visit order from P and $Q^{(r)}$ is maintained. Hence, by induction, $P^{(r)}$ satisfies property (E).

Repeat with P replaced by $P^{(r)}$. Finally $|P| = m, \{r_1, \ldots, r_q\} = M$.

Remark 7.C.1 The above-described method appears to be similar to an algorithm presented in Arian & Levy [12]. Simulation results in [12] suggest that the visit numbers r can best be selected from $M \setminus \{r_1, \ldots, r_q\}$ in descending order.

7.D Proof of Lemma 7.4.1

Lemma 7.4.1

The set of linear equations

$$T_k = \rho_{P_k} (1 + \epsilon_{P_k}) \sum_{l=1}^m h_{kl} T_l + (1 + \delta_{P_k}) s_{P_k}, \qquad k = 1, \dots, m,$$
(7.36)

has a unique solution; this solution is non-negative.

Proof

Define the matrix A by $A_{kl} := \rho_{P_k}(1 + \epsilon_{P_k})h_{kl}$, $k, l = 1, \ldots, m$, and the vector b by $b_k := (1 + \delta_{P_k})s_{P_k}$, $k = 1, \ldots, m$. Then the set of linear equations (7.36) may be rewritten as (I - A)T = b.

Since A is a non-negative irreducible matrix, A has a real eigenvalue μ , which is strictly maximal in absolute value, cf. Seneta [163] pp. 3-4. For μ holds $\min_{1 \le l \le m} \sum_{k=1}^{m} A_{kl} \le \mu \le \max_{1 \le l \le m} \sum_{k=1}^{m} A_{kl}$, cf. [163] p. 8. $\sum_{k=1}^{m} A_{kl} = \sum_{i=1}^{n} \sum_{\{k:P_k=i\}} \rho_{P_k}$ $(1+\epsilon_{P_k}) h_{kl} = \rho + \sum_{i=1}^{n} \epsilon_i \rho_i$, as $\sum_{\{k:P_k=i\}} h_{kl} = 1, l = 1, \dots, m$, so $\mu = \rho + \sum_{i=1}^{n} \epsilon_i$ ρ_i . Since b is a non-negative vector, $\rho + \sum_{i=1}^{n} \epsilon_i \rho_i < 1$ implies that (I-A)T = bhas a unique solution $T = (I-A)^{-1}b \ge 0$, cf. [163] p. 30.

CHAPTER 8

OPTIMAL ALLOCATION OF CUSTOMER TYPES TO SERVERS

8.1 INTRODUCTION

In the present chapter, we consider a model consisting of several parallel servers, which process jobs generated at several distinct sources. Such a model may arise quite naturally in a situation where a pool of resources is available for performing various kinds of activities. Examples may be found in distributed computer systems, flexible manufacturing systems, and telecommunication networks. Another example may be found in a situation where a pool of repair crews is available to perform maintenance activities at various installations.

In such a situation, there is usually some freedom of decision as to which server is to process which job at what time. So there is typically a need for a scheduling strategy, i.e., a collection of decision instructions for scheduling the jobs. At a global level, decisions need to be made about which server is to process which job. Subsequently, at a local level, decisions need to be made about the order of service. In this chapter, we mainly focus on the global scheduling problem; we hardly touch on the local scheduling problem. Locally, the order of service is assumed not to discriminate between the sources from which the jobs originated.

The main function of a global scheduling strategy is load sharing; a strategy should make the servers cooperate in sharing the load of the system so as to optimize the system performance. Load sharing is also frequently referred to as load balancing. The term load balancing arises from the intuition that to optimize the system performance, the load should be balanced among the servers. In this chapter, however, we find that the load, in the sense of traffic intensity, should *not* be completely balanced in general.

Wang & Morris [186] give a comprehensive survey of the overwhelming variety of approaches to load sharing in the literature. They identify some fundamentally distinguishing features of load sharing strategies. A first distinction refers to the side that takes the initiative in scheduling the jobs. In source-initiative policies, the sources decide to which server to route a job. In server-initiative policies, the servers decide from which source to get a job. Consequently in source-initiative policies decisions are typically made at arrival epochs, whereas in server-initiative policies decisions are usually made at service completion epochs (possibly at arrival epochs when servers are idle). Moreover, in sourceinitiative policies queues tend to form at the servers, whereas in server-initiative policies queues tend to form at the sources. Of course, there are also policies conceivable in which both the sources and the servers participate in allocating the jobs.

A second distinction refers to the amount of information that is used in allocating the jobs. In purely static policies, only information is used about the basic characteristics of the system, like the traffic intensities. In dynamic policies, also information is used about the actual state of the system, like the queue lengths. Evidently, the performance of the system may, in principle, improve substantially by using such information in allocating the jobs. However, gathering such information and implementing a sophisticated dynamic allocation strategy may involve a considerable communication overhead and complicate the operation of the system significantly. Therefore, dynamic policies are not necessarily preferable to static policies.

In this chapter, we assume that customers are allocated to the servers in a probabilistic manner; upon arrival customers are sent to one of the servers according to a matrix of routing probabilities. Such a load sharing strategy is commonly referred to as random splitting. In the taxonomy of Wang & Morris [186], random splitting belongs to the class of static source-initiative load sharing strategies. We are interested in the problem of finding a random splitting that minimizes a weighted sum of the mean waiting times.

The novelty of the model lies in the combination of heterogeneous servers (i.e. different service rates), heterogeneous sources (i.e. different service times), and a fairly general cost function. Buzacott & Shanthikumar [64] consider a version of the problem with homogeneous servers and the overall mean waiting time as performance measure, which we will discuss later on in greater technical detail. Buzen & Chen [65] consider a variant of the problem with a single source and the overall mean sojourn time as performance criterion. A natural approach to deal with heterogeneous servers and heterogeneous sources might be to aggregate the different sources into a single source, and then use the results of Buzen & Chen. Each server would thus handle a traffic mix of the same, heterogeneous, composition, but of possibly different intensity, depending on the processing rate of the servers. In this chapter, however, we find that each server should handle a traffic mix as homogeneous as possible. For homogeneous sources, Boxma & Combé [41] show that 'pattern' allocation outperforms probabilistic allocation, but that the optimal routing probabilities of Buzen & Chen provide a reasonable indication for the optimal occurrence fractions in 'pattern' allocation. It is likely that similar observations hold for heterogeneous sources.

8.2 Model description

Tantawi & Towsley [178], [179] and De Souza e Silva & Gerla [81] consider optimal load balancing models of distributed computer systems consisting of a number of heterogeneous host computers connected by a communication network. A job may be either processed at the host to which it arrives or transferred to another host. In the latter case, a transferred job incurs a communication delay in addition to the queueing delay at the host on which it is processed. The assumptions in [178], [179], and [81] on the service requirements of jobs are however somewhat restrictive.

In some situations, it may be desirable that job classes are not split among different servers. In flexible manufacturing systems e.g., such a splitting may be undesirable because handling a particular job class usually requires special expensive tools. In case job classes are not split among different servers, a load sharing strategy is commonly referred to as source partitioning. In view of the practical relevance, we are specifically interested in finding an optimal source partitioning. For general partitioning problems, Anily & Federgruen [10] identify analytical properties of the cost function under which an optimal solution has a simple structure, thus allowing for a simple solution method. They mention the problem of finding an optimal source partitioning as an example for which the mean number of waiting customers as cost function (which by Little's law is nothing but a specific weighted sum of the mean waiting times) does *not* have such analytical properties.

The remainder of the chapter is organized as follows. In Section 8.2, we present a detailed model description. We then consider the problem of finding an optimal random splitting. In Section 8.3, we expose the structure of an optimal allocation, and in Section 8.4 we describe for some special cases in detail how the structure may be exploited in actually determining an optimal allocation. In Section 8.5, we consider the problem of finding an optimal source partitioning. We show the problem to be NP-hard, and indicate how the structure of an optimal non-deterministic allocation may be used as a heuristic guideline in searching for an optimal deterministic allocation. In Section 8.6, we conclude with some remarks and suggestions for further research.

8.2 MODEL DESCRIPTION

The model under consideration consists of n customer types, attended by m parallel non-identical servers. Customers arrive according to Poisson processes. The arrival rate of type-*i* customers is λ_i , i = 1, ..., n. The total arrival rate is $\lambda := \sum_{i=1}^{n} \lambda_i$. Upon arrival customers are routed to one of the servers. Type-*i* customers are routed to server *j* with probability x_{ij} , i = 1, ..., n, j = 1, ..., m. The matrix $x = (x_{ij})$ of routing probabilities will be referred to as the (probabilistic) allocation of the customer types to the servers. In case the matrix of routing probabilities is 0-1, the (probabilistic) allocation will be referred to as deterministic. When processed at server *j*, type-*i* customers require ser-

vice times having distribution $F_{ij}(t) = F_i(\mu_j t)$, i.e., type-*i* customers require amounts of service having distribution $F_i(t)$, while server *j* has processing rate μ_j . In other words, the servers may have different characteristics, the customer types may have different characteristics, but servers cannot 'specialize' in some of the customer types. Denote by β_i and $\beta_i^{(2)}$ the first and second moment of $F_i(t)$, i = 1, ..., n. We assume $F_i(0) < 1$, so $\beta_i > 0$, $\beta_i^{(2)} > 0$, i = 1, ..., n. Define the traffic intensity associated with type-*i* customers as $\rho_i := \lambda_i \beta_i$, i = 1, ..., n. The total traffic intensity is $\rho := \sum_{i=1}^n \rho_i$. The order of service is assumed not to discriminate between the various customer types. Furthermore, all arrival, service, and routing processes are assumed to be mutually independent.

The queues that form at the servers are ordinary M/G/1 queues. The arrival rate at server j is $\sum_{i=1}^{n} x_{ij}\lambda_i$. Customers that are routed to server j require service times having distribution $\sum_{i=1}^{n} x_{ij}\lambda_iF_i(\mu_j t)/\sum_{i=1}^{n} x_{ij}\lambda_i$ with first moment $\left[\sum_{i=1}^{n} x_{ij}\lambda_i\beta_i\right]/\left[\mu_j\sum_{i=1}^{n} x_{ij}\lambda_i\right]$ and second moment $\left[\sum_{i=1}^{n} x_{ij}\lambda_i\beta_i^{(2)}\right]/\left[\mu_j\sum_{i=1}^{n} x_{ij}\lambda_i\right]$, $j = 1, \ldots, m$. Define the traffic intensity at server j as $\sum_{i=1}^{n} x_{ij}\lambda_i\beta_i$, $j = 1, \ldots, m$. Necessary and sufficient ergodicity conditions are

$$\sum_{i=1}^{n} x_{ij} \lambda_i \beta_i < \mu_j, \qquad j = 1, \dots, m.$$
(8.1)

Denote by $\mu := \sum_{j=1}^{m} \mu_j$ the total processing rate of the servers. Summing (8.1) with respect to j = 1, ..., m yields $\rho < \mu$, as $\sum_{j=1}^{m} x_{ij} = 1, i = 1, ..., n$.

Throughout the chapter, $\rho < \mu$ is assumed to hold.

We are interested in the problem of finding an allocation that minimizes a weighted sum of the mean waiting times. Therefore, we first derive a formula that describes the mean waiting times as function of the allocation matrix $x = (x_{ij})$. Denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer, $i = 1, \ldots, n$, i.e., the time from its arrival to the start of its service. Denote by \mathbf{V}_j the waiting time of an arbitrary customer that is routed to server j, $j = 1, \ldots, m$.

As the order of service is assumed not to discriminate between the various customer types,

$$\mathbf{E}\mathbf{W}_{i} = \sum_{j=1}^{m} x_{ij} \mathbf{E}\mathbf{V}_{j}, \qquad i = 1, \dots, n.$$
(8.2)

8.3 Finding an optimal random splitting

As the queues that form at the servers are ordinary M/G/1 queues,

$$\mathbf{E}\mathbf{V}_{j} = \frac{\sum_{i=1}^{n} \lambda_{i} \beta_{i}^{(2)} x_{ij}}{2\mu_{j} \left(\mu_{j} - \sum_{i=1}^{n} \lambda_{i} \beta_{i} x_{ij}\right)}, \qquad j = 1, \dots, m.$$
(8.3)

Let c_i represent the waiting cost per unit of time of a type-*i* customer, $i = 1, \ldots, n$. We assume $c_i > 0, i = 1, \ldots, n$. The mean total waiting cost per unit of time amounts to $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i$. Using (8.2) and (8.3),

$$\sum_{i=1}^{n} c_i \lambda_i \mathbf{E} \mathbf{W}_i = \sum_{j=1}^{m} \frac{\left(\sum_{i=1}^{n} \lambda_i c_i x_{ij}\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}\right)}{2\mu_j \left(\mu_j - \sum_{i=1}^{n} \lambda_i \beta_i x_{ij}\right)}.$$
(8.4)

8.3 FINDING AN OPTIMAL RANDOM SPLITTING

In this section, we consider the problem of finding an optimal random splitting, i.e., a probabilistic allocation of the customer types to the servers that minimizes the mean total waiting cost per unit of time. Using (8.1) and (8.4), we formulate the problem as follows.

Problem (I).

minimize

$$e \qquad f(x) = \sum_{j=1}^{m} \frac{\left(\sum_{i=1}^{n} \lambda_i c_i x_{ij}\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}\right)}{\mu_j \left(\mu_j - \sum_{i=1}^{n} \lambda_i \beta_i x_{ij}\right)} \tag{8.5}$$

subject to

n

$$\sum_{i=1}^{m} \lambda_i \beta_i x_{ij} < \mu_j, \qquad j = 1, \dots, m;$$
$$\sum_{j=1}^{m} x_{ij} = 1, \qquad i = 1, \dots, n;$$
$$x_{ij} \ge 0, \qquad i = 1, \dots, n, \ j = 1, \dots, m.$$

Problem (I) is a non-linear programming problem. It is easily verified that the objective function $f(\cdot)$ is not convex, so that it is not guaranteed that there exists a unique Kuhn-Tucker point. Moreover, finding a Kuhn-Tucker point is not quite straightforward.

All in all there is not an obvious way of solving problem (I). Nevertheless, if one were merely interested in computing an optimal allocation for some given parameters, then one could in principle proceed to solving problem (I)

by standard non-convex programming techniques. That is however not what we are interested in here. What we are primarily interested in, is obtaining some insight into the structural properties of an optimal allocation. We will show that an optimal solution of problem (I) indeed has a very characteristic structure. As secondary motivation, the structural properties not only provide some insight, but are also very useful in computing an optimal allocation. In Section 8.4, we will describe how in cases with identical servers where all the customer types are in a sense ordered, the structure may be exploited in a very simple manner in actually computing an optimal solution of problem (I). In these cases, there exists a unique Kuhn-Tucker point that has the structure of an optimal solution. So it is guaranteed that this Kuhn-Tucker point is the optimal solution. Moreover, finding this Kuhn-Tucker point is comparatively straightforward in these cases. In cases where not all the customer types are ordered, the structure may still be exploited in determining an optimal solution of problem (I), but not in such a simple manner. In Section 8.5, we indicate how the knowledge of the structure of an optimal solution of problem (I) may also be used as a guideline in heuristically solving the NP-hard integer version of problem (I) (integer x_{ij} 's).

We now expose the structure of an optimal allocation x^* . We first introduce some notation. For a given allocation x, define $K_j(x) := \{i \mid x_{ij} > 0\}$ to be the index set of the customer types (partially) allocated to server j. Define $A_j(x) := \left\{ \left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i} \right) \mid i \in K_j(x) \right\}$ to be the set of $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i} \right)$ -values corresponding to the customer types allocated to server j. Denote $P_j(x) :=$ int(conv($A_j(x)$)), with int(conv(\cdot)) denoting the interior of the convex hull. The set $P_j(x)$ may be interpreted as the global range of $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i} \right)$ -values corresponding to the customer types allocated to server j. Denote

$$B_j(x) := \left[\sum_{i=1}^n \lambda_i \beta_i^{(2)} x_{ij}\right] / \left[\mu_j \left(\mu_j - \sum_{i=1}^n \lambda_i \beta_i x_{ij}\right)\right],$$
$$C_j(x) := \left[\sum_{i=1}^n \lambda_i c_i x_{ij}\right] / \left[\mu_j \left(\mu_j - \sum_{i=1}^n \lambda_i \beta_i x_{ij}\right)\right].$$

The numbers $B_j(x)$ and $C_j(x)$ may be interpreted as measures for the $\beta_i^{(2)}/\beta_i$ -weight' and the c_i/β_i -weight' associated with the customer types allocated to server j.

We now expose the structure of an optimal allocation x^* in terms of the corresponding sets $P_j(x^*)$. Intuitively, it is to be expected that an optimal allocation will satisfy one of the following two (in general mutually exclusive) 'extremal' properties.

8.3 Finding an optimal random splitting

(i). Each server handles a traffic mix of the same composition (e.g. $x_{ij}^* = \mu_j/\mu$, i = 1, ..., n, j = 1, ..., m), so that the traffic mix at each server is completely heterogeneous;

(ii). Each server handles a traffic mix as homogeneous as possible, so that different servers deal with traffic mixes of a completely different composition. The next Lemma says that an optimal allocation in fact satisfies the second property (so the first one is *not* satisfied in general).

Lemma 8.3.1 $P_{j'}(x^*) \cap P_{j''}(x^*) = \emptyset \text{ for } j' \neq j''.$ In other words, if $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \in P_j(x^*)$, then $x_{ij}^* = 1.$

Proof

See Appendix 8.A.

Lemma 8.3.1 suggests that the customer types should be clustered according to the corresponding $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right)$ -values. As a consequence, different servers will deal with different traffic mixes. The next Lemma says that different traffic mixes may however not involve an arbitrarily different ' $\beta_i^{(2)}/\beta_i$ -weight' and ' c_i/β_i -weight'.

Lemma 8.3.2 If $B_{j'}(x^*) \ge B_{j''}(x^*), C_{j'}(x^*) \ge C_{j''}(x^*),$ then $\mu_{j'}B_{j'}(x^*)C_{j'}(x^*) \le \mu_{j''}B_{j''}(x^*)C_{j''}(x^*).$

Proof

See Appendix 8.B.

Lemma 8.3.2 states that if one server carries both larger $B_j(x^*)$ and $C_j(x^*)$ than another, then it cannot carry larger $\mu_j B_j(x^*)C_j(x^*)$ as well.

In the remainder of this section, as well as in the next section, we consider the case of identical servers, i.e., $\mu_j = \mu/m$, $j = 1, \ldots, m$. Lemma 8.3.2 then states that it is no longer possible that one server carries both larger $B_j(x^*)$ and $C_j(x^*)$ than another.

Corollary 8.3.1 $B_{j'}(x^*) \ge B_{j''}(x^*) \iff C_{j'}(x^*) \le C_{j''}(x^*).$

We now assume that the servers are indexed such that $B_{j'}(x^*) \ge B_{j''}(x^*)$, $C_{j'}(x^*) \le C_{j''}(x^*)$ for j' < j''.

Lemma 8.3.3 Assume $x_{i'j'}^* > 0, x_{i''j''}^* > 0.$ If $\frac{c_{i'}}{\beta_{i'}} \leq \frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}} \geq \frac{\beta_{i''}^{(2)}}{\beta_{i''}}, \left(\frac{c_{i'}}{\beta_{i'}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}}\right) \neq \left(\frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i''}^{(2)}}{\beta_{i''}}\right), \text{ then } j' \leq j''.$

Proof

See Appendix 8.C.

Lemma 8.3.3 states that expensive, calm (cheap, wild) customer types with large (small) c_i/β_i and small (large) $\beta_i^{(2)}/\beta_i$ should be sent to servers with small (large) $B_j(x^*)$ and large (small) $C_j(x^*)$, thus experiencing a small (large) waiting time. (Note that $B_j(x)$ is in fact twice the mean waiting time at server j for allocation x.) Lemma 8.3.3 does however not indicate what should be done with expensive but wild (cheap but calm) customer types with large (small) c_i/β_i and large (small) $\beta_i^{(2)}/\beta_i$. Indeed, it depends not only on their own individual c_i/β_i and $\beta_i^{(2)}/\beta_i$, but also on some other less easily quantifiable factors, whether they should be sent to servers with small $B_j(x^*)$ and large $C_j(x^*)$, or with large $B_j(x^*)$ and small $C_j(x^*)$.

Lemma 8.3.3 allows us to strengthen the statements on the clustering of the customer types in Lemma 8.3.1. We first introduce some additional notation. Define

$$\begin{aligned} Q_j(x) &:= \bigcup_{i \in K_j(x)} \left\{ (y, z) : y \le \frac{c_i}{\beta_i}, z \ge \frac{\beta_i^{(2)}}{\beta_i}, (y, z) \neq \left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \right\}, \\ R_j(x) &:= \bigcup_{i \in K_j(x)} \left\{ (y, z) : y \ge \frac{c_i}{\beta_i}, z \le \frac{\beta_i^{(2)}}{\beta_i}, (y, z) \neq \left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \right\}. \end{aligned}$$

Denote $S_j(x) := Q_j(x) \cap R_j(x), T_j(x) := P_j(x) \cup S_j(x)$. See Figure 8.1, where the bold dots constitute the set $A_j(x)$. The area inside the dotted lines corresponds to the set $P_j(x)$. The rectangular area represents the set $S_j(x)$.

Lemma 8.3.4

$$\begin{split} T_{j'}(x^*) \cap T_{j''}(x^*) &= \emptyset \text{ for } j' \neq j''.\\ \text{In other words, if } \left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \in S_j(x^*), \text{ then } x^*_{ij} = 1. \end{split}$$

Proof See Appendix 8.D.

Lemma 8.3.4 suggests that in the case of identical servers, the customer types should be clustered according to the corresponding $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right)$ -values in an even stronger sense than stated earlier in Lemma 8.3.1.





The optimality of clustering as exposed in the previous Lemma's suggests that the optimal routing probabilities are almost all equal to either 0 or 1. Although the settings are quite different, the latter observation strongly reminds of the vertex-allocation theorem for the optimal routing of single customer chains in closed product-form networks, saying that each customer should consistently select the same server for each request type, rather than choose probabilistically, cf. Cheng & Muntz [67], Tripathi & Woodside [180], Woodside & Tripathi [189].

8.4 The case of ordered customer types

In this section, we show how the structure, as characterized in the Lemma's of the previous section, may be exploited in computing an optimal allocation. We make the following assumption.

Assumption 8.4.1

The customer types are ordered such that $\frac{c_{i'}}{\beta_{i'}} \leq \frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}} \geq \frac{\beta_{i''}^{(2)}}{\beta_{i''}}, \left(\frac{c_{i'}}{\beta_{i'}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}}\right) \neq \left(\frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i''}^{(2)}}{\beta_{i''}}\right) \text{ for } i' < i''.$



FIGURE 8.2. The case of ordered customer types.

Note that if Assumption 8.4.1 is satisfied, then $P_j(x) \subseteq S_j(x)$, so that $T_j(x) = S_j(x)$. Lemma 8.3.4 then provides a very strong characterization of an optimal allocation, see Figure 8.2, where the rectangles represent the sets $T_j(x)$ for an instance with m = 3 servers and n = 16 customer types. The fact that the sets $T_j(x)$ do not intersect, completely determines their 'position', so that only the problem remains to determine their 'size'. In cases where the customer types are not ordered, the structure may still be exploited in actually determining an optimal allocation, but not in such a simple manner.

Theoretically speaking, Assumption 8.4.1 is somewhat restrictive. However, there are several cases of practical interest that satisfy Assumption 8.4.1.

Case i. $c_i/\beta_i = \gamma, i = 1, \ldots, n$.

In other words, the waiting costs per unit of time are proportional to the mean service times. This is the case when the goal is minimizing the mean amount of waiting work, $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$. Minimizing the mean amount of waiting work is equivalent to minimizing the mean total amount of work, since the difference, the mean amount of work in service always equals $\frac{1}{2} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)}$, irrespective of the allocation x. When the customer types have the same mean service time, minimizing the mean amount of waiting work is also equivalent to minimizing the overall mean waiting time.

150

8.4 The case of ordered customer types

For $c_i/\beta_i = \gamma$, i = 1, ..., n, Corollary 8.3.1 reduces to

$$\frac{\sum\limits_{i=1}^n \lambda_i \beta_i^{(2)} x_{ij'}^*}{\mu/m - \sum\limits_{i=1}^n \lambda_i \beta_i x_{ij'}^*} \ge \frac{\sum\limits_{i=1}^n \lambda_i \beta_i^{(2)} x_{ij''}^*}{\mu/m - \sum\limits_{i=1}^n \lambda_i \beta_i x_{ij''}^*} \Longleftrightarrow \sum\limits_{i=1}^n \lambda_i \beta_i x_{ij'}^* \le \sum\limits_{i=1}^n \lambda_i \beta_i x_{ij''}^*.$$

In particular,

$$\sum_{i=1}^{n} \lambda_{i} \beta_{i} x_{ij'}^{*} = \sum_{i=1}^{n} \lambda_{i} \beta_{i} x_{ij''}^{*} \iff \sum_{i=1}^{n} \lambda_{i} \beta_{i}^{(2)} x_{ij'}^{*} = \sum_{i=1}^{n} \lambda_{i} \beta_{i}^{(2)} x_{ij''}^{*}.$$
 (8.6)

For $c_i/\beta_i = \gamma$, i = 1, ..., n, the set $T_j(x)$ reduces to the line-segment

$$\left\{ (\gamma, z) \mid \min_{i \in K_j(x)} \frac{\beta_i^{(2)}}{\beta_i} < z < \max_{i \in K_j(x)} \frac{\beta_i^{(2)}}{\beta_i} \right\}.$$

Thus, Lemma 8.3.4 says that the customer types should be clustered according to the corresponding $\beta_i^{(2)}/\beta_i$ -values. This means that

$$\left(\sum_{i=1}^n \lambda_i \beta_i x_{ij}^*, \sum_{i=1}^n \lambda_i \beta_i^{(2)} x_{ij}^*\right) = \left(\frac{1}{m} \sum_{i=1}^n \lambda_i \beta_i, \frac{1}{m} \sum_{i=1}^n \lambda_i \beta_i^{(2)}\right),$$

 $j = 1, \ldots, m$, can only hold when all the customer types not only have the same $c_i/\beta_i = \gamma$, but also happen to have the same $\beta_i^{(2)}/\beta_i$, which is not the case in general. In view of (8.6), we may thus conclude that $\sum_{i=1}^n \lambda_i \beta_i x_{ij'}^* \neq \sum_{i=1}^n \lambda_i \beta_i x_{ij''}^*$, i.e., the total load should *not* be completely balanced in general.

Case ii. $\beta_i^{(2)} / \beta_i = \delta, i = 1, ..., n.$

In other words, the mean residual service times are constant. This is the case when the customer types have the same service time characteristics, but different priorities, reflected in different waiting costs per unit of time. For $\beta_i^{(2)}/\beta_i = \delta$, i = 1, ..., n, Corollary 8.3.1 reduces to

$$\sum_{i=1}^n \lambda_i \beta_i x_{ij'}^* \ge \sum_{i=1}^n \lambda_i \beta_i x_{ij''}^* \iff \frac{\sum_{i=1}^n \lambda_i c_i x_{ij'}^*}{\mu/m - \sum_{i=1}^n \lambda_i \beta_i x_{ij'}^*} \le \frac{\sum_{i=1}^n \lambda_i c_i x_{ij''}^*}{\mu/m - \sum_{i=1}^n \lambda_i \beta_i x_{ij''}^*}$$

In particular,

$$\sum_{i=1}^{n} \lambda_i \beta_i x_{ij'}^* = \sum_{i=1}^{n} \lambda_i \beta_i x_{ij''}^* \iff \sum_{i=1}^{n} \lambda_i c_i x_{ij'}^* = \sum_{i=1}^{n} \lambda_i c_i x_{ij''}^*.$$
 (8.7)

For $\beta_i^{(2)}/\beta_i = \delta$, i = 1, ..., n, the set $T_j(x)$ reduces to the line-segment

$$\left\{ (y,\delta) \mid \min_{i \in K_j(x)} \frac{c_i}{\beta_i} < y < \max_{i \in K_j(x)} \frac{c_i}{\beta_i} \right\}.$$

Thus, Lemma 8.3.4 says that the customer types should be clustered according to the corresponding c_i/β_i -values (which strongly reminds of the $c\mu$ -rule, cf. [151], although the $c\mu$ -rule in fact refers to the order of service of customer types, rather than the clustering of customer types). This implies that

$$\left(\sum_{i=1}^n \lambda_i \beta_i x_{ij}^*, \sum_{i=1}^n \lambda_i c_i x_{ij}^*\right) = \left(\frac{1}{m} \sum_{i=1}^n \lambda_i \beta_i, \frac{1}{m} \sum_{i=1}^n \lambda_i c_i\right),$$

 $\sum_{i=1}^{n} \lambda_i \beta_i x_{ij''}^*$, i.e., the total load should *not* be completely balanced in general.

Case iii. $c_i = c, i = 1, ..., n, \beta_{i'} \leq \beta_{i''} \iff \beta_{i'}^{(2)} / \beta_{i'} \leq \beta_{i''}^{(2)} / \beta_{i''}.$

In other words, the waiting costs per unit of time are constant. This is the case when the goal is minimizing the overall mean waiting time. Moreover, a larger mean service time corresponds to a larger mean residual service time. For the majority of meaningful service time distributions this is indeed the case. Lemma 8.3.4 then says that the customer types should be clustered according to the corresponding β_j -values. Again, it may verified that the total load should not be completely balanced, unless the values of λ_i , β_i , $\beta_i^{(2)}$ happen to satisfy some very specific relationships.

Remark 8.4.1

Buzacott & Shanthikumar [64] consider the problem of finding an optimal allocation in the case $c_i = c = 1, i = 1, ..., n$, i.e., the goal is minimizing the mean overall waiting time. In addition, they require that the total load be balanced, i.e., $\sum_{i=1}^{n} \rho_i x_{ij} = \rho/m, j = 1, ..., m$. They show that if the agreeability condition $\beta_{i'} \leq \beta_{i''} \iff \beta_{i'}^{(2)}/\beta_{i'} \leq \beta_{i''}^{(2)}/\beta_{i''}$ is satisfied, then the customer types should be clustered according to the corresponding β_i -values, as we also concluded in Case iii. above, without requiring that the total load be balanced. In fact, we concluded in Case iii. that the total load should not be completely balanced in general.

We now describe a method for determining an optimal allocation in cases that satisfy Assumption 8.4.1. Here we sketch the main idea of the method. In Appendix 8.E, we describe the method in greater detail.

Denote by $x_j = (x_{1j}, \ldots, x_{nj})$ the allocation to server j. From Lemma 8.3.4 we know that the structure of an optimal allocation is such that (i) $x_j^* = (0, \ldots, 0, x_{i'_j j}^*, 1, \ldots, 1, x_{i''_j j}^*, 0, \ldots, 0)$ with $i''_j = \max\{i \mid x_{ij}^* > 0\}, i''_j = i'_{j+1}$.

8.4 The case of ordered customer types

This knowledge, together with (ii) $\left\{\frac{\partial f(x)}{\partial x_{i''_j j}}\right\}_{|x=x^*} = \left\{\frac{\partial f(x)}{\partial x_{i'_{j+1} j+1}}\right\}_{|x=x^*}$, allows us to essentially reduce the search for an optimal allocation in $n \times m$ dimensions to a binary search with regard to $s_1 = \sum_{i=1}^n x_{i1}$.

Step 1. Determine a lower and an upper bound for s_1^* .

Step 2. Make an estimate s_1 for s_1^* , somewhere in between lower and upper bound, which directly determines x_1 .

Step 3. Given x_j , determine x_{j+1} , for j = 1, 2, ..., m-1, from (i), (ii). In Appendix 8.E, we show that through (i), (ii) x_j uniquely determines x_{j+1} , j = 1, 2, ..., m-1.

Step 4. Sooner or later one either runs out of servers or out of customer types. If one runs out of servers, then apparently $s_1 < s_1^*$, so then replace the old lower bound by s_1 . If one runs out of customer types, then apparently $s_1 > s_1^*$, so then replace the old upper bound by s_1 . Repeat the procedure until lower and upper bound are considered to be sufficiently close.

We now consider some examples illustrating that the load should indeed not be completely balanced in general. We assume that there are m = 4 servers and n = 4 customer types. We take $(\mu_1, \mu_2, \mu_3, \mu_4) = (1, 1, 1, 1), (\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \alpha(8, 8, 1, 1), (\beta_1, \beta_2, \beta_3, \beta_4) = (1, 2, 4, 8), so <math>(\rho_1, \rho_2, \rho_3, \rho_4) = 4\alpha(2, 4, 1, 2)$. We take successively $\alpha = 0.01, \alpha = 0.05, \alpha = 0.10, \alpha = 0.11$, so successively $\rho = 0.36, \rho = 1.80, \rho = 3.6, \rho = 3.96$. We assume that $\beta_i^{(2)} = \kappa \beta_i^2, i = 1, \ldots, n$, so that the agreeability condition is trivially satisfied. Note that κ influences the value of the objective function $f(\cdot)$ linearly, so that an optimal allocation is independent of κ . We compare the value of the objective function $f(\cdot)$ with $\kappa = 1$ for each of the following three allocations:

(i). x^S , the completely symmetric allocation, i.e., $x_{ij}^S = 1/m$, i = 1, ..., n, j = 1, ..., m;

(ii). x^B , the optimal allocation with $r_j(x^B) = \sum_{i=1}^n \rho_i x_{ij}^B = \rho/m, \ j = 1, \dots, m;$

(iii). x^* , the true optimal allocation computed by the method of Appendix 8.E; as well as the $r_j(x^*) = \sum_{i=1}^n \rho_i x_{ij}^*$ for the latter allocation. Table 8.1 contains the results for the case $c_i = c = 1, i = 1, ..., n$, like in Buzacott & Shanthikumar [64], i.e., f(x) measures (twice) the overall mean waiting time for allocation x.

α	$f(x^S)$	$f(x^B)$	$f(x^*)$	r_1^*	r_2^*	r_3^*	r_4^*
0.01	0.05934	0.03747	0.03728	0.0876	0.0916	0.0986	0.0822
0.05	2.4545	1.5500	1.5468	0.4463	0.4502	0.4673	0.4362
0.10	54.100	34.100	34.086	0.8999	0.8980	0.9030	0.8992
0.11	653.40	412.61	412.46	0.9900	0.9897	0.9903	0.9900

TABLE 8.1. Value of objective function for x^B , x^S , and x^* with $c_j = c = 1$.

Table 8.1 supports the conclusion that in general the load should not be completely balanced, but suggests that for $c_i = c, i = 1, ..., n$, the quality of x^B tends to match that of x^* .

Table 8.2 contains the results for the case $c_i = \beta_i$, i = 1, ..., n, i.e., f(x) measures (twice) the mean amount of waiting work for allocation x.

α	$f(x^S)$	$f(x^B)$	$f(x^*)$	r_1^*	r_2^*	r_3^*	r_4^*
0.01	0.11868	0.11868	0.10477	0.0553	0.0647	0.1140	0.1260
0.05	4.9091	4.9091	4.3542	0.3366	0.4105	0.5000	0.5528
0.10	108.00	108.00	94.941	0.8464	0.9024	0.9162	0.9351
0.11	1306.8	1306.8	1149.7	0.9840	0.9904	0.9917	0.9938

TABLE 8.2. Value of objective function for x^B , x^S , and x^* with $c_j = \beta_j$.

Table 8.2 suggests that for $c_i \neq c$, i = 1, ..., n, the quality of x^B in comparison with x^* tends to deteriorate.

8.5 FINDING AN OPTIMAL SOURCE PARTITIONING

In this section, we consider the problem of finding an optimal source partitioning, i.e., a deterministic allocation of the customer types to the servers that minimizes the mean total waiting cost per unit of time. Using (8.1) and (8.4), we formulate the problem as follows.

Problem (II).

minimize

ze
$$f(x) = \sum_{j=1}^{m} \frac{\left(\sum_{i=1}^{n} \lambda_i c_i x_{ij}\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}\right)}{\mu_j \left(\mu_j - \sum_{i=1}^{n} \lambda_i \beta_i x_{ij}\right)}$$
(8.8)

subject to

n

$$\sum_{i=1}^{m} \lambda_i \beta_i x_{ij} < \mu_j, \qquad j = 1, \dots, m;$$
$$\sum_{j=1}^{m} x_{ij} = 1, \qquad i = 1, \dots, n;$$
$$x_{ij} \in \{0, 1\}, \qquad i = 1, \dots, n, \ j = 1, \dots, m.$$

Problem (II) is a non-linear integer programming problem. Finding a feasible solution of problem (II) is equivalent to packing n elements of size ρ_1, \ldots, ρ_n in m bins of capacity μ_1, \ldots, μ_m , which is known to be NP-hard for $m \ge 2$, even in the case $\mu_j = \mu/m$, cf. Garey & Johnson [105]. Finding an optimal solution is of course at least as hard as finding a feasible solution of problem (II). Thus, even in the case of identical servers, there is not likely to be an effi-

cient way of solving problem (II), which motivates a heuristic approach. Since even finding a feasible solution of problem (II) is NP-hard, we have to take for

8.5 Finding an optimal source partitioning

granted however that heuristic methods cannot be guaranteed to yield even a feasible solution. Globally speaking, the larger n and the larger $\mu - \rho$, the more feasible solutions there are and the more likely heuristic methods are to yield a feasible solution of problem (II). From a practical point of view, neither small n nor very small $\mu - \rho$ are of great concern with respect to solving problem (II). For small n, the NP-hardness of problem (II) does not really matter, so that it may be wiser to decide solving problem (II) by an enumerative method. For very small $\mu - \rho$, the system will be critically loaded anyway, so that it may be wiser in the design of the system to decide employing an extra server.

Below we indicate how the structure of an optimal solution of problem (I), the non-integer version of problem (II) $(x_{ij}$'s non-integer), may be used as a guideline in heuristically solving problem (II). To illustrate the specific complexity of problem (II), in comparison with problem (I), we first consider the special case of identical servers, i.e., $\mu_j = \mu/m, j = 1, \ldots, m$, and 'almost' identical customer types, i.e., $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) = (\gamma, \delta), i = 1, \ldots, n.$

Minimizing f(x) then reduces to minimizing g(x) defined as

$$g(x):=\sum_{j=1}^m rac{\left(\sum\limits_{i=1}^n
ho_i x_{ij}
ight)^2}{\mu/m-\sum\limits_{i=1}^n
ho_i x_{ij}}.$$

In a sense, minimizing g(x) amounts to making $\sum_{i=1}^{n} \rho_i x_{ij}$ for all $j = 1, \ldots, m$ as equal as possible. In particular, if $\sum_{i=1}^{n} \rho_i x_{ij} = \rho/m$ for all $j = 1, \ldots, m$, then g(x) is certainly minimal, cf. Lemma 8.3.2. So an optimal solution of problem (I) is simply $x_{ij}^* = 1/m$, $i = 1, \ldots, n$, $j = 1, \ldots, m$. Due to the integrality conditions, an optimal solution of problem (II) is however not so simple to obtain.

Minimizing g(x) is strongly related to minimizing h(x) defined as

$$h(x) := \max_{j=1,...,m} \sum_{i=1}^{n} \rho_i x_{ij}.$$

Just like minimizing g(x), minimizing h(x) in a sense amounts to making $\sum_{i=1}^{n} \rho_i x_{ij}$ for all $j = 1, \ldots, m$ as equal as possible. In particular, if $\sum_{i=1}^{n} \rho_i x_{ij} = \rho/m$ for all $i = 1, \ldots, n$, then h(x) is certainly minimal, again just like g(x). For $m \leq 2$, minimizing h(x) is completely equivalent to minimizing g(x). In machine scheduling, minimizing h(x) is known as minimizing the makespan of n jobs of length ρ_1, \ldots, ρ_n on m parallel identical machines, which is known to be NP-hard for $m \geq 2$, cf. Garey & Johnson [105]. (Of course this is not very surprising, knowing that packing is already NP-hard; minimizing the makespan is

equivalent to packing the jobs in m bins of capacity as small as possible, which is of course at least as hard as packing the jobs in m bins of given capacity.) A prominent family of heuristics for minimizing the makespan is the class of list scheduling rules. Characteristically of list scheduling rules, jobs are successively selected in order of appearance on some prespecified list, to be assigned to the machine with the least total processing time already assigned. The worst-case ratio for a list scheduling rule is 2 - 1/m, cf. [110]. Evidently, the list ordering is the critical factor for the performance of a list scheduling rule. There is e.g. always a list ordering for which the list scheduling rule yields an optimal schedule. The worst-case instances suggest that the performance of a list scheduling rule may be better when jobs are selected in order of non-increasing ρ_i , which features the LPT (Longest Processing Time) rule. Indeed, the worst-case ratio for the LPT rule is 4/3 - 1/(3m), cf. [111]. Of course, a worst-case ratio renders an inherently sombre picture, which not necessarily reflects the average performance. Probabilistic analysis reveals that the average performance of the LPT rule is indeed considerably better than the worst-case ratio may suggest, cf. [70].

Although minimizing h(x) is closely related to minimizing g(x), measures of the good performance with regard to h(x) do not immediately carry over to measures of a good performance with regard to g(x). The crux is that g(x) is substantially more sensitive to suboptimality than h(x); the larger the total traffic intensity, the more sensitive.

So far, we considered the special case of identical servers and 'almost' identical customer types, in which it was easy to conclude that the total traffic stream should be balanced among the servers. Due to the integrality conditions, it was however not so easy to accomplish an exactly balanced division. Returning now to the general case, we cannot expect the picture to be brighter. On the contrary, in the general case it is not even known how exactly the total traffic stream should be divided among the servers, not to mention the problem of accomplishing a desirable division.

We now indicate how the structure of an optimal solution of problem (I) may be used as a guideline in heuristically solving problem (II). Since the cost structure of problem (II) and problem (I) do not differ, there is no reason to believe that the structure of an optimal solution of problem (II) and problem (I) will dramatically differ. Globally speaking, the larger n and the larger $\mu - \rho$, the more feasible integer allocations there are and the more the structure of an optimal integer allocation is likely to resemble the structure of an optimal non-integer allocation. As mentioned earlier, from a practical point of view, neither small n nor very small $\mu - \rho$ are of great concern with respect to solving problem (II). There are several options as to how the knowledge of an optimal solution of problem (I), the non-integer version of problem (II), may be used in heuristically solving problem (II).

Option 1.

A first option is to construct an integer allocation, starting from an optimal non-integer allocation. One may e.g. somehow round an optimal non-integer allocation x^* to an integer allocation x^{**} , taking into account the condition $\sum_{j=1}^{m} x_{ij}^{**} = 1$, $i = 1, \ldots, n$, i.e., $x_{ij}^{**} = 1$ for $j = j_i$, $x_{ij}^{**} = 0$ for $j \neq j_i$, for some j_i with $x_{ij_i}^* > 0$, $j = 1, \ldots, m$. As mentioned earlier, the optimality of clustering suggests that an optimal non-integer allocation is in fact 'almost' integer. Thus, rounding may be expected to yield quite acceptable results. The main drawback is that an optimal non-integer allocation is needed, which is not so simple to obtain in cases with non-identical servers or where the customer types are not ordered.

Option 2.

A second option, which to some extent meets the drawback of the first option, is to construct an integer allocation, not starting from an optimal allocation *itself*, but from the *structure*. One may e.g. select the best from all integer allocations that satisfy Lemma 8.3.1, i.e., $P_{j'}(x^*) \cap P_{j''}(x^*) = \emptyset$ for $j' \neq j''$. It is easily verified that the best of all integer allocations that satisfy Lemma 8.3.1 is at least as good as the best integer allocation obtained from rounding an optimal non-integer allocation. The main drawback is that such a procedure, although polynomial in n for fixed m, may prove to be rather time-consuming.

Option 3.

A third option, which to some extent meets the drawback of the second option, is to construct an integer allocation, again starting from the structure of an optimal integer allocation, but not so rigorously. One may e.g. select the best from not all but some proper subclass of integer allocations that satisfy Lemma 8.3.1, i.e., integer allocations that satisfy $U_{j'}(x) \cap U_{j''}(x) = \emptyset$ for $j' \neq j''$ for some $U_j(x) \supseteq P_j(x)$, $j = 1, \ldots, m$. One may define e.g. $U_j(x) := \begin{bmatrix} \min_{i \in K_j(x)} c_i/\beta_i, \max_{i \in K_j(x)} c_i/\beta_i \end{bmatrix} \times \begin{bmatrix} \min_{i \in K_j(x)} \beta_i^{(2)}/\beta_i, \max_{i \in K_j(x)} \beta_i^{(2)}/\beta_i \end{bmatrix}$, thus blowing up the sets $P_j(x)$ to rectangles, so that indeed $P_j(x) \subseteq U_j(x)$, $j = 1, \ldots, m$.

8.6 Concluding remarks and suggestions for further research

We have considered the problem of finding an allocation that minimizes the mean total waiting cost per unit of time. We have shown that the customer types should be clustered according to the corresponding $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right)$ -values, and we described for some special cases in detail how that property may be

exploited in computing an optimal allocation. In other cases (e.g. when the customer types are not ordered in the sense of Assumption 8.4.1), that property may still be exploited in calculating an optimal allocation, but not in such a simple manner. An interesting topic for further research might be to develop efficient (heuristic) methods for this. Furthermore, we have considered the problem of finding an optimal deterministic allocation. We have shown the problem to be NP-hard, and indicated how the structure of an optimal non-deterministic allocation may be used as a heuristic guideline in searching for an optimal deterministic allocation. An interesting topic for further research might be to investigate the quality of the proposed heuristics, either in a theoretical framework or by numerical experiments. In either way, the quality of the heuristics may be judged by comparison with the optimal non-deterministic allocation, which provides a lower bound for the optimal deterministic allocation.

In this chapter, we have focused on the global scheduling problem; we have hardly touched on the local scheduling problem. We assumed the order of service not to discriminate between the various customer types. When the order of service does discriminate between the various customer types, the expressions for the mean waiting times are more complicated. However, for $c_i/\beta_i = \gamma$, $i = 1, \ldots, n$, for some γ , the results of this chapter still hold. Minimizing the mean total waiting cost per unit of time $\sum_{i=1}^{n} c_i \lambda_i E \mathbf{W}_i$ then amounts to minimizing the mean amount of waiting work $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$. Minimizing the mean amount of waiting work is equivalent to minimizing the mean total amount of work is not influenced by the local scheduling strategy, which property is frequently referred to as work conservation. As a consequence, minimizing $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$ is completely insensitive to the local scheduling strategy.

In this chapter, we implicitly assumed the various customer types to be served without any interruptions. In some situations changing over from one customer type to another may however require a non-negligible switch-over time (e.g., tool changing in a flexible manufacturing system, traveling in the case of a repair crew visiting various installations). When the switch-over times are non-negligible, the expressions for the mean waiting times are more complicated. For some local scheduling strategies, so-called pseudo-conservation laws provide comparatively simple expressions for $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$, but these expressions appear to be less amenable to optimization procedures than the objective function (8.4).

In this chapter, we assumed that any customer type could, in principle, be allocated to any server. In some situations, it may however occur that some customer types cannot be allocated to some servers, or that some customer types cannot be combined. Such restrictions may be translated into additional constraints on the routing probabilities, the x_{ij} 's. It would be interesting to investigate how those restrictions on the x_{ij} 's affect the structure of an optimal allocation.

APPENDICES

8.A Proof of Lemma 8.3.1

Lemma 8.3.1

 $P_{j'}(x^*) \cap P_{j''}(x^*) = \emptyset$ for $j' \neq j''$. In other words, if $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \in P_j(x^*)$, then $x_{ij}^* = 1$.

Proof

Suppose not. So, by definition of $P_j(x^*)$, there exist $k_0, j', j'', j' \neq j''$, such that

$$\left(\frac{c_{k_0}}{\beta_{k_0}}, \frac{\beta_{k_0}^{(2)}}{\beta_{k_0}}\right) = \sum_{k \in K_{j''}(x^*)} \alpha_k \left(\frac{c_k}{\beta_k}, \frac{\beta_k^{(2)}}{\beta_k}\right),$$
(8.9)

with $x_{k_0j'}^* > 0, x_{kj''}^* > 0, \alpha_k \ge 0, k \in K_{j''}(x^*), \sum_{k \in K_{j''}(x^*)} \alpha_k = 1.$ Moreover,

not all the points $\left(\frac{c_k}{\beta_k}, \frac{\beta_k^{(2)}}{\beta_k}\right)$ with $\alpha_k > 0$ lie on a line, i.e., there is no linear

equality that is satisfied by all the points $\left(\frac{c_k}{\beta_k}, \frac{\beta_k^{(2)}}{\beta_k}\right)$ with $\alpha_k > 0$. Now consider $\epsilon \left\{\frac{\partial f(x)}{\partial x_{k_0j''}} - \frac{\partial f(x)}{\partial x_{k_0j'}}\right\}_{|x=x^*}$ and $\epsilon \left\{\frac{\partial f(x)}{\partial x_{kj'}} - \frac{\partial f(x)}{\partial x_{kj''}}\right\}_{|x=x^*}$, which measure the first-order effect on f(x) around x^* of transferring a fraction ϵ of

customer type k_0 from server j' to server j'' and transferring a fraction ϵ of customer type k from server j'' to server j', respectively, $k \in K_{j''}(x^*)$. Since x^* is an optimal solution of problem (I), the first-order effect on f(x) around x^* of transferring a customer type from one server to another cannot be negative, i.e., $\left\{\frac{\partial f(x)}{\partial x_{k_0j''}} - \frac{\partial f(x)}{\partial x_{k_0j'}}\right\}_{|x=x^*} \ge 0, \left\{\frac{\partial f(x)}{\partial x_{kj'}} - \frac{\partial f(x)}{\partial x_{kj''}}\right\}_{|x=x^*} \ge 0, \ k \in K_{j''}(x^*),$

as may also be formally verified from the Kuhn-Tucker conditions. From (8.9),

$$\lambda_{k_0}(c_{k_0},\beta_{k_0}^{(2)},\beta_{k_0}) = \sum_{k \in K_{j''}(x^*)} \alpha_k \frac{\lambda_{k_0} \beta_{k_0}}{\lambda_k \beta_k} \lambda_k(c_k,\beta_k^{(2)},\beta_k).$$

So

$$\left\{\frac{\partial f(x)}{\partial x_{k_0j''}} - \frac{\partial f(x)}{\partial x_{k_0j'}}\right\}_{|x=x^*} = -\sum_{k \in K_{j''}(x^*)} \alpha_k \frac{\lambda_{k_0} \beta_{k_0}}{\lambda_k \beta_k} \left\{\frac{\partial f(x)}{\partial x_{kj'}} - \frac{\partial f(x)}{\partial x_{kj''}}\right\}_{|x=x^*}$$

160

8.B Proof of Lemma 8.3.2

$$\begin{split} &\operatorname{As}\left\{\frac{\partial f(x)}{\partial x_{k_0j^{\prime\prime}}} - \frac{\partial f(x)}{\partial x_{k_0j^{\prime}}}\right\}_{|x=x^*} \ge 0, \alpha_k \left\{\frac{\partial f(x)}{\partial x_{kj^{\prime}}} - \frac{\partial f(x)}{\partial x_{kj^{\prime\prime}}}\right\}_{|x=x^*} \ge 0, k \in K_{j^{\prime\prime}}(x^*), \\ & \text{we conclude that } \alpha_k \left\{\frac{\partial f(x)}{\partial x_{kj^{\prime}}} - \frac{\partial f(x)}{\partial x_{kj^{\prime\prime}}}\right\}_{|x=x^*} = 0, k \in K_{j^{\prime\prime}}(x^*). \\ & \text{Differentiating } f(\cdot) \text{ once,} \end{split}$$

$$\frac{\partial f(x)}{\partial x_{ij}} = \lambda_i c_i B_j(x) + \lambda_i \beta_i^{(2)} C_j(x) + \lambda_i \beta_i \mu_j B_j(x) C_j(x).$$
(8.10)

So,
$$\frac{c_k}{\beta_k} (B_{j'}(x^*) - B_{j''}(x^*)) + \frac{\beta_k^{(2)}}{\beta_k} (C_{j'}(x^*) - C_{j''}(x^*)) = \mu_{j''} B_{j''}(x^*) C_{j''}(x^*)$$

 $- \mu_{j'} B_{j'}(x^*) C_{j'}(x^*)$ for all the points $\left(\frac{c_k}{\beta_k}, \frac{\beta_k^{(2)}}{\beta_k}\right)$ with $\alpha_k > 0$, i.e., there is a linear equality that is satisfied by all the points $\left(\frac{c_k}{\beta_k}, \frac{\beta_k^{(2)}}{\beta_k}\right)$ with $\alpha_k > 0$.

Contradiction.

8.B PROOF OF LEMMA 8.3.2

Lemma 8.3.2 If $B_{j'}(x^*) \ge B_{j''}(x^*), C_{j'}(x^*) \ge C_{j''}(x^*),$ then $\mu_{j'}B_{j'}(x^*)C_{j'}(x^*) \leq \mu_{j''}B_{j''}(x^*)C_{j''}(x^*).$

Proof

Suppose not, i.e.,

$$B_{j'}(x^*) \ge B_{j''}(x^*), C_{j'}(x^*) \ge C_{j''}(x^*),$$

$$\mu_{j'}B_{j'}(x^*)C_{j'}(x^*) > \mu_{j''}B_{j''}(x^*)C_{j''}(x^*).$$
(8.11)

Take k_0 such that $x_{k_0j'}^* > 0$. Now consider $\epsilon \left\{ \frac{\partial f(x)}{\partial x_{k_0j'}} - \frac{\partial f(x)}{\partial x_{k_0j'}} \right\}_{|x=x^*}$, which measures the first-order effect on f(x) around x^* of transferring a fraction ϵ of customer type k_0 from server

of f(x) around x of transferring a fraction t of customer type no non-correction t of the first-order effect on f(x) around x^* of transferring a customer type from one server to another cannot be negative, i.e., $\left\{\frac{\partial f(x)}{\partial x_{k_0j''}} - \frac{\partial f(x)}{\partial x_{k_0j'}}\right\}_{|x=x^*} \ge 0$, as may also be formally verified from the Kuhn-Tucker conditions.

From (8.10), if (8.11) were to hold,

$$\left\{\frac{\partial f(x)}{\partial x_{k_0j''}} - \frac{\partial f(x)}{\partial x_{k_0j'}}\right\}_{|x=x^*} =$$

Chapter 8 Optimal allocation of customer types to servers

$$\lambda_{k_0} c_{k_0} \left(B_{j''}(x^*) - B_{j'}(x^*) \right) + \lambda_{k_0} \beta_{k_0}^{(2)} \left(C_{j''}(x^*) - C_{j'}(x^*) \right) + \\\lambda_{k_0} \beta_{k_0} \left(\mu_{j''} B_{j''}(x^*) C_{j''}(x^*) - \mu_{j'} B_{j'}(x^*) C_{j'}(x^*) \right) < 0.$$

Contradiction.

8.C PROOF OF LEMMA 8.3.3

Lemma 8.3.3

Assume
$$x_{i'j'}^* > 0, x_{i''j''}^* > 0.$$

If $\frac{c_{i'}}{\beta_{i'}} \le \frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}} \ge \frac{\beta_{i''}^{(2)}}{\beta_{i''}}, \left(\frac{c_{i'}}{\beta_{i'}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}}\right) \ne \left(\frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i''}^{(2)}}{\beta_{i''}}\right), \text{ then } j' \le j''.$

Proof

Suppose not, i.e., $x^*_{i'j'} > 0, \, x^*_{i''j''} > 0, \, j' \neq j'',$

$$\frac{c_{i'}}{\beta_{i'}} \le \frac{c_{i''}}{\beta_{i''}}, \quad \frac{\beta_{i'}^{(2)}}{\beta_{i'}} \ge \frac{\beta_{i''}^{(2)}}{\beta_{i''}}, \quad \left(\frac{c_{i'}}{\beta_{i'}}, \frac{\beta_{i'}^{(2)}}{\beta_{i'}}\right) \ne \left(\frac{c_{i''}}{\beta_{i''}}, \frac{\beta_{i''}^{(2)}}{\beta_{i''}}\right), \tag{8.12}$$

$$B_{j'}(x^*) \leq B_{j''}(x^*), \ C_{j'}(x^*) \geq C_{j''}(x^*).$$

Now consider $\epsilon \left\{ \frac{\partial f(x)}{\partial x_{i'j''}} - \frac{\partial f(x)}{\partial x_{i'j'}} \right\}_{|x=x^*}$ and $\epsilon \left\{ \frac{\partial f(x)}{\partial x_{i''j'}} - \frac{\partial f(x)}{\partial x_{i''j''}} \right\}_{|x=x^*}$, which measure the first-order effect on f(x) around x^* of transferring a fraction ϵ of customer type i' from server j' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from server j'' to server j'' and a fraction ϵ of customer type i'' from one server to another cannot be negative, i.e., $\left\{ \frac{\partial f(x)}{\partial x_{i'j'}} - \frac{\partial f(x)}{\partial x_{i'j'}} \right\}_{|x=x^*} \ge 0, \left\{ \frac{\partial f(x)}{\partial x_{i''j'}} - \frac{\partial f(x)}{\partial x_{i''j''}} \right\}_{|x=x^*} \ge 0.$ From (8.10),

$$\begin{split} \lambda_{i^{\prime\prime}}\beta_{i^{\prime\prime}} \left\{ \frac{\partial f(x)}{\partial x_{i^{\prime}j^{\prime\prime}}} - \frac{\partial f(x)}{\partial x_{i^{\prime}j^{\prime}}} \right\} + \lambda_{i^{\prime}}\beta_{i^{\prime}} \left\{ \frac{\partial f(x)}{\partial x_{i^{\prime\prime}j^{\prime\prime}}} - \frac{\partial f(x)}{\partial x_{i^{\prime\prime}j^{\prime\prime}}} \right\} = \\ \lambda_{i^{\prime}}\lambda_{i^{\prime\prime}}\beta_{i^{\prime\prime}}\beta_{i^{\prime\prime}} \left\{ \left(\frac{c_{i^{\prime}}}{\beta_{i^{\prime}}} - \frac{c_{i^{\prime\prime}}}{\beta_{i^{\prime\prime}}} \right) \left(B_{j^{\prime\prime}}(x) - B_{j^{\prime}}(x) \right) + \\ \left(\frac{\beta_{i^{\prime}}^{(2)}}{\beta_{i^{\prime}}} - \frac{\beta_{i^{\prime\prime}}^{(2)}}{\beta_{i^{\prime\prime}}} \right) \left(C_{j^{\prime\prime}}(x) - C_{j^{\prime}}(x) \right) \right\}. \end{split}$$

8.C Proof of Lemma 8.3.3

So, if (8.12) were to hold, then

$$\lambda_{i''}\beta_{i''}\left\{\frac{\partial f(x)}{\partial x_{i'j''}}-\frac{\partial f(x)}{\partial x_{i'j'}}\right\}_{|x=x^*}+\lambda_{i'}\beta_{i'}\left\{\frac{\partial f(x)}{\partial x_{i''j'}}-\frac{\partial f(x)}{\partial x_{i''j''}}\right\}_{|x=x^*}\leq 0.$$

Contradiction, unless

$$\left\{\frac{\partial f(x)}{\partial x_{i'j''}} - \frac{\partial f(x)}{\partial x_{i'j'}}\right\}_{|x=x^*} = 0, \left\{\frac{\partial f(x)}{\partial x_{i''j'}} - \frac{\partial f(x)}{\partial x_{i''j''}}\right\}_{|x=x^*} = 0.$$

Now consider $\epsilon^2 \sum_{i_1,i_2=i',i''} \sum_{j_1,j_2=j',j''} \left\{ \frac{\partial^2 f(x)}{\partial x_{i_1j_1} \partial x_{i_2j_2}} \alpha_{i_1j_1} \alpha_{i_2j_2} \right\}_{|x=x^*}$ for $\alpha_{i'j'} = -\lambda_{i''}\beta_{i''}, \alpha_{i'j'j'} = \lambda_{i'}\beta_{i'} + \alpha, \alpha_{i''j''} = -\lambda_{i'}\beta_{i'} - \alpha$, which measures the second-order effect on f(x) around x^* of transferring a fraction $\epsilon \lambda_{i''}\beta_{i''}$ of

 $\begin{aligned} &-\lambda_{i''}\beta_{i''}, \alpha_{i'j''} = \lambda_{i''}\beta_{i''}, \alpha_{i''j'} = \lambda_{i'}\beta_{i'} + \alpha, \alpha_{i''j''} = -\lambda_{i'}\beta_{i'} - \alpha, \text{ which measures} \\ &\text{the second-order effect on } f(x) \text{ around } x^* \text{ of transferring a fraction } \epsilon\lambda_{i''}\beta_{i''} \text{ of} \\ &\text{customer type } i' \text{ from server } j' \text{ to server } j'' \text{ and a fraction } \epsilon(\lambda_{i'}\beta_{i'} + \alpha) \text{ of} \\ &\text{customer type } i'' \text{ from server } j'' \text{ to server } j'. \text{ Since } x^* \text{ is an optimal solution} \\ &\text{of problem (I), while the first-order effects on } f(x) \text{ around } x^* \text{ of transferring} \\ &\text{customer types from one server to another are zero, the second-order effect} \\ &\text{cannot be negative, i.e., } \sum_{i_1,i_2=i',i''} \sum_{j_1,j_2=j',j''} \left\{ \frac{\partial^2 f(x)}{\partial x_{i_1j_1}\partial x_{i_2j_2}} \alpha_{i_1j_1}\alpha_{i_2j_2} \right\}_{|x=x^*} \ge 0. \\ &\text{Differentiating } f(\cdot) \text{ twice,} \end{aligned}$

$$\frac{\partial^2 f(x)}{\partial x_{i_1 j_1} \partial x_{i_2 j_2}} = \delta_{j_1 j_2} \left\{ \frac{\lambda_{i_1} \lambda_{i_2} \left(c_{i_1} \beta_{i_2}^{(2)} + c_{i_2} \beta_{i_1}^{(2)} \right) + 2\lambda_{i_2} \beta_{i_2} \frac{\partial f(x)}{\partial x_{i_1 j_1}}}{\mu/m - \sum_{i=1}^n \lambda_i \beta_i x_{i_j_1}} + \frac{\lambda_{i_1} \lambda_{i_2} \left((\beta_{i_1} c_{i_2} - \beta_{i_2} c_{i_1}) \sum_{i=1}^n \lambda_i \beta_i^{(2)} x_{i_j_1} + \left(\beta_{i_1} \beta_{i_2}^{(2)} - \beta_{i_2} \beta_{i_1}^{(2)} \right) \sum_{i=1}^n \lambda_i c_i x_{i_j_1} \right)}{\left(\mu/m - \sum_{i=1}^n \lambda_i \beta_i x_{i_j_1} \right)^2}$$

with $\delta_{j_1j_2}$ the Kronecker delta. So, as $\frac{\partial f(x)}{\partial x_{i'j'}} - \frac{\partial f(x)}{\partial x_{i'j'}} = 0$, $\frac{\partial f(x)}{\partial x_{i''j'}} - \frac{\partial f(x)}{\partial x_{i''j''}} = 0$, $j' \neq j''$,

$$\begin{split} \sum_{i_{1},i_{2}=i',i''} \sum_{j_{1},j_{2}=j',j''} \frac{\partial^{2} f(x)}{\partial x_{i_{1}j_{1}} \partial x_{i_{2}j_{2}}} \alpha_{i_{1}j_{1}} \alpha_{i_{2}j_{2}} = \\ 2 \left\{ \frac{1}{\mu/m - \sum_{i=1}^{n} \lambda_{i} \beta_{i} x_{ij'}} + \frac{1}{\mu/m - \sum_{i=1}^{n} \lambda_{i} \beta_{i} x_{ij''}} \right\} \times \\ \left\{ \lambda_{i'}^{2} \lambda_{i''}^{2} \beta_{i'}^{2} \beta_{i''}^{2} \left(\frac{c_{i''}}{\beta_{i''}} - \frac{c_{i'}}{\beta_{i'}} \right) \left(\frac{\beta_{i''}^{(2)}}{\beta_{i''}} - \frac{\beta_{i'}^{(2)}}{\beta_{i'}} \right) + \alpha \lambda_{i'} \lambda_{i''}^{2} \beta_{i'} \beta_{i''}^{2} \times \end{split}$$

$$\left(\left(\frac{c_{i^{\prime\prime}}}{\beta_{i^{\prime\prime}}}-\frac{c_{i^{\prime}}}{\beta_{i^{\prime}}}\right)\left(\frac{\beta_{i^{\prime\prime}}^{(2)}}{\beta_{i^{\prime\prime}}}+B_{j^{\prime}}(x)\right)+\left(\frac{\beta_{i^{\prime\prime}}^{(2)}}{\beta_{i^{\prime\prime}}}-\frac{\beta_{i^{\prime}}^{(2)}}{\beta_{i^{\prime}}}\right)\left(\frac{c_{i^{\prime\prime}}}{\beta_{i^{\prime\prime}}}+C_{j^{\prime}}(x)\right)\right)+\alpha^{2}\left(\lambda_{i^{\prime\prime}}^{2}c_{i^{\prime\prime}}\beta_{i^{\prime\prime}}^{(2)}+\lambda_{i^{\prime\prime}}\beta_{i^{\prime\prime}}\frac{\partial f(x)}{\partial x_{i^{\prime\prime}j^{\prime}}}\right)\right\}.$$

So, if (8.12) were to hold, then $\sum_{i_1, i_2=i', i''} \sum_{j_1, j_2=j', j''} \left\{ \frac{\partial^2 f(x)}{\partial x_{i_1j_1} \partial x_{i_2j_2}} \alpha_{i_1j_1} \alpha_{i_2j_2} \right\}_{|x=x^*}$ < 0 for some α . Contradiction.

8.D Proof of Lemma 8.3.4

Lemma 8.3.4

 $T_{j'}(x^*) \cap T_{j''}(x^*) = \emptyset \text{ for } j' \neq j''.$ In other words, if $\left(\frac{c_i}{\beta_i}, \frac{\beta_i^{(2)}}{\beta_i}\right) \in T_j(x^*)$, then $x_{ij}^* = 1$.

Proof

Suppose not. From Lemma 8.3.1, $P_{j'}(x^*) \cap P_{j''}(x^*) = \emptyset$. So, by definition of $T_j(x^*)$, there exist $j', j'', j' \neq j'', i'_{j'}, i''_{j'} \in K_{j'}(x^*), i'_{j''}, i''_{j''} \in K_{j''}(x^*)$ (possibly either $i'_{j'} = i''_{j'}$ or $i'_{j''} = i''_{j''}$), such that

$$\frac{c_{i'_{j'}}}{\beta_{i'_{j'}}} \leq \frac{c_{i''_{j'}}}{\beta_{i''_{j'}}}, \frac{\beta_{i'_{j'}}^{(2)}}{\beta_{i'_{j'}}} \geq \frac{\beta_{i''_{j}}^{(2)}}{\beta_{i''_{j'}}}, \left(\frac{c_{i'_{j'}}}{\beta_{i'_{j'}}}, \frac{\beta_{i'_{j}}^{(2)}}{\beta_{i''_{j'}}}\right) \neq \left(\frac{c_{i''_{j'}}}{\beta_{i''_{j'}}}, \frac{\beta_{i''_{j}}^{(2)}}{\beta_{i''_{j'}}}\right),$$
$$\frac{c_{i'_{j''}}}{\beta_{i''_{j''}}} \leq \frac{c_{i''_{j''}}}{\beta_{i''_{j''}}}, \frac{\beta_{i'_{j''}}^{(2)}}{\beta_{i''_{j''}}} \geq \frac{\beta_{i''_{j'}}^{(2)}}{\beta_{i''_{j''}}}, \left(\frac{c_{i'_{j''}}}{\beta_{i''_{j''}}}, \frac{\beta_{i''_{j'}}^{(2)}}{\beta_{i''_{j''}}}\right) \neq \left(\frac{c_{i''_{j'}}}{\beta_{i''_{j''}}}, \frac{\beta_{i''_{j'}}^{(2)}}{\beta_{i''_{j''}}}\right)$$

According to Lemma 8.3.3, on the one hand, $j' \leq j''$, as $x^*_{i'_{i'},j'} > 0, x^*_{i'_{i''},j''} > 0$,

$$\frac{c_{i'_{j'}}}{\beta_{i'_{j'}}} \leq \frac{c_{i''_{j''}}}{\beta_{i''_{j''}}}, \frac{\beta_{i'_{j'}}^{(2)}}{\beta_{i'_{j'}}} \geq \frac{\beta_{i''_{j''}}^{(2)}}{\beta_{i'_{j''}}}, \left(\frac{c_{i'_{j'}}}{\beta_{i'_{j'}}}, \frac{\beta_{i'_{j'}}^{(2)}}{\beta_{i'_{j'}}}\right) \neq \left(\frac{c_{i''_{j''}}}{\beta_{i''_{j''}}}, \frac{\beta_{i''_{j''}}^{(2)}}{\beta_{i''_{j''}}}\right);$$

on the other hand, $i' \ge i''$, as $x^*_{i''_{i'}j'} > 0$, $x^*_{i'_{i''}j''} > 0$,

$$\frac{c_{i_{j'}'}}{\beta_{i_{j'}'}} \ge \frac{c_{i_{j''}}}{\beta_{i_{j''}'}}, \frac{\beta_{i_{j''}'}^{(2)}}{\beta_{i_{j''}'}} \le \frac{\beta_{i_{j''}}^{(2)}}{\beta_{i_{j''}'}}, \left(\frac{c_{i_{j''}'}}{\beta_{i_{j'}'}}, \frac{\beta_{i_{j'}'}^{(2)}}{\beta_{i_{j'}'}}\right) \ne \left(\frac{c_{i_{j''}}}{\beta_{i_{j''}}}, \frac{\beta_{i_{j''}}^{(2)}}{\beta_{i_{j''}'}}\right)$$

So j' = j''. Contradiction.

8.E A method for determining an optimal al-LOCATION

In this appendix, we describe a method for determining an optimal allocation in cases that satisfy Assumption 8.4.1. In Section 8.4, we sketched the main idea of the method. Here we describe the method in greater detail.

Lemma 8.E.1
a.
$$\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{i1}^* \ge \frac{1}{m} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} \ge \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{im}^*$$
.
b. $\sum_{i=1}^{n} \lambda_i c_i x_{i1}^* \le \frac{1}{m} \sum_{i=1}^{n} \lambda_i c_i \le \sum_{i=1}^{n} \lambda_i c_i x_{im}^*$.

Proof

Proof of a. The servers are indexed such that

$$\frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^*}{\mu/m - \sum_{i=1}^{n} \lambda_i \beta_i x_{ij}^*} \ge \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^*}{\mu/m - \sum_{i=1}^{n} \lambda_i \beta_i x_{ij+1}^*}$$

So

$$\frac{\mu}{m} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^* - \left(\sum_{i=1}^{n} \lambda_i \beta_i x_{ij+1}^*\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^*\right) \ge$$

$$\frac{\mu}{m} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^* - \left(\sum_{i=1}^{n} \lambda_i \beta_i x_{ij}^*\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^*\right).$$
(8.13)

On the one hand

r

$$\frac{\sum\limits_{i=1}^n \lambda_i \beta_i^{(2)} x_{ij}^*}{\sum\limits_{i=1}^n \lambda_i \beta_i x_{ij}^*} \geq \min_{i \in K_j(x^*)} \frac{\beta_i^{(2)}}{\beta_i};$$

on the other hand

$$\frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^*}{\sum_{i=1}^{n} \lambda_i \beta_i x_{ij+1}^*} \le \max_{i \in K_{j+1}(x^*)} \frac{\beta_i^{(2)}}{\beta_i}.$$

According to Lemma 8.3.3, $\min_{i \in K_j(x^*)} \frac{\beta_i^{(2)}}{\beta_i} \ge \max_{i \in K_{j+1}(x^*)} \frac{\beta_i^{(2)}}{\beta_i}.$ So

$$\left(\sum_{i=1}^{n} \lambda_i \beta_i x_{ij+1}^*\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^*\right) \ge$$
(8.14)

$$\left(\sum_{i=1}^{n} \lambda_i \beta_i x_{ij}^*\right) \left(\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^*\right).$$

Combining (8.13), (8.14), $\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^* \ge \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij+1}^*$. Furthermore, $\sum_{j=1}^{m} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{ij}^* = \sum_{i=1}^{n} \lambda_i \beta_i^{(2)}$, as $\sum_{j=1}^{m} x_{ij}^* = 1, i = 1, ..., n$. So $\sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{i1}^* \ge \frac{1}{m} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} \ge \sum_{i=1}^{n} \lambda_i \beta_i^{(2)} x_{im}^*$. *Proof of b.* Follows immediately from symmetry considerations.

166

8.E A method for determining an optimal allocation

and increasing in s_{j+1} . So through (i)-(iv), x_j uniquely determines x_{j+1} , unless $\frac{\partial f(x)}{\partial x_{i_j j}} > \frac{\partial f(x)}{\partial x_{i_j j+1}}$ even for $x_{nj+1} = 1$, to which we return in the next step.

Step 4. Sooner or later one either runs out of servers, i.e., $\frac{\partial f(x)}{\partial x_{i_{m-1}m-1}} = \frac{\partial f(x)}{\partial x_{i_{m-1}m}}$, but $x_{nm} < 1$, or out of customer types, i.e., $x_{nj+1} = 1$, but $\frac{\partial f(x)}{\partial x_{i_jj}} > \frac{\partial f(x)}{\partial x_{i_jj+1}}$.

If one runs out of servers, then apparently $s_1 < s_1^*$, so then replace s_1^l by s_1 . If one runs out of customer types, then apparently $s_1 > s_1^*$, so then replace s_1^u by s_1 . One may, of course, make a more sophisticated estimate for s_1^* than the estimate suggested above. One may e.g. use information about how soon one either ran out of servers or out of customer types. Repeat the procedure until lower and upper bound are considered to be sufficiently close.

CHAPTER 9

Polling systems with multiple coupled servers

9.1 INTRODUCTION

In the present chapter, we consider polling systems with multiple coupled servers. So far, there are hardly any exact results known for multiple-server polling systems, apart from some mean value results for global performance measures, like cycle times. In this chapter, we explore the class of systems that allow an exact analysis.

Polling systems with multiple servers have received remarkably little attention in the vast literature on polling systems. One of the first studies is Morris & Wang [153] in which the servers are assumed to be independent, i.e., the servers visit the queues independently of each other, each server according to some cyclic schedule. A very interesting phenomenon observed by Morris & Wang is the tendency for the servers to cluster if they follow identical routes, especially in heavy traffic. The phenomenon may be visualized as follows. A trailing server will tend to move fast, as it only encounters recently served queues, whereas a leading server will tend to be slowed down by queues that have not been served for a while, so that the servers tend to form bunches while constantly leapfrogging over one another.

Browne & Weiss [58] is one of the few studies in which the servers are assumed to be coupled, i.e., the servers always visit the queues together. They obtain index-type rules for determining the visit order that minimizes the mean length of individual cycles for both the exhaustive and the gated service discipline. Browne et al. [56] derive the mean waiting time for a completely symmetric two-queue system with an infinite number of coupled servers and deterministic service times. Browne & Kella [57] obtain the busy-period distribution for a two-queue system with an infinite number of coupled servers, exhaustive service, and deterministic service times at one queue and general service times at the other queue.

Levy & Yechiali [143] and Kao & Narayanan [122] study the joint distribution

of the queue length and the number of busy servers for a Markovian multipleserver queue, where the servers individually go on vacation when there are no waiting customers left. Mitrany & Avi-Itzhak [152] and Neuts & Lucantoni [155] analyze the joint distribution of the queue length and the number of busy servers for a Markovian multiple-server queue, where servers break down at exponential intervals and then get repaired. For references on an approximative analysis of models with multiple queues and models with independent servers, we refer to the introduction of the next chapter.

All these studies unanimously point out that multiple-server polling systems, combining the complexity of single-server polling systems and multiple-server systems, are extraordinarily hard to analyze. In fact, none of the studies (except [56], [57] for very specific two-queue infinite-server cases) presents any exact results for systems with multiple queues, apart from some mean-value results for global performance measures, like cycle times.

In this chapter, we consider the case of coupled servers. We are mainly interested in exploring the class of systems that allow an exact analysis. For these systems, we present distributional results for the waiting time, the marginal queue length, and the joint queue length at polling epochs. The motivation for considering the case of coupled servers is threefold. First, the dependence in the position of the servers does not play any complicating role then. Second, in some situations the servers may in fact happen to be physically coupled. Third, the coupled server case may also be relevant for the study of the independent server case. The tendency for the servers to cluster provides e.g. an indication that they tend to behave as if they were coupled.

The remainder of the chapter is organized as follows. In Section 9.2, we consider a single-queue multiple-server system with service interruptions, which is not only interesting in its own right, but also useful for the study of a multipleserver polling system. As described in Section 2.1, in isolation a specific queue in a polling system may be viewed as a single-queue system with service interruptions, the intervisit periods constituting the service interruptions. Results for single-queue systems with service interruptions may thus be used to obtain results for the marginal distributions in polling systems. We then return to the multiple-server polling system. In Section 9.3, we present a detailed model description. In Section 9.4, we relate the probability generating function (pgf) of the joint queue length at the beginning of a visit to the pgf of the joint queue length at the end of the previous visit. Next, we relate the pgf of the joint queue length distribution at the end of a visit to the pgf of the joint queue length at the beginning of a visit. Thus, we obtain 2n equations involving 2n pgf's, with n the number of queues. In Section 9.5, we identify some cases for which these pgf's can actually be solved from these equations. These cases include several single-queue systems with a varying number of servers, two-queue two-server systems with exhaustive service and exponential service times, as well as infinite-server systems with an arbitrary number of queues, exhaustive or gated service, and deterministic service times. In Section 9.6, we conclude with some remarks and suggestions for further research.
9.2 An M/M/m queue with coupled servers AND SERVICE INTERRUPTIONS

In this section, we consider an M/M/m queue with coupled servers and service interruptions. The service interruptions are assumed to result from some interfering process that from time to time may keep the servers from working, even when there are customers present. Service preemption due to service interruptions is allowed. The service interruptions may be intervoven with the arrival and service processes in an arbitrarily complex manner, but may not anticipate on the arrival and service times of future customers. In particular, the *durations* of successive service interruptions are allowed to be dependent. We abstract here from what kind of interfering process causes the service interruptions. In the context of polling models, a service interruption typically models the intervisit period. In the setting of performability models, a service interruption usually represents a down-period of the system. A period during which none of the servers is busy, because of a service interruption or because there are no customers present, will be called a non-serving interval. A period during which at least one of the servers is busy, will be called a serving interval.

Fuhrmann & Cooper [102] consider an M/G/1 queue with service interruptions. Under rather mild assumptions, they prove a decomposition property of the queue length distribution. Using concepts from the theory of branching processes, they show that the queue length distribution can be expressed as the convolution of the distribution of the following two quantities:

(i). the queue length at an arbitrary epoch in the 'corresponding' M/G/1 queue without service interruptions;

(ii). the queue length at an arbitrary epoch in a non-serving interval.

The 'corresponding' M/G/1 queue without service interruptions is an ordinary M/G/1 queue with similar traffic characteristics, of which the queue length distribution is simply known from the Pollaczek-Khintchine formula. To find the queue length distribution at an arbitrary epoch, it thus suffices to find the queue length distribution in a non-serving interval, which is quite often relatively simple. By the distributional form of Little's law, the queue length decomposition also translates into a decomposition of the waiting time. Under somewhat milder assumptions than Fuhrmann & Cooper, Boxma [38] proves a similar decomposition of the amount of work in the system. Browne & Kella [57] analyze the queue length distribution in an $M/G/\infty$ queue with vacations. They observe that for deterministic service times a Fuhrmann & Cooper-like decomposition property holds, but not for exponential service times.

We now analyze the queue length distribution in the M/M/m queue under consideration. Although for m = 1 the amount of work is somewhat easier to study than the queue length, for m > 1 we need to focus on the queue length, as the amount of work then no longer completely determines the number of busy servers. We make the following assumptions. (i). During a serving interval, there are no servers idling while there are customers waiting, i.e., if there are l customers present during a serving interval, then there are $\min(l,m)$ servers working, just like in an ordinary M/M/m queue.

(ii). The order in which customers enter service is independent of their service times.

Under the above assumptions, we will show that the queue length distribution can be expressed into the distribution of (conceptually) the same two quantities as in the M/G/1 queue with service interruptions, but not in the same simple convolution form. However, to find the queue length distribution at an arbitrary epoch, it still suffices to find the queue length distribution in a nonserving interval. Under some additional assumptions, we will also show how the queue length decomposition translates into a decomposition of the waiting time.

We first introduce some notation. Let λ be the arrival rate and let μ be the service rate. Define $\rho := \lambda/\mu$. Denote by N and N_I the total number of customers present (including customers in service) at, respectively, an arbitrary epoch and an arbitrary epoch in a non-serving interval. Denote by $N_{M/M/m}^{(l)}$ the number of customers at an arbitrary epoch in the 'corresponding' M/M/m queue, given that the number of customers is at least $l, l \geq 0$. The 'corresponding' M/M/m queue is an ordinary M/M/m queue with arrival rate λ and service rate μ . For $l \leq m-1$,

$$E(z^{\mathbf{N}_{M/M/m}^{(l)}}) =$$

$$\left[\sum_{k=l}^{m-1} \frac{\rho^{k}}{k!} + \frac{\rho^{m}}{m!} \frac{m}{m-\rho}\right]^{-1} \left[\sum_{k=l}^{m-1} z^{k} \frac{\rho^{k}}{k!} + z^{m} \frac{\rho^{m}}{m!} \frac{m}{m-\rho z}\right].$$
(9.1)

For $l \geq m-1$,

$$E(z^{N_{M/M/m}^{(l)}}) = z^{l} \frac{m - \rho}{m - \rho z}.$$
(9.2)

Lemma 9.2.1 expresses the distribution of N into the distribution of N_I and $N_{M/M/m}^{(l)}$.

Lemma 9.2.1

$$E(z^{\mathbf{N}}) =$$

$$\gamma \left[\sum_{l=0}^{m-2} \frac{E(z^{\mathbf{N}_{M/M/m}^{(l)}}) \Pr\{\mathbf{N}_{I} = l\}}{\Pr\{\mathbf{N}_{M/M/m}^{(l)} = l\}} + \frac{m}{m-\rho z} \sum_{l=m-1}^{\infty} z^{l} \Pr\{\mathbf{N}_{I} = l\} \right],$$
(9.3)

9.2 An M/M/m queue with coupled servers and service interruptions

with ·

$$\gamma = \left[\sum_{l=0}^{m-2} \frac{\Pr\{\mathbf{N}_I = l\}}{\Pr\{\mathbf{N}_{M/M/m}^{(l)} = l\}} + \frac{m}{m-\rho} \sum_{l=m-1}^{\infty} \Pr\{\mathbf{N}_I = l\}\right]^{-1}.$$
 (9.4)

Proof

See Appendix 9.A.

Remark 9.2.1

For $\Pr{\{N_I = 0\}} = 1$, i.e., in a non-serving interval there are never any customers present, (9.3) and (9.4) reduce to

$$\mathbf{E}(z^{\mathbf{N}}) = \left[\sum_{l=0}^{m-1} \frac{\rho^{l}}{l!} + \frac{\rho^{m}}{m!} \frac{m}{m-\rho}\right]^{-1} \left[\sum_{l=0}^{m-1} z^{l} \frac{\rho^{l}}{l!} + z^{m} \frac{\rho^{m}}{m!} \frac{m}{m-\rho z}\right],$$

which is of course just the queue length distribution at an arbitrary epoch in the corresponding M/M/m queue without service interruptions.

Remark 9.2.2

For m = 1, (9.3) and (9.4) reduce to

$$\mathbf{E}(z^{\mathbf{N}}) = \frac{1-\rho}{1-\rho z} \mathbf{E}(z^{\mathbf{N}_I}),$$

which is the Fuhrmann-Cooper decomposition for an M/M/1 queue with service interruptions.

For $m = \infty$, (9.3) and (9.4) reduce to

$$\mathbf{E}(z^{\mathbf{N}}) = \left[\sum_{l=0}^{\infty} \Pr\{\mathbf{N}_{I} = l\} \frac{l!}{\rho^{l}} \sum_{k=l}^{\infty} \frac{\rho^{k}}{k!}\right]^{-1} \left[\sum_{l=0}^{\infty} \Pr\{\mathbf{N}_{I} = l\} \frac{l!}{\rho^{l}} \sum_{k=l}^{\infty} z^{k} \frac{\rho^{k}}{k!}\right].$$
Recognizing that $\frac{l!}{\rho^{l}} \sum_{k=l}^{\infty} z^{k} \frac{\rho^{k}}{k!} = z^{l} + \rho \int_{u=0}^{z} u^{l} \mathrm{e}^{(z-u)\rho} du,$

$$E(z^{\mathbf{N}}) =$$

$$\left[1 + \rho \int_{u=0}^{1} E(u^{\mathbf{N}_{A}}) e^{(1-u)\rho} du\right]^{-1} \left[E(z^{\mathbf{N}_{I}}) + \rho \int_{u=0}^{z} E(u^{\mathbf{N}_{I}}) e^{(z-u)\rho} du\right],$$
(9.5)

which will be useful later on.

Lemma 9.2.1 implies that to find the distribution of N, it suffices to find the distribution of N_I , as the distribution of $N_{M/M/m}^{(l)}$ is known from (9.1). From a methodological point of view, however, it is more natural to analyze the queue length at either the beginning or the end of non-serving intervals than to study N_I , the queue length at an arbitrary epoch in a non-serving interval. Therefore, we now relate the distribution of N_I to the queue length distribution at such embedded epochs. Denote by $N_{\text{begin}}^{(k)}$ and $N_{\text{end}}^{(k)}$ the queue length at, respectively, the beginning and the end of the k-th non-serving interval. Denote by N_{begin} , $N_{\text{end}}^{(k)}$ a pair of stochastic variables with as joint distribution the stationary joint distribution of $N_{\text{begin}}^{(k)}$, $N_{\text{end}}^{(k)}$.

Lemma 9.2.2 relates the distribution of N_I to the distribution of N_{begin} and N_{end} . Lemma 9.2.2

$$\Pr\{\mathbf{N}_{I} = l\} = \frac{\Pr\{\mathbf{N}_{\text{begin}} \le l\} - \Pr\{\mathbf{N}_{\text{end}} \le l\}}{\mathrm{EN}_{\text{end}} - \mathrm{EN}_{\text{begin}}}.$$
(9.6)

Written in terms of pgf's,

$$\mathbf{E}(z^{\mathbf{N}_{I}}) = \frac{\mathbf{E}(z^{\mathbf{N}_{begin}}) - \mathbf{E}(z^{\mathbf{N}_{end}})}{(1-z)(\mathbf{E}\mathbf{N}_{end} - \mathbf{E}\mathbf{N}_{begin})}.$$
(9.7)

Proof

The proof is completely similar to the proof of Lemma 2.2.1. Note that the proof of Lemma 2.2.1 does not rely on the assumption of a single server as implicitly made in Chapter 2.

To illustrate Lemma 9.2.2, we now give two simple examples.

Example 9.2.1

Consider an exhaustive vacation system, where the servers together go on vacation when the system is empty. Then $E(z^{N_{begin}}) = 1$. Let V be the length of an arbitrary vacation. Let $v(\omega) = E(e^{-\omega V})$ for $\operatorname{Re} \omega \ge 0$. If the system is still empty when the servers return from vacation, then in the multiple-vacation case the servers again go on vacation, i.e., $E(z^{N_{end}}) = \frac{v(\lambda(1-z)) - v(\lambda)}{1 - v(\lambda)}$, whereas in the single-vacation case the servers just remain idling, awaiting a customer to arrive, i.e., $E(z^{N_{end}}) = v(\lambda(1-z)) - v(\lambda)(1-z)$. Summarizing, in the multiple-vacation case

$$\mathbf{E}(z^{\mathbf{N}_I}) = \frac{1 - v(\lambda(1-z))}{(1-z)\lambda \mathbf{E}\mathbf{V}},\tag{9.8}$$

whereas in the single-vacation case

$$E(z^{N_I}) = \frac{1 - v(\lambda(1-z)) + v(\lambda)(1-z)}{(1-z)(\lambda E V + v(\lambda))}.$$
(9.9)

Browne & Kella [57] analyze an exhaustive $M/M/\infty$ system with multiple vacations. By a direct method, they find $E(z^N) = H_V(z)/H_V(1)$, with

$$H_{\mathbf{V}}(z) = \frac{1 - v(\lambda(1-z))}{(1-z)\lambda E\mathbf{V}} + \rho \int_{u=0}^{z} \frac{1 - v(\lambda(1-u))}{(1-u)\lambda E\mathbf{V}} e^{(z-u)\rho} du,$$

which agrees with (9.5), (9.8).

Example 9.2.2

Consider an exhaustive system with a stochastic K-policy, cf. Bisdikian [19], where service is interrupted when the system is empty, while service is resumed as soon as a stochastic number of K customers have accumulated again. Then

$$E(z^{N_I}) = \frac{1 - E(z^K)}{(1 - z)EK}.$$
(9.10)

Browne & Kella [57] also study an exhaustive $M/M/\infty$ system with a deterministic K-policy. By a direct method, they find $E(z^N) = H_K(z)/H_K(1)$, with

$$H_K(z) = \frac{1 - z^K}{1 - z} + \rho \sum_{k=0}^{K-1} \int_{u=0}^{z} u^k e^{(z - u)\rho} du,$$

which agrees with (9.5), (9.10).

We now show how the queue length decomposition translates into a decomposition of the waiting time. In addition to (i) and (ii), we make the following assumptions.

(iii). Customers enter service in order of arrival.

(iv). The waiting time of customers is independent of arrivals after their own arrival.

Denote by **W** and **R** the waiting and the sojourn time, respectively, of an arbitrary customer. Denote by **L** the number of waiting customers at an arbitrary epoch. The familiar relationship $E(z^{N}) = E(e^{-\lambda(1-z)\mathbf{R}})$ does not hold here, as customers do not necessarily leave in order of arrival. However, what *does* hold under the assumptions (iii) and (iv), is the relationship $E(z^{L}) = E(e^{-\lambda(1-z)\mathbf{W}})$. What thus remains to be done, is to relate the distribution of **L** to the distribution of **N**.

Lemma 9.2.3

$$\mathbf{E}(z^{\mathbf{L}}) = \tag{9.11}$$

Chapter 9 Polling systems with multiple coupled servers

$$\mathbf{E}(z^{\mathbf{N}}) + (1 - p_I) \left([z^{-m} - 1] \mathbf{E}(z^{\mathbf{N}_E}) + \sum_{k=0}^{m-1} [1 - z^{k-m}] \Pr\{\mathbf{N}_E = k\} \right),\$$

with

$$E(z^{N_E}) = \frac{E(z^N) - p_I E(z^{N_I})}{1 - p_I},$$

$$p_I = \left[\sum_{l=0}^{\infty} \frac{\Pr\{N_I = l\}}{\Pr\{N_{M/M/m}^{(l)} = l\}}\right]^{-1}.$$
(9.12)

Proof

Denote by N_E the number of customers present at an arbitrary epoch in a serving interval. Denote by p_I the fraction of time occupied by non-serving intervals. Then

$$E(z^{N}) = E(z^{N_{I}})p_{I} + E(z^{N_{E}})(1 - p_{I}), \qquad (9.13)$$

$$E(z^{L}) = E(z^{N_{I}})p_{I} + E(z^{[N_{E}-m]^{+}})(1-p_{I}), \qquad (9.14)$$

with $[x]^+ = \max(0, x)$. Comparing (9.13), (9.14), using that

$$E(z^{[N_E-m]^+}) = z^{-m}E(z^{N_E}) + \sum_{k=0}^{m-1} [1 - z^{k-m}] Pr\{N_E = k\}$$
(9.15)

yields (9.11).

Because of the PASTA property, p_I equals the probability that an arbitrary customer arrives in a non-serving interval. In the proof of Lemma 9.2.1, we introduced the notion of a fundamental period. We showed that in a fundamental period exactly 1 customer is served that arrived in a non-serving interval. Denote by **M** the number of customers served in a fundamental period. Then

$$p_I = \frac{1}{\mathrm{E}\mathbf{M}}.\tag{9.16}$$

From the proof of Lemma 9.2.1,

$$\mathbf{EM} = \sum_{l=0}^{\infty} \frac{\Pr\{\mathbf{N}_{I} = l\}}{\Pr\{\mathbf{N}_{M/M/m}^{(l)} = l\}}.$$
(9.17)

Substituting (9.17) into (9.16) completes the proof.

9.3 MODEL DESCRIPTION

We now return to the polling system with multiple coupled servers. We first present a detailed model description. The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by m coupled servers. For the specification of the arrival, service, and switch-over processes, we refer to the description of the 'basic model' in Section 1.3.

The server pool visits the queues in a strictly cyclic order, Q_1, \ldots, Q_n . As soon as the servers arrive at Q_i , they start serving type-*i* customers, as prescribed by the service discipline. For now, we do not specify the service discipline any further. In fact, what we are mainly interested in, is exploring the class of service disciplines that allow an exact analysis. As soon as the servers have finished serving type-*i* customers, as prescribed by the service discipline, they move to Q_{i+1} .

9.4 The joint queue length distribution I

In this section, we relate the pgf of the joint queue length distribution at the beginning of a visit to Q_i to the pgf of the joint queue length distribution at the end of a visit to Q_{i-1} . Next, we also relate the pgf of the joint queue length distribution at the end of a visit to Q_i to the pgf of the joint queue length distribution at the *end* of a visit to Q_i . Thus, we obtain 2n equations involving 2n pgf's. In the next section, we identify some cases in which these pgf's can actually be solved from these equations.

We first introduce some notation. Denote by \mathbf{X}_{ih} and \mathbf{Y}_{ih} stochastic variables with distribution the stationary queue length distribution at Q_h at, respectively, the beginning and the end of a visit to Q_i , $h = 1, \ldots, n$, $i = 1, \ldots, n$. Define

$$\begin{split} F_i(z) &:= \mathrm{E}(z_1^{\mathbf{X}_{i1}} \dots z_n^{\mathbf{X}_{in}}) \\ G_i(z) &:= \mathrm{E}(z_1^{\mathbf{Y}_{i1}} \dots z_n^{\mathbf{Y}_{in}}) \end{split}$$

for $z = (z_1, \ldots, z_n)$, $|z_h| \le 1$, $h = 1, \ldots, n$, $i = 1, \ldots, n$. We first relate $F_i(\cdot)$ to $G_{i-1}(\cdot)$, and subsequently $G_i(\cdot)$ to $F_i(\cdot)$. Thus, we obtain an expression for $G_i(\cdot)$ in terms of $G_{i-1}(\cdot)$, which recursively yields a functional equation for $G_i(\cdot)$.

Define

$$d_i(z) := \sigma_i(\sum_{h=1}^n \lambda_h (1 - z_h))$$
(9.18)

for $z = (z_1, ..., z_n)$, $|z_h| \le 1$, h = 1, ..., n, i = 1, ..., n. Then

$$F_i(z) = G_{i-1}(z)d_{i-1}(z), (9.19)$$

where $d_0(\cdot)$ and $G_0(\cdot)$ are to be understood as $d_n(\cdot)$ and $G_n(\cdot)$, respectively.

Remark 9.4.1

In accordance with the model description, we assume here that the switchover times are non-zero, and that the servers keep switching when the system is empty. In case the switch-over times are zero, or in case the servers stop switching when the system is empty, (9.19) should be modified into

$$F_i(z) = G_{i-1}(z)d_{i-1}(z) + G_{i-1}(0)d_{i-1}(z)[e_i(z) - 1]$$

with

$$e_i(z) = \sum_{h \neq i} \frac{\lambda_h}{\lambda} z_h + \frac{\lambda_i}{\lambda} \theta_i(z).$$

The function $\theta_i(z)$ depends on what happens when an arriving type-*i* customer sees the servers idling at Q_i .

We now relate $G_i(\cdot)$ to $F_i(\cdot)$.

$$G_{i}(z) = \sum_{l_{1}=0}^{\infty} \dots \sum_{l_{n}=0}^{\infty} E(z_{1}^{\mathbf{Y}_{i_{1}}} \dots z_{n}^{\mathbf{Y}_{i_{n}}} | (\mathbf{X}_{i_{1}}, \dots, \mathbf{X}_{i_{n}}) = (l_{1}, \dots, l_{n}))$$
(9.20)
$$\Pr\{(\mathbf{X}_{i_{1}}, \dots, \mathbf{X}_{i_{n}}) = (l_{1}, \dots, l_{n})\}.$$

Evidently, it is the service discipline at Q_i that decides whether or not the righthand side of (9.20) can be expressed into $F_i(\cdot)$. As discussed in Section 1.4, Fuhrmann [99] and Resing [159] consider the class of service disciplines (in single-server systems) that satisfy the following property:

Property 9.4.1

If there are k_i customers present at Q_i at the start of a visit, then during the course of the visit each of these k_i customers will effectively be replaced in an i.i.d. manner by a random population having pgf $h_i(z)$, which may be any *n*-dimensional pgf.

Formally,

$$E(z_1^{\mathbf{Y}_{i_1}} \dots z_n^{\mathbf{Y}_{i_n}} \mid (\mathbf{X}_{i_1}, \dots, \mathbf{X}_{i_n}) = (l_1, \dots, l_n)) =$$

$$z_1^{l_1} \dots z_{i-1}^{l_{i-1}} (\eta_i(z))^{l_i} z_{i+1}^{l_{i+1}} \dots z_n^{l_n}.$$
(9.21)

Substituting (9.21) into (9.20),

$$G_i(z) = F_i(z_1, \dots, z_{i-1}, \eta_i(z), z_{i+1}, \dots, z_n).$$
(9.22)

Using the theory of multi-type branching processes, both Fuhrmann and Resing show that the class of service disciplines that satisfy Property 9.4.1, like exhaustive and gated, allows a relatively simple exact analysis, basically due to

the relatively simple form of (9.22). The results suggest that service disciplines that violate Property 9.4.1 defy an exact analysis, except for some special cases, like two-queue cases and completely symmetric cases.

In multiple-server systems, there are no non-trivial service disciplines that satisfy Property 9.4.1. However, some service disciplines *do* satisfy the following somewhat milder property than Property 9.4.1:

Property 9.4.2

If there are k_i customers present at Q_i at the start of a visit, then during the course of the visit one of these k_i customers will effectively be replaced by a random population having pgf $\eta_i^{(1)}(z)$, while each of the other customers will effectively be replaced in an i.i.d. manner by a random population having pgf $\eta_i(z)$. Formally,

$$E(z_1^{\mathbf{Y}_{i_1}} \dots z_n^{\mathbf{Y}_{i_n}} \mid (\mathbf{X}_{i_1}, \dots, \mathbf{X}_{i_n}) = (l_1, \dots, l_n)) =$$

$$z_1^{l_1} \dots z_{i-1}^{l_{i-1}} \eta_i^{(l_i)}(z) z_{i+1}^{l_{i+1}} \dots z_n^{l_n},$$
(9.23)

with

$$\eta_i^{(l_i)}(z) = 1, \qquad l_i = 0,$$
(9.24)

$$\eta_i^{(l_i)}(z) = \eta_i^{(1)}(z)(\eta_i(z))^{l_i-1}, \qquad l_i > 0.$$
(9.25)

Below we will describe some multiple-server systems with service disciplines that satisfy Property 9.4.2. We will then also briefly indicate some circumstances that may occur in single-server systems, under the influence of which in principle simple service disciplines violate Property 9.4.1, but still satisfy Property 9.4.2.

Substituting (9.23), (9.24), (9.25) into (9.20),

$$G_{i}(z) = F_{i}(z_{1}, \dots, z_{i-1}, \eta_{i}(z), z_{i+1}, \dots, z_{n}) \frac{\eta_{i}^{(1)}(z)}{\eta_{i}(z)}$$

$$+ F_{i}(z_{1}, \dots, z_{i-1}, 0, z_{i+1}, \dots, z_{n}) [1 - \frac{\eta_{i}^{(1)}(z)}{\eta_{i}(z)}].$$
(9.26)

Define

$$a_i(z) := (z_1, \dots, z_{i-1}, \eta_i(z), z_{i+1}, \dots, z_n);$$
(9.27)

$$b_i(z) := (z_1, \dots, z_{i-1}, 0, z_{i+1}, \dots, z_n);$$
(9.28)

$$c_i(z) := \frac{\eta_i^{(1)}(z)}{\eta_i(z)}$$
(9.29)

for $z = (z_1, ..., z_n)$, $|z_h| \le 1$, h = 1, ..., n, i = 1, ..., n. Then (9.26) may be written as

$$G_i(z) = F_i(a_i(z))c_i(z) + F_i(b_i(z))[1 - c_i(z)].$$
(9.30)

In view of the results of Fuhrmann and Resing, one can in general not expect that the class of service disciplines that satisfy Property 9.4.2, but not Property 9.4.1, i.e., with $c_i(z) \neq 1$, allows an exact analysis, except possibly for some special cases. In the next section, we will identify some of those cases.

We now describe some multiple-server systems with service disciplines that satisfy Property 9.4.2. Property 9.4.2 says that during the course of a visit to Q_i , one of the customers initially present gets replaced by a different population than all the others. This suggests that either only one of the customers initially present at Q_i actually gets served, or that all of them get served, but that one of them keeps the servers busy for a different time than all the others. Keeping this in mind, we consider a class of service disciplines that are parametrized by two vectors (p_1, \ldots, p_n) and (q_1, \ldots, q_n) with the following interpretation. If there are any customers present at Q_i at the start of a visit, then one of them is served anyway, while the others are served with probability q_i . Customers arriving at Q_i during the course of a visit are served with probability p_i . The case $q_i = 0$ contains both the semi-exhaustive service discipline $(p_i = 1)$ and the 1-limited service discipline $(p_i = 0)$. The case $q_i = 1$ includes both the exhaustive service discipline $(p_i = 1)$ and the gated service discipline $(p_i = 0)$. Denote $\kappa_i := \lambda_i p_i$.

Let $\mathbf{T}_{i}^{(k)}$ be the length of a busy period starting with k customers present in an ordinary M/G/m queue with arrival rate κ_i and service time distribution $B_i(\cdot)$. Let $\tau_i^{(k)}(\omega) = \mathrm{E}(\mathrm{e}^{-\omega \mathbf{T}_i^{(k)}})$ for $\mathrm{Re}\,\omega \geq 0$. Define

$$\alpha_i(z) := \sum_{h \neq i} \lambda_h (1 - z_h) + \lambda_i (1 - p_i) (1 - z_i)$$

for $z = (z_1, \ldots, z_n)$, $|z_h| \le 1$, $h = 1, \ldots, n$, $i = 1, \ldots, n$. For the above-defined class of service disciplines,

$$\eta_i^{(l_i)}(z) = \sum_{k_i=1}^{l_i} \begin{pmatrix} l_i - 1\\ k_i - 1 \end{pmatrix} q_i^{k_i - 1} (1 - q_i)^{l_i - k_i} \tau_i^{(k_i)}(\alpha_i(z)) z_i^{l_i - k_i}, \qquad (9.31)$$

with the interpretation of $\eta_i^{(l_i)}(z)$ as in (9.25). For $q_i = 0$, (9.31) satisfies (9.24) and (9.25) in Property 9.4.2 with $\eta_i(z) = z_i$, $\eta_i^{(1)}(z) = \tau_i^{(1)}(\alpha_i(z))$. If $\tau_i^{(k_i)}(\cdot)$ is of the form

$$\tau_i^{(k_i)}(\omega) = 1, \qquad k_i = 0,$$
(9.32)

$$\tau_i^{(k_i)}(\omega) = \tau_i^{(1)}(\omega)(\tau_i(\omega))^{k_i - 1}, \qquad k_i > 0,$$
(9.33)

for some LST $\tau_i(\cdot)$, then (9.31) satisfies (9.24) and (9.25) also for $q_i > 0$, with $\eta_i(z) = q_i \tau_i(\alpha_i(z)) + (1 - q_i)z_i, \ \eta_i^{(1)}(z) = \tau_i^{(1)}(\alpha_i(z)).$ We now give two examples where $\tau_i^{(k_i)}(\cdot)$ is of the form (9.32), (9.33).

Example 9.4.1

Assume that the service times at Q_i are exponentially distributed with parameter $\mu_i = 1/\beta_i$. Let $\mathbf{U}_i^{(k,l)}$ be the time needed in an M/M/m queue with arrival rate κ_i and service rate μ_i to reduce the queue length from k to $l, k \ge l$. Let $\phi_i^{(k,l)}(\omega) = \mathbf{E}(e^{-\omega \mathbf{U}_i^{(k,l)}})$ for $\operatorname{Re} \omega \ge 0, k \ge l$. So $\tau_i^{(k_i)}(\omega) = \phi_i^{(k_i,0)}(\omega)$.

Let \mathbf{U}_i be the length of a busy period in an M/M/1 queue with arrival rate κ_i and service rate $m\mu_i$. Let $\phi_i(\omega) = \mathbf{E}(e^{-\omega \mathbf{U}_i})$ for $\operatorname{Re} \omega \ge 0$. Because of the properties of the exponential distribution,

$$\phi_i^{(k,0)}(\omega) = \prod_{l=1}^k \phi_i^{(l,l-1)}(\omega),$$

$$\phi_i^{(l,l-1)}(\omega) = \phi_i(\omega), \qquad l \ge m.$$

The transforms $\phi_i(\cdot), \phi_i^{(1,0)}(\cdot), \ldots, \phi_i^{(m-1,m-2)}(\cdot)$ may be determined by solving a set of linear equations, cf. Medhi [149] pp. 138-140. In particular,

$$\phi_i(\omega) = \frac{\kappa_i + \omega + m\mu_i - \sqrt{(\kappa_i + \omega + m\mu_i)^2 - 4m\kappa_i\mu_i}}{2\kappa_i}$$
$$\phi_i^{(m-1,m-2)}(\omega) = \frac{(m-1)\phi_i(\omega)}{m - \phi_i(\omega)}.$$

For the gated service discipline, i.e., $p_i = 0$,

$$\phi_i(\omega) = \frac{m\mu_i}{m\mu_i + \omega},$$

$$\phi_i^{(l,l-1)}(\omega) = \frac{l\mu_i}{l\mu_i + \omega}, \qquad l \le m.$$

Summarizing, for m = 1, $\tau_i^{(k_i)}(\cdot)$ is of the form (9.32), (9.33) with $\tau_i(\omega) = \tau_i^{(1)}(\omega) = \phi_i(\omega)$. Also, for m = 2, $\tau_i^{(k_i)}(\cdot)$ is of the form (9.32), (9.33) with $\tau_i(\omega) = \phi_i(\omega)$, $\tau_i^{(1)}(\omega) = \phi_i^{(1,0)}(\omega)$, $\phi_i^{(1,0)}(\omega) = \frac{\phi_i(\omega)}{2 - \phi_i(\omega)}$. For m > 2, $\tau_i^{(k_i)}(\cdot)$ is no longer of the form (9.32), (9.33).

Example 9.4.2

Assume that the service times at Q_i are deterministic. Let \mathbf{V}_i be the length of a busy period in an ordinary M/D/m queue with arrival rate κ_i and service time β_i . Let $\psi_i(\omega) = \mathrm{E}(\mathrm{e}^{-\omega \mathbf{V}_i})$ for $\mathrm{Re} \ \omega \ge 0$. For $m = \infty, \tau_i^{(k_i)}(\cdot)$ is of the form (9.32), (9.33) with $\tau_i(\omega) = 1, \tau_i^{(1)}(\omega) = \psi_i(\omega)$, $\psi_i(\omega) = \frac{(\omega + \lambda_i)\mathrm{e}^{-\beta_i(\omega + \lambda_i)}}{\omega + \lambda_i \mathrm{e}^{-\beta_i(\omega + \lambda_i)}}$, cf. Stadje [170].

Remark 9.4.2

There are also some circumstances that may occur in single-server systems, under the influence of which in principle simple service disciplines violate Property 9.4.1, but still satisfy Property 9.4.2. An obvious example arises when one of the customers served during a visit, e.g. the first one or the last one, requires an exceptional service time. Another example is provided by a set-up time or a shut-down time that is only incurred when at least one customer is served during a visit.

9.5 The joint queue length distribution II

In the previous section, we obtained under assumption of Property 9.4.2 a set of 2n equations (9.19), (9.30) involving the 2n pgf's $F_i(z)$, $G_i(z)$, i = 1, ..., n. In this section, we identify some cases in which these pgf's can actually be solved from these equations. Obviously, it suffices to find either $F_i(z)$ or $G_i(z)$ for an arbitrary *i*, as the remaining $F_i(z)$, $G_i(z)$, i = 1, ..., n, can then easily be found from (9.19), (9.30).

Substituting (9.19) into (9.30),

$$G_i(z) = G_{i-1}(a_i(z))d_{i-1}(a_i(z))c_i(z)$$
(9.34)

+
$$G_{i-1}(b_i(z))d_{i-1}(b_i(z))[1-c_i(z)],$$
 (9.35)

where $d_0(\cdot)$ and $G_0(\cdot)$ are to be understood as $d_n(\cdot)$ and $G_n(\cdot)$, respectively. Applying (9.34) *n* times, we obtain a functional equation for $G_i(\cdot)$. For n = 1 we find, using the definitions (9.18), (9.27), (9.28), (9.29),

$$G(z) = G(\eta(z))\sigma(\lambda(1-\eta(z)))\frac{\eta^{(1)}(z)}{\eta(z)} + G(0)\sigma(\lambda)[1-\frac{\eta^{(1)}(z)}{\eta(z)}].$$
(9.36)

Here (as well as in the sequel) the redundant indices are omitted.

For n = 2 we find

$$\begin{split} G_{i}(z) &= \\ G_{i}(a_{i-1}(a_{i}(z)))d_{i-1}(a_{i-1}(a_{i}(z)))c_{i-1}(a_{i}(z))d_{i}(a_{i}(z))c_{i}(z) + \\ G_{i}(b_{i-1}(a_{i}(z)))d_{i-1}(b_{i-1}(a_{i}(z)))[1 - c_{i-1}(a_{i}(z))]d_{i}(a_{i}(z))c_{i}(z) + \\ G_{i}(a_{i-1}(b_{i}(z)))d_{i-1}(a_{i-1}(b_{i}(z)))c_{i-1}(b_{i}(z))d_{i}(b_{i}(z))[1 - c_{i}(z)] + \\ G_{i}(b_{i-1}(b_{i}(z)))d_{i-1}(b_{i-1}(b_{i}(z)))[1 - c_{i-1}(b_{i}(z))]d_{i}(b_{i}(z))[1 - c_{i}(z)]. \end{split}$$

Using the definitions (9.18), (9.27), (9.28), (9.29),

$$G_{1}(z_{1}, z_{2}) = (9.37)$$

$$G_{1}(\eta_{1}(z), \eta_{2}(\eta_{1}(z), z_{2}))\sigma_{2}\{\gamma(\eta_{1}(z), \eta_{2}(\eta_{1}(z), z_{2}))\}\sigma_{1}\{\gamma(\eta_{1}(z), z_{2})\}$$

$$\times \frac{\eta_{2}^{(1)}(\eta_{1}(z), z_{2})}{\eta_{2}(\eta_{1}(z), z_{2})}\frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)} +$$

$$G_{1}(\eta_{1}(z), 0)\sigma_{2}\{\gamma(\eta_{1}(z), 0)\}\sigma_{1}\{\gamma(\eta_{1}(z), z_{2})\}[1 - \frac{\eta_{2}^{(1)}(\eta_{1}(z), z_{2})}{\eta_{2}(\eta_{1}(z), z_{2})}]\frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)} +$$

$$G_{1}(0, \eta_{2}(0, z_{2}))\sigma_{2}\{\gamma(0, \eta_{2}(0, z_{2}))\}\sigma_{1}\{\gamma(0, z_{2})\}\frac{\eta_{2}^{(1)}(0, z_{2})}{\eta_{2}(0, z_{2})}[1 - \frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)}] +$$

$$G_{1}(0, 0)\sigma_{2}\{\gamma(0, 0)\}\sigma_{1}\{\gamma(0, z_{2})\}[1 - \frac{\eta_{2}^{(1)}(0, z_{2})}{\eta_{2}(0, z_{2})}][1 - \frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)}],$$

(similarly with the indices interchanged) with $\gamma(z_1, z_2) = \lambda_1(1-z_1) + \lambda_2(1-z_2)$. Remember that $\eta_i(z)$ is an *n*-dimensional pgf so that $|\eta_i(z)| \leq 1$ for $z = (z_1, \ldots, z_n), |z_h| \leq 1, h = 1, \ldots, n, i = 1, \ldots, n$.

In general, we obtain a functional equation for $G_i(\cdot)$ containing 2^n arguments in the right-hand side. So, in accordance with the results of Fuhrmann and Resing, in general the functional equation cannot be solved. In fact, solving the functional equation only stands a chance in cases where 'enough' of the 2^n arguments in the right-hand side reduce either to z or to a constant. We will now indicate some of those cases.

Case I. n = 1 queue, $\eta(z) = z$.

This covers the case q = 0 described in the previous section, i.e., only one of the customers present at the start of a visit is served, while customers arriving during the course of a visit are served with probability p. Rewriting (9.36),

$$G(z)[z - \sigma(\lambda(1-z))\eta^{(1)}(z)] = G(0)\sigma(\lambda)[z - \eta^{(1)}(z)].$$
(9.38)

Letting $z \to 1$ in (9.38),

$$G(0) = \frac{1}{\sigma(\lambda)} \frac{1 - (\eta^{(1)})'(1) - \lambda s}{1 - (\eta^{(1)})'(1)},$$

with $(\eta^{(1)})'(1) = \frac{d\eta^{(1)}(z)}{dz}$. Apparently, the stability condition is $\lambda s + (\eta^{(1)})'(1) < 1$. Note that $\lambda s + (\eta^{(1)})'(1)$ is the mean increase of the queue length between the start of two successive visits when the system is not empty, which should indeed be less than 1 to ensure stability.

Case II. n = 1 queue, $\eta(z) \neq z$.

This covers the case q > 0 described in the previous section, i.e., one of the customers present at the start of a visit is served anyway, the others are served with probability q > 0, while customers arriving during the course of a visit are served with probability p, moreover assuming that there are either two servers and exponential service times, cf. Example 9.4.1, or an infinite number of servers and deterministic service times, cf. Example 9.4.2.

Writing $e(z) = \eta(z)$, $f(z) = \sigma(\lambda(1 - \eta(z)))\eta^{(1)}(z)/\eta(z)$, $g(z) = \sigma(\lambda)[1 - \eta^{(1)}(z)/\eta(z)]$ in (9.36),

$$G(z) = G(e(z))f(z) + G(0)g(z).$$
(9.39)

Define

$$e^{(0)}(z) := z;$$

 $e^{(k)}(z) := e(e^{(k-1)}(z)); \qquad k = 1, 2, ...,$

for $|z| \leq 1$.

Iterating (9.39) K times,

$$G(z) = G(e^{(K+1)}(z)) \prod_{k=0}^{K} f(e^{(k)}(z))$$

$$+ G(0) \sum_{k=0}^{K} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z)).$$
(9.40)

The next lemma establishes the convergence of (9.40) for $K \to \infty$ under the condition $\eta'(1) < 1$.

Lemma 9.5.1

If
$$\eta'(1) < 1$$
, then
i. $\lim_{K \to \infty} e^{(K+1)}(z) = 1$ for all z with $|z| \le 1$;
ii. $\prod_{k=0}^{\infty} f(e^{(k)}(z))$ converges for all z with $|z| \le 1$;
iii. $\sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))$ converges for all z with $|z| \le 1$.

9.5 The joint queue length distribution II

Proof See Appendix 9.B.

Apparently, the stability condition is $\eta'(1) < 1$. Note that $\eta'(1)$ is the mean number of customers by which each of the customers present at the start of a visit, except one, gets replaced in the course of the visit, which should indeed be less than 1 to ensure stability. In Example 9.4.1, $\eta'(1) = \frac{(1-p)\rho}{2-p\rho} < 1$ iff $\rho < 2$, irrespective of p. In Example 9.4.2, $\eta'(1) = 0$, also irrespective of p. If $\eta'(1) < 1$ then, letting $K \to \infty$ in (9.40),

$$G(z) = \prod_{k=0}^{\infty} f(e^{(k)}(z)) + G(0) \sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z)).$$
(9.41)

Putting z = 0 in (9.41),

$$G(0) = \frac{\prod_{k=0}^{\infty} f(e^{(k)}(0))}{1 - \sum_{k=0}^{\infty} g(e^{(k)}(0)) \prod_{l=0}^{k-1} f(e^{(l)}(0))}$$

Case III. n = 2 queues, $\eta_i(z) = z_i$, i = 1, 2.

This covers the case $q_i = 0$ described in the previous section, i.e., only one of the customers present at Q_i at the start of a visit is served, while customers arriving at Q_i during the course of a visit are served with probability p_i , i = 1, 2. Equation (9.37) reduces to

$$\begin{aligned} G_1(z_1, z_2)[z_1 z_2 - \sigma_1(\gamma(z))\sigma_2(\gamma(z))\eta_1^{(1)}(z)\eta_2^{(1)}(z)] &= \\ G_1(z_1, 0)\sigma_1(\gamma(z))\sigma_2(\gamma(z_1, 0))\eta_1^{(1)}(z)[z_2 - \eta_2^{(1)}(z)] + \\ G_1(0, z_2)\sigma_1(\gamma(0, z_2))\sigma_2(\gamma(0, z_2))[z_1 - \eta_1^{(1)}(z)]\eta_2^{(1)}(0, z_2) + \\ G_1(0, 0)\sigma_1(\gamma(0, z_2))\sigma_2(\gamma(0))[z_1 - \eta_1^{(1)}(z)][z_2 - \eta_2^{(1)}(0, z_2)] \end{aligned}$$

For $p_i = 0$, i.e., $\eta_i^{(1)}(z) = \beta_i(\gamma(z))$, i = 1, 2, the problem of solving the above functional equation may be formulated as a boundary value problem, cf. Boxma & Groenendijk [42].

Case IV. n = 2 queues, $\eta_i(z)$, $\eta_i^{(1)}(z)$ do not depend on z_i , i = 1, 2. This occurs in Example 9.4.1 for $p_i = 1$, i = 1, 2, i.e., two servers, exponential service times, exhaustive service.

If $\eta_i(z)$, $\eta_i^{(1)}(z)$ do not depend on z_i , i = 1, 2, then the complete right-hand side of (9.37) does not depend on z_i . In other words, $G_i(z)$ does not depend on

 z_i , reflecting that Q_i is empty at the completion of a visit to Q_i when $p_i = 1$. So, equation (9.37) may be replaced by

$$H_1(z_2) = H_1(e_1(z_2))f_1(z_2) + H_1(\eta_2(0))g_1(z_2) + H_1(0)h_1(z_2), \qquad (9.42)$$

with

$$\begin{array}{lll} e_{1}(z_{2}) &:= & \eta_{2}(\eta_{1}(z), z_{2}); \\ f_{1}(z_{2}) &:= & \sigma_{2}\{\gamma(\eta_{1}(z), \eta_{2}(\eta_{1}(z), z_{2}))\}\sigma_{1}\{\gamma(\eta_{1}(z), z_{2}), \frac{\eta_{1}^{(1)}(z)}{\eta_{2}(\eta_{1}(z), z_{2})}, \frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)}; \\ g_{1}(z_{2}) &:= & \sigma_{2}\{\gamma(0, \eta_{2}(0, z_{2}))\}\sigma_{1}\{\gamma(0, z_{2})\}\frac{\eta_{2}^{(1)}(0, z_{2})}{\eta_{2}(0, z_{2})}[1 - \frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)}]; \\ h_{1}(z_{2}) &:= & \sigma_{2}\{\gamma(\eta_{1}(z), 0)\}\sigma_{1}\{\gamma(\eta_{1}(z), z_{2})\}[1 - \frac{\eta_{2}^{(1)}(\eta_{1}(z), z_{2})}{\eta_{2}(\eta_{1}(z), z_{2})}]\frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)} \\ & + & \sigma_{2}\{\gamma(0, 0)\}\sigma_{1}\{\gamma(0, z_{2})\}[1 - \frac{\eta_{2}^{(1)}(0, z_{2})}{\eta_{2}(0, z_{2})}][1 - \frac{\eta_{1}^{(1)}(z)}{\eta_{1}(z)}]; \\ H_{1}(z_{2}) &:= & G_{1}(z_{1}, z_{2}). \end{array}$$

Define

$$e_1^{(0)}(y) := y;$$

 $e_1^{(k)}(y) := e_1(e_1^{(k-1)}(y)); \qquad k = 1, 2, \dots,$

for $|y| \leq 1$.

Iterating (9.42) K times, writing $z_2 = y$,

$$H_{1}(y) = H_{1}(e_{1}^{(K+1)}(y)) \prod_{k=0}^{K} f_{1}(e_{1}^{(k)}(y))$$

$$+ H_{1}(\eta_{2}(0)) \sum_{k=0}^{K} g_{1}(e_{1}^{(k)}(y)) \prod_{l=0}^{k-1} f_{1}(e_{1}^{(l)}(y))$$

$$+ H_{1}(0) \sum_{k=0}^{K} h_{1}(e_{1}^{(k)}(y)) \prod_{l=0}^{k-1} f_{1}(e_{1}^{(l)}(y)).$$
(9.43)

The next lemma establishes the convergence of (9.43) for $K \to \infty$ under the condition $e_1'(1) < 1$.

Lemma 9.5.2

If
$$e'_{1}(1) < 1$$
, then
i. $\lim_{K \to \infty} e_{1}^{(K+1)}(y) = 1$ for all y with $|y| \le 1$;
ii. $\prod_{k=0}^{\infty} f_{1}(e_{1}^{(k)}(y))$ converges for all y with $|y| \le 1$;

iii.
$$\sum_{k=0}^{\infty} g_1(e_1^{(k)}(y)) \prod_{l=0}^{k-1} f_1(e_1^{(l)}(y)) \text{ converges for all } y \text{ with } |y| \le 1;$$

iv.
$$\sum_{k=0}^{\infty} h_1(e_1^{(k)}(y)) \prod_{l=0}^{k-1} f_1(e_1^{(l)}(y)) \text{ converges for all } y \text{ with } |y| \le 1.$$

Proof

Similar to the proof of Lemma 9.5.1.

Apparently the stability condition is $e'_1(1) < 1$. Note that $e'_1(1)$ is the mean number of type-1 customers by which each of the type-1 customers present at the start of a cycle, except one, gets replaced during the course of the cycle. In Example 9.4.1, when $p_i = 1$, $e'_1(1) = \eta'_1(1) \eta'_2(1) = \frac{\rho_1}{2 - \rho_1} \frac{\rho_2}{2 - \rho_2} = 0$

 $\frac{\rho_1\rho_2}{4-2\rho+\rho_1\rho_2} < 1 \text{ iff } \rho < 2.$ If $e'_1(1) < 1$ then, letting $K \to \infty$ in (9.43),

$$H_{1}(y) = \prod_{k=0}^{\infty} f_{1}(e_{1}^{(k)}(y))$$

$$+ H_{1}(\eta_{2}(0)) \sum_{k=0}^{\infty} g_{1}(e_{1}^{(k)}(y)) \prod_{l=0}^{k-1} f_{1}(e_{1}^{(l)}(y))$$

$$+ H_{1}(0) \sum_{k=0}^{\infty} h_{1}(e_{1}^{(k)}(y)) \prod_{l=0}^{k-1} f_{1}(e_{1}^{(l)}(y)).$$
(9.44)

Putting y = 0 and $y = \eta_2(0)$ in (9.44), we obtain a pair of linear equations for the unknown constants $H_1(0)$ and $H_1(\eta_2(0))$.

Case V. n = 2 queues, $\eta_i(z) = 1$, i = 1, 2. This occurs in Example 9.4.2, i.e., an infinite number of servers, deterministic service times, all the customers present at the start of a visit are served, while customers arriving during the course of a visit are served with probability p_i , i = 1, 2.

Equation (9.37) reduces to

$$G_{1}(z_{1}, z_{2}) =$$

$$\sigma_{1}(\lambda_{2}(1 - z_{2}))\eta_{2}^{(1)}(1, z_{2})\eta_{1}^{(1)}(z) +$$

$$G_{1}(1, 0)\sigma_{2}(\lambda_{2})\sigma_{1}(\lambda_{2}(1 - z_{2}))[1 - \eta_{2}^{(1)}(1, z_{2})]\eta_{1}^{(1)}(z) +$$

$$G_{1}(0, 1)\sigma_{2}(\lambda_{1})\sigma_{1}(\lambda_{1} + \lambda_{2}(1 - z_{2}))\eta_{2}^{(1)}(0, z_{2})[1 - \eta_{1}^{(1)}(z)] +$$

$$G_{1}(0, 0)\sigma_{2}(\lambda_{1} + \lambda_{2})\sigma_{1}(\lambda_{1} + \lambda_{2}(1 - z_{2}))[1 - \eta_{2}^{(1)}(0, z_{2})][1 - \eta_{1}^{(1)}(z)].$$
(9.45)

Putting z = (1,0), z = (0,1), and z = (0,0) in (9.45), we obtain a set of three linear equations for the unknown constants $G_1(1,0)$, $G_1(0,1)$, and $G_1(0,0)$.

Case VI. n = 2 queues, $\eta_1(z) = z_1, \ \eta_2(z) = 1$.

This covers the case $q_1 = 0$, $q_2 = 1$ described in the previous section, i.e., one of the customer present at Q_1 at the start of a visit is served, customers arriving at Q_1 during the course of a visit are served with probability p_1 , all the customers present at Q_2 at the start of a visit are served, customers arriving at Q_2 during the course of a visit are served with probability p_2 , moreover assuming that there are an infinite number of servers and deterministic service times at Q_2 , cf. Example 9.4.2.

Equation (9.37) reduces to

$$G_{1}(z_{1}, z_{2}) =$$

$$G_{1}(z_{1}, 1)\sigma_{2}(\gamma(z_{1}, 1))\sigma_{1}(\gamma(z))\eta_{2}^{(1)}(z)\frac{\eta_{1}^{(1)}(z)}{z_{1}} +$$

$$G_{1}(z_{1}, 0)\sigma_{2}(\gamma(z_{1}, 0))\sigma_{1}(\gamma(z))[1 - \eta_{2}^{(1)}(z)]\frac{\eta_{1}^{(1)}(z)}{z_{1}} +$$

$$G_{1}(0, 1)\sigma_{2}(\gamma(0, 1))\sigma_{1}(\gamma(0, z_{2}))\eta_{2}^{(1)}(0, z_{2})[1 - \frac{\eta_{1}^{(1)}(z)}{z_{1}}] +$$

$$G_{1}(0, 0)\sigma_{2}(\gamma(0, 0))\sigma_{1}(\gamma(0, z_{2}))[1 - \eta_{2}^{(1)}(0, z_{2})][1 - \frac{\eta_{1}^{(1)}(z)}{z_{1}}].$$
(9.46)

Setting $z_2 = 0$ and $z_2 = 1$ in (9.46), we find expressions for $G_1(z_1, 0)$ and $G_1(z_1, 1)$ containing the unknown constants $G_1(0, 0)$ and $G_1(0, 1)$. Putting $z_1 = 0$ in those expressions, we obtain a pair of linear equations for these constants.

Case VII. general n, $\eta_i(z) = 1$, i = 1, ..., n. Similar to Case V.

Case VIII. general n, $\eta_1(z) = z_1$, $\eta_i(z) = 1$, $i \neq 1$. Similar to Case VI.

9.6 Concluding Remarks and suggestions for further research

So far, we have focused on the joint queue length distribution at embedded epochs. In Section 9.4, we have obtained under assumption of Property 9.4.2 a set of 2n equations (9.19), (9.30) for the associated pgf's $F_i(z)$, $G_i(z)$,

i = 1, ..., n. In Section 9.5, we have identified some cases in which these pgf's can actually be solved from these equations. These cases include several singlequeue systems with a varying number of servers, two-queue two-server systems with exhaustive service and exponential service times, as well as infinite-server systems with an arbitrary number of queues, exhaustive or gated service, and deterministic service times.

To conclude, we now briefly discuss the derivation of the marginal queue length distribution at an arbitrary epoch from the joint queue length distribution at embedded epochs. Denote by N_i the queue length at Q_i at an arbitrary epoch. As stated in the introduction, in isolation a particular queue in a polling system may be viewed as a single-queue system with service interruptions, the intervisit periods constituting the service interruptions. In Section 9.2, we have shown how in such a system with service interruptions and exponential service times, the queue length distribution at an arbitrary epoch may be expressed into the queue length distribution at the beginning and the end of a service interruption. In case the assumptions of Section 9.2 are satisfied, one may thus obtain the marginal queue length distribution at Q_i from the queue length distribution at the beginning and the end of a visit to Q_i , given by $E(z^{X_{ii}}) = F_i(1, \ldots, 1, z, 1, \ldots, 1)$ and $E(z^{Y_{ii}}) = G_i(1, \ldots, 1, z, 1, \ldots, 1)$, respectively, with z as *i*-th argument. Consider e.g. the two-queue two-server system with exhaustive service and exponential service times, for which we obtained $F_i(z)$ and $G_i(z)$ in Case IV of the previous section. For such a system, using Lemma 9.2.1 and Lemma 9.2.2,

$$\mathbf{E}(z^{\mathbf{N}_{i}}) = \left[\frac{2}{2-\rho_{i}} + \frac{\rho_{i}}{2-\rho_{i}} \operatorname{Pr}\{\mathbf{N}_{i|I} = 0\}\right]^{-1} \\ \times \left[\frac{2}{2-\rho_{i}z} \mathbf{E}(z^{\mathbf{N}_{i|I}}) + \frac{\rho_{i}z}{2-\rho_{i}z} \operatorname{Pr}\{\mathbf{N}_{i|I} = 0\}\right],$$

with

$$\mathbf{E}(z^{\mathbf{N}_{i|I}}) = \frac{1 - \mathbf{E}(z^{\mathbf{X}_{ii}})}{(1 - z)\mathbf{E}\mathbf{X}_{ii}}$$

In Section 9.2, we have also shown how subsequently the waiting-time distribution may be related to the marginal queue length distribution by using Lemma 9.2.3. In case the assumptions of Section 9.2 are not satisfied, one may quite often still obtain the marginal queue length distribution from the joint queue length distribution at the beginning and the end of a visit by developing ad hoc methods. We do however not pursue the matter any further, leaving it as an interesting topic for further research. APPENDICES

9.A PROOF OF LEMMA 9.2.1

Lemma 9.2.1

 $E(z^N) =$

$$\gamma \left[\sum_{l=0}^{m-2} \frac{\mathrm{E}(z^{\mathbf{N}_{M/M/m}^{(l)}}) \mathrm{Pr}\{\mathbf{N}_{I}=l\}}{\mathrm{Pr}\{\mathbf{N}_{M/M/m}^{(l)}=l\}} + \frac{m}{m-\rho z} \sum_{l=m-1}^{\infty} z^{l} \mathrm{Pr}\{\mathbf{N}_{I}=l\} \right],$$

with

$$\gamma = \left[\sum_{l=0}^{m-2} \frac{\Pr\{\mathbf{N}_{I} = l\}}{\Pr\{\mathbf{N}_{M/M/m}^{(l)} = l\}} + \frac{m}{m-\rho} \sum_{l=m-1}^{\infty} \Pr\{\mathbf{N}_{I} = l\}\right]^{-1}$$

Proof

Define a vacation customer to be a customer arriving in a non-serving interval. Consider now a vacation customer C arriving at some time u. Suppose that C sees l customers upon arrival; so the queue length just after u equals l + 1, $l \geq 0$. Let **T** be the first epoch in a serving interval after u, at which the queue length reaches the level l + 1 again. Let U be the first epoch after u, at which the queue length drops to the level l. Suppose that the interval $[\mathbf{T}, \mathbf{U}]$ contains K distinct non-serving intervals starting at the consecutive epochs $\mathbf{U}_1, \ldots, \mathbf{U}_{\mathbf{K}}, \mathbf{K} \geq 0$. Let \mathbf{N}_k be the queue length just after the epoch \mathbf{U}_k . Let \mathbf{T}_k be the first epoch in a serving interval after \mathbf{U}_k , at which the queue length reaches the level N_k again. The interval [T, U], exclusive of the intervals $[\mathbf{U}_1, \mathbf{T}_1], \ldots, [\mathbf{U}_{\mathbf{K}}, \mathbf{T}_{\mathbf{K}}]$, is called a fundamental period at level *l*. Note that we have thus established a 1-1 correspondence between fundamental periods at level l and vacation customers that see l customers upon arrival. (For m = 1, one can establish the 1-1 correspondence in an elegant way by choosing the order of service to be non-preemptive LCFS, cf. Fuhrmann & Cooper [102]; the vacation customer is then the 'ancestor' of the customers served in the fundamental period. For m > 1, one cannot establish the 1-1 correspondence in such an elegant way, as the customers then do not necessarily leave in order of service.) The notion of a fundamental period is illustrated in Figure 9.1, with N(t) denoting the queue length at time t. Parallel to the time axis, the non-serving intervals are indicated by dotted lines. The serving intervals constituting a fundamental period at level l = 1 are indicated by bold lines.



FIGURE 9.1. A fundamental period at level l = 1.

Consider now an arbitrary tagged customer as it departs from the system. Denote by N_D the number of customers that the tagged customer leaves behind. By virtue of the PASTA property and an up- & down crossing argument, N_D has the same distribution as N. Denote by L_D the level of the fundamental period in which the tagged customer is served. (Note here that the fundamental periods together constitute a partitioning of the serving intervals.)

$$E(z^{N}) = E(z^{N_{D}}) = \sum_{l=0}^{\infty} E(z^{N_{D}} \mid \mathbf{L}_{D} = l) \Pr\{\mathbf{L}_{D} = l\}.$$
(9.47)

Define a fundamental period at level l in the corresponding M/M/m queue to be a period ranging from an epoch when the queue length jumps to the level l + 1 to an epoch when the queue length drops to the level l. Because of the memoryless property of the exponential service time distribution, a fundamental period at level l in the queue with service interruptions is stochastically indistinguishable from a fundamental period at level l in the corresponding M/M/m queue. So, given that $\mathbf{L}_D = l$, \mathbf{N}_D has the same distribution as the number of customers that an arbitrary customer leaves behind as it departs from the corresponding M/M/m queue in a fundamental period at level l. By virtue of the (conditional) PASTA property and an up- & down crossing argument, this number has again the same distribution as $\mathbf{N}_{M/M/m}^{(l)}$, the queue length at an arbitrary epoch in the corresponding M/M/m queue, given that the queue length is at least l.

$$E(z^{N_D} \mid L_D = l) = E(z^{N_{M/M/m}^{(i)}}).$$
 (9.48)

Denote by L the level of an arbitrary fundamental period. Remember that we have established a 1-1 correspondence between fundamental periods at level l and vacation customers that see l customers upon arrival. So L has the same distribution as the number of customers seen by an arbitrary arriving vacation customer. Because of the PASTA property, this number has again the same distribution as N_I . Denote by M_l the number of customers served in a fundamental period at level l. Then

...

$$\Pr\{\mathbf{L}_D = l\} = \frac{\Pr\{\mathbf{L} = l\} \mathbf{E} \mathbf{M}_l}{\sum\limits_{k=0}^{\infty} \Pr\{\mathbf{L} = k\} \mathbf{E} \mathbf{M}_k} = \frac{\Pr\{\mathbf{N}_I = l\} \mathbf{E} \mathbf{M}_l}{\sum\limits_{k=0}^{\infty} \Pr\{\mathbf{N}_I = k\} \mathbf{E} \mathbf{M}_k}.$$
(9.49)

In a fundamental period at level l, exactly 1 customer is served that leaves behind l customers as it departs from the system. So EM_l equals the reciprocal of the probability that an arbitrary customer leaves behind l customers as it departs from the system in a fundamental period at level l:

$$\mathbf{EM}_{l} = \frac{1}{\Pr\{\mathbf{N}_{M/M/m}^{(l)} = l\}}.$$
(9.50)

Summarizing,

$$E(z^{N}) = \gamma \left[\sum_{l=0}^{\infty} \frac{E(z^{N_{M/M/m}^{(l)}}) \Pr\{N_{I} = l\}}{\Pr\{N_{M/M/m}^{(l)} = l\}} \right],$$
(9.51)

with

$$\gamma = \left[\sum_{k=0}^{\infty} \frac{\Pr\{\mathbf{N}_I = k\}}{\Pr\{\mathbf{N}_{M/M/m}^{(k)} = k\}}\right]^{-1}.$$
(9.52)

Substituting (9.2) into (9.51) and (9.52) completes the proof.

9.B PROOF OF LEMMA 9.5.1

Lemma 9.5.1

If
$$\eta'(1) < 1$$
, then
i. $\lim_{K \to \infty} e^{(K+1)}(z) = 1$ for all z with $|z| \le 1$;
ii. $\prod_{k=0}^{\infty} f(e^{(k)}(z))$ converges for all z with $|z| \le 1$;
iii. $\sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))$ converges for all z with $|z| \le 1$.

Proof

Proof of i. Since $e(z) = \eta(z)$ is a pgf,

$$|1 - e(z)| \le \eta'(1) |1 - z|.$$
(9.53)

By induction,

$$|1 - e^{(k)}(z)| \le (\eta'(1))^k |1 - z|, \qquad k \ge 0.$$
So $\lim_{K \to \infty} e^{(K+1)}(z) = 1.$
(9.54)

9.B Proof of Lemma 9.5.1

Proof of ii.

According to the theory of infinite products, cf. Titchmarsh [185] p. 18, $\prod_{k=0}^{\infty} f(e^{(k)}(z)) \text{ converges iff } \sum_{k=0}^{\infty} [1 - f(e^{(k)}(z))] \text{ converges.}$ Let $\Gamma(z)$ be the straight contour in the complex plane from z to 1.

According to the theory of complex functions,

$$|1 - f(z)| = |f(1) - f(z)| = |\int_{u \in \Gamma(z)} df(u)| \le M(z) |1 - z|,$$
(9.55)

with

$$M(z) = \max_{u \in \Gamma(z)} \left| \frac{df(u)}{du} \right| < \infty,$$

as f(u) is continuously-differentiable on $|u| \leq 1$. Using (9.54), (9.55),

$$\sum_{k=0}^{\infty} |1 - f(e^{(k)}(z))| \le \sum_{k=0}^{\infty} M(z)(\eta'(1))^k |1 - z| = \frac{M(z)}{1 - \eta'(1)} |1 - z| < \infty.$$

So $\prod_{\substack{k=0\\Proof of iii.}}^{\infty} f(e^{(k)}(z))$ converges.

Note that
$$\sum_{k=0}^{\infty} |g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))| \le L \sum_{k=0}^{\infty} |g(e^{(k)}(z)| \text{ with } L = \max_{k\geq 0} |\prod_{l=0}^{k-1} f(e^{(l)}(z))|.$$
 Since $\prod_{k=0}^{\infty} f(e^{(k)}(z))$ converges, $\max_{k\geq 0} |\prod_{l=0}^{k-1} f(e^{(l)}(z))| < \infty.$
So to prove that $\sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))$ converges, it suffices to prove that $\sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))$

 $\sum_{k=0}^{k} g(e^{(k)}(z)) \text{ converges.}$

Let $\Gamma(z)$ be the straight contour in the complex plane from z to 1. Similarly to (9.55), noting that g(1) = 0,

$$|g(z)| \le N(z) |1 - z|, \tag{9.56}$$

with

$$N(z) = \max_{u \in \Gamma(z)} \left| \frac{dg(u)}{du} \right| < \infty,$$

as g(u) is also continuously-differentiable on $|u| \leq 1$. Using (9.54), (9.56),

$$\sum_{k=0}^{\infty} |g(e^{(k)}(z))| \le \sum_{k=0}^{\infty} N(z)(\eta'(1))^k |1-z| = \frac{N(z)}{1-\eta'(1)} |1-z| < \infty.$$

So $\sum_{k=0}^{\infty} g(e^{(k)}(z)) \prod_{l=0}^{k-1} f(e^{(l)}(z))$ converges.

CHAPTER 10

WAITING-TIME APPROXIMATIONS FOR MULTIPLE-SERVER POLLING SYSTEMS

10.1 INTRODUCTION

In the previous chapter, we analyzed multiple-server polling systems in which the servers are assumed to be *coupled*, i.e., the servers visit the queues always together. In the present chapter, we consider systems in which the servers are assumed to be *independent*, i.e., each of the servers visits the queues according to its own cyclic schedule. We derive waiting-time approximations for such systems with the exhaustive and gated service discipline.

As mentioned in the previous chapter, polling systems with multiple servers have received remarkably little attention in the vast literature on polling systems. One of the first studies is Morris & Wang [153]. They obtain the mean cycle time of each server and the mean intervisit time to a queue, and derive approximate expressions for the mean sojourn time for both a gated-type and a limited-type service discipline. A very interesting phenomenon observed by Morris & Wang is the tendency for the servers to cluster if they follow identical routes, especially in heavy traffic. Numerical experiments indicate that the bunching of servers is likely to deteriorate the system performance. Obviously, the bunching of servers is alleviated if they follow different routes. Therefore, Morris & Wang advocate the use of 'dispersive' schedules to improve the system performance.

In references [18], [121], [123], [157], [190] mean response time approximations are developed to analyze the performance of LAN's with multiple token rings. Mean response time approximations oriented to LAN's with a multiple slotted ring are contained in references [11], [18], [144], [190], [192]. Ajmone Marsan et al. [3], [4], [5] derive the mean cycle time and bounds for the mean waiting times in symmetric systems for the exhaustive, gated, and 1-limited service

discipline. In [2] they illustrate how Petri-net techniques may be used to study Markovian multiple-server polling systems.

Gamse & Newell [107] obtain approximate expressions for the mean round-trip time for a multiple-elevator facility. They make a comparison of some control options of multiple parallel elevators and – although there are some distinguishing features in the model description – find a similar tendency to form bunches as observed by Morris & Wang. For references on an exact analysis of models with a single queue, or models with coupled servers, we refer to the introduction of the previous chapter.

All these studies unanimously point out that multiple-server polling systems, combining the complexity of single-server polling systems and multiple-server systems, are extraordinarily hard to analyze. In fact, none of the studies (except [56], [57] for very specific two-queue infinite-server cases) presents any exact results for systems with multiple queues, apart from some mean-value results for global performance measures, like cycle times.

In this chapter, we consider the case of independent servers, each of which visits the queues according to its own cyclic schedule. In view of the mathematical intractability, we are interested in deriving waiting-time approximations for asymmetric systems with the exhaustive and gated service discipline. Most of the existing approximations are only developed for systems with the 1-limited service discipline or are restricted to completely symmetric systems. Moreover, most of the approximations completely ignore the considerable influence of the visit order on the waiting times or simply assume the visit order to be dispersive.

The remainder of the chapter is organized as follows. We present a detailed model description in Section 10.2. In Section 10.3, some preliminary results are obtained for the mean interarrival times of the various servers at the various queues, which will repeatedly be used throughout the subsequent sections. In Sections 10.4, 10.5, 10.6, we derive waiting-time approximations for asymmetric systems with the exhaustive and gated service discipline. Considering the merits and drawbacks of existing approximations, we intend (i) to use pseudo-conservation-like concepts, which have proven to be a very useful instrument in the single-server case, and (ii) to take into account the visit orders of the servers, which in the multiple-server case, through the clustering effects, appear to have a major impact on the waiting times. In Section 10.7, the approximations are tested for a wide range of parameter combinations. In Section 10.8, we conclude with some remarks and suggestions for further research.

10.2 MODEL DESCRIPTION

The model under consideration consists of n queues, Q_1, \ldots, Q_n , each of infinite capacity, attended by m identical servers, S_1, \ldots, S_m . For the specification of the arrival, service, and switch-over processes, we refer to the description of

10.3 The server interarrival time

the 'basic model' in Section 1.3.

The servers move from queue to queue in a cyclic manner. Server j visits the queues in the order $Q_{\pi_j(1)}, \ldots, Q_{\pi_j(n)}$, with $(\pi_j(1), \ldots, \pi_j(n))$ a permutation of $(1, \ldots, n), j = 1, \ldots, m$.

The servers visit the queues independently of each other, under the restriction that at most m_i servers may visit Q_i simultaneously. In view of the latter restriction, a server arrival will be called effective if there are less than m_i other servers already busy at Q_i . If an arrival at Q_i is not effective, then the server starts switching to the next queue immediately. If an arrival at Q_i is effective, then the server starts serving type-*i* customers (possibly none), as prescribed by the service discipline at Q_i . At each queue, the service discipline may either be exhaustive or gated. Under the exhaustive service discipline, a server leaves the queue when there are no waiting customers left. Under the gated service discipline, a server leaves the queue when there are no waiting customers left whose arrival fell before the last server arrival. In other words, at each server arrival, an imaginary gate opens to let waiting customers pass through. At each queue, customers are taken into service in order of arrival. As soon as the server finishes serving type-*i* customers, as prescribed by the service discipline at Q_i , it starts switching to the next queue as specified in its schedule.

Finally some words on the stability conditions. Necessary conditions are of course that $\rho < m$, $\rho_i < m_i$, i = 1, ..., n. We strongly conjecture that these conditions are in fact also sufficient for service disciplines, like exhaustive and gated, that do not impose any (probabilistic) parametric restriction on the number of customers served during a server visit. Throughout the chapter, the stability conditions are assumed to hold.

10.3 The server interarrival time

In this section, we derive some preliminary results for the mean interarrival time of the various servers at the various queues, which will repeatedly be used throughout the subsequent sections in obtaining waiting-time approximations. We first introduce some notation. Define r_{ij} as the load carried by S_j at Q_i , i.e., the fraction of time that S_j is busy at Q_i , $i = 1, \ldots, n$, $j = 1, \ldots, m$. The total load carried by S_j is $r_j = \sum_{i=1}^n r_{ij}$. In general, the fractions r_{ij} are unknown. However, the balance between the carried and the offered load at Q_i implies $\sum_{j=1}^m r_{ij} = \rho_i$, $i = 1, \ldots, n$.

Denote by \mathbf{A}_{ij} (\mathbf{A}_{ij}^*) the interarrival (effective interarrival) time of S_j at Q_i , i.e., the time between two consecutive arrivals (effective arrivals) of S_j at Q_i , $i = 1, \ldots, n, j = 1, \ldots, m$. Denote by p_{ij} the probability that an arbitrary arrival of S_j at Q_i is effective, i.e., the probability that at an arbitrary arrival of S_j there are less than m_i other servers already busy at Q_i , $i = 1, \ldots, n$, j = 1, ..., m. Obviously, for $m_i = m$, $p_{ij} = 1$; for $m_i < m$ the probabilities p_{ij} are however not known.

Applying a traffic balance argument,

$$\mathbf{E}\mathbf{A}_{ij} = \frac{s}{1 - r_j},\tag{10.1}$$

independent of *i*. As $p_{ij} = \mathbf{E}\mathbf{A}_{ij}/\mathbf{E}\mathbf{A}_{ij}^*$,

$$\mathbf{EA}_{ij}^{*} = \frac{s/p_{ij}}{1 - r_{j}}.$$
(10.2)

A question that arises here quite naturally, is whether or not all the servers will carry the same load. If two servers follow the same visit order, then by symmetry considerations both will also carry the same load of course. In particular, if all the servers follow the same visit order, then $r_{ij} = \rho_i/m$, $j = 1, \ldots, m$. Numerical experiments indicate that, even when the servers follow different visit orders, at each individual queue the load carried by each of the servers tends to differ only slightly, although in case of highly asymmetric system configurations the differences may increase somewhat. However, as observed in [150], [153], even in case of highly asymmetric system configurations the *total* load carried by each of the servers does not appear to differ significantly.

The above observation may be explained as follows. Suppose that the total load r_1 carried by S_1 is larger than the total load r_2 carried by S_2 . So by (10.1) also the mean interarrival time \mathbf{EA}_{i1} of S_1 is larger than the mean interarrival time \mathbf{EA}_{i2} of S_2 . In other words, S_2 visits the queues more frequently than S_1 , so that S_2 is also likely (but not absolutely sure) to meet more work at the queues than S_1 . So the total load r_2 carried by S_2 is likely to be larger than the total load r_1 carried by S_1 , in contradiction with the initial supposition. The above explanation does not exclude that some minor differences may occur in the total load carried by each of the servers. However, the reasoning supports the observation that such differences cannot grow dramatically.

Denote by \mathbf{A}_i (\mathbf{A}_i^*) the server interarrival (effective server interarrival) time at Q_i , i.e., the time between two consecutive server arrivals (effective server arrivals) at Q_i , i = 1, ..., n. Denote by p_i the probability that an arbitrary server arrival at Q_i is effective, i.e., the probability that at an arbitrary server arrival there are less than m_i other servers busy at Q_i , i = 1, ..., n. Obviously, for $m_i = m$, $p_i = 1$; for $m_i < m$ the probabilities p_i are however not known. The \mathbf{A}_i (\mathbf{A}_i^*) process is the superposition of the \mathbf{A}_{ij} (\mathbf{A}_{ij}^*) processes. So

$$\frac{1}{\mathbf{E}\mathbf{A}_i} = \sum_{j=1}^m \frac{1}{\mathbf{E}\mathbf{A}_{ij}} \qquad \left(\frac{1}{\mathbf{E}\mathbf{A}_i^*} = \sum_{j=1}^m \frac{1}{\mathbf{E}\mathbf{A}_{ij}^*}\right), \qquad i = 1, \dots, n.$$

So, from (10.1),

$$\mathbf{EA}_i = \frac{s}{m - \rho},\tag{10.3}$$

which is again like (10.1) independent of *i*. Moreover, the mean server interarrival time is completely insensitive to how the total load is divided among the individual servers.

As $p_i = \mathrm{E}\mathbf{A}_i / \mathrm{E}\mathbf{A}_i^*$,

$$\mathbf{E}\mathbf{A}_{i}^{*} = \frac{s/p_{i}}{m-\rho}, \qquad i = 1, \dots, n.$$
 (10.4)

Note that the mean-value results obtained here for the (effective) server interarrival time also hold for the (effective) server interdeparture time. (A departure is called effective if it corresponds to an effective arrival.)

10.4 The waiting time

In this section, we derive waiting-time approximations for systems with the exhaustive and gated service discipline. We first introduce some notation. Denote by \mathbf{W}_i the waiting time of an arbitrary type-*i* customer, $i = 1, \ldots, n$. For any non-negative continuous stochastic variable \mathbf{T} , denote by \mathbf{RT} a stochastic variable with as distribution the residual-lifetime distribution of \mathbf{T} , i.e.,

$$\Pr{\mathbf{RT} < t} = \frac{1}{\mathrm{ET}} \int_{u=0}^{t} (1 - \Pr{\mathbf{T} < u}) du, t \ge 0.$$

For reference, we first briefly review the single-server case. The usual approach to obtain waiting-time approximations may be outlined as follows. To start with, one derives an (approximative) relationship of the form

$$\mathbf{E}\mathbf{W}_i \approx \gamma_i \mathbf{E}\mathbf{R}\mathbf{C}_i, \qquad i = 1, \dots, n,$$
 (10.5)

with C_i either the interarrival or the interdeparture time at Q_i , depending on the service discipline at Q_i . The symbol γ_i represents some coefficient in terms of the system parameters, which reflects the influence of the service discipline at Q_i .

For the exhaustive service discipline,

$$\mathbf{EW}_i = (1 - \rho_i) \mathbf{ERD}_i, \qquad i = 1, \dots, n, \tag{10.6}$$

with \mathbf{D}_i the server inter*departure* time at Q_i , cf. [88], [113]. (An alternative relationship for the exhaustive service discipline is $\mathbf{EW}_i = \frac{\lambda_i \beta_i^{(2)}}{2(1-\rho_i)} + \mathbf{ERI}_i$, with \mathbf{I}_i the intervisit time at Q_i , cf. [62].) For the gated service discipline,

$$\mathbf{E}\mathbf{W}_i = (1+\rho_i)\mathbf{E}\mathbf{R}\mathbf{A}_i,\tag{10.7}$$

with, as before, \mathbf{A}_i the server inter*arrival* time at Q_i , cf. [88], [113]. To proceed, one turns to approximating \mathbf{ERC}_i (or \mathbf{ERI}_i). Since \mathbf{ERC}_i = $E(\mathbf{C}_i)^2/2E\mathbf{C}_i$ (similarly for \mathbf{ERI}_i), cf. [95], where $\mathbf{EC}_i = s/(1-\rho)$ (respectively $\mathbf{EI}_i = (1-\rho_i)s/(1-\rho)$), it remains to approximate $E(\mathbf{C}_i)^2$ (or $E(\mathbf{I}_i)^2$) by using some 'additional' information. One approach, followed by Bux & Truong [62] in the case of exhaustive service and deterministic switch-over times, is to derive an exact formula for $E(\mathbf{I}_i^2)$ in the case of two queues, subsequently applying a 'heuristic extrapolation' to the case of an arbitrary number of queues. Another approach, proposed by Everitt [88], and further elaborated on by Groenendijk [113] is to approximate \mathbf{ERC}_i in a direct manner, by invoking a so-called pseudo-conservation law, which provides an exact explicit expression for a weighted sum of the mean waiting times, typically $\sum_{i=1}^n \rho_i \mathbf{EW}_i$, cf. also Chapter 2. Substituting (10.5) into a pseudo-conservation law, assuming the table approximate the latter

ing $\mathbf{ERC}_i \approx \mathbf{ERC}$, yields an approximation for \mathbf{ERC}_i . Note that the latter method in particular yields an exact expression for the mean waiting time in completely symmetric systems.

We now return to the multiple-server case. The usual approach to obtain waiting-time approximations in the multiple-server case may be sketched as follows, cf. [18], [121], [123], [153], [157], [190]. Like in the single-server case, one starts by deriving an (approximative) relationship of the form

$$\mathbf{E}\mathbf{W}_i \approx \gamma_i \mathbf{E}\mathbf{R}\mathbf{C}_i^*,\tag{10.8}$$

with \mathbf{C}_{i}^{*} either the effective server interarrival or interdeparture time at Q_{i} . At that stage, the complications already start, since for most service disciplines at best a very rough approximation for γ_i can be found. Next, like in the single-server case, one proceeds by approximating ERC_i^* . The complications then grow even worse, since there is very little 'additional' information available that can be used, neither in the form of exact results for special cases, nor in the global form of a pseudo-conservation law. Having little choice left, one typically considers the C_i^* -process as resulting from the C_i -process after a 'filtering' with probability p_i (the probability of an arrival at Q_i being effective), and then the C_i -process in its turn as the superposition of the C_{ij} -processes, with j indicating server j. Subsequently, one approximates p_i and fits some distribution to the C_{ij} -processes, assuming that the C_{ij} -processes are independent and identically distributed. The motivation for fitting some particular distribution to the C_{ij} -processes is at best questionable, but is usually even completely lacking. What is worse however, is that the assumption that the C_{ij} -processes are independent and identically distributed, completely ignores the tendency for the servers to cluster, which immediately explains why the resulting approximations only appear to be reasonably accurate for dispersive schedules, or under conditions (like $m_i = 1$, or 1-limited service) with dispersive effects, cf. [18], [121], [123], [153], [157], [190].

We now describe an alternative approach to derive waiting-time approximations. From now on, we focus on the case $m_i = m$, which we consider to be the most interesting case; in the last section of the chapter, we briefly discuss

10.4 The waiting time

the case $m_i = 1$. Considering the above-mentioned objections, we intend

(i) to take into account the visit orders of the servers, which in the multipleserver case, through the clustering effects, appear to have a major impact on the waiting times;

(ii) to avoid considering cycle-time processes, instead using pseudo-conservationlike concepts, which have proven to be a very useful instrument in the singleserver case.

Denote by q_i the steady-state probability that at least one of the servers is busy at Q_i . In general, the probabilities q_i are unknown. However, $\rho_i/m \leq q_i \leq \min\{\rho_i, 1\}$. To derive an approximative relationship of the form $EW_i \approx \gamma_i ERC_i$, we assume that the customers experience the presence of multiple servers as if there is a single server processing at speed $\alpha_i = \rho_i/q_i$, the exact average processing speed at Q_i .

For the exhaustive service discipline we then obtain from (10.6), replacing ρ_i by ρ_i/α_i ,

$$\mathbf{EW}_i \approx (1 - q_i) \mathbf{ERD}_i,\tag{10.9}$$

with \mathbf{D}_i the server interdeparture time at Q_i .

Similarly, we obtain from (10.7) for the gated service discipline,

$$\mathbf{EW}_i \approx (1+q_i) \mathbf{ERA}_i,\tag{10.10}$$

with A_i , as before, the server interarrival time at Q_i .

In the multiple-server case, it is no longer reasonable to assume that the residual server interdeparture (interarrival) times are approximately equal, since the degree of clustering may differ significantly from queue to queue. Instead, we assume that the residual server interdeparture (interarrival) times are proportional to the average processing speed $\alpha_i = \rho_i/q_i$, which may be seen as a measure for the degree of clustering at Q_i , i.e.,

$$\operatorname{ERD}_i \approx \operatorname{ERD}\rho_i/q_i,$$
(10.11)

and

$$\mathbf{ERA}_i \approx \mathbf{ERA}\rho_i/q_i,$$
 (10.12)

with ERD and ERA some unknown constants. Note that in case m = 1, $q_i = \rho_i$, so that (10.11) and (10.12) reduce to ERD_i \approx ERD and ERA_i \approx ERA, respectively, the usual assumptions in the single-server case.

To complete the derivation of the approximations, it suffices to (i) find an expression for the weighted sum $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$ and (ii) determine the probabilities q_i , which we will do in Sections 10.5 and 10.6, respectively.

Chapter 10 Waiting-time approximations for multiple-server polling systems

10.5 Approximating the weighted sum $\sum_{i=1}^{n} \rho_i E \mathbf{W}_i$

In this section, we describe a method for approximating $\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i$. Denote by **V** the steady-state total amount of work in the system. Applying Brumelle's formula [60],

$$\sum_{i=1}^{n} \rho_i \mathbf{E} \mathbf{W}_i = \mathbf{E} \mathbf{V} - \frac{1}{2} \sum_{i=1}^{n} \lambda_i \beta_i^{(2)}.$$
 (10.13)

So to find an expression for $\sum_{i=1}^{n} \rho_i \mathbf{EW}_i$, it suffices to find an expression for the mean amount of work EV. For reference, we first briefly review the single-server case, where the crucial property that facilitates the determination of EV is work decomposition, which in its turn builds on the fundamental property of work conservation, cf. also Chapter 2. To illuminate these concepts, denote by \mathbf{V}^0 the steady-state total amount of work in the 'corresponding M/G/1 system'. The 'corresponding M/G/1 system' is a single-server system with similar traffic characteristics but with zero switch-over times, i.e., without any interruptions by the switch-over process. Denote by \mathbf{Y} the steady-state amount of work in the original system in a switching interval. Then the following work decomposition property holds:

$$\mathbf{V} \stackrel{a}{=} \mathbf{V}^0 + \mathbf{Y},\tag{10.14}$$

with $\stackrel{d}{=}$ indicating equality in distribution. When the amount of work in a switching interval is always zero, we may recognize in (10.14) the underlying property of *work conservation*, which in fact holds even in sample-path sense. Note that \mathbf{EV}^0 is simply known from the Pollaczek-Khintchine formula. For a broad class of service disciplines, including gated and exhaustive, \mathbf{EY} may be determined along the lines of [38], [42]. Taking expectations in (10.14), substituting into (10.13), then yields a so-called *pseudo-conservation law* for the mean waiting times.

We now return to the multiple-server case, where deriving a pseudo-conservation law in an exact way involves serious complications. A simple interchange argument shows that in the multiple-server case, a strict work conservation property in sample-path sense only holds if all the customers have the same (deterministic) service time. A weaker work conservation property in stochastic sense only holds if all the customers have the same service time distribution and the service discipline is regardless of the actual service times. Hence, since work conservation may be seen as the basis for work decomposition, it is not very likely that a property like (10.14) holds in the multiple-server case. Even if it were, we would face the problem that \mathbf{EV}^{0} is not known in the multiple-server case, not to mention the problem of determining \mathbf{EY} , so that the chances of deriving a pseudo-conservation law in an exact way appear to be negligible.

10.5 Approximating the weighted sum $\sum_{i=1}^{n} \rho_i EW_i$ 203

Instead, we therefore derive an approximative pseudo-conservation law. Although a work decomposition property probably does not hold, we can always write

$$\mathbf{EV} = \mathbf{EV}^0 + \mathbf{EY},$$

with \mathbf{V}^0 denoting the steady-state total amount of work in the 'corresponding M/G/m system', and \mathbf{Y} representing a stochastic variable whose mean satisfies the above equality, but which further remains unspecified. (The 'corresponding M/G/m system' is defined analogously as in the single-server case.) To approximate \mathbf{EV} , we consider two auxiliary single-server systems with similar characteristics, for which the work decomposition property *does* hold, viz:

(i). the ' λ/m system', i.e., a single-server system with identical characteristics, but with the arrival rate decreased by a factor m;

(ii). the ' β/m system', i.e., a single-server system with identical characteristics, but with the service rate increased by a factor m.

For these auxiliary systems, we adopt the notational convention introduced for the original system.

Applying (10.14) to the two auxiliary systems,

$$\mathbf{V}_{\lambda/m} \stackrel{d}{=} \mathbf{V}_{\lambda/m}^0 + \mathbf{Y}_{\lambda/m}, \ \mathbf{V}_{\beta/m} \stackrel{d}{=} \mathbf{V}_{\beta/m}^0 + \mathbf{Y}_{\beta/m}.$$

From the Pollaczek-Khintchine formula,

$$\mathrm{E}\mathbf{V}^{0}_{\lambda/m} = \mathrm{E}\mathbf{V}^{0}_{\beta/m} = \frac{1}{m} \frac{\sum_{i=1}^{n} \lambda_{i} \beta_{i}^{(2)}}{2(1-\hat{\rho})},$$

with $\hat{\rho} = \rho/m$. From [38], [42],

$$\mathbf{E}\mathbf{Y}_{\lambda/m} = \frac{1}{m} \mathbf{E}\mathbf{Y}_{\beta/m} = \hat{\rho} \frac{s^{(2)}}{2s} + \frac{s}{2(1-\hat{\rho})} \left[\hat{\rho}^2 - \sum_{i \in E} \hat{\rho}_i^2 + \sum_{i \in G} \hat{\rho}_i^2 \right],$$

with $\hat{\rho}_i = \rho_i/m$. The symbols *E* and *G* indicate the index sets of the queues with the exhaustive and gated service discipline, respectively.

To approximate EV^0 , we assume that the ratio of the mean amount of work in a multiple-server system and a single-server system with similar characteristics and proportional load, is rather insensitive to the service time distribution, i.e.,

$$\frac{\mathrm{E}\mathbf{V}^{0}}{\mathrm{E}\mathbf{V}^{0}_{\lambda/m}} = \frac{\mathrm{E}\mathbf{V}^{0}}{\mathrm{E}\mathbf{V}^{0}_{\beta/m}} \approx \gamma(\rho),$$

with $\gamma(\rho)$ denoting the known value of the ratio in question in case of identically exponentially distributed service times. In other words,

$$\mathrm{EV}_0 \approx \gamma(\rho) \mathrm{EV}_{\lambda/m}^0 = \gamma(\rho) \mathrm{EV}_{\beta/m}^0,$$

with

$$\gamma(\rho) = \frac{(m-\rho)\sum_{l=0}^{m-1} l\frac{\rho^l}{l!} + \frac{\rho^m}{m!}m^2 + \frac{\rho^m}{m!}\frac{m\rho}{m-\rho}}{\rho\sum_{l=0}^{m-1} \frac{\rho^l}{l!} + \frac{\rho^m}{m!}\frac{m\rho}{m-\rho}}.$$
(10.15)

To approximate EY, we assume

 $\mathbf{E}\mathbf{Y} \approx (1-\alpha)\zeta_{\lambda/m}(\rho)\mathbf{E}\mathbf{Y}_{\lambda/m} + \alpha\zeta_{\beta/m}(\rho)\mathbf{E}\mathbf{Y}_{\beta/m},$

with α indicating whether the comparison with the ' λ/m system' or with the ' β/m system' is most appropriate. The interpretation of the coefficients $\zeta_{\lambda/m}(\rho)$ and $\zeta_{\beta/m}(\rho)$ is similar to that of the factor $\gamma(\rho)$ introduced above.

If the server clustering is strong, which will occur especially in heavy traffic if the servers follow identical routes, then the system will tend to behave as the ' β/m system', i.e., $\alpha \uparrow 1$ for a high degree of clustering. It also suggests choosing $\zeta_{\beta/m}(\rho) = 1$ (note from (10.15) that also $\gamma(\rho) \downarrow 1$ when $\rho \uparrow m$). On the other hand, if the server clustering is weak, which will occur in light traffic, or if a dispersive schedule is used, then the comparison with the ' λ/m system' is probably more appropriate, i.e., $\alpha \downarrow 0$ for a low degree of clustering. Choosing $\zeta_{\lambda/m}(\rho)$ in this case is however not so easy. In light traffic, the switch-over times will tend to dominate the behavior of the system. In fact, if the total switch-over time incurred during a cycle is deterministic, $\mathbf{EW}_i \downarrow s/(m+1)$ for $\rho \downarrow 0$. If the total switch-over time during a cycle is exponentially distributed, $\mathbf{EW}_i \downarrow s/m$ for $\rho \downarrow 0$. Interpolating, we obtain $\mathbf{EW}_i \downarrow (m+s^{(2)}/s^2 - 1)s/m(m+1)$ for $\rho \downarrow 0$, implying that $\mathbf{EY} = \rho(m+s^{(2)}/s^2 - 1)s/m(m+1) + O(\rho^2)$ for $\rho \downarrow 0$. Note that $\mathbf{EY}_{\lambda/m} = \rho s^{(2)}/2ms + O(\rho^2)$ for $\rho \downarrow 0$. So $\mathbf{EY}/\mathbf{EY}_{\lambda/m} \to \frac{(m+s^{(2)}/s^2 - 1)s/m(m+1)}{s^{(2)}/2ms} = 2(1+(m-1)s^2/s^{(2)})/(m+1)$

for $\rho \downarrow 0$. In other words, $\zeta_{\lambda/m}(\rho) \rightarrow 2(1+(m-1)s^2/s^{(2)})/(m+1)$ for $\rho \downarrow 0$. On the other hand, in heavy traffic, the switch-over times occupy only a negligible fraction of time, implying that $\zeta_{\lambda/m}(\rho) \rightarrow \gamma(\rho)$ for $\rho \uparrow m$. Interpolating, we obtain $\zeta_{\lambda/m}(\rho) \approx 2(\rho/m)(1+(m-1)s^2/s^{(2)})/(m+1) + \gamma(\rho)(1-\rho/m)$. To choose α , we consider again $\alpha_{\mu} = \alpha_{\mu}/a_{\mu}$ the superscent processing speed at ρ .

To choose α , we consider again $\alpha_i = \rho_i/q_i$, the average processing speed at Q_i , as a measure for the degree of clustering at Q_i . We define

$$\alpha := \frac{1}{n} \sum_{i=1}^{n} \frac{\alpha_i - \rho_i / \left(1 - (1 - \rho_i / m)^m\right)}{m - \rho_i / \left(1 - (1 - \rho_i / m)^m\right)}.$$
(10.16)

Note that for a high degree of clustering, i.e., $q_i \downarrow \rho_i/m$, $\alpha \uparrow 1$. On the other hand, for a low degree of clustering, i.e., $q_i \uparrow 1 - (1 - \rho_i/m)^m$, $\alpha \downarrow 0$. Concluding,

$$E\mathbf{V} \approx \frac{\gamma(\rho)}{m} \frac{\sum_{i=1}^{n} \lambda_i \beta_i^{(2)}}{2(1-\hat{\rho})}$$
(10.17)

$$+ \left((1-\alpha) \left(2\frac{\rho}{m} \frac{1+(m-1)s^2/s^{(2)}}{m+1} + \gamma(\rho) \left(1-\frac{\rho}{m} \right) \right) + \alpha m \right) \\ \times \left(\hat{\rho} \frac{s^{(2)}}{2s} + \frac{s}{2(1-\hat{\rho})} \left[\hat{\rho}^2 - \sum_{i \in E} \hat{\rho}_i^2 + \sum_{i \in G} \hat{\rho}_i^2 \right] \right),$$

with $\gamma(\rho)$ and α as in (10.15) and (10.16), respectively. Substituting (10.17) into (10.13) yields an approximative pseudo-conservation law. Subsequently substituting (10.9), (10.10), (10.11), (10.12) into (10.13) yields waiting-time approximations, still containing the probabilities q_i , which we will determine in the next section. Note that the method in particular yields an exact expression for the mean waiting time in completely symmetric systems with exponential service times and zero switch-over times.

10.6 Approximating the probabilities q_i

In this section, we describe a method for approximating the probabilities q_i that at least one of the servers is busy at Q_i , i = 1, ..., n. We first introduce some notation. Denote by $\mathbf{H}_j(t)$ the entry in the polling table of S_j at time t. Indicate by $\mathbf{Z}_j(t)$ whether S_j is switching $(\mathbf{Z}_j(t) = 0)$ or serving $(\mathbf{Z}_j(t) = 1)$ at time t. So, if $(\mathbf{H}_j(t), \mathbf{Z}_j(t)) = (h, 0)$, then S_j is switching to $Q_{\pi_j(h)}$ at time t; if $(\mathbf{H}_j(t), \mathbf{Z}_j(t)) = (h, 1)$, then S_j is serving at $Q_{\pi_j(h)}$ at time t. Denote by (\mathbf{H}, \mathbf{Z}) a pair of stochastic variables with as joint distribution the joint stationary distribution of $(\mathbf{H}(t), \mathbf{Z}(t))$ with $(\mathbf{H}(t), \mathbf{Z}(t)) = (\mathbf{H}_1(t), \ldots, \mathbf{H}_m(t), \mathbf{Z}_1(t), \ldots, \mathbf{Z}_m(t))$.

We now describe a method for approximating the distribution of (\mathbf{H}, \mathbf{Z}) . Note that the probabilities q_i follow immediately from the distribution of (\mathbf{H}, \mathbf{Z}) as

$$q_i = 1 - \Pr\{(\pi_j(\mathbf{H}_j), \mathbf{Z}_j) \neq (i, 1), j = 1, \dots, m\}.$$
(10.18)

In fact, it is not difficult to approximate each of the marginal distributions of $(\mathbf{H}_j, \mathbf{Z}_j)$, $j = 1, \ldots, m$. As observed in Section 10.3, at each individual queue the load carried by each of the servers tends to differ only rather slightly, i.e., $r_{ij} \approx \rho_i/m$, $i = 1, \ldots, n$, $j = 1, \ldots, m$. So

$$\Pr\{(\mathbf{H}_j, \mathbf{Z}_j) = (h, 1)\} = r_{\pi_j(h)j} \approx \frac{\rho_{\pi_j(h)}}{m}.$$
(10.19)

Also, from (10.1),

$$\Pr\{(\mathbf{H}_j, \mathbf{Z}_j) = (h, 0)\} = \frac{s_{\pi_j(h)}}{\mathrm{E}\mathbf{C}_j} \approx (1 - \frac{\rho}{m}) \frac{s_{\pi_j(h)}}{s}, \qquad (10.20)$$

with EC_j denoting the mean cycle time of S_j . However, it is considerably harder to approximate the *simultaneous* distribution of $(\mathbf{H}, \mathbf{Z}) = (\mathbf{H}_1, \ldots, \mathbf{H}_m, \mathbf{Z}_1, \ldots, \mathbf{Z}_m)$, which is actually needed in (10.18). There are three types of transitions in (\mathbf{H}, \mathbf{Z}) . First,

$$(h, z) \to (h + e_j, z - e_j), \qquad z_j = 1,$$
 (10.21)

representing a departure of S_j from $Q_{\pi_j(h_j)}$, which does not result from an instantaneous passage; here e_j represents the *j*-th *m*-dimensional unit vector; $h_j + 1$ is in fact to be understood as $(h_j \mod n) + 1$. Second,

$$(h, z) \to (h, z + e_i), \qquad z_i = 0,$$
 (10.22)

representing an arrival of S_j at $Q_{\pi_j(h_j)}$, which does not lead to an instantaneous passage.

Third,

$$(h,z) \to (h+e_j,z), \qquad z_j = 0,$$
 (10.23)

representing an instantaneous passage of S_j at $Q_{\pi_j(h_j)}$.

Note that $\{(\mathbf{H}(t), \mathbf{Z}(t)), t \geq 0\}$ is not a Markov process, since the transitions are not independent of the past. To approximate the simultaneous distribution of (\mathbf{H}, \mathbf{Z}) , we will however deal with the process as if it is Markov, i.e., as if the transitions in (\mathbf{H}, \mathbf{Z}) occur at a constant rate, independent of the past. The distribution of (\mathbf{H}, \mathbf{Z}) may then be determined, as soon as the transition rates $\mu_{(h,z)\to(h',z')}$ are specified, which we might do as follows. First,

$$\mu_{(h,z)\to(h+e_j,z-e_j)} = (m-\rho)/(\rho_{\pi_j(h_j)}s), \qquad z_j = 1,$$
(10.24)

i.e., a departure of S_j from $Q_{\pi_j(h_j)}$ (which does not result from an instantaneous passage) occurs at a rate reciprocal to the approximate mean visit time of S_j at $Q_{\pi_j(h_j)}$ (i.e. $r_{\pi_j(h_j),j} \ge \mathbf{A}_{\pi_j(h_j),j} \approx \rho_{\pi_j(h_j)} s/(m-\rho)$). Second,

$$\mu_{(h,z)\to(h,z+e_j)} = 1/s_{\pi_j(h_j)}, \qquad z_j = 0, \tag{10.25}$$

i.e., an arrival of S_j at $Q_{\pi_j(h_j)}$ (which does not lead to an instantaneous passage) occurs at a rate reciprocal to the mean switch-over time into $Q_{\pi_j(h_j)}$. Third,

$$\mu_{(h,z)\to(h+e_j,z)} = 0, \qquad z_j = 0, \tag{10.26}$$

i.e., an instantaneous passage of S_j at $Q_{\pi_j(h_j)}$ (only occurring when there are no waiting customers at $Q_{\pi_j(h_j)}$, which cannot be deduced from (\mathbf{H}, \mathbf{Z})) does not occur. Note that in light traffic, instantaneous passages in fact *do* frequently occur, since a server arrival is likely to be followed by a concurrent server departure, which might suggest replacing (10.26) by

$$\mu_{(h,z)\to(h+e_j,z)} = 1/s_{\pi_j(h_j)}, \qquad z_j = 0, \tag{10.27}$$
when $\rho \downarrow 0$. However, when $\rho \downarrow 0$ in light traffic, combining (10.25) with (10.24) has a similar effect as using (10.27) would have.

Because of the homogeneity in the transition rates, we would obtain from (10.24), (10.25), (10.26)

$$\Pr\{(\mathbf{H}, \mathbf{Z}) = (h, z)\} = \prod_{j=1}^{m} \Pr\{(\mathbf{H}_j, \mathbf{Z}_j) = (h_j, z_j)\},$$
(10.28)

where $\Pr\{(\mathbf{H}_j, \mathbf{Z}_j) = (h_j, z_j)\}$ satisfies (10.19), (10.20). In other words, we would obtain complete independence in the server position distribution, while in fact we attempted to capture the tendency for the servers to cluster.

The driving force behind the tendency for the servers to cluster is that the time that a server visits a queue, depends on the time that the queue has not been visited by one of the other servers, so that the servers, somewhat depending on the visit orders, tend to be driven together. Once driven together, the servers do not disperse, as long as the visit orders do not direct them to different queues. To capture these phenomena, we slightly modify the transitions for the states in which more than one server is busy at the same queue simultaneously. For these states, we replace the transitions where *one* server leaves the queue, by a single transition of the same rate where *all* the visiting servers leave the queue simultaneously, reflecting that actually all the servers will tend to leave relatively shortly after one another.

The transition rates being specified, the distribution of (\mathbf{H}, \mathbf{Z}) may then be determined by solving the balance equations, supplemented with the normalization condition. Because of the inhomogeneity introduced in the transition rates, it is no longer possible to give the simultaneous distribution as explicitly as in (10.28), but it is easily verified from the balance equations that the marginal distribution $\Pr\{(\mathbf{H}_j, \mathbf{Z}_j) = (h_j, z_j)\}$ still satisfies (10.19), (10.20).

More detailed clustering measures

Remember that we approximated the distribution of (\mathbf{H}, \mathbf{Z}) in the first place to determine the probabilities q_i that at least one of the servers is busy at Q_i , $i = 1, \ldots, n$. In their turn, we used the probabilities q_i to determine $\alpha_i = \rho_i/q_i$, the average processing speed at Q_i , as a measure for the degree of clustering at Q_i . Having approximated the simultaneous distribution of (\mathbf{H}, \mathbf{Z}) , we may however refine the latter estimate for the degree of clustering, to deal with situations in which the average processing speed does not provide a good indication for the degree of clustering (e.g. in case of a lightly-loaded queue preceded by a heavily-loaded queue). In the remainder of the present section, we briefly discuss the definition of those alternatives. In the next section, when testing the resulting waiting-time approximations, we will examine the impact of implementing these alternatives.

Denote by $P_{ij}^{(0)}(h, z)$ and $P_{ij}^{(1)}(h, z)$ the conditional probability that $(\mathbf{H}, \mathbf{Z}) = (h, z)$ just after an arrival of S_j at Q_i and after a departure of S_j from Q_i , respectively. These conditional probabilities follow immediately from the dis-

tribution of (\mathbf{H}, \mathbf{Z}) . Denote by $\mathbf{T}_{ij}^{(0)}(h_j, z_j)$ and $\mathbf{T}_{ij}^{(1)}(h_j, z_j)$ the entrance time into $(\mathbf{H}_j, \mathbf{Z}_j) = (h_j, z_j)$ just after an arrival of S_j at Q_i and after a departure of S_j from Q_i , respectively. The mean values of these entrance times are given by

$$\begin{aligned} \mathbf{ET}_{ij}^{(b)}(h_j, z_j) &= \frac{r_{ij}s(1-b)}{1-r_j} + \sum_{k=\pi_j^{-1}(i)+1}^{h_j-1} \left(s_{\pi_j(k)} + \frac{r_{\pi_j(k)j}s}{1-r_j} \right) + s_{\pi_j(h_j)}z_j \\ &\approx \frac{\rho_i s(1-b)}{m-\rho} + \sum_{k=\pi_j^{-1}(i)+1}^{h_j-1} \left(s_{\pi_j(k)} + \frac{\rho_{\pi_j(k)}s}{m-\rho} \right) + s_{\pi_j(h_j)}z_j, \end{aligned}$$

$$\begin{split} b &= 0, 1, \text{ with } k = \pi_j^{-1}(i) \text{ such that } \pi_j(k) = i. \\ \text{For given } (h, z) &= (h_1, \ldots, h_m, z_1, \ldots, z_m), \text{ let } \mathbf{ET}_{ij_l}^{(b)}(h_{j_l}, z_{j_l}), \ l = 1, \ldots, m, \\ \text{be the mean entrance times } \mathbf{ET}_{ij}^{(b)}(h_j, z_j), \ j = 1, \ldots, m, \text{ ordered in decreasing magnitude. Let } \Delta_{il}^{(b)}(h, z) = \left(\mathbf{ET}_{ij_{l-1}}^{(b)}(h_{j_{l-1}}, z_{j_{l-1}}) - \mathbf{ET}_{ij_l}^{(b)}(h_{j_l}, z_{j_l})\right), \ l = 1, \ldots, m, \\ \text{with } \mathbf{ET}_{ij_0}^{(b)}(h_{j_0}, z_{j_0}) = \mathbf{EC}, \ \mathbf{EC} = s/(m-\rho). \\ \text{For given } (h, z), \\ \Delta_{il}^{(b)}(h, z) \text{ represents the mean of the } l\text{-th of the } m \text{ most recent server inter-arrival } (b = 0) \text{ or interdeparture } (b = 1) \text{ times at } Q_i, \ l = 1, \ldots, m. \\ \text{Denote } \Delta_i^{(b)}(h, z) := \sum_{l=1}^m \left(\Delta_{il}^{(b)}(h, z)\right)^2. \\ \text{The ordinary sum of } \Delta_{il}^{(b)}(h, z), \ l = 1, \ldots, m, \\ \text{being always equal to } \mathbf{EC}, \text{ the sum of the } squares \text{ provides a good indication for the } spacing of the server arrivals at or departures from } Q_i. \\ \text{Having this in mind, we define} \end{split}$$

$$\delta_i^{(b)} := \sum_{j=1}^m \sum_{(h,z)} P_{ij}^{(b)}(h,z) \Delta_i^{(b)}(h,z) / (\text{EC})^2$$
(10.29)

for b = 1 and b = 0 as a measure for the *local* degree of clustering at Q_i under the exhaustive and gated service discipline, respectively. If the degree of clustering at Q_i is high, then for the states (h, z) with large $P_{ij}^{(b)}(h, z)$, one of the $\Delta_{il}^{(b)}(h, z)$'s is approximately equal to EC, while all the other $\Delta_{il}^{(b)}(h, z)$'s are approximately equal to 0, so that $\delta_i^{(b)} \approx m$. On the other hand, if the degree of clustering at Q_i is low, i.e., the $\Delta_{il}^{(b)}(h, z)$'s are the distances between approximately homogeneously distributed points on [0, EC], then $\delta_i^{(b)} \approx m\kappa$, with $\kappa = \int_{1=x_0 \ge \dots \ge x_m = 0} \sum_{l=1}^m (x_{l-1} - x_l)^2 dx_1 \dots dx_{m-1}$. If the servers even tend

to repel each other, i.e., all the $\Delta_{il}^{(b)}(h, z)$'s are approximately equal to EC/m, then $\delta_i^{(b)} \approx 1$.

We may also refine the measure (10.16) for the global degree of clustering. In

the spirit of (10.29), we define

$$\delta := \frac{\sum_{(h,z)} \left(\Pr\{(\mathbf{H}, \mathbf{Z}) = (h, z)\} - \prod_{l=1}^{m} P_l(h_l, z_l) \right) \sum_{j=1}^{m} \Delta_{\pi_j(h_j)}^{(z_j)}(h, z)}{m(\mathbf{E}\mathbf{C})^2 - \sum_{(h,z)} \left(\prod_{l=1}^{m} P_l(h_l, z_l) \right) \sum_{j=1}^{m} \Delta_{\pi_j(h_j)}^{(z_j)}(h, z)},$$
(10.30)

with $P_l(h_l, z_l) = \Pr\{(\mathbf{H}_l, \mathbf{Z}_l) = (h_l, z_l)\}$. Note that for a high degree of clustering, i.e., for the states (h, z) with large $\Pr\{(\mathbf{H}, \mathbf{Z}) = (h, z)\}, \Delta_{\pi_j(h_j)}^{(z_j)}(h, z) \approx 1, \delta \uparrow 1.$ On the other hand, for a low degree of clustering, i.e., $\Pr\{(\mathbf{H}, \mathbf{Z}) = (h, z)\} \approx \prod_{l=1}^{m} \Pr\{(\mathbf{H}_l, \mathbf{Z}_l) = (h_l, z_l)\}, \delta \downarrow 0.$

10.7NUMERICAL RESULTS

We have tested the waiting-time approximations for a wide range of parameter combinations, by comparison with either simulation results, or exact numerical results obtained from the power-series algorithm (PSA), as elaborated upon in [150]. The large class of models considered in this chapter has forced us to limit the examples that are used to demonstrate the accuracy of the approximations. Thus, we have restricted ourselves to four-queue models with two servers and with exponentially distributed service and switch-over times. Numerous numerical experiments have indicated that for these models the accuracy of the approximations is acceptable, even for rather asymmetric and heavily-loaded systems. In particular, the approximations rightly capture the clustering effects of the visit order, whereas most of the existing approximations completely ignore the considerable influence of the visit order on the waiting times.

The results of the numerical experiments are summarized below. Limited numerical experience (which is however not reported in any further detail below) suggests that the approximations perform similarly for non-exponentially distributed service and switch-over times). We reemphasize that the models considered here are very complex, containing single-server polling models and ordinary multiple-server models as special cases, while the visit order constitutes an additional complicating factor. The accuracy of the approximations should be judged from this perspective.

In order to test the accuracy of the approximations for a wide variety of models, we have considered various variants of a set of models, in which the ratios between the arrival rates (λ_1 : λ_2 : λ_3 : λ_4), and the mean service times (β_1 , β_2 , β_3, β_4), respectively, are given as follows:

- (1: 1: 1: 1); (1.0, 1.0, 1.0, 1.0);I.
- II. (1: 1: 3: 3); (1.0, 1.0, 1.0, 1.0);
- III. (2: 2: 5: 5); (1.0, 1.0, 0.4, 0.4);
- IV. (1: 1: 1: 1); (0.5, 0.5, 1.5, 1.5);

- V. (1: 1: 3: 3); (0.5, 1.5, 0.5, 1.5);
- VI. (1: 1: 9: 1); (0.5, 0.5, 0.5, 2.5).

By convention, the queues are numbered such that π_1 , the visit order of S_1 , is always (1, 2, 3, 4). The visit order of S_2 is considered for the cases $\pi_2 = (1, 2, 3, 4)$, $\pi_2 = (1, 2, 4, 3)$, and $\pi_2 = (1, 4, 3, 2)$. For each of the models, all switch-over times are assumed to have mean s/n = s/4, with either s = 0 or s = 1. (In evaluating the approximations, we actually took $s = 10^{-6}$, since the formal definition of (\mathbf{H}, \mathbf{Z}) is restricted to the case of non-zero switch-over times.) The value of the total load is either $\rho = 0.8$, $\rho = 1.6$, or $\rho = 1.8$. In all considered cases, we assume $m_i = m = 2$.

For the models listed above, Tables 10.1.A to 10.6.B show the results for exhaustive service at each of the queues. Tables 10.7.A to 10.8.B show the results for the models II and V with gated service. The rows indicated by ' α ' contain the approximations obtained with $\alpha_i = \rho_i/q_i$ as a measure for the local degree of clustering at Q_i , and α as in (10.16) as a measure for the global degree of clustering. The rows marked with ' δ ' give the approximations with α_i replaced by $\delta_i^{(1)}$ as in (10.29) for exhaustive service or $\delta_i^{(0)}$ for gated service, and α replaced by δ as in (10.30). The rows indicated by 'exact' contain the 'exact' mean waiting times obtained from either the PSA (for exhaustive service) or simulation (for gated service). (We implemented the PSA only for Bernoulli service, including exhaustive service as special case, but in principle the method may also be developed to compute the mean waiting times for gated service.)

$(\lambda_1, \ldots, \lambda_n)$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.25, 0.25); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); s = 0.0.$					
(EW_1, EW_2, EW_3, EW_4)						
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)		
	exact	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)		
0.8	α	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)		
	δ	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)	(0.19, 0.19, 0.19, 0.19)		
	exact	(1.78, 1.78, 1.78, 1.78)	(1.78, 1.85, 1.74, 1.74)	(1.78, 1.78, 1.78, 1.78)		
1.6	α	(1.78, 1.78, 1.78, 1.78)	(1.76, 1.93, 1.71, 1.71)	(1.78, 1.78, 1.78, 1.78)		
	δ	(1.78, 1.78, 1.78, 1.78)	(1.78, 1.92, 1.71, 1.71)	(1.78, 1.78, 1.78, 1.78)		
	exact	(4.26, 4.26, 4.26, 4.26)	(4.41, 4.66, 3.98, 3.98)	(4.26, 4.26, 4.26, 4.26)		
1.8	α	(4.26, 4.26, 4.26, 4.26)	(4.19, 4.77, 4.04, 4.04)	(4.26, 4.26, 4.26, 4.26)		
	δ	(4.26, 4.26, 4.26, 4.26)	(4.25, 4.71, 4.04, 4.04)	(4.26, 4.26, 4.26, 4.26)		

TABLE 10.1.A. The mean waiting times for Model I with s = 0.0; exhaustive service.

$ (\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.25, 0.25); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); s = 1.0.$					
$(\mathbf{E}\mathbf{W}_1, \mathbf{E}\mathbf{W}_2, \mathbf{E}\mathbf{W}_3, \mathbf{E}\mathbf{W}_4)$						
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)		
	exact	(0.77, 0.77, 0.77, 0.77)	(0.77, 0.77, 0.77, 0.77)	(0.76, 0.76, 0.76, 0.76)		
0.8	α	(0.78, 0.78, 0.78, 0.78)	(0.78, 0.78, 0.77, 0.77)	(0.78, 0.78, 0.78, 0.78)		
	δ	(0.78, 0.78, 0.78, 0.78)	(0.77, 0.78, 0.77, 0.77)	(0.77, 0.77, 0.77, 0.77)		
	exact	(3.29, 3.29, 3.29, 3.29)	(3.24, 3.39, 3.09, 3.09)	(3.16, 3.16, 3.16, 3.16)		
1.6	α	(3.36, 3.36, 3.36, 3.36)	(3.14, 3.43, 3.05, 3.05)	(3.12, 3.12, 3.12, 3.12)		
	δ	(3.52, 3.52, 3.52, 3.52)	(3.24, 3.49, 3.11, 3.11)	(3.14, 3.14, 3.14, 3.14)		
	exact	(7.55, 7.55, 7.55, 7.55)	(7.52, 7.86, 6.38, 6.38)	(6.87, 6.87, 6.87, 6.87)		
1.8	α	(7.58, 7.58, 7.58, 7.58)	(6.79, 7.72, 6.54, 6.54)	(6.75, 6.75, 6.75, 6.75)		
	δ	(7.87, 7.87, 7.87, 7.87)	(7.09, 7.85, 6.74, 6.74)	(6.83, 6.83, 6.83, 6.83)		

TABLE 10.1.B. The mean waiting times for Model I with s = 1.0; exhaustive service.

Tables 10.1.A and 10.1.B show that the approximations for Model I are very accurate. In particular, Table 10.1.A confirms that for completely symmetric systems (including 'symmetric' visit order combinations, i.e., $\pi_2 = (1, 2, 3, 4)$ or $\pi_2 = (1, 4, 3, 2)$), the approximations are exact for exponentially distributed service times and zero switch-over times.

 $(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); s = 0.0.$

			$(\mathbf{E}\mathbf{W}_1, \mathbf{E}\mathbf{W}_2, \mathbf{E}\mathbf{W}_3, \mathbf{E}\mathbf{W}_4$)
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)
	exact	(0.21, 0.21, 0.18, 0.18)	(0.21, 0.21, 0.18, 0.18)	(0.21, 0.21, 0.18, 0.18)
0.8	α	(0.20, 0.20, 0.19, 0.19)	(0.20, 0.20, 0.19, 0.19)	(0.20, 0.20, 0.19, 0.19)
	δ	(0.21, 0.21, 0.18, 0.19)	(0.21, 0.21, 0.18, 0.18)	(0.21, 0.21, 0.18, 0.18)
	exact	(2.24, 2.30, 1.63, 1.60)	(2.09, 2.26, 1.65, 1.65)	(2.17, 2.17, 1.65, 1.65)
1.6	α	(2.03, 1.92, 1.67, 1.76)	(1.99, 2.06, 1.70, 1.70)	(2.00, 2.00, 1.70, 1.70)
	δ	(2.18, 2.13, 1.63, 1.68)	(2.05, 2.19, 1.66, 1.66)	(2.08, 2.08, 1.68, 1.68)
	exact	(5.51, 5.92, 3.90, 3.75)	(4.82, 5.36, 4.00, 4.00)	(5.11, 5.11, 4.00, 4.00)
1.8	α	(5.06, 4.83, 3.96, 4.12)	(4.83, 5.14, 4.02, 4.02)	(4.86, 4.86, 4.06, 4.06)
	δ	(5.41, 5.37, 3.86, 3.92)	(5.00, 5.43, 3.95, 3.95)	(5.04, 5.04, 4.00, 4.00)

TABLE 10.2.A. The mean waiting times for Model II with s = 0.0; exhaustive service.

$(\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); s = 1.0.$				
$(\mathbf{E}\mathbf{W}_1, \mathbf{E}\mathbf{W}_2, \mathbf{E}\mathbf{W}_3, \mathbf{E}\mathbf{W}_4)$)		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.83, 0.82, 0.74, 0.74)	(0.81, 0.81, 0.73, 0.73)	(0.81,0.81,0.74,0.74)	
0.8	α	(0.82, 0.81, 0.75, 0.76)	(0.81, 0.81, 0.74, 0.74)	(0.81, 0.81, 0.75, 0.75)	
	δ	(0.86, 0.85, 0.74, 0.75)	(0.83, 0.84, 0.73, 0.73)	(0.83, 0.83, 0.73, 0.73)	
	exact	(3.98, 4.04, 2.92, 2.88)	(3.58, 3.86, 2.88, 2.88)	(3.69, 3.69, 2.89, 2.89)	
1.6	α	(3.72, 3.53, 3.56, 3.23)	(3.42, 3.55, 2.92, 2.92)	(3.43, 3.43, 2.93, 2.93)	
	δ	(4.23, 4.14, 3.16, 3.26)	(3.63, 3.87, 2.94, 2.94)	(3.65, 3.65, 2.95, 2.95)	
	exact	(9.04, 9.69, 6.53, 6.32)	(7.27, 8.04, 6.44, 6.44)	(7.73, 7.73, 6.45, 6.45)	
1.8	α	(8.69, 8.29, 6.79, 7.07)	(7.53, 8.01, 6.27, 6.27)	(7.54, 7.54, 6.30, 6.30)	
	δ	(9.60, 9.43, 6.96, 7.12)	(8.05, 8.74, 6.36, 6.36)	(8.06, 8.06, 6.40, 6.40)	

TABLE 10.2.B. The mean waiting times for Model II with s = 1.0; exhaustive service.

Tables 10.2.A and 10.2.B show that the results are still accurate when the arrival rates are fairly asymmetric, even for heavily-loaded systems.

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.625, 0.625); \ (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 0.4, 0.4); \ s = 0.0.$					
			$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.13, 0.12, 0.13, 0.13)	(0.13, 0.12, 0.13, 0.13)	(0.13, 0.13, 0.13, 0.13)	
0.8	α	(0.13, 0.13, 0.13, 0.13)	(0.13, 0.14, 0.13, 0.13)	(0.13, 0.13, 0.13, 0.13)	
	δ	(0.13, 0.13, 0.13, 0.13)	(0.13, 0.14, 0.13, 0.13)	(0.13, 0.13, 0.13, 0.13)	
	exact	(1.25, 1.14, 1.22, 1.34)	(1.27, 1.20, 1.24, 1.24)	(1.21, 1.21, 1.26, 1.26)	
1.6	α	(1.24, 1.24, 1.24, 1.24)	(1.23, 1.35, 1.20, 1.20)	(1.24, 1.24, 1.24, 1.24)	
	δ	(1.24, 1.24, 1.24, 1.24)	(1.25, 1.34, 1.20, 1.20)	(1.24, 1.24, 1.24, 1.24)	
	exact	(3.01, 2.75, 2.99, 3.22)	(3.15, 3.14, 2.83, 2.83)	(2.94, 2.94, 3.01, 3.01)	
1.8	α	(2.98, 2.98, 2.98, 2.98)	(2.94, 3.34, 2.83, 2.83)	(2.98, 2.98, 2.98, 2.98)	
	δ	(2.98, 2.98, 2.98, 2.98)	(2.98, 3.30, 2.83, 2.83)	(2.98, 2.98, 2.98, 2.98)	

TABLE 10.3.A. The mean waiting times for Model III with s = 0.0; exhaustive service.

$(\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.625, 0.625); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 0.4, 0.4); s = 1.0.$					
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$				
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)		
	exact	(0.72, 0.70, 0.70, 0.72)	(0.71, 0.70, 0.70, 0.70)	(0.70, 0.70, 0.70, 0.70)		
0.8	α	(0.72, 0.72, 0.72, 0.72)	(0.72, 0.73, 0.72, 0.72)	(0.71, 0.71, 0.71, 0.71)		
	δ	(0.72, 0.72, 0.72, 0.72)	(0.71, 0.72, 0.71, 0.71)	(0.71, 0.71, 0.71, 0.71)		
	exact	(2.80, 2.60, 2.79, 2.99)	(2.76, 2.72, 2.60, 2.60)	(2.59, 2.59, 2.65, 2.65)		
1.6	α	(2.83, 2.83, 2.83, 2.83)	(2.61, 2.85, 2.54, 2.54)	(2.59, 2.59, 2.59, 2.59)		
	δ	(2.98, 2.98, 2.98, 2.98)	(2.71, 2.92, 2.60, 2.60)	(2.61, 2.61, 2.61, 2.61)		
	exact	(6.43, 5.97, 6.38, 6.70)	(6.42, 6.60, 5.24, 5.24)	(5.55, 5.55, 5.63, 5.63)		
1.8	α	(6.30, 6.30, 6.30, 6.30)	(5.53, 6.29, 5.33, 5.33)	(5.47, 5.47, 5.47, 5.47)		
	δ	(6.59, 6.59, 6.59, 6.59)	(5.81, 6.43, 5.52, 5.52)	(5.55, 5.55, 5.55, 5.55)		

TABLE 10.3.B. The mean waiting times for Model III with s = 1.0; exhaustive service.

In the cases considered in Tables 10.3.A and 10.3.B, the arrival rates and the service rates are rather asymmetric, but the load offered to each of the queues is the same. By construction, the approximated ratios of the mean waiting times only depend on the λ_i 's and β_i 's through the ρ_i 's. Since the ρ_i 's are all equal here, the approximated mean waiting times are also all equal for 'symmetric' visit order combinations, i.e., $\pi_2 = (1, 2, 3, 4)$ or $\pi_2 = (1, 4, 3, 2)$. The numerical results show that the *true* ratios of the mean waiting times *do* depend on the individual λ_i 's and β_i 's, but that the accuracy of the approximated mean waiting times is still acceptable.

$(\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.25, 0.25); (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 0.5, 1.5, 1.5); s = 0.0.$					
$(\mathbf{E}\mathbf{W}_1, \mathbf{E}\mathbf{W}_2, \mathbf{E}\mathbf{W}_3, \mathbf{E}\mathbf{W}_4)$						
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)		
	exact	(0.25, 0.27, 0.23, 0.21)	(0.25, 0.27, 0.22, 0.22)	(0.26, 0.26, 0.22, 0.22)		
0.8	α	(0.25, 0.25, 0.23, 0.24)	(0.25, 0.26, 0.23, 0.23)	(0.25, 0.25, 0.23, 0.23)		
	δ	(0.26, 0.26, 0.23, 0.23)	(0.26, 0.26, 0.23, 0.23)	(0.26, 0.26, 0.23, 0.23)		
	exact	(2.79, 3.07, 2.05, 1.88)	(2.55, 2.92, 2.03, 2.03)	(2.74, 2.74, 2.03, 2.03)		
1.6	α	(2.53, 2.40, 2.08, 2.20)	(2.48, 2.57, 2.12, 2.12)	(2.50, 2.50, 2.13, 2.13)		
	δ	(2.72, 2.66, 2.03, 2.10)	(2.57, 2.74, 2.08, 2.08)	(2.60, 2.60, 2.10, 2.10)		
	exact	(7.00, 7.58, 4.84, 4.49)	(5.90, 6.77, 4.99, 4.99)	(6.38, 6.38, 4.96, 4.96)		
1.8	α	(6.33, 6.03, 4.95, 5.15)	(6.04, 6.43, 5.03, 5.03)	(6.08, 6.08, 5.08, 5.08)		
	δ	(6.68, 6.56, 4.84, 4.96)	(6.25, 6.78, 4.93, 4.93)	(6.30, 6.30, 5.00, 5.00)		

TABLE 10.4.A The mean waiting times for Model IV with s = 0.0; exhaustive service.

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.25, 0.25, 0.25, 0.25); (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 0.5, 1.5, 1.5); s = 1.0.$					
$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$					
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.87, 0.88, 0.79, 0.78)	(0.85, 0.88, 0.78, 0.78)	(0.86, 0.86, 0.78, 0.78)	
0.8	α	(0.87, 0.86, 0.79, 0.81)	(0.86, 0.87, 0.79, 0.79)	(0.86, 0.86, 0.79, 0.79)	
	δ	(0.91, 0.90, 0.78, 0.80)	(0.88, 0.89, 0.78, 0.78)	(0.89, 0.89, 0.78, 0.78)	
	exact	(4.56, 4.91, 3.35, 3.13)	(4.05, 4.53, 3.28, 3.28)	(4.28, 4.28, 3.28, 3.28)	
1.6	α	(4.23, 4.01, 3.47, 3.67)	(3.92, 4.06, 3.35, 3.35)	(3.93, 3.93, 3.36, 3.36)	
	δ	(4.78, 4.67, 3.56, 3.68)	(4.14, 4.42, 3.35, 3.35)	(4.17, 4.17, 3.37, 3.37)	
	exact	(10.6, 11.2, 7.23, 7.00)	(8.34, 9.51, 7.41, 7.41)	(9.04, 9.04, 7.39, 7.39)	
1.8	α	(9.95, 9.49, 7.78, 8.10)	(8.74, 9.30, 7.27, 7.27)	(8.75, 8.75, 7.32, 7.32)	
	δ	(10.9, 10.8, 7.92, 8.11)	(9.30, 10.10, 7.35, 7.35)	(9.32, 9.32, 7.40, 7.40)	

TABLE 10.4.B. The mean waiting times for Model IV with s = 1.0; exhaustive service.

In Model IV, the service times are asymmetric, whereas the arrival rates are the same. Tables 10.4.A and 10.4.B show that the accuracy of the results is

10.7 Numerical results

acceptable, even in heavily-loaded systems.

$(\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); \ (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 1.5, 0.5, 1.5); \ s = 0.0.$				
$(\mathbf{E}\mathbf{W}_1, \mathbf{E}\mathbf{W}_2, \mathbf{E}\mathbf{W}_3, \mathbf{E}\mathbf{W}_4)$)	
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.27, 0.25, 0.26, 0.21)	(0.28, 0.26, 0.26, 0.21)	(0.27, 0.25, 0.26, 0.21)	
0.8	α	(0.27, 0.25, 0.25, 0.23)	(0.27, 0.25, 0.25, 0.23)	(0.26, 0.26, 0.25, 0.23)	
	δ	(0.29, 0.26, 0.26, 0.22)	(0.28, 0.26, 0.26, 0.22)	(0.27, 0.27, 0.26, 0.22)	
	exact	(3.24, 2.65, 2.99, 1.67)	(3.13, 2.84, 2.76, 1.71)	(3.02, 2.81, 2.78, 1.72)	
1.6	α	(2.75, 2.63, 2.61, 1.90)	(2.72, 2.65, 2.48, 1.94)	(2.59, 2.72, 2.42, 1.95)	
	δ	(3.20, 2.75, 2.76, 1.76)	(3.08, 2.82, 2.54, 1.82)	(2.65, 2.91, 2.50, 1.85)	
	exact	(8.26, 6.68, 7.53, 3.73)	(7.51, 7.17, 6.59, 4.03)	(7.12, 7.14, 6.61, 4.06)	
1.8	α	(7.01, 6.64, 6.62, 4.28)	(6.77, 6.74, 6.18, 4.42)	(6.36, 6.87, 6.04, 4.46)	
	δ	(7.97, 6.84, 6.86, 4.02)	(7.74, 7.08, 6.34, 4.14)	(6.50, 7.29, 6.26, 4.24)	

TABLE 10.5.A. The mean waiting times for Model V with s = 0.0; exhaustive service.

$(\lambda_1, \ldots, \lambda_n)$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 1.5, 0.5, 1.5); s = 1.0.$				
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$			
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.91, 0.85, 0.86, 0.74)	(0.90, 0.86, 0.85, 0.74)	(0.89, 0.86, 0.85, 0.74)	
0.8	α	(0.89, 0.85, 0.85, 0.75)	(0.88, 0.85, 0.84, 0.75)	(0.88, 0.85, 0.84, 0.75)	
	δ	(0.98, 0.89, 0.90, 0.74)	(0.94, 0.88, 0.86, 0.73)	(0.91, 0.89, 0.85, 0.73)	
	exact	(5.12, 4.28, 4.68, 2.69)	(4.82, 4.42, 4.19, 2.74)	(4.58, 4.41, 4.22, 2.76)	
1.6	α	(4.43, 4.23, 4.20, 3.06)	(4.19, 4.08, 3.81, 2.98)	(3.99, 4.19, 3.73, 3.00)	
	δ	(5.43, 4.67, 4.69, 2.98)	(4.88, 4.47, 4.03, 2.89)	(4.19, 4.60, 3.95, 2.93)	
1.8	exact	(11.9, 11.0, 11.1, 5.44)	(10.6, 10.3, 9.15, 5.89)	(9.88, 10.5, 9.27, 5.89)	
	α	(10.6, 10.0, 10.0, 6.47)	(9.53, 9.48, 8.69, 6.22)	(8.95, 9.67, 8.50, 6.28)	
	δ	(12.6, 10.8, 10.8, 6.34)	(11.3, 10.3, 9.24, 6.03)	(9.43, 10.6, 9.08, 6.15)	

TABLE 10.5.B. The mean waiting times for Model V with s = 1.0; exhaustive service.

In the models considered in Tables 10.5.A and 10.5.B, the arrival rates as well as the service times are asymmetric. In these cases, the approximations are less accurate than in the cases considered above, but still acceptable. In Models I-IV, both approximations yielded similar results, but here the δ -approximation tends to outperform the α -approximation. Apparently, the latter fails to detect the clustering at the lightly-loaded queues that are visited after the heavily-loaded Q_4 .

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 1.125, 0.125); (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 0.5, 0.5, 2.5); s = 0.0.$					
			$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4$)	
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.21, 0.22, 0.18, 0.19)	(0.23, 0.23, 0.18, 0.19)	(0.23, 0.23, 0.18, 0.19)	
0.8	α	(0.24, 0.24, 0.20, 0.23)	(0.24, 0.24, 0.21, 0.22)	(0.24, 0.24, 0.21, 0.22)	
	δ	(0.26, 0.26, 0.20, 0.23)	(0.26, 0.26, 0.20, 0.22)	(0.26, 0.25, 0.20, 0.22)	
	exact	(2.96, 3.17, 1.65, 2.04)	(2.99, 3.25, 1.62, 2.08)	(3.13, 3.12, 1.62, 2.08)	
1.6	α	(2.45, 2.31, 1.70, 2.39)	(2.43, 2.47, 1.77, 2.23)	(2.48, 2.41, 1.77, 2.23)	
	δ	(2.94, 2.88, 1.61, 2.34)	(2.60, 2.75, 1.70, 2.28)	(2.76, 2.54, 1.70, 2.28)	
	exact	(7.94, 8.50, 3.78, 4.91)	(7.54, 8.18, 3.83, 5.05)	(8.05, 7.75, 3.84, 5.07)	
1.8	α	(9.14, 8.57, 5.21, 8.22)	(8.29, 8.55, 5.46, 7.92)	(8.50, 8.18, 5.47, 7.93)	
	δ	(10.4, 10.3, 5.00, 8.01)	(8.88, 9.43, 5.21, 8.09)	(9.39, 8.66, 5.23, 8.11)	

TABLE 10.6.A. The mean waiting times for Model VI with s = 0.0; exhaustive service.

$ (\lambda_1,$	$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 1.125, 0.125); \ (\beta_1, \beta_2, \beta_3, \beta_4) = (0.5, 0.5, 0.5, 2.5); \ s = 1.0.$				
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$			
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)	
	exact	(0.84, 0.84, 0.69, 0.79)	(0.86, 0.86, 0.69, 0.77)	(0.86, 0.85, 0.69, 0.77)	
0.8	α	(0.86, 0.85, 0.72, 0.81)	(0.86, 0.86, 0.73, 0.79)	(0.86, 0.86, 0.73, 0.79)	
	δ	(0.95, 0.94, 0.72, 0.84)	(0.92, 0.92, 0.72, 0.80)	(0.93, 0.90, 0.72, 0.80)	
	exact	(4.69, 5.05, 2.67, 3.35)	(4.63, 4.95, 2.62, 3.36)	(4.93, 4.86, 2.62, 3.34)	
1.6	α	(4.05, 3.81, 2.80, 3.95)	(3.85, 3.90, 2.80, 3.52)	(3.93, 3.81, 2.80, 3.52)	
	δ	(5.19, 5.09, 2.84, 4.13)	(4.29, 4.52, 2.80, 3.75)	(4.55, 4.18, 2.80, 3.75)	
	exact	(11.4, 12.5, 5.63, 5.70)	(9.99, 11.1, 5.70, 7.36)	(11.0, 9.97, 5.74, 7.36)	
1.8	α	(13.9, 13.0, 7.93, 12.5)	(11.7, 12.1, 7.71, 11.2)	(12.0, 11.5, 7.72, 11.2)	
	δ	(16.7, 16.4, 7.98, 12.8)	(13.2, 14.0, 7.75, 12.0)	(13.9, 12.9, 7.76, 12.0)	

TABLE 10.6.B. The mean waiting times for Model VI with s = 1.0; exhaustive service.

Model VI is a typical example of a very asymmetric system. In such cases, the accuracy of all waiting-time approximations in the literature degrades significantly, even in single-server systems. Tables 10.6.A and 10.6.B show that the accuracy of the waiting-time approximation presented in this chapter also degrades when the model is very asymmetric, but remains acceptable as long as the load is not too high.

We now check the accuracy of the approximation for multiple-server systems with gated service at all queues.

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); \ (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); \ s = 0.0.$				
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)
0.8	exact	(0.18, 0.17, 0.19, 0.20)	(0.18, 0.17, 0.19, 0.20)	(0.18, 0.18, 0.19, 0.19)
	α	(0.16, 0.16, 0.20, 0.20)	(0.16, 0.16, 0.20, 0.20)	(0.16, 0.16, 0.20, 0.20)
	δ	(0.17, 0.17, 0.20, 0.20)	(0.17, 0.17, 0.20, 0.20)	(0.17, 0.17, 0.20, 0.20)
1.6	exact	(1.54, 1.52, 1.82, 1.85)	(1.46, 1.51, 1.85, 1.86)	(1.46, 1.46, 1.86, 1.86)
	α	(1.38, 1.33, 1.90, 1.94)	(1.29, 1.32, 1.94, 1.94)	(1.29, 1.29, 1.94, 1.94)
	δ	(1.56, 1.54, 1.84, 1.87)	(1.44, 1.53, 1.88, 1.88)	(1.44, 1.44, 1.89, 1.89)
1.8	exact	(3.60, 3.59, 4.34, 4.40)	(3.22, 3.42, 4.46, 4.46)	(3.29, 3.29, 4.47, 4.47)
	α	(3.40, 3.28, 4.53, 4.61)	(3.03, 3.16, 4.65, 4.65)	(3.03, 3.03, 4.67, 4.67)
	δ	(3.72, 3.67, 4.43, 4.48)	(3.36, 3.63, 4.52, 4.52)	(3.36, 3.36, 4.56, 4.56)

TABLE 10.7.A. The mean waiting times for Model II with s = 0.0; gated service.

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.375); (\beta_1, \beta_2, \beta_3, \beta_4) = (1.0, 1.0, 1.0, 1.0); s = 1.0.$				
		$(\mathbf{EW}_1, \mathbf{EW}_2, \mathbf{EW}_3, \mathbf{EW}_4)$		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)
0.8	exact	(0.84, 0.83, 0.91, 0.92)	(0.83, 0.83, 0.91, 0.91)	(0.83, 0.83, 0.91, 0.91)
	'α	(0.75, 0.75, 0.95, 0.96)	(0.75, 0.75, 0.95, 0.95)	(0.75, 0.75, 0.95, 0.95)
	δ	(0.83, 0.82, 0.94, 0.95)	(0.80, 0.81, 0.93, 0.93)	(0.80, 0.80, 0.93, 0.93)
1.6	exact	(3.85, 3.79, 4.56, 4.61)	(3.27, 3.44, 4.28, 4.28)	(3.35, 3.35, 4.35, 4.35)
	α	(3.25, 3.12, 4.47, 4.58)	(2.80, 2.88, 4.21, 4.21)	(2.80, 2.80, 4.21, 4.21)
	δ	(3.95, 3.88, 4.66, 4.73)	(3.22, 3.43, 4.22, 4.22)	(3.22, 3.22, 4.21, 4.21)
1.8	exact	(8.38, 8.31, 10.2, 10.2)	(6.47, 6.88, 9.33, 9.37)	(6.94, 6.94, 9.84, 9.84)
	α	(7.69, 7.42, 10.2, 10.4)	(6.00, 6.26, 9.22, 9.22)	(5.96, 5.96, 9.19, 9.19)
	δ	(8.91, 8.81, 10.6, 10.7)	(6.96, 7.53, 9.37, 9.37)	(6.89, 6.89, 9.37, 9.37)

TABLE 10.7.B. The mean waiting times for Model II with s = 1.0; gated service.

10.7 Numerical results

$(\lambda_1, \lambda_2, \lambda_3, \lambda_4) = \rho(0.125, 0.125, 0.375, 0.37)$			$(0.375); \ (\beta_1, \beta_2, \beta_3, \beta_4) = ($	(0.5, 1.5, 0.5, 1.5); s = 0.0.
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)
0.8	exact	(0.20, 0.21, 0.21, 0.25)	(0.20, 0.22, 0.21, 0.24)	(0.20, 0.21, 0.21, 0.24)
	α	(0.17, 0.20, 0.20, 0.27)	(0.17, 0.20, 0.19, 0.27)	(0.17, 0.21, 0.19, 0.27)
	δ	(0.20, 0.21, 0.22, 0.26)	(0.20, 0.22, 0.21, 0.26)	(0.19, 0.22, 0.21, 0.26)
1.6	exact	(1.71, 1.91, 1.86, 2.46)	(1.68, 1.94, 1.81, 2.52)	(1.54, 1.91, 1.80, 2.49)
	α	(1.39, 1.74, 1.73, 2.64)	(1.28, 1.68, 1.61, 2.71)	(1.23, 1.72, 1.60, 2.71)
	δ	(1.77, 1.93, 1.93, 2.47)	(1.64,1.91,1.74,2.55)	(1.40, 1.95, 1.72, 2.57)
1.8	exact	(3.91, 4.45, 4.36, 5.87)	(3.76, 4.54, 4.04, 6.18)	(3.35, 4.62, 4.11, 6.37)
	α	(3.46, 4.28, 4.28, 6.24)	(2.96, 4.04, 3.84, 6.52)	(2.83, 4.10, 3.80, 6.52)
	δ	(4.19, 4.60, 4.59, 5.95)	(3.85, 4.55, 4.12, 6.16)	(3.21, 4.62, 4.08, 6.22)

TABLE 10.8.A. The mean waiting times for Model V with s = 0.0; gated service.

$(\lambda_1, \ldots, \lambda_n)$	$\lambda_2, \lambda_3, \lambda_4$	$(\mu_4) = ho(0.125, 0.125, 0.375, 0.375)$	$(0.375); \ (\beta_1, \beta_2, \beta_3, \beta_4) = ($	0.5, 1.5, 0.5, 1.5); s = 1.0.
		$(\mathrm{E}\mathbf{W}_1,\mathrm{E}\mathbf{W}_2,\mathrm{E}\mathbf{W}_3,\mathrm{E}\mathbf{W}_4)$		
ρ	π_2	(1, 2, 3, 4)	(1, 2, 4, 3)	(1, 4, 3, 2)
0.8	exact	(0.85, 0.87, 0.89, 0.99)	(0.85, 0.87, 0.88, 0.98)	(0.84, 0.88, 0.90, 1.00)
	α	(0.70, 0.80, 0.80, 1.12)	(0.69, 0.80, 0.79, 1.11)	(0.69, 0.80, 0.79, 1.11)
	δ	(0.84, 0.89, 0.90, 1.07)	(0.81, 0.88, 0.85, 1.06)	(0.78, 0.88, 0.85, 1.06)
1.6	exact	(3.75, 4.18, 4.14, 5.42)	(3.51, 4.11, 3.73, 5.39)	(3.16, 4.09, 3.76, 5.40)
	α	(2.95, 3.69, 3.68, 5.61)	(2.54, 3.36, 3.22, 5.40)	(2.46, 3.43, 3.19, 5.41)
	δ	(4.06, 4.42, 4.42, 5.65)	(3.41, 3.98, 3.62, 5.32)	(2.91, 4.05, 3.56, 5.33)
1.8	exact	(8.56, 9.66, 9.56, 12.7)	(7.15, 8.84, 7.54, 12.2)	(6.26, 9.10, 7.79, 12.7)
	α	(7.07, 8.76, 8.75, 12.8)	(5.42, 7.41, 7.05, 12.0)	(5.18, 7.52, 6.98, 12.0)
	δ	(9.12, 10.0, 10.0, 13.0)	(7.45, 8.80, 7.97, 11.9)	(6.17, 8.88, 7.84, 12.0)

TABLE 10.8.B. The mean waiting times for Model V with s = 1.0; gated service.

Tables 10.7.A to 10.8.B show similar results as for the corresponding models with exhaustive service: the accuracy is acceptable for systems which are not too asymmetric, even for heavily-loaded systems in which the switch-over times are significant.

Discussion of the numerical results

We have tested the accuracy of the waiting-time approximations extensively for a broad set of parameter combinations, viz., for lightly-, medium- and heavilyloaded systems, with symmetric and asymmetric arrival and service rates, with negligible and non-negligible switch-over times, and with varying visit order combinations. In general, the results are fairly accurate. Because the clustering effects of the visit order are explicitly taken into account, the results are still accurate in cases where the server bunching is significant, whereas most of the existing approximations completely ignore the influence of the visit order on the waiting times.

As discussed extensively in Sections 10.4, 10.5, and 10.6, the waiting-time approximations presented in this chapter are based on a series of assumptions, each of which inherently forms a source of inaccuracy. The first source of inaccuracy stems from the estimation of the ratios between the mean waiting times which, in turn, is composed of a number of approximations for (i) the mean waiting times in terms of the mean residual cycle times (cf. (10.9), (10.10)), (ii) the ratios between the mean residual cycle times (cf. (10.11), (10.12)), and (iii) the value of q_i (cf. (10.18)). The second error source is the estimation of

the mean amount of work in the system (cf. (10.17)). Extensive simulation experiments have been performed to check the impact of each of these error sources on the finally observed error in the approximated mean waiting times. Inspection of the numerical results has revealed that the estimation of the mean amount of work in the system, EV, according to (10.17), is rather accurate. The error in the estimation of EV is typically less than 5% in fairly symmetric systems, even under heavy traffic, and remains well below 10% for rather asymmetric systems, even when the offered load is high.

The main source of inaccuracy stems from the estimation of the ratios between the mean waiting times. The approximation of q_i , i.e., the probability that at least one of the servers is busy at Q_i , is quite accurate in many cases, also when the clustering effect is significant, with errors typically below 10%. We found that q_i is underestimated in most of the cases. This is probably due to the fact that the clustering effect in the approximative approach is somewhat exaggerated because of the assumption that all the visiting servers depart from a queue simultaneously. The approximation of q_i may become inaccurate for very asymmetric systems under a heavy-traffic scenario. Other inaccuracies stem from the approximation of the mean waiting times in terms of the mean residual cycle times (cf. (10.9), (10.10)). For systems with the exhaustive service discipline, the mean waiting times are usually underestimated according to (10.9) (where q_i and ERD_i are taken to be their respective true (simulated) values), whereas in case of the gated service discipline, the mean waiting times are somewhat overestimated according to (10.10). Apparently, in the case of exhaustive service the approximation (10.9) is too optimistic and, in the case of gated service, the approximation (10.10) is rather pessimistic. However, in both cases the *ratios* between the overestimated mean waiting times appear to be rather robust with respect to these errors, so that the errors resulting from (10.9) and (10.10) only have a marginal impact on the finally obtained waiting-time approximations. The ratios between the mean residual cycle times are estimated by the ratios between the estimated average processing rates according to (10.11) and (10.12). Numerical experimentation has indicated that the quality of these estimations is quite good, with errors typically up to 10%, except for very asymmetric systems under heavy traffic.

In the general approach developed in Sections 10.4, 10.5, and 10.6, we used $\alpha_i = \rho_i/q_i$ as a measure for the local degree of clustering and α in (10.16) as a measure for the global degree of clustering, where both degrees of clustering are based on estimation of the average processing speed. However, for situations in which the average processing speed does not provide a good indication for the degree of clustering, we defined in the second part of Section 10.6 alternative clustering measures, viz., sum-of-square-like spacing measures for the positions of the servers in the system (cf. (10.29), (10.30)). Comparing the accuracy of the waiting-time approximations based on both clustering measures (in the tables indicated by α and δ) has not indicated a clear superiority of one of the two measures; the α -approximation is 122 out of the 468 times more than 10% off; the δ -approximation 96 times.

Summarizing, in general the approximations presented in this chapter lead to fairly accurate results when the system load is not too high and the system parameters are not too asymmetric. Apparently, the approximations cover the main characteristics of the extremely complicated behavior of multiple-server polling systems. When the system is very asymmetric and the offered load is very high, the accuracy of the approximations may however degrade.

10.8 Concluding Remarks and suggestions for further Research

In the present chapter, we have focused on the case $m_i = m$, i.e., all the m servers may visit Q_i simultaneously. It would be interesting to derive waitingtime approximations for the case $m_i < m$, in particular for $m_i = 1$. In some respects, the analysis will be somewhat facilitated then. Formulae (10.9), (10.10) e.g. are exact for $m_i = 1$. Also, the probabilities q_i that show up in these formulae are simply known to be ρ_i for $m_i = 1$. In certain other respects, the analysis will however be more complicated. The average processing speed $\alpha_i = \rho_i/q_i$ will always be equal to 1 for $m_i = 1$, so that it can no longer be used as a measure for the degree of clustering at Q_i . The more detailed measures $\delta_i^{(b)}$ can still be used. It will be harder, however, to approximate the simultaneous distribution of (\mathbf{H}, \mathbf{Z}) needed to determine these measures, since for $m_i < m$ there are also instantaneous passages through states with more than m_i servers at Q_i . The derivation of an approximative pseudo-conservation law will also be considerably harder.

In the present chapter, we have considered systems in which each of the servers visits the queues cyclically, in a fixed order, and where the switch-over times only depend on the next queue to be visited. It would be interesting to explore the same ideas to derive approximations for the mean waiting times in multiple-server polling systems with a non-cyclic polling table, with probabilistic server routing, or where the switch-over times also depend on the previous queue visited.

BIBLIOGRAPHY

- Ackroyd, M.H. (1985). Numerical computation of delays in clocked schedules. AT & T Techn. J. 64, 617-631.
- [2] Ajmone Marsan, M., Donatelli, S., Neri, F. (1990). GSPN models of Markovian multiserver multiqueue systems. *Perf. Eval.* 11, 227-240.
- [3] Ajmone Marsan, M., Donatelli, S., Neri, F. (1991). Multiserver multiqueue systems with limited service and zero walk time. In: Proc. INFO-COM '91, 1178-1188.
- [4] Ajmone Marsan, M., De Moraes, L.F., Donatelli, S., Neri, F. (1990). Analysis of symmetric nonexhaustive polling with multiple servers. In: *Proc. INFOCOM '90*, 284-295.
- [5] Ajmone Marsan, M., De Moraes, L.F., Donatelli, S., Neri, F. (1992). Cycles and waiting times in symmetric exhaustive and gated multiserver multiqueue systems. In: Proc. INFOCOM '92, 2315-2324.
- [6] Altman, E., Foss, S. (1993). Polling on a graph with general arrival and service time distribution. Report INRIA Sophia Antipolis.
- [7] Altman, E., Khamisy A., Yechiali, U. (1992). On elevator polling with globally gated regime. Queueing Systems 11, Special Issue on Polling Models, 85-90.
- [8] Altman, E., Konstantopoulos, P., Liu, Z. (1992). Stability, monotonicity and invariant quantities in general polling systems. *Queueing Systems* 11, Special Issue on Polling Models, 35-57.
- [9] Altman, E., Yechiali, U. (1994). Polling in a closed network. Prob. Eng. Inf. Sc. 8, 327-343.
- [10] Anily, S., Federgruen, A. (1991). Structured partitioning problems. Oper. Res. 39, 130-149.
- [11] Arem, B. van (1990). Queueing Models for Slotted Transmission Systems. Ph.D. Thesis, Twente University, Enschede.

- [12] Arian, Y., Levy, Y. (1992). Algorithms for generalized round robin routing. Oper. Res. Lett. 12, 313-319.
- [13] Athreya, K.B., Ney, P.E. (1972). Branching Processes (Springer, Berlin).
- [14] Avi-Itzhak, B., Maxwell, W.L., Miller, W. (1965). Queuing with alternating priorities. Oper. Res. 13, 306-318.
- [15] Baker, J.E., Rubin, I. (1987). Polling with a general-service order table. *IEEE Trans. Commun.* **35**, 283-288.
- [16] Bertsekas, D., Gallager, R. (1992). Data Networks (Prentice-Hall, Englewood Cliffs, NJ, 2nd ed.).
- [17] Bertsimas, D.J., Van Ryzin, G. (1993). Stochastic and dynamic vehicle routing with general demand and interarrival time distributions. Adv. Appl. Prob. 25, 947-978.
- [18] Bhuyan, L.N., Ghosal, D., Yang, Q. (1989). Approximate analysis of single and multiple ring networks. *IEEE Trans. Comput.* 38, 1027-1040.
- [19] Bisdikian, C. (1993). The random N-policy. Preprint.
- [20] Bisdikian, C., Merakos, L. (1992). Output process from a continuous token-ring local area network. *IEEE Trans. Commun.* 40, 1796-1799.
- [21] Blanc, J.P.C. (1990). A numerical approach to cyclic-service queueing models. *Queueing Systems* 6, 173-188.
- [22] Blanc, J.P.C. (1991). The power-series algorithm applied to cyclic polling systems. Commun. Stat. - Stoch. Mod. 7, 527-545.
- [23] Blanc, J.P.C., Van der Mei, R.D. (1995). Optimization of polling systems with Bernoulli schedules. *Perf. Eval.* 21, 139-158.
- [24] Blanc, J.P.C., Van der Mei, R.D. (1994). The power-series algorithm applied to polling systems with a dormant server. In: *The Fundamen*tal Role of Teletraffic in the Evolution of Telecommunications Networks, Proc. ITC-14, eds. J. Labetoulle, J.W. Roberts (North-Holland, Amsterdam), 865-874.
- [25] Borovkov, A.A., Schassberger, R. (1994). Ergodicity of a polling network. Stoch. Proc. Appl. 50, 253-262.
- [26] Borst, S.C. (1993). A polling system with a dormant server. CWI Report BS-R9313.
- [27] Borst, S.C. (1994). A pseudo-conservation law for a polling system with a dormant server. In: The Fundamental Role of Teletraffic in the Evolution of Telecommunications Networks, Proc. ITC-14, eds. J. Labetoulle, J.W. Roberts (North-Holland, Amsterdam), 729-742.

- [28] Borst, S.C. (1995). A globally gated polling system with a dormant server. Prob. Eng. Inf. Sc. 9, 239-254.
- [29] Borst, S.C. (1995). Polling systems with multiple coupled servers. Queueing Systems 20, 369-393.
- [30] Borst, S.C. (1995). Optimal probabilistic allocation of customer types to servers. In: Proc. 1995 ACM SIGMETRICS & Performance '95, 116-125.
- [31] Borst, S.C., Boxma, O.J. (1994). Polling models with and without switchover times. CWI Report BS-R9421. To appear in *Oper. Res.*
- [32] Borst, S.C., Boxma, O.J., Combé, M.B. (1992). Collection of customers: a correlated M/G/1 queue. In: Perf. Eval. Review 20, 47-59 (Proc. 1992 ACM SIGMETRICS & Performance '92).
- [33] Borst, S.C., Boxma, O.J., Combé, M.B. (1993). An M/G/1 queue with customer collection. Commun. Stat. – Stoch. Mod. 9, 341-371.
- [34] Borst, S.C., Boxma, O.J., Harink, J.H.A., Huitema, G.B. (1994). Optimization of fixed time polling schemes. *Telecommunication Systems* 3, 31-59.
- [35] Borst, S.C., Boxma, O.J., Levy, H. (1995). The use of service limits for efficient operation of multi-station single-medium communication systems. *IEEE/ACM Trans. Netw.* 3, 602-612.
- [36] Borst, S.C., Van der Mei, R.D. (1994). Waiting-time approximations for multiple-server polling systems. CWI Report BS-R9428. To appear in *Perf. Eval.*
- [37] Boxma, O.J. (1985). Two symmetric queues with alternating service and switching times. In: *Proc. Performance '84*, ed. E. Gelenbe (North-Holland, Amsterdam), 409-431.
- [38] Boxma, O.J. (1989). Workloads and waiting times in single-server queues with multiple customer classes. *Queueing Systems* 5, 185-214.
- [39] Boxma, O.J. (1991). Analysis and optimization of polling systems. In: Queueing, Performance and Control in ATM, eds. J.W. Cohen, C.D. Pack (North-Holland, Amsterdam), 173-183.
- [40] Boxma, O.J., Combé, M.B. (1993). The correlated M/G/1 queue. AEU
 47, Special Issue on Teletraffic Theory and Engineering in Memory of Félix Pollaczek, 330-335.
- [41] Boxma, O.J., Combé, M.B. (1994). Optimization of static traffic allocation policies. *Theor. Comput. Sc.* 125, 17-43.

- [42] Boxma, O.J., Groenendijk, W.P. (1987). Pseudo-conservation laws in cyclic service systems. J. Appl. Prob. 24, 949-964.
- [43] Boxma, O.J., Groenendijk, W.P. (1988). Two queues with alternating service and switching times. In: *Queueing Theory and its Applications Liber Amicorum for J.W. Cohen*, eds. O.J. Boxma, R. Syski (North-Holland, Amsterdam), 261-282.
- [44] Boxma, O.J., Groenendijk, W.P. (1988). Waiting times in discrete-time cyclic-service systems. *IEEE Trans. Commun.* **36**, 164-170.
- [45] Boxma, O.J., Groenendijk, W.P., Weststrate, J.A. (1990). A pseudoconservation law for service systems with a polling table. *IEEE Trans. Commun.* 38, 1865-1870.
- [46] Boxma, O.J., Kelbert, M. (1994). Stochastic bounds for a polling system. Ann. Oper. Res. 48, Special Issue on Queueing Networks, ed. N.M. van Dijk, 295-310.
- [47] Boxma, O.J., Levy, H., Weststrate, J.A. (1990). Optimization of polling systems. In: *Proc. Performance '90*, eds. P.J.B. King, I. Mitrani, R.J. Pooley (North-Holland, Amsterdam), 349-361.
- [48] Boxma, O.J., Levy, H., Weststrate, J.A. (1991). Efficient visit frequencies for polling tables: minimization of waiting cost. *Queueing Systems* 9, 133-162.
- [49] Boxma, O.J., Levy, H., Weststrate, J.A. (1993). Efficient visit orders for polling systems. *Perf. Eval.* 18, 103-123.
- [50] Boxma, O.J., Levy, H., Yechiali, U. (1992). Cyclic reservation schemes for efficient operation of multiple-queue single-server systems. Ann. Oper. Res. 35, 187-208.
- [51] Boxma, O.J., Meister, B.W. (1987). Waiting-time approximations for cyclic-service systems with switchover times. *Perf. Eval.* 7, 299-308.
- [52] Boxma, O.J., Meister, B.W. (1987). Waiting-time approximations in multi-queue systems with cyclic-service. *Perf. Eval.* 7, 59-70.
- [53] Boxma, O.J., Weststrate, J.A. (1989). Waiting times in polling systems with Markovian server routing. In: Messung, Modellierung und Bewertung von Rechensystemen und Netzen, eds. G. Stiege, J.S. Lie (Springer, Berlin), 89-104.
- [54] Boxma, O.J., Weststrate, J.A., Yechiali, U. (1993). A globally gated polling system with server interruptions, and applications to the repairman problem. *Prob. Eng. Inf. Sc.* 7, 187-208.

- [55] Bozer, Y.A., Srinivasan, M.M. (1991). Tandem configurations for automated guided vehicle systems and the analysis of single-vehicle loops. *IIE Trans.* 23, 72-82.
- [56] Browne, S., Coffman, E.G. Jr., Gilbert, E.N., Wright, P.E.W. (1992). Gated, exhaustive, parallel service. Prob. Eng. Inf. Sc. 6, 217-239.
- [57] Browne, S., Kella, O. (1995). Parallel service with vacations. Oper. Res. 43, 870-878.
- [58] Browne, S., Weiss, G. (1992). Dynamic priority rules when polling with multiple parallel servers. *Oper. Res. Lett.* **12**, 129-137.
- [59] Browne, S., Yechiali, U. (1989). Dynamic priority rules for cyclic-type queues. Adv. Appl. Prob. 21, 432-450.
- [60] Brumelle, S.L. (1971). On the relation between customer and time averages in queues. J. Appl. Prob. 8, 508-520.
- [61] Bunday, B.D., El-Badri, W.K. (1988). The efficiency of M groups of N machines served by a travelling robot: comparison of two models. Int. J. Prod. Res. 26, 299-308.
- [62] Bux, W., Truong, H.L. (1983). Mean-delay approximations for cyclicservice queueing systems. *Perf. Eval.* **3**, 187-196.
- [63] Buyukkoc, C., Varaiya, P., Walrand, J. (1985). The cμ rule revisited. Adv. Appl. Prob. 17, 237-238.
- [64] Buzacott, J.A., Shanthikumar, J.G. (1992). Design of manufacturing systems using queueing models. Queueing Systems 12, Special Issue on Queueing Models of Manufacturing Systems, 135-213.
- [65] Buzen, J.P., Chen, P.P.-S. (1974). Optimal load balancing in memory hierarchies. In: *Proc. IFIP 1974*, ed. J.L. Rosenfeld (North-Holland, Amsterdam), 271-275.
- [66] Chang, K.C., Sandhu, D. (1992). Mean waiting time approximations in cyclic-service systems with exhaustive limited service policy. *Perf. Eval.* 15, 21-40.
- [67] Cheng, W.C., Muntz, R.R. (1990). Optimal routing for closed queueing networks. In: *Proc. Performance '90*, eds. P.J.B. King, I. Mitrani, R.J. Pooley (North-Holland, Amsterdam), 3-17.
- [68] Choudhury, G.L. (1990). Polling with a general service order table: gated service. In: Proc. INFOCOM '90, 268-276.
- [69] Coffman, E.G. Jr., Gilbert, E.N. (1986). A continuous polling system with constant service times. *IEEE Trans. Inform. Theory* **33**, 584-591.

- [70] Coffman, E.G. Jr., Lueker, G.S., Rinnooy Kan, A.H.G. (1988). Asymptotic methods in the probabilistic analysis of sequencing and packing heuristics. *Mgmt. Sc.* **34**, 266-290.
- [71] Coffman, E.G. Jr., Mitrani, I., Fayolle, G. (1988). Two queues with alternating service periods. In: *Proc. Performance '87*, eds. P.-J. Courtois, G. Latouche (North-Holland, Amsterdam), 227-239.
- [72] Coffman, E.G. Jr., Stolyar, A. (1993). Continuous polling on graphs. Prob. Eng. Inf. Sc. 7, 209-226.
- [73] Cohen, J.W. (1982). The Single Server Queue (North-Holland, Amsterdam, 2nd ed.).
- [74] Cohen, J.W., Boxma, O.J. (1981). The M/G/1 queue with alternating service formulated as a Riemann-Hilbert boundary value problem. In: *Proc. Performance '81*, ed. F.J. Kylstra (North-Holland, Amsterdam), 181-199.
- [75] Cohen, J.W. (1988). A two-queue model with semi-exhaustive alternating service. eds. P.-J. Courtois, G. Latouche (North-Holland, Amsterdam), 19-37.
- [76] Combé, M.B. (1994). Modelling dependence between interarrival and service times with Markovian arrival processes. CWI Report BS-R9412.
- [77] Cooper, R.B., Murray, G. (1969). Queues served in cyclic order. Bell Syst. Techn. J. 48, 675-689.
- [78] Cooper, R.B. (1970). Queues served in cyclic order: waiting times. Bell Syst. Techn. J. 49, 399-413.
- [79] Cooper, R.B., Niu, S.-C., Srinivasan, M.M. (1992). A decomposition theorem for polling models: the switchover times are effectively additive. To appear in *Oper. Res.*
- [80] De Souza e Silva, E., Gail, H.R., Muntz, R.R. (1994). Polling systems with server timeouts. Preprint.
- [81] De Souza e Silva, E., Gerla, M. (1985). Load balancing in distributed systems with multiple classes and site constraints. In: *Proc. Performance* '84, ed. E. Gelenbe (North-Holland, Amsterdam), 17-33.
- [82] Doshi, B.T. (1985). Analysis of clocked schedules high priority tasks. AT & T Techn. J. 64, 633-660.
- [83] Eisenberg, M. (1971). Two queues with changeover times. Oper. Res. 19, 386-401.

- [84] Eisenberg, M. (1972). Queues with periodic service and changeover times. Oper. Res. 20, 440-451.
- [85] Eisenberg, M. (1979). Two queues with alternating service. SIAM J. Appl. Math. 36, 287-303.
- [86] Eisenberg, M. (1994). The polling system with a stopping server. Queueing Systems 18, 387-431.
- [87] Everitt, D.E. (1986). A conservation-type law for the token ring with limited service. Br. Telecom Techn. J. 4, 51-61.
- [88] Everitt, D.E. (1986). Simple approximations for token rings. IEEE Trans. Commun. 34, 719-721.
- [89] Everitt, D.E. (1989). An approximation procedure for cyclic service queues with limited service. In: *Performance of Distributed and Parallel Systems*, eds. T. Hasegawa, H. Takagi, Y. Takahashi (North-Holland, Amsterdam), 141-156.
- [90] Fabian, O., Levy, H. (1994). Pseudo-cyclic policies for multi-queue single server systems. Ann. Oper. Res. 48, Special Issue on Queueing Networks, ed. N.M. van Dijk, 127-152.
- [91] Ferguson, M.J. (1985). Mean waiting time for a token ring with nodal dependent overheads. In: *Teletraffic Issues in an Advanced Information Society, Proc. ITC-11*, ed. M. Akiyama (North-Holland, Amsterdam), 634-640.
- [92] Ferguson, M.J. (1986). Mean waiting time for a token ring with station dependent overheads. In: Local Area & Multiple Access Networks, ed. R.L. Pickholtz (Computer Science Press, Rockville, MD), 43-67.
- [93] Ferguson, M.J., Aminetzah, Y.J. (1985). Exact results for nonsymmetric token ring systems. *IEEE Trans. Commun.* **33**, 223-231.
- [94] Fournier, L., Rosberg, Z. (1991). Expected waiting times in cyclic service systems under priority disciplines. *Queueing Systems* 9, 419-439.
- [95] Franken, P., König, D., Arndt, U., Schmidt, V. (1982). Queues and Point Processes (Wiley, New York).
- [96] Fredericks, A.A., Farrell, B.L., DeMaio, D.F. (1985). Approximate analysis of a generalized clocked schedule. AT & T Techn. J. 64, 597-615.
- [97] Fricker, C., Jaïbi, M.R. (1994). Monotonicity and stability of periodic polling models. *Queueing Systems* 15, 211-238.
- [98] Fricker, C., Jaïbi, M.R. (1994). Stability of polling models with Markovian routing. Report INRIA Cedex 2278.

- [99] Fuhrmann, S.W. (1981). Performance analysis of a class of cyclic schedules. Bell Laboratories technical memorandum 81-59531-1.
- [100] Fuhrmann, S.W. (1985). Symmetric queues served in cyclic order. Oper. Res. Lett. 4, 139-144.
- [101] Fuhrmann, S.W. (1992). A decomposition result for a class of polling models. Queueing Systems 11, Special Issue on Polling Models, 109-120.
- [102] Fuhrmann, S.W., Cooper, R.B. (1985). Application of decomposition principle in M/G/1 vacation model to two continuum cyclic queueing models especially token-ring LANs. AT & T Techn. J. 64, 1091-1099.
- [103] Fuhrmann, S.W., Cooper, R.B. (1985). Stochastic decompositions in the M/G/1 queue with generalized vacations. Oper. Res. 33, 1117-1129.
- [104] Fuhrmann, S.W., Wang, Y.T. (1988). Analysis of cyclic service systems with limited service: bounds and approximations. *Perf. Eval.* 9, 35-54.
- [105] Garey, M.R., Johnson, D.S. (1979). Computers and Intractability: a Guide to the Theory of NP-Completeness (Freeman, San Francisco).
- [106] Gamse, B., Newell, G.F. (1982). An analysis of elevator operation in moderate height buildings - I. A single elevator. *Transp. Res. B* 16, 303-319.
- [107] Gamse, B., Newell, G.F. (1982). An analysis of elevator operation in moderate height buildings - II. Multiple elevators. Transp. Res. B 16, 321-335.
- [108] Georgiadis, L., Szpankowski, W. (1992). Stability of token passing rings. Queueing Systems 11, Special Issue on Polling Models, 7-34.
- [109] Gersht, A.M., Marbukh, V.V. (1975). Queueing systems with readjustment. Engin. Cyb. 13, 55-65.
- [110] Graham, R.L. (1966). Bounds for certain multiprocessing anomalies. Bell Syst. Techn. J. 45, 1563-1581.
- [111] Graham, R.L. (1969). Bounds on multiprocessing timing anomalies. SIAM J. Appl. Math. 17, 263-269.
- [112] Grillo, D. (1990). Polling mechanism models in communication systems - some application examples. In: Stochastic Analysis of Computer and Communication Systems, ed. H. Takagi (North-Holland, Amsterdam), 659-698.
- [113] Groenendijk, W.P. (1989). Waiting-time approximations for cyclic service systems with mixed service strategies. In: *Teletraffic Science for New Cost-Effective Systems, Networks and Services, ITC 12*, ed. M. Bonatti (North-Holland, Amsterdam), 1434-1441.

- [114] Groenendijk, W.P. (1990). Conservation Laws in Polling Systems. Ph.D. Thesis, University of Utrecht, Utrecht.
- [115] Gupta, D., Srinivasan, M.M. (1996). When should a roving server be patient? Mgmt. Sc. 42, 437-451.
- [116] Hajek, B. (1985). Extremal splittings of point processes. Math. Oper. Res. 10, 543-556.
- [117] Harink, J.H.A., Cramer, P., Huitema, G.B. (1992). Optimization of polling call records from switches. Report TI-RA-92-435, PTT Research, Groningen.
- [118] Hofri, M., Rosberg, Z. (1987). Packet delay under the Golden Ratio weighted TDM policy in a multiple-access channel. *IEEE Trans. Inform. Theory* 33, 341-349.
- [119] Hofri, M., Ross, K.W. (1987). On the optimal control of two queues with server set-up times and its analysis. SIAM J. Comput. 16, 399-420.
- [120] Itai, A., Rosberg, Z. (1984). A Golden Ratio control policy for a multipleaccess channel. *IEEE Trans. Autom. Control* 29, 712-718.
- [121] Kamal, A.E., Hamacher, V.C. (1989). Approximate analysis of nonexhaustive multiserver polling systems with applications to local area networks. *Comput. Netw. ISDN Syst.* 17, 15-27.
- [122] Kao, E.P.C., Narayanan, K.S. (1991). Analyses of an M/M/N queue with servers' vacations. *Eur. J. Oper. Res.* 54, 256-266.
- [123] Karmarkar, V.V., Kuhl, J.G. (1989). An integrated approach to distributed demand assignment in multiple-bus local networks. *IEEE Trans. Comput.* 38, 679-695.
- [124] Keilson, J., Servi, L.D. (1986). Oscillating random walk models for GI/G/1 vacation systems with Bernoulli schedules. J. Appl. Prob. 23, 790-802.
- [125] Keilson, J., Servi, L.D. (1990). The distributional form of Little's law and the Fuhrmann-Cooper decomposition. *Oper. Res. Lett.* 9, 239-247.
- [126] Khamisy, A., Altman, E., Sidi, M. (1992). Polling systems with synchronization constraints. Ann. Oper. Res. 35, 231-267.
- [127] Kim, W.B., Königsberg, E. (1987). The efficiency of two groups of N machines served by a single robot. J. Oper. Res. Soc. 38, 523-538.
- [128] Kleinrock, L., Levy, H. (1988). The analysis of random polling systems. Oper. Res. 36, 716-732.

- [129] Konheim, A.G., Levy, H. (1992). Efficient analysis of polling systems. In: Proc. INFOCOM '92, 2325-2331.
- [130] Königsberg. E., Mamer, J. (1982). The analysis of production systems. Int. J. Prod. Res. 20, 1-16.
- [131] Koole, G.M. (1994). Assigning a single server to inhomogeneous queues with switching costs. CWI Report BS-R9405.
- [132] Kroese, D.P., Schmidt, V. (1992). A continuous polling system with general service times. Ann. Appl. Prob. 2, 906-927.
- [133] Kroese, D.P., Schmidt, V. (1994). Single-server queues with spatially distributed arrivals. *Queueing Systems* 17, 317-345.
- [134] Kruskal, J.B. (1969). Work-scheduling algorithms: a non-probabilistic queueing study (with possible applications to No. 1 ESS). Bell Syst. Techn. J. 48, 2963-2974.
- [135] Kühn, P.J. (1979). Multiqueue systems with nonexhaustive cyclic service. Bell Syst. Techn. J. 58, 671-698.
- [136] Lee, D.-S., Sengupta, B. (1992). A reservation based cyclic server queue with limited service. In: *Perf. Eval. Review* 20, 70-77 (Proc. 1992 ACM SIGMETRICS & Performance '92).
- [137] Lee, D.-S., Sengupta, B. (1992). An approximate analysis of a cyclic server queue with limited service and reservations. *Queueing Systems* 11, 153-178.
- [138] Levy, H. (1988). Optimization of polling systems: the fractional exhaustive service method. Report Tel Aviv University, Tel Aviv.
- [139] Levy, H. (1989). Analysis of cyclic polling systems with binomial gated service. In: *Performance of Distributed and Parallel Systems*, eds. T. Hasegawa, H. Takagi, Y. Takahashi (North-Holland, Amsterdam), 127-139.
- [140] Levy, H., Kleinrock, L. (1991). Polling systems with zero switch-over periods: a general method for analyzing the expected delay. *Perf. Eval.* 13, 97-107.
- [141] Levy, H., Sidi, M. (1990). Polling systems: applications, modelling and optimization. *IEEE Trans. Commun.* 38, 1750-1760.
- [142] Levy, H., Sidi, M., Boxma, O.J. (1990). Dominance relations in polling systems. Queueing Systems 6, 155-171.
- [143] Levy, Y., Yechiali, U. (1976). An M/M/s queue with servers' vacations. INFOR 14, 153-163.

- [144] Loucks, W.M., Hamacher, V.C., Preiss, B.R., Wong, L. (1985). Shortpacket transfer performance in local area ring networks. *IEEE Trans. Comput.* 34, 1006-1014.
- [145] Liu, Z., Nain, P., Towsley, D. (1992). On optimal polling policies. Queueing Systems 11, Special Issue on Polling Models, 59-83.
- [146] Lucantoni, D.M. (1991). New results on the single server queue with a batch Markovian arrival process. Commun. Stat. Stoch. Mod. 7, 1-46.
- [147] Lucantoni, D.M. (1993). The BMAP/G/1 queue: a tutorial. In: Models and Techniques for the Performance Evaluation of Computer and Communication Systems, eds. L. Donatiello, R. Nelson (Springer, Berlin), 330-358.
- [148] Mack, C., Murphy, T., Webb, N.L. (1957). The efficiency of N machines unidirectionally patrolled by one operative when walking times and repair times are constants. J. Roy. Stat. Soc. B 19, 166-172.
- [149] Medhi, J. (1991). Stochastic Models in Queueing Theory (Academic Press, San Diego).
- [150] Mei, R.D., van der, Borst, S.C. (1994). Analysis of multiple-server polling systems by means of the power-series algorithm. CWI Report BS-R9410. To appear in *Commun. Stat. - Stoch. Mod.*
- [151] Meilijson, I., Yechiali, U. (1977). On optimal right-of-way policies at a single-server station when insertion of idle times is permitted. *Stoch. Proc. Appl.* 6, 25-32.
- [152] Mitrany, I.L., Avi-Itzhak, B. (1968). A many-server queue with server interruptions. Oper. Res. 16, 628-638.
- [153] Morris, R.J.T., Wang, Y.T. (1984). Some results for multi-queue systems with multiple cyclic servers. In: *Performance of Computer-Communication Systems*, eds. W. Bux, H. Rudin (North-Holland, Amsterdam), 245-258.
- [154] Neuts, M.F. (1989). Structured Stochastic Matrices of M/G/1 Type and Their Applications (Marcel Dekker, New York).
- [155] Neuts, M.F., Lucantoni, D.M. (1979). A Markovian queue with N servers subject to breakdowns and repairs. Mgmt. Sc. 25, 849-861.
- [156] Panwar, S.S., Philips, T.K., Chen, M.-S. (1992). Golden Ratio scheduling for flow control with low buffer requirements. *IEEE Trans. Commun.* 40, 765-772.

- [157] Raith, T. (1985). Performance analysis of multibus interconnection networks in distributed systems. In: *Teletraffic Issues in an Advanced Information Society, Proc. ITC-11*, ed. M. Akiyama (North-Holland, Amsterdam), 662-668.
- [158] Resing, J.A.C. (1990). Asymptotic Results in Feedback Systems. Ph.D. Thesis, Technical University Delft, Delft.
- [159] Resing, J.A.C. (1993). Polling systems and multitype branching processes. Queueing Systems 13, 409-426.
- [160] Sarkar, D., Zangwill, W.I. (1989). Expected waiting time for nonsymmetric cyclic queueing systems - Exact results and applications. Mgmt. Sc. 35, 1463-1474.
- [161] Sarkar, D., Zangwill, W.I. (1991). Variance effects in cyclic production systems. Mgmt. Sc. 37, 444-453.
- [162] Schassberger, R. (1995). Stability of polling networks with statedependent server routing. *Prob. Eng. Inf. Sc.* 9, 539-550.
- [163] Seneta, E. (1981). Non-negative Matrices and Markov Chains (Springer, New York, 2nd ed.).
- [164] Servi, L.D. (1986). Average delay approximation of M/G/1 cyclic service queue with Bernoulli schedule. *IEEE J. Sel. Areas Commun.* 4, 813-822.
- [165] Shimogawa, S., Takahashi, Y. (1992). A note on the pseudo-conservation law for a multi-queue with local priority. Queueing Systems 11, Special Issue on Polling Models, 145-151.
- [166] Sidi, M., Levy, H. (1990). Customer routing in polling systems. In: Proc. Performance '90, eds. P.J.B. King, I. Mitrani, R.J. Pooley (North-Holland, Amsterdam), 319-331.
- [167] Sidi, M., Levy, H., Fuhrmann, S.W. (1992). A queueing network with a single cyclically roving server. Queueing Systems 11, Special Issue on Polling Models, 121-144.
- [168] Srinivasan, M.M. (1988). An approximation for mean waiting times in cyclic server systems with nonexhaustive service. *Perf. Eval.* 9, 17-33.
- [169] Srinivasan, M.M., Niu, S.-C., Cooper, R.B. (1995). Relating polling models with nonzero and zero switchover times. *Queueing Systems* 19, 149-168.
- [170] Stadje, W. (1985). The busy period of the queueing system $M/G/\infty$. J. Appl. Prob. 22, 697-704.

- [171] Takács, L. (1968). Two queues attended by a single server. Oper. Res. 16, 639-650.
- [172] Takagi, H. (1986). Analysis of Polling Systems (The MIT Press, Cambridge, MA).
- [173] Takagi, H. (1991). Application of polling models to computer networks. Comp. Netw. ISDN Syst. 22, 193-211.
- [174] Takagi, H. (1991). Queueing Analysis, Vol. 1 (North-Holland, Amsterdam).
- [175] Takagi, H. (1990). Queueing analysis of polling models: an update. In: Stochastic Analysis of Computer and Communication Systems, ed. H. Takagi (North-Holland, Amsterdam), 267-318.
- [176] Takagi, H. (1994). Queueing analysis of polling models: progress in 1990-1993. To appear in: Frontiers in Queueing: Models, Methods and Problems, ed. J.H. Dshalalow (CRC Press).
- [177] Takine, T., Hasegawa, T. (1992). On the M/G/1 queue with multiple vacations and gated service discipline. JORSJ 35, 217-235.
- [178] Tantawi, A.N., Towsley, D. (1985). A general model for optimal static load balancing in star network configurations. In: *Proc. Performance* '84, ed. E. Gelenbe (North-Holland, Amsterdam), 277-291.
- [179] Tantawi, A.N., Towsley, D. (1985). Optimal static load balancing in distributed computer systems. J. ACM 32, 445-465.
- [180] Tripathi, S.K., Woodside, C.M. (1988). A vertex-allocation theorem for resources in queueing networks. J. ACM 35, 221-230.
- [181] Tedijanto, T.E. (1990). Nonexhaustive policies in polling systems and vacation models. Ph.D. Thesis, University of Maryland, College Park, MD.
- [182] Tedijanto, T.E. (1991). Stochastic comparisons in vacation models. Commun. Stat. - Stoch. Mod. 7, 125-135.
- [183] Tedijanto, T.E. (1992). A note on the comparison between Bernoulli and limited policies in vacation models. *Perf. Eval.* 15, 89-97.
- [184] Tijms, H.C. (1986). Stochastic Modelling and Analysis: a Computational Approach (Wiley, Chichester).
- [185] Titchmarsh, E.C. (1939). The Theory of Functions (Oxford University Press, London, 2nd ed.).
- [186] Wang, Y.-T., Morris, R.J.T. (1985). Load sharing in distributed systems. IEEE Trans. Comput. 34, 204-217.

- [187] Watson, K. S. (1985). Performance evaluation of cyclic service strategies - a survey. In: Proc. Performance '84, ed. E. Gelenbe (North-Holland, Amsterdam), 521-533.
- [188] Weststrate, J.A. (1992). Analysis and Optimization of Polling Models. Ph.D. Thesis, University of Tilburg, Tilburg.
- [189] Woodside, C.M., Tripathi, S.K. (1986). Optimal allocation of file servers in a local network environment. *IEEE Trans. Softw. Eng.* 12, 844-848.
- [190] Yang, Q., Ghosal, D., Bhuyan, L.N. (1986). Performance analysis of multiple token ring and multiple slotted ring networks. In: Proc. 1986 Comput. Netw. Symp., 79-86.
- [191] Yechiali, U. (1991). Optimal dynamic control of polling systems. In: Queueing, Performance and Control in ATM, eds. J.W. Cohen, C.D. Pack (North-Holland, Amsterdam), 205-217.
- [192] Zafirovic-Vukotic, M., Niemegeers, I.G., Valk, D.S. (1988). Performance modelling of slotted ring protocols in HSLAN's. *IEEE J. Sel. Areas Commun.* 6, 1011-1024.

CWI TRACTS

- 1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.
- 2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility, 1984.
- 3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.
- 4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
- 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
- 6 A.P.W. Böhm. Dataflow computation. 1984.
- 7 A. Blokhuis. Few-distance sets. 1984.
- 8 M.H. van Hoorn. Algorithms and approximations for queueing systems. 1984.
- 9 C.P.J. Koymans. *Models of the lambda calculus.* 1984.
- 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.
- 11 N.M. van Dijk. Controlled Markov processes; timediscretization. 1984.
- 12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.
- 13 D. Grune. On the design of ALEPH. 1985.
- 14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
- 15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
- 16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.
- 17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.
- 18 A.D.M. Kester. Some large deviation results in statistics. 1985.
- 19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.
- 20 B.F. Schriever. Order dependence. 1986.
- 21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.
- 22 J.C.S.P. van der Woude. Topological dynamix. 1986.
- 23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.
- 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.
- 25 A.W.J. Kolen. Tree network and planar rectilinear location theory, 1986.
- 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
- 27 A.J.M. van Engelen. Homogeneous zerodimensional absolute Borel sets. 1986.
- 28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.
- 29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.
- 30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

- 31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.
- 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.
- 33 O.J. Vrieze. Stochastic games with finite state and action spaces. 1987.
- 34 P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.
- 35 M.L. Eaton. Lectures on topics in probability inequalities. 1987.
- 36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.). Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.
- 37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.
- 38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.
- 39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.
- 40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems. 1987.
- 41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.
- 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, formalisms, and tools. 1987.
- 43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.
- 44 A.W. van der Vaart. Statistical estimation in large parameter spaces. 1988.
- 45 S.A. van de Geer. Regression analysis and empirical processes. 1988.
- 46 S.P. Spekreijse. Multigrid solution of the steady Euler equations. 1988.
- 47 J.B. Dijkstra. Analysis of means in some nonstandard situations. 1988.
- 48 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988.
- 49 F.W. Wubs. Numerical solution of the shallowwater equations. 1988.
- 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.
- 51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.
- 52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.
- 53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequen-
- tial quadratic programming in function space. 1988. 54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.
- 55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.
- 56 J.C. Ebergen. Translating programs into delayinsensitive circuits. 1989.
- 57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.
- 58 M.C.M. de Gunst. A random model for plant cell population growth. 1989.
- 59 D. van Dulst. Characterizations of Banach spaces not containing l^1 . 1989.
- H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

- 61 P. de Jong. Central limit theorems for generalized multilinear forms. 1989.
- 62 V.J. de Jong. A specification system for statistical software. 1989.
- 63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.
- 64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989.
- 65 B.M.M. de Weger. Algorithms for diaphantine equations. 1989.
- 66 A. Jung. Cartesian closed categories of domains. 1989.
- 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.
- 68 H.J. Woerdeman. Matrix and operator extensions. 1989.
- 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.
- 70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedicated to Gijs de Leve. 1990.
- 71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.
- 72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.
- 73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.
- 74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991.
- 75 M.W.P. Savelsbergh. Computer aided routing. 1992.
- 76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991.
- 77 A.J. van Es. Aspects of nonparametric density estimation. 1991.
- 78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.
- 79 J.J. Lodder. Towards a symmetrical theory of generalized functions. 1991.
- 80 S.A. Smulders. Control of freeway traffic flow. 1996.
- 81 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992.
- 82 F. Thuijsman. Optimality and equilibria in stochastic games. 1992.
- 83 R.J. Kooman. Convergence properties of recurrence sequences. 1992.
- 84 A.M. Cohen (ed.). Computational aspects of Lie group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991.
- 85 V. de Valk. One-dependent processes. 1994.
- 86 J.A. Baars, J.A.M. de Groot. On topological and linear equivalence of certain function spaces. 1992.
- 87 A.F. Monna. The way of mathematics and mathematicians, 1992.

- 88 E.D. de Goede. Numerical methods for the threedimensional shallow water equations. 1993.
- 89 M. Zwaan. Moment problems in Hilbert space with applications to magnetic resonance imaging. 1993.
- 90 C. Vuik. The solution of a one-dimensional Stefan problem. 1993.
- 91 E.R. Verheul. Multimedians in metric and normed spaces, 1993.
- 92 J.L.M. Maubach. Iterative methods for non-linear partial differential equations. 1994.
- 93 A.W. Ambergen. Statistical uncertainties in posterior probabilities. 1993.
- 94 P.A. Zegeling. Moving-grid methods for timedependent partial differential equations. 1993.
- 95 M.J.C. van Pul. Statistical analysis of software reliability models. 1993.
- 96 J.K. Scholma. A Lie algebraic study of some integrable systems associated with root systems. 1993.
- 97 J.L. van den Berg. Sojourn times in feedback and processor sharing queues. 1993.
- 98 A.J. Koning. Stochastic integrals and goodness-offit tests, 1993.
- 99 B.P. Sommeijer. Parallelism in the numerical integration of initial value problems. 1993.
- 100 J. Molenaar. Multigrid methods for semiconductor device simulation. 1993.
- 101 H.J.C. Huijberts. Dynamic feedback in nonlinear synthesis problems. 1994.
- 102 J.A.M. van der Weide. Stochastic processes and point processes of excursions. 1994.
- 103 P.W. Hemker, P. Wesseling (eds.). Contributions to multigrid. 1994.
- 104 I.J.B.F. Adan. A compensation approach for queueing problems. 1994.
- 105 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 1. 1994.
- 106 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 2. 1994.
- 107 R.A. Trompert. Local uniform grid refinement for time-dependent partial differential equations. 1995.
- 108 M.N.M. van Lieshout. Stochastic geometry models in image analysis and spatial statistics. 1995.
- 109 R.J. van Glabbeek. Comparative concurrency semantics and refinement of actions. 1996.
- 110 W. Vervaat (ed.). *Probability and lattices.* To appear.
- 111 I. Helsloot. Covariant formal group theory and some applications. 1995.
- 112 R.N. Bol. Loop checking in logic programming. 1995.
- 113 G.J.M. Koole. Stochastic scheduling and dynamic programming. 1995.
- 114 M.J. van der Laan. Efficient and inefficient estimation in semiparametric models. 1995.
- 115 S.C. Borst. Polling Models. 1996.

MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963.

2 A.R. Bloemena. Sampling from a graph. 1964.

3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part II: probabilistic background. 1964.
5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

W.R. van Zwet. Convex transformations of random variables. 1964

8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964. 10 E.M. de Jager. Applications of distributions in mathematical physics. 1964.

11 A.B. Paalman-de Miranda. *Topological semigroups*, 1964. 12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. *Formal properties of newspaper Dutch*.

A. va 1965.

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print: replaced by MCT 54.

14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and

topology. 1968. 22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1. 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part I. 1968.

27 European Meeting 1968. Selected statistical papers, part II. 1968. 28 J. Oosterhoff. Combination of one-sided statistical tests.

1969

29 J. Verhoeff. Error detecting decimal codes. 1969

30 H. Brandt Corstius. *Exercises in computational linguistics*. 1970.

31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.

33 F.W. Steutel. Preservations of infinite divisibility under mix-ing and related topics. 1970.

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971.

36 J. Grasman. On the birth of boundary layers. 1971.

37 J.W. de Bakker, G.A. Blaauw, A.J.W. Duijvest, IDTL. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium, 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972. 42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

Jormaa manipulation systems, part 1: the tangange: 1975. 47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973. 48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok. Connected orderable spaces. 1974.

50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.). *Revised report on the algorithmic language ALGOL* 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974.

53 M.J. Faber. Metrizability in generalized ordered spaces. 1974.

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974.

56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatorial geometry. 1974.

57 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.

58 W. Albers. Asymptotic expansions and the deficiency con-cept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975.

63 J.W. de Bakker (ed.). Foundations of computer science. 1975.

64 W.J. de Schipper. Symmetric closed categories. 1975.

65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes, 1976.

processes. 1976. 72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979. 74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977.

76 M. Rem. Associons and the closure statements. 1976.

77 W.C.M. Kallenberg. Asymptotic optimality of likelihood ratio tests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Empirical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary problems. 1977

81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976.

82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

Steine II, part 2, 1970.
Sa L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.
H.L.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallmann spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program-ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and ana-lytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteristic. 1979

92 P.J. Weeda. Finite generalized Markov programming. 1979. 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977.

94 A. Bijlsma. Simultaneous approximations in transcendental number theory. 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978. 96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations research 1978

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bice 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part ings bic 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces. 1978

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978. 104 J.M. van Wouwe. GO-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979.

109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part 1: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part II: an implemen-tation model. 1979.

112 H.C.P. Berbee. Random walks with stationary increments and renewai theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979.

proteins with restricted alternatives. 1979. http://www.y.for.testing 114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 1. 1979. 116 P.C. Baayen, J. van Mill (eds.). Topological structures II, part 2. 1979.

117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large-scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph theory. 1980.

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

equations. 1980. 123 I. Yuhász. Cardinal functions in topology - ten years later.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980.

126 G. van der Hoek. Reduction methods in nonlinear pro-gramming. 1980. 127 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures.*

1980.

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.
132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location estiators. 1981

134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

(eds.). Formal methods in the study of language, part I. 1981.

(cds.), Formal methods in the study of language, part Hort 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.), Formal methods in the study of language, part II. 1981. 137 J. Telgen. Redundancy and linear programs. 1981.

138 H.A. Lauwerier. Mathematical models of epidemics. 1981.

139 J. van der Wal. Stochastic dynamic programming, succes sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1982

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems. 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a sympo-sium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes. 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequencies. 1982.

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired services. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.162 J.J. Dik. Tests for preference. 1983.

163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.

164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983.

166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.